Traffic Benchbook–Fourth Edition

Content Derived from former MJI's Traffic Benchbook, Vols. 1-3 and Juvenile Traffic Benchbook



MICHIGAN JUDICIAL INSTITUTE © 2025

Michigan Supreme Court

- The Honorable Elizabeth T. Clement, *Chief Justice*
- The Honorable Elizabeth M. Welch, MJI Supervising Justice
- The Honorable Brian K. Zahra, *Justice*
- The Honorable Richard H. Bernstein, Justice
- The Honorable Megan K. Cavanagh, Justice
- The Honorable Kyra Harris Bolden, Justice
- The Honorable Kimberly A. Thomas, Justice
- The Honorable Thomas P. Boyd, State Court Administrator
- Elizabeth Rios-Jones, Deputy State Court Administrator
- Alicia Moon, Supreme Court General Counsel
- Larry S. Royster, Supreme Court Chief of Staff and Supreme Court Clerk

Michigan Judicial Institute Staff

- John R. Nizol, *Director*
- Amanda Clark, Judicial Education Analyst
- Anne M. DeMarco, *Program Assistant*
- Rachael Drenovsky, Learning Center Coordinator
- Amy Feinauer, Program Assistant
- Aaron D. Hanke, Research Attorney
- Michael Hennessy, *E-Learning Instructional Designer and LMS Administrator*
- Phoenix Hummel, *Publications Manager*
- Amanda Joseph, Administrative Assistant
- Heather Leidi, Operations Specialist
- Adam Pfau, Education Analyst
- Lindsay E. Poetz, Research Attorney Specialist
- Janaya Royal, Administrative Specialist
- Rebekah L. T. Sellers, Research Attorney
- Jonah Sjoquist, Continuing Judicial Education Staff Attorney
- Peter C. Stathakis, *Program Manager*
- Prestina Whitlow, Administrative Assistant

Statements in this benchbook represent the professional judgment of the author and are not intended to be authoritative statements by the justices of the Michigan Supreme Court. This benchbook was initially published in 2015, and the text has been revised, reordered, and updated through April 16, 2025.

Note on Precedential Value

"A panel of the Court of Appeals must follow the rule of law established by a prior published decision of the Court of Appeals issued on or after November 1, 1990, that has not been reversed or modified by the Supreme Court, or by a special panel of the Court of Appeals as provided in this court rule." MCR 7.215(J)(1).

Several cases in this book have been reversed, vacated, or overruled in part and/ or to the extent that they contained a specific holding on one issue or another. Generally, trial courts are bound by decisions of the Court of Appeals "until another panel of the Court of Appeals or [the Supreme] Court rules otherwise[.]" *In re Hague*, 412 Mich 532, 552 (1982). While a case that has been fully reversed, vacated, or overruled is no longer binding precedent, it is less clear when an opinion is not reversed, vacated, or overruled in its entirety. Some cases state that "an overruled proposition in a case is no reason to ignore all other holdings in the case." People v Carson, 220 Mich App 662, 672 (1996). See also Stein v Home-Owners Ins Co, 303 Mich App 382, 389 (2013) (distinguishing between reversals in their entirety and reversals in part); Graham v Foster, 500 Mich 23, 31 n 4 (2017) (because the Supreme Court vacated a portion of the Court of Appeals decision, "that portion of the Court of Appeals' opinion [had] no precedential effect and the trial court [was] not bound by its reasoning"). But see Dunn v Detroit Inter-Ins Exch, 254 Mich App 256, 262 (2002), citing MCR 7.215(J)(1) and stating that "a prior Court of Appeals decision that has been reversed on other grounds has no precedential value. . . . [W]here the Supreme Court reverses a Court of Appeals decision on one issue and does not specifically address a second issue in the case, no rule of law remains from the Court of Appeals decision." See also People v James, 326 Mich App 98 (2018) (citing *Dunn* and MCR 7.215()(1) and stating that the decision, "People v Crear, 242 Mich App 158, 165-166 (2000), overruled in part on other grounds by *People v Miller*, 482 Mich 540 (2008), . . . [was] not binding"). Note that *Stein* specifically distinguished its holding from the *Dunn* holding because the precedent discussed in *Dunn* involved a reversal in its entirety while the precedent discussed in *Stein* involved a reversal in part.

The Michigan Judicial Institute endeavors to present accurate, binding precedent when discussing substantive legal issues. Because it is unclear how subsequent case history may affect the precedential value of a particular opinion, trial courts should proceed with caution when relying on cases that have negative subsequent history. The analysis presented in a case that is not binding may still be persuasive. See generally, *Dunn*, 254 Mich App at 264-266. MJI's *Traffic Benchbook* was first published in 1993, and a revised edition was released in 1993 through a project funded by the Michigan Office of Highway Safety Planning and the United States Department of Transportation. A third edition was released in 2005. The *Traffic Benchbook* was previously divided into Volume 1: Civil Infractions and Misdemeanor Traffic Offenses, Volume 2: Recreational Vehicles, and Volume 3: Drunk Driving and Driving While License Suspended Violations, and Felonies. The material has been reorganized into two benchbooks: the *Traffic Benchbook* and the *Recreational Vehicles Benchbook*. This version of the *Traffic Benchbook* was authored by MJI Research Attorney Alessa Boes and MJI Research Attorney Lisa Schmitz, and was edited by MJI Publications Manager Sarah Roth. The authors of this edition were greatly assisted by an editorial advisory committee whose members reviewed draft text and provided valuable feedback. The members of the editorial advisory committee were:

- Michael Balian, Balian & Busse, PLLC
- Sergeant Jill Bennett, Michigan State Police, Training Division
- Colin Boes, Trial Court Services, Forms and Manuals Analyst
- Andrea Erratt, Conservation Officer, Michigan Department of Natural Resources Law Enforcement Division
- Lee Ann Gaspar, Court Liaison, Michigan Department of State
- Hon. Phyllis McMillen, 6th Circuit Court
- Bradene Moore, Michigan Judicial Institute, Judicial Education Analyst
- Julia Norton, Trial Court Services, Management Analyst
- Hon. Charles Pope, 14-B District Court
- Magistrate Tammi Rodgers, 86th District Court
- Kenneth Stecker, Prosecuting Attorneys Association of Michigan

The **Michigan Judicial Institute (MJI)** was created in 1977 by the Michigan Supreme Court. MJI is responsible for providing educational programs and written materials for Michigan judges and court personnel. In addition to formal seminar offerings, MJI is engaged in a broad range of publication activities, services, and projects that are designed to enhance the professional skills of all those serving in the Michigan court system. MJI welcomes comments and suggestions. Please send them to **Michigan Judicial Institute**, **Hall of Justice**, **P.O. Box 30048, Lansing, MI 48909. (517) 373-7171.**

Table of Contents

Cover and Acknowledgments

Title Page i	
Michigan Supreme Court ii	
Michigan Judicial Institute Staff iii	
Note on Precedential Value iv	
Foreword and Acknowledgmentsv	

Chapter 1: Procedures for Traffic Civil Infractions, Misdemeanors, and Felonies

1.1	Scope Note 1-3
1.2	Filings by Incarcerated Individuals 1-3
1.3	Attempts 1-3
	A.MVC Provisions1-3B.General Attempt Statute1-5
1.4	Probation and Delayed Sentencing 1-6
1.5	Taking Pleas Under Advisement 1-8
	A.Felony Cases Generally1-8B.Misdemeanor Cases Generally1-8C.Traffic Exception1-8
1.6	Restitution 1-8
1.7	Investigative Stops 1-9
1.8	Courts With Jurisdiction Over Civil Infraction Actions 1-9
1.9	Venue for Civil Infraction Actions 1-10
1.10	Time Guidelines for Processing Civil Infraction Actions 1-10
1.11	Authority to Conduct Civil Infraction Actions 1-10
	A.District Court or Municipal Judge1-10B.District Court Magistrate1-11
1.12	Technical Requirements of Traffic Citations and Parking Violation
	Notices 1-12
	A. Contents of the Citation 1-12
	B. Signed Under Oath
1.13	Initiating a Civil Infraction Action
	 A. Authority to Issue a Citation
1.14	Special Requirements for Nonresidents 1-19
1.15	Defendant's Options When a Citation Is Issued 1-19

	A. Admit Responsibility Without Explanation	
	B. Admit Responsibility With Explanation	
1 1/	C. Deny Responsibility	
1.16	Informal Hearings	
	A. Conducting the HearingB. No Representation at Hearing	
	C. Discovery	
	D. Notice and Subpoenas	
	E. Determining Responsibility	
	F. Appeals	
1.17	Formal Hearings	
	A. Conducting the Hearing	
	B. Representation PermittedC. Discovery	
	D. Notice and Subpoenas	
	E. Determining Responsibility	
	F. Appeals	
1.18	Entering Default Judgment for Failure to Answer a Citation or Appea	
	a Scheduled Hearing	1-28
1.19	Setting Aside a Default Judgment	1-29
1.20	Civil Fines, Costs, and Assessments for Civil Infractions	1-30
	A. Civil Fines	1-31
	B. Costs	
	C. Assessments	
1.21	Waiving Civil Fines, Costs, and Assessments	
	A. Mandatory WaiverB. Discretionary Waiver	
1 22	Civil Contempt Proceedings for Nonpayment of Monies Owed	
1.22	Treatment, Education, and Rehabilitation Programs	
1.23	Courts With Jurisdiction Over Misdemeanor Traffic Offenses	
	Jurisdiction and Duties of District Court Magistrates	
1.20	Initiating a Misdemeanor Action by Issuing a Citation A. Citation Requirements	
	A. Citation RequirementsB. Accepting a Plea Based on a Citation	
	C. Failure to Appear	
	D. Restrictions on the Issuance of Citations	
	E. Summons	
4.07	F. Arraignment	
1.27	General Criminal Penalty for Misdemeanor Traffic Offenses	
	A. Generally	
1.28	B. Exceptions Rebuttable Presumption Regarding Jail or Probation Sentence	
1.20	Minimum State Costs for Misdemeanor Traffic Offenses	
1.27	Costs	
1.30	Failure to Comply with Sentence	
1.31	Courts With Jurisdiction Over Felony Traffic Offenses	
		. 00

1.33	Initiating a Felony Action	1-54
1.34	General Criminal Penalties for Felony Traffic Offenses	1-55
1.35	Felony Sentencing	1-55
	A. Calculating the Minimum Sentence Range	
1.36	B. Departures Minimum State Costs for Felony Traffic Offenses	
	-	
1.37	Costs	
1.38	Felony in Which a Motor Vehicle Was Used	
1.39	Authority	
1.40	Multiple Sanctions for the Same Offense	1-61
1.41	Abstracts of Convictions	1-62
1.42	Points	1-64
1.43	License Denial	1-65
1.44	License Revocation	1-66
1.45	License Suspension	1-67
	A. Mandatory Suspension Under § 319	1-67
	B. Mandatory Suspension for Failure to Answer Citation/Notice to	
	Appear or Failure to Comply with Order/Judgment	
	C. Mandatory Additional Like Periods of Suspension Under § 904D. No Appeal to the Secretary of State	
	E. Improper Suspension	
1.46	Out-of-State Offenses and Nonresidents	
1.47	Vehicle Immobilization	1-72
	A. Circumstances Under Which Immobilization is Permitted	
	B. Exceptions	1-72
	C. Period of Immobilization	
	D. Manner of ImmobilizationE. Immobilization Suspension	
	F. Immobilization Reinstatement	
1.48	Vehicle Impoundment	
1.49	Vehicle Forfeiture	
1.50	Registration Denial	
1.50	Basic Driver Improvement Course	
1.01	A. Secretary of State Responsibilities	
	B. Eligibility Requirements	
	C. Cost of Course and Allocation of Cost	1-81

Chapter 2: Juvenile Traffic Proceedings

2.1	Scope Note	2-2
2.2	Jurisdiction Over Juveniles Accused of Committing a Criminal Traffic	
	Violation	2-2
	A. Continued Jurisdiction	2-2
	B. Age Calculation	2-2
2.3	Jurisdiction Over Juveniles Accused of Committing a Civil Infraction	2-3

2.4	Venue and Transfer of Case 2-3
2.5	Court Rules Applicable to Cases Under the Juvenile Code 2-3
2.6	Investigative Stops 2-4
2.7	Taking a Juvenile Into Custody 2-4
2.8	Required Procedures for Traffic-Related Offenses Contained in the
	Michigan Penal Code 2-5
2.9	Special Procedures Applicable in Juvenile Traffic Cases 2-5
	A. Commencing Proceedings 2-6
	B. Notice
	C. Required Notice When a Juvenile is Charged With a Felony in Which a Motor Vehicle Was Used
	 D. Preliminary Inquiries and Preliminary Hearings
2.10	Required Procedures Following a Juvenile's Failure to Respond to a Motor
	Vehicle Violation
	A. Dispositions
	B. Required Reporting to the Secretary of State 2-10
2.11	Restitution, Crime Victim's Rights Assessment, and Costs 2-11
2.12	Setting Aside Adjudications 2-11
2.13	Holmes Youthful Trainee Act

Chapter 3: Civil Infraction Nonmoving Violations

3.1	Introduction and Scope Note 3-2
3.2	Causing or Permitting an Unlicensed Minor to Drive
	A. Statutory Authority 3-2
	B. Penalties 3-2
3.3	Failing to Change Address on Driver's License
	A. Statutory Authority 3-2
	B. Penalties
	C. Sanctions
~ .	D. Issues
3.4	Operating a Vehicle in Violation of Graduated Licensing
	Requirements
	A. Statutory Authority 3-3
	B. Penalties
	C. Sanctions
3.5	Failing to Change Address on Registration or Title 3-8
	A. Statutory Authority 3-8
	B. Penalties
3.6	Failing to Assure Title Transfer When Vehicle Is Sold 3-8
	A. Statutory Authority
	B. Penalties
	C. Issues
3.7	Invalid or No Registration Plate 3-9
	A. Statutory Authority
	B. Penalties 3-10

3.8	No Proof of Insurance	3-10
	A. Statutory Authority	3-10
	B. Penalties	
	C. Sanctions	
3.9	No Proof of Registration	3-12
	A. Statutory Authority	
	B. Penalties	
3.10	Registration Plate Violations	3-13
	A. Statutory Authority	3-13
	B. Penalties	
	C. Issues	
3.11	Abandoning a Vehicle and Failing to Redeem It Before Disposition	
	A. Statutory Authority	
	B. Penalties	3-15
3.12	Child Restraint System and Safety Belt System Violations	3-15
	A. Statutory Authority	
	B. Penalties	3-18
3.13	Equipment Violations	3-19
	A. Statutory Authority	3-19
	B. Equipment Violations	
	C. Penalties	
	D. Issues	
3.14	Parking, Stopping, or Standing	
	A. Statutory Authority	
	B. Parking, Standing, and Stopping Violations	
	C. Exceptions to Parking, Stopping, and Standing Violations	
	D. Parking, Standing, and Stopping Violation Proceedings	
	E. Liability for Parking, Standing, and Stopping ViolationsF. Penalties	
2.15		
3.15	Soliciting Business at the Scene of a Motor Vehicle Accident or Disa	
	Vehicle	
	A. Statutory Authority	
0.4.	B. Penalties	
3.16	Transporting or Possessing Alcohol in a Motor Vehicle by a Person L	
	Than 21 Years of Age	3-31

Chapter 4: Civil Infraction Moving Violations

4.1	Introduction and Scope Note	4-2
4.2	Careless or Negligent Driving	4-2
	A. Statutory Authority	
	B. Penalties	
	C. Sanctions	4-2
	D. Issues	4-3
4.3	Failing to Remove a Motor Vehicle From Roadway After an Accident	4-4
	A. Statutory Authority	4-4
	B. Penalties	4-4

	С.	Issues	. 4-4
4.4	Fai	iling to Stop for School Bus	. 4-5
	Α.	Statutory Authority	. 4-5
	Β.	Exception	
	С.	Penalties	
	D.	Sanctions	
4.5	Fo	llowing a Fire Truck Too Closely	
	Α.	Statutory Authority	
	B.	Penalties	
	С.	Sanctions	
4.6		proper Passing of a Stationary Emergency Vehicle	
	A. B	Statutory Authority Penalties	
	ь. С.	Sanctions	
4.7		proper Use of Emergency Lights	
4.7	A.	Statutory Authority	
	R.	Penalties	
	С.	Sanctions	
4.8	Int	erference With View, Control, or Operation of Vehicle	
	Α.	Statutory Authority	
	Β.	Penalties	
	С.	Sanctions	4-14
4.9	Ov	ertaking and Passing	4-15
	Α.	Statutory Authority	
	Β.	Overtaking and Passing Violations:	
	C.	Penalties	
4 10	D.	Sanctions	
4.10		rmitting Minor to Ride in Pickup Truck Bed	
	A. B.	Statutory Authority Exceptions	
	С.	Penalties	
	D.	Sanctions	
4.11	Ra	ilroad Crossings	4-19
	Α.	Statutory Authority	
	Β.	Penalties	
	С.	Sanctions	4-20
4.12	Rig	ht-of-Way or Failure to Yield	4-21
	Α.	Statutory Authority	
	Β.	Right-of-Way or Failure to Yield Violations	
	C. D	Penalties Sanctions	
	F.	Issues	
4.13		eed Violations	
	A.	Statutory Authority: Lawful Highway Speeds	
	В.	Statutory Authority: Maximum and Minimum Speed Limits and	
		General Speed Limit	4-33
	С.	Statutory Authority: Modified Speed Limits	
	D.	Special Citation Requirements	
	Ε.	Evidence in a Speed Case	4-35

	F.	Penalties	
	G.	Sanctions	4-40
4.14	Sto	p and Go, Signs and Signals	4-42
	Α.	Statutory Authority: Stop Signs	4-43
	В.	Statutory Authority: Traffic Lights or Signals	4-44
	С.	Stop and Go, Sign and Signal Violations	4-46
	D.	Penalties	4-46
	Ε.	Sanctions	4-46
4.15	Ho	Iding or Using a Mobile Electronic Device While Driving	4-47
	Α.	Statutory Authority	4-47
	В.	Exceptions	4-47
	С.	Penalties	4-49
	D.	Sanctions	4-50
4.16	Tur	ning and Signaling	4-51
	Α.	Statutory Authority: Turning at Intersection	
	В.	Statutory Authority: Signals for Stopping or Turning	
	С.	Turning and Signaling Violations	
	D.	Penalties	
	E.	Sanctions	
	F.	Issues	4-53
4.17	Usi	ng a Cellular Telephone While Operating a Motor Vehicle (Kelsey	ľS
	law	/)	4-53
	Α.	Statutory Authority	4-53
	В.	Exceptions	4-54
	C.	Penalties	4-54
	D.	Sanctions	4-54
4.18	Wr	ong Side or Wrong Way	4-55
	Α.	Statutory Authority: "Keep to the Right" Rule	4-55
	В.	Exceptions to the "Keep to the Right" Rule	
	С.	Statutory Authority: One-Way and Two-Way Traffic	4-56
	D.	Wrong Side or Wrong Way Violations	4-56
	Ε.	Penalties	
	F .	Sanctions	4-57

Chapter 5: Misdemeanor Nonmoving Violations

5.1	Introduction and Scope Note	5-3
5.2	Failing to Give Information and Aid at the Scene of an Accident	5-3
	A. Statutory Authority	5-3
	B. Penalties	
	C. Issues	5-4
5.3	Failing to Report Accident Involving Death, Personal Injury, or Prop	erty
	Damage of \$1,000 or More	5-5
	A. Statutory Authority	5-5
	B. Penalties	5-5
	C. Issues	5-5
5.4	Garage or Repair Shop Failing to Report Evidence of Accident or	
	Bullet	5-6

	A. Statutory Authority	
5.5	Leaving the Scene of an Accident Resulting in Damage to Fixtures That Are	
	Upon or Adjacent to a Highway 5-	
	A. Statutory Authority	
	B. Penalties 5-	6
5.6	Leaving the Scene of or Failing to Report an Accident Resulting in Persona	I
	Injury 5-	
	A. Statutory Authority 5-	
	 B. Relevant Jury Instruction	
	D. Sanctions	
5.7	Leaving the Scene of or Failing to Report an Accident Resulting in Vehicle	
	Damage Only 5-	
	A. Statutory Authority 5-	8
	B. Relevant Jury Instruction	
	C. Penalties	
5.8	Leaving the Scene of or Failing to Report an Accident With an Attended o	
0.0	Unattended Vehicle	
	A. Statutory Authority	
	B. Penalties	
	C. Sanctions 5-	9
5.9	Allowing Another to Operate in Violation of the Michigan Vehicle	
	Code 5-1	
	A. Statutory Authority	
	B. Penalties 5-1 C. Issues 5-1	
5.10	Allowing Another Person to Operate a Motor Vehicle With a Suspended	
0.10	Revoked License	
	A. Statutory Authority	
	B. Relevant Jury Instruction 5-1	
	C. Penalties	
	D. Sanctions	
5.11	Possessing a Reproduced, Altered, Counterfeit, Forged, or Duplicate	2
0.11	License	2
	A. Statutory Authority	
	B. Penalties	
5.12	Reporting a False Address Change to the Secretary of State 5-1	3
	A. Statutory Authority 5-1	
	B. Penalties	
F 10	C. Sanctions 5-1 Reproducing, Altering, Counterfeiting, Forging, or Duplicating a License,	
0.13	or Using Such License, With Intent to Commit a Misdemeanor 5-1	
	A. Statutory Authority	
	B. Penalties	

5.14	Unlawful Use or Display of License	5-14
	A. Statutory Authority	
	B. Penalties	
	C. Sanctions	
5.15	Allowing Another to Drive or Move an Unregistered Vehicle	
	A. Statutory Authority	
	B. Penalties	
5.16	Failing to Apply for Registration and Certificate of Title	
	A. Statutory Authority	
	B. Penalties	
	C. Sanctions	
5.17	Failing to Transfer Title	
	A. Statutory Authority	
F 10	B. Penalties	
5.18	Forging Proof of Insurance	
	A. Statutory Authority	
F 10	B. Penalties	
5.19	Producing False Evidence of Motor Vehicle Insurance	
	A. Statutory AuthorityB. Penalties	
5.20		0-19
5.20		
	Certificate of Title, or Using Such Certificate of Title, With Intent to	F 20
	Commit a Crime	
	A. Statutory AuthorityB. Penalties	
5.21		
0.21	Temporary Registration Violations	
	A. Statutory AuthorityB. Penalties	
5 22	Unlawful Lending or Use of Title, Registration Certificate, Plate, or	521
0.22	Permit	5 21
	A. Statutory Authority	
	B. Penalties	
5.23	Failing to Answer Citation, Appear in Court, or Comply With an Orde	
0.20	Judgment	
	A. Statutory Authority	
	B. Sanctions	
5.24	Failing to Disclose Odometer Mileage	
	A. Statutory Authority	
	B. Penalties	
	C. Issues	5-23
5.25	Falsifying or Improperly Disposing of a Citation	5-24
	A. Statutory Authority	
	B. Penalties	5-24
5.26	Improper Use of Disabled Person Identification	5-24
	A. Statutory Authority	
	B. Penalties	
	C. Issues	5-25

5.27	Traffic Control Device Violations 5	5-26	
	A. Statutory Authority 5	5-26	
	B. Penalties 5	5-26	
5.28	Transporting or Possessing Alcohol in a Motor Vehicle by a Person Le	SS	
	Than 21 Years of Age 5	5-27	
	A. Statutory Authority 5	5-27	
	B. Penalties 5		
	C. Sanctions 5	5-28	
	D. Issues 5	5-28	
5.29	Minor Purchasing, Consuming, or Possessing Alcohol, or Having Any		
	Bodily Alcohol Content/Furnishing Fraudulent Identification to a		
	Minor, or Minor Using Fraudulent Identification to Purchase Alcoholic		
	Liquor 5	5-28	
	A. Statutory Authority 5	5-29	
		5-29	
	C. Issues 5	5-31	
5.30	Transporting or Possessing Open Alcohol in a Vehicle	5-34	
	A. Statutory Authority 5	5-34	
	B. Penalties 5	5-34	
	C. Sanctions 5	5-35	
	D. Issues 5	5-35	

Chapter 6: Misdemeanor Moving Violations

6.1	Introduction and Scope Note
6.2	Disobeying the Direction of a Police Officer Who Is Regulating Traffic 6-2
	A. Statutory Authority
	B. Penalties 6-2
	C. Sanctions 6-2
6.3	Drag Racing 6-3
	A. Statutory Authority
	B. Penalties 6-3
	C. Sanctions
	D. Issues
6.4	Driving in Violation of a Restricted License
	A. Statutory Authority
	B. Penalties
	C. Sanctions 6-5
6.5	Driving or Moving an Unregistered Vehicle
	A. Statutory Authority
	B. Penalties
6.6	Driving With an Invalid License
	A. Statutory Authority
	B. Penalties
	C. Sanctions
6.7	Failing to Stop for School Crossing Guard
	A. Statutory Authority

	Β.	Penalties	
	C.	Sanctions	
(0	D.	Issues	
6.8		ling to Yield to Handicapped Individual	
	A. B.	Statutory Authority Penalties	
	Б. С.	Sanctions	
6.9	•••	proper Operation of a Golf Cart	
0.7	A.		
	л. В.	Penalties	
	C.	Issues	
6.10	Im	proper Use of Emergency Vehicle	6-11
	Α.	Statutory Authority	
	Β.	Penalties	6-12
	С.	Sanctions	
	D.	Issues	
6.11		oving Violation Causing Injury to Person in Work Zone or School E	
	Zor	пе	
	Α.	Statutory Authority	
	B.	Penalties	
(1)	C.	Sanctions	0-13
0.12		oving Violation Causing Injury to Person Operating Farm	
	. '	uipment	
	A.	Statutory Authority Penalties	
	В. С.	Sanctions	
6.13	· · ·	oving Violation Causing Serious Impairment of a Body Function	
0.15	A.	Statutory Authority	
	В.	Relevant Jury Instruction	
	C.	Penalties	
	D.	Sanctions	6-15
	Ε.	Issues	
6.14	Mc	oving Violation Causing Death	6-17
	Α.	Statutory Authority	
	Β.	Relevant Jury Instruction	
	C. D.	Penalties Sanctions	
6 15		erating Motor Vehicle While License Is Suspended or Revoked	
6.15	Ор А.		
	A. B.	Statutory Authority Relevant Jury Instruction	
	C.	Penalties	
	D.	Sanctions	
	Ε.	Issues	6-20
6.16	Ор	erating a Motor Vehicle While Unlicensed	6-21
	Α.	Statutory Authority	6-21
	Β.	Penalties	
	С.	Sanctions	
6.17	Ор	erating a Motor Vehicle Without a License in Possession	6-22

A. Statutory Authority	6-22
B. Penalties	6-22
Reckless Driving	6-23
A. Statutory Authority	6-23
B. Relevant Jury Instruction	6-23
C. Penalties	6-23
D. Sanctions	6-24
E. Issues	6-24
Reckless Driving Causing Miscarriage, Stillbirth, or Death	6-25
A. Statutory Authority	6-25
Unlawful Use of an Automobile Without Intent to Steal (Joyriding) .	6-26
A. Statutory Authority	6-26
5	
	 A. Statutory Authority

Chapter 7: Felony Nonmoving Violations

7.1	Introduction and Scope Note	. 7-2
7.2	Allowing A Person Who is Unlicensed or Whose License is Suspende	d or
	Revoked to Operate a Motor Vehicle Causing Serious Impairment o	fa
	Body Function or Death	. 7-2
	A. Statutory Authority	. 7-2
	B. Relevant Jury Instructions	
	C. Criminal Penalties	
	D. Sanctions	
	E. Issues	
7 0	F. Related Misdemeanor	
7.3	Altering, Forging, or Falsifying Documents	
	A. Statutory Authority B. Criminal Penalties	
	B. Criminal PenaltiesC. Sanctions	
7.4	Buying, Selling, or Using a Portable Signal Preemption Device	
7.4	A. Statutory Authority	
	B. Criminal Penalties	
	C. Sanctions	
	D. Related Misdemeanor	
7.5	False Certification	. 7-8
	A. Statutory Authority	. 7-8
	B. Criminal Penalties	
	C. Sanctions	. 7-9
7.6	False Statement in Application for Certificate of Title or in Assignme	nt of
	Title; Possession of Stolen Vehicle	. 7-9
	A. Statutory Authority	. 7-9

	B. Relevant Jury Instructions	
	D. Sanctions	
	E. Issues	
7.7	Odometer Tampering	
	C. Issues	
7.8	Police Officer Knowingly Making N	Naterially False Statement in Citation as
	Perjury	
	C. Issues	
7.9	Possessing, Selling, or Offering for	⁻ Sale a Stolen, False, or Counterfeit
	Certificate of Insurance	
	A. Statutory Authority	

Chapter 8: Felony Moving Violations

8.1	Introduction and Scope Note	. 8-2
8.2	Failure to Stop at Signal of Police Officer ("Fleeing and Eluding")	. 8-2
	A. Statutory Authority	. 8-2
	B. Relevant Jury Instructions	
	C. Criminal Penalties	
	D. Sanctions	. 8-4
	E. Issues	
	F. Related Misdemeanor	. 8-7
8.3	Failure to Use Due Care and Caution Causing Injury or Death to	
	Emergency Response Personnel	. 8-7
	A. Statutory Authority	. 8-7
	B. Criminal Penalties	. 8-8
	C. Sanctions	
	D. Related Civil Infraction	. 8-9
8.4	Leaving the Scene of an Accident Resulting in Serious Impairment of	fa
	Body Function or Death	. 8-9
	5	
	5	. 8-9
	5	
	A. Statutory Authority	8-11
	A. Statutory AuthorityB. Relevant Jury Instructions	8-11 8-11
	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions E. Issues 	8-11 8-11 8-11 8-12
	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions 	8-11 8-11 8-11 8-12
8.5	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions E. Issues 	8-11 8-11 8-11 8-12 8-15
8.5	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions E. Issues F. Related Misdemeanors 	8-11 8-11 8-11 8-12 8-15 8-15
8.5	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions E. Issues F. Related Misdemeanors Moving Violation Causing Death in Work Zone or School Bus Zone 	8-11 8-11 8-12 8-12 8-15 8-15 8-15
8.5	 A. Statutory Authority B. Relevant Jury Instructions C. Criminal Penalties D. Sanctions E. Issues F. Related Misdemeanors Moving Violation Causing Death in Work Zone or School Bus Zone A. Statutory Authority 	8-11 8-11 8-12 8-15 8-15 8-15 8-15 8-15
8.5	 A. Statutory Authority B. Relevant Jury Instructions	8-11 8-11 8-12 8-15 8-15 8-15 8-15 8-15 8-15

	Husbandry	8-16
	A. Statutory Authority	8-16
	B. Criminal Penalties	8-16
	C. Sanctions	
	D. Related Misdemeanor	8-17
8.7	Operating a Motor Vehicle While Unlicensed or With Suspended/	
	Revoked License	8-17
	A. Statutory Authority	8-17
	B. Criminal Penalties	
	C. Sanctions	8-19
	D. Issues	
	E. Related Misdemeanor	
8.8	Reckless Driving Causing Death	
	A. Statutory Authority	
	B. Relevant Jury Instruction	
	C. Criminal Penalties	
	D. Sanctions	
	E. Issues	
8.9	Reckless Driving Causing Serious Impairment of a Body Function	
	A. Statutory Authority	
	B. Relevant Jury Instruction	
	C. Criminal Penalties	
	D. Sanctions	
_	E. Issues	8-26
8.10	Malicious Destruction of Trees, Shrubs, Plants, or Soil with a Motor	
	Vehicle	
	A. Statutory Authority	
	B. Criminal Penalties	
	C. Sanctions	
8.11	Manslaughter with a Motor Vehicle	
	A. Statutory Authority	
	B. Relevant Jury Instructions	
	C. Criminal Penalties	
	D. Sanctions	
0.10	E. Issues	
8.12	Unlawfully Driving Away an Automobile	
	A. Statutory Authority	
	B. Relevant Jury Instructions	
	C. Criminal Penalties	
	D. Sanctions	
	E. Issues	0-3Z

Chapter 9: Section 625 Offenses

9.1	Sco	pe Note	9-2
9.2	Spe	cial Procedures for Section 625 Offenses	9-2
	Α.	Warrantless Arrest	9-2
	В.	Arraignment	9-3

	C. Pretrial Conference	
	D. Plea Provision in Section 625(16)	
	E. Advice to the Accused Before Accepting Plea	
	F. Final Adjudication	
	G. Order Screening and Assessment for Rehabilitative Services Bel	
	Imposing Sentence	
	H. Issuance of a Restricted License	
	I. Appeals to Circuit Court	
9.3	DWI/Sobriety Court and the Specialty Court Interlock Program	
	A. Guiding Principles	
	B. Requirements for Placement	
	C. Restricted Licensing for Program Participants	
9.4	Operating While Intoxicated (OWI)—Section 625(1)	9-15
	A. Statutory Authority	9-15
	B. Relevant Jury Instructions	
	C. Penalties	
	D. Sanctions	
	E. Issues	9-19
9.5	Permitting or Authorizing Certain Individuals to Drive—Section	
	625(2)	9-23
	A. Statutory Authority	9-23
	B. Penalties	
9.6	Operating While Visibly Impaired (OWVI)—Section 625(3)	9-24
	A. Statutory Authority	
	B. Relevant Jury Instructions	
	C. Penalties	
	D. Sanctions	
	E. Issues	
9.7	OWI or OWVI Causing Death—Section 625(4)	9-28
	A. Statutory Authority	
	B. Relevant Jury Instructions	
	C. Penalties	
	D. Sanctions	
	E. Issues	
9.8	OWI or OWVI Causing Serious Impairment of a Body Function—Sec	
	625(5)	
	A. Statutory Authority	
	B. Relevant Jury Instructions	
	C. Penalties	
	D. Sanctions	
	E. Issues	
9.9	Zero Tolerance Violations—Section 625(6)	
	A. Statutory Authority	
	B. Penalties	
	C. Sanctions	
	D. Issues	
9.10	Child Endangerment—Section 625(7)	
2.10	A. Statutory Authority	
	B. Penalties	
		7-00

	C. Sanctions	9-40
9.11	Operating with the Presence of Certain Controlled Substances in Bo	dy—
	Section 625(8)	9-41
	A. Statutory Authority	9-41
	B. Relevant Jury Instruction	
	C. Penalties	
	D. Sanctions	
	E. Issues	
9.12	Ignition Interlock Device	
	A. Generally	
	B. Vehicle Impoundment	
9.13	Preliminary Chemical Breath Analysis (PBT) & Chemical Tests of Bloc	
	Urine, or Breath	9-48
	A. Preliminary Chemical Breath Analysis (PBT)	
	B. Chemical Test	
9.14	Roadside Drug Testing	9-61
	A. Preliminary Oral Fluid Analysis	
	B. Authority to Arrest	
	C. Admissibility of Preliminary Oral Fluid Analysis	
9.15	Testimony Regarding Field Sobriety Tests	9-63
9.16	Drug Recognition Evaluator or Expert Testimony	9-63
9.17	Obtaining or Transferring a Vehicle to Circumvent Vehicle or License	è
	Sanctions	9-64
	A. Unlawful Acquisition of a Vehicle	9-64
	B. Unlawful Transfer of a Vehicle	9-65

Glossary

Chapter 1: Procedures for Traffic Civil Infractions, Misdemeanors, and Felonies

1.1	Scope Note	1-3
1.2	Filings by Incarcerated Individuals	. 1-3
1.3	Attempts	1-3
1.4	Probation and Delayed Sentencing	1-6
1.5	Taking Pleas Under Advisement	1-8
1.6	Restitution	1-8
1.7	Investigative Stops	1-9
Part	A—Civil Infractions	
1.8	Courts With Jurisdiction Over Civil Infraction Actions	1-9
1.9	Venue for Civil Infraction Actions	1-10
1.10	Time Guidelines for Processing Civil Infraction Actions	1-10
1.11	Authority to Conduct Civil Infraction Actions	1-10
1.12	Technical Requirements of Traffic Citations and Parking Violation No	tices
		1-12
1.13	Initiating a Civil Infraction Action	1-15
1.14	Special Requirements for Nonresidents	1-19
1.15	Defendant's Options When a Citation Is Issued	1-19
1.16	Informal Hearings	1-22
1.17	Formal Hearings	1-26
1.18	Entering Default Judgment for Failure to Answer a Citation or Appear	r for
	a Scheduled Hearing	1-28
1.19	Setting Aside a Default Judgment	1-29
1.20	Civil Fines, Costs, and Assessments for Civil Infractions	1-30
1.21	Waiving Civil Fines, Costs, and Assessments	1-36
1.22	Civil Contempt Proceedings for Nonpayment of Monies Owed	1-38
1.23	Treatment, Education, and Rehabilitation Programs	1-39
_		

ction
JUON

)
1.25 Jurisdiction and Duties of District Court Magistrates 1-41	
1.26 Initiating a Misdemeanor Action by Issuing a Citation 1-42)
1.27 General Criminal Penalty for Misdemeanor Traffic Offenses	}
1.28 Rebuttable Presumption Regarding Jail or Probation Sentence 1-49)
1.29 Minimum State Costs for Misdemeanor Traffic Offenses 1-50)
1.30 Costs)
1.31 Failure to Comply with Sentence 1-52)
Part C—Felonies	
1.32 Courts With Jurisdiction Over Felony Traffic Offenses 1-53	}
1.33 Initiating a Felony Action 1-54	ŀ
1.34 General Criminal Penalties for Felony Traffic Offenses 1-55)
1.35 Felony Sentencing 1-55)
1.36 Minimum State Costs for Felony Traffic Offenses 1-57	1
1.37 Costs 1-58	}
1.38 Felony in Which a Motor Vehicle Was Used 1-59)
Part D—General Overview of Licensing Sanctions for Civil Infractions,	
Misdemeanor, and Felony Traffic Offenses	
1.39 Authority 1-61	
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-62	 <u>2</u>
1.39 Authority1-611.40 Multiple Sanctions for the Same Offense1-61	 <u>2</u>
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-62	 2
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-64	1 2 1 5
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-65	
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-651.44License Revocation1-661.45License Suspension1-671.46Out-of-State Offenses and Nonresidents1-71	 2
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-651.44License Revocation1-661.45License Suspension1-67	 2
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-651.44License Revocation1-661.45License Suspension1-671.46Out-of-State Offenses and Nonresidents1-71	
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-651.44License Revocation1-661.45License Suspension1-671.46Out-of-State Offenses and Nonresidents1-711.47Vehicle Immobilization1-72	
1.39Authority1-611.40Multiple Sanctions for the Same Offense1-611.41Abstracts of Convictions1-621.42Points1-641.43License Denial1-651.44License Revocation1-661.45License Suspension1-671.46Out-of-State Offenses and Nonresidents1-711.47Vehicle Immobilization1-721.48Vehicle Impoundment1-75	

1.1 Scope Note

This chapter discusses the jurisdictional and procedural requirements related to civil infractions involving traffic or parking violations, misdemeanor traffic offenses, and felony traffic offenses. It also provides an overview of the criminal penalties and licensing sanctions generally associated with violations of the Michigan Vehicle Code (MVC).¹ A more detailed discussion of criminal penalties and licensing sanctions is provided in the chapters covering specific civil infractions, misdemeanors, and felonies. Procedures regarding juveniles are discussed in Chapter 2.

1.2 Filings by Incarcerated Individuals

A pleading or other document filed by an unrepresented individual who is incarcerated in prison or jail "must be deemed timely filed if it was deposited in the institution's outgoing mail on or before the filing deadline. Proof of timely filing may include a receipt of mailing, a sworn statement setting forth the date of deposit and that postage has been prepaid, or other evidence (such as a postmark or date stamp) showing that the document was timely deposited and that postage was prepaid." MCR 1.112.

1.3 Attempts

Attempted traffic offenses may be governed by either the MVC, MCL 257.204b, or by the general attempt statute, MCL 750.92.

A. MVC Provisions

MCL 257.204b establishes sanctions and penalties for attempted traffic offenses as follows:

"(1) When assessing points, taking licensing or registration actions, or imposing other sanctions under [the MVC] for a conviction of an attempted violation of a law of this state, a local ordinance substantially corresponding to a law of this state, or a law of another state substantially corresponding to a law of this state, is a local ordinance substantially corresponding to a law of this state, or a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantially corresponding to a law of this state, a local ordinance substantial state state substantial state st

¹The MVC is the principal source of Michigan traffic law. "Except as otherwise provided in [MCL 257.901(3)], it is a misdemeanor for a person to violate [the MVC], unless that violation is by [the MVC] or other law of this state declared to be a felony or a civil infraction." MCL 257.901(1). "Except as otherwise provided in [the MVC], a violation of [the MVC] by the owner of a commercial quadricycle arising out of the ownership or operation of the commercial quadricycle is a civil infraction." MCL 257.901(3). Except as otherwise provided in the Revised Judicature Act, the MVC governs civil infractions involving a traffic or parking violation. MCL 600.113(2).

the secretary of state or the court shall treat the conviction the same as if it were a conviction for the completed offense.

(2) The court shall impose a criminal penalty for a conviction of an attempted violation of this act or a local ordinance substantially corresponding to a provision of this act in the same manner as if the offense had been completed."

MCL 257.204b distinguishes between attempted offenses for purposes of imposing licensing or vehicle sanctions and for purposes of imposing criminal penalties.

Licensing and vehicle sanctions—MCL 257.204b(1) applies to attempted violations of any Michigan law, local ordinance substantially corresponding to a Michigan law, or substantially corresponding law from another state for which licensing or vehicle sanctions are imposed under the MVC. Under MCL 257.204b(1), any attempted offense that results in licensing or vehicle sanctions under the MVC must be treated as a completed offense for purposes of imposing sanctions.

Criminal penalties—MCL 257.204b(2) requires courts to treat attempted violations of "this act," i.e., of the MVC, or a substantially corresponding local ordinance, as completed offenses for purposes of imposing criminal penalties. Thus, MCL 257.204b(2) does not apply to attempted traffic offenses not codified in the MVC, such as unlawful driving away a motor vehicle under MCL 750.413. Criminal penalties for offenses not codified in the MVC are governed by the general attempt statute, MCL 750.92. See *People v Cervi*, 270 Mich App 603, 618 (2006) (general attempt statute applies only where there is no express provision for attempt in the statute under which the defendant is charged).

Moreover, MCL 257.204b(2) only addresses the penalties for attempted violations of the MVC; it does not criminalize them. *People v Burton*, 252 Mich App 130, 136 (2002) (addressing former MCL 257.625(17), which contained substantially similar language as MCL 257.204b(2)). In *Burton*, the defendant challenged his convictions of attempted violations of MCL 257.625 and MCL 257.904 on the grounds that the MVC did not criminalize attempts. *Burton*, 252 Mich App at 133. The Court of Appeals agreed, stating that the MVC "continues to treat violations of the code . . . *as if* they were completed offenses for purposes of punishment, but it does not specifically proscribe and include attempted violations within the bounds of the code." *Id.* at 136 n 7. In the absence of a statutory provision specifically criminalizing attempts, attempted violations of the MVC should be tried under the general attempt statute.

B. General Attempt Statute

The general attempt statute, MCL 750.92, applies to attempted traffic offenses that are not covered by MCL 257.204b. The general attempt statute provides:

"Attempt to commit crime—Any person who shall attempt to commit an offense prohibited by law, and in such attempt shall do any act towards the commission of such offense, but shall fail in the perpetration, or shall be intercepted or prevented in the execution of the same, when no express provision is made by law for the punishment of such attempt, shall be punished as follows . . . "

The elements of attempt are: "(1) an attempt to commit an offense prohibited by law, and (2) any act towards the commission of the intended offense." *People v Thousand*, 465 Mich 149, 164 (2001). Moreover, "the elements of attempt under [MCL 750.92 include] 'an intent to do an act or bring about certain consequences which would in law amount to a crime; and . . . an act in furtherance of that intent which, as it is most commonly put, goes beyond mere preparation." *Thousand*, 465 Mich at 164, quoting *People v Jones*, 443 Mich 88, 100 (1993). Attempt is a specific intent crime. *People v Langworthy*, 416 Mich 630, 644–645 (1982). When factually appropriate, the trial court should instruct the jury that it may convict the defendant of attempt even if the evidence proves that the crime was completed. CJI2d 9.1.

MCL 750.92 sets forth the punishment for a conviction of attempt:²

"2. If the offense so attempted to be committed is punishable by imprisonment in the state prison for life, or for 5 years or more, the person convicted of such attempt shall be guilty of a felony, punishable by imprisonment in the state prison not more than 5 years or in the county jail not more than 1 year;

3. If the offense so attempted to be committed is punishable by imprisonment in the state prison for a term less than 5 years, or imprisonment in the county jail or by fine, the offender convicted of such attempt shall be guilty of a misdemeanor, punishable by imprisonment in the state prison or reformatory not more than 2 years or in any county jail not more than 1 year or by a fine not to exceed 1,000 dollars; but in no case shall the imprisonment exceed 1/2 of the greatest

² MCL 750.92(1) applies to offenses punishable by death and is not included below.

punishment which might have been inflicted if the offense so attempted had been committed."

1.4 Probation and Delayed Sentencing

MCL 771.1(1) permits the court to place the defendant on probation:

"In all prosecutions for felonies, misdemeanors, or ordinance violations other than murder, treason, criminal sexual conduct in the first or third degree, armed robbery, or major controlled substance offenses, if the defendant has been found guilty upon verdict or plea^[3] and the court determines that the defendant is not likely again to engage in an offensive or criminal course of conduct and that the public good does not require that the defendant suffer the penalty imposed by law, the court may place the defendant on probation under the charge and supervision of a probation officer." MCL 771.1(1).

"[A] defendant may decline a sentence of probation and instead seek a sentence of incarceration." *People v Bensch*, 328 Mich App 1, 13 (2019).

If the defendant is placed on probation, "the court may delay sentencing the defendant for not more than 1 year to give the defendant an opportunity to prove to the court his or her eligibility for probation or other leniency compatible with the ends of justice and the defendant's rehabilitation[.]" MCL 771.1(2). "When sentencing is delayed, the court shall enter an order stating the reason for the delay upon the court's records. The delay in passing sentence does not deprive the court of jurisdiction to sentence the defendant at any time during the period of delay." *Id.*

"[T]he plain language of MCL 771.1(2) does not deprive a sentencing judge of jurisdiction if a defendant is not sentenced within one year after the imposition of a delayed sentence[.]" *People v Smith (Ryan),* 496 Mich 133, 143 (2014).⁴ Rather, "[a]fter the one-year statutory limitation elapses, sentencing may no longer be delayed for the purpose of permitting a defendant the opportunity to prove that he [or she] is worthy of leniency, and the judge is required to sentence [the] defendant as provided by law." *Id.* at 142.

³ MCR 6.302 sets forth the procedures for pleas of guilty and nolo contendere in cases cognizable in the circuit court or court of equivalent jurisdiction. See MCR 6.001(A). MCR 6.302(F) provides that "[t]he court may take the plea under advisement."

⁴Overruling *People v McLott,* 70 Mich App 524 (1976), *People v Turner,* 92 Mich App 485 (1979), *People v Dubis,* 158 Mich App 504 (1987), *and People v Boynton,* 185 Mich App 669 (1990), "to the extent they hold that a court loses jurisdiction to sentence a defendant as a remedy for a violation of MCL 771.1(2)." *Smith (Ryan),* 496 Mich at 142.

Committee Tip: Note that the MVC requires convictions to be abstracted as required by MCL 257.732. MCL 257.8a defines conviction to include "[a] final conviction, the payment of a fine, a plea of guilty or nolo contendere if accepted by the court, or a finding of guilt for a criminal law violation or a juvenile adjudication, probate court disposition, or juvenile disposition for a violation that if committed by an adult would be a crime, regardless of whether the penalty is rebated or suspended." (Emphasis added.) See also People v McCann, 314 Mich App 605, 613 (2016) (noting that "[a]Ithough a trial judge has discretion to delay sentencing[,] . . . the [MVC] regards [a guilty] plea ... as a conviction[]"). Abstracting requirements are not mentioned in MCL 771.1, and thus, do not appear to be implicated by that statute. Accordingly, a court should send timely abstracts to the Secretary of State for *convictions* under the MVC, regardless of whether probation or a delayed sentence has been ordered. See Section 1.41 for abstracting requirements in cases involving MVC violations.

For more information on sentencing, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2.* For more information on probation, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 3,* Chapter 2.

[&]quot;[A] trial court may not require the Secretary of State to amend driving records when a conviction is dismissed following [a] guilty plea and delayed sentencing[under MCL 771.1;]" although MCL 257.732(1)(b) of the MVC "requires a trial court to forward abstracts to the Secretary of State following the dismissal of charges, . . . it does not command the secretary to take specific action in response[,]" and MCL 257.732(22) prohibits a court from ordering the expunction of a Secretary of State record of a reportable offense that has been set aside or dismissed. *McCann*, 314 Mich App at 614 (citations omitted).

1.5 Taking Pleas Under Advisement

A. Felony Cases Generally

In a felony case, the court may take a plea under advisement and must make a verbatim record of the proceedings. MCR 6.001(A); MCR 6.302(F). For more information on pleas, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 6.

B. Misdemeanor Cases Generally

MCR 6.302(F) is not applicable to misdemeanor pleas. See MCR 6.001(B). In addition, no statute or court rule language has been identified that authorizes district courts to take a misdemeanor guilty plea under advisement. As a result, the State Court administrative Office (SCAO) continues to "recommend[] that courts discontinue the practice of taking [these] matters under advisement." See the *Michigan Supreme Court Report*, October 1998. For more information on misdemeanor pleas, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook*, *Vol. 1*, Chapter 6.

C. Traffic Exception

MCL 257.732(21) states:

"Notwithstanding any other law of this state, a court shall not take under advisement an offense committed by an individual while operating a motor vehicle for which this act requires a conviction or civil infraction determination to be reported to the secretary of state. A conviction or civil infraction determination that is the subject of this subsection must not be masked, delayed, diverted, suspended, or suppressed by a court. Upon a conviction or civil infraction determination, the conviction or civil infraction determination must immediately be reported to the secretary of state in accordance with this section."

1.6 Restitution

Restitution is mandatory for an offender convicted of a felony, misdemeanor, or ordinance violation. MCL 769.1a(2); MCL 780.766(2); MCL 780.794(2); MCL 780.826(2). Restitution is also mandatory "[f]or an offense that is resolved by assignment of the defendant to youthful trainee status, by a delayed sentence or deferred judgment of guilt, or in another way that is not an acquittal or unconditional dismissal[.]" MCL 780.766(2); MCL 780.826(2). See also MCL 780.794(2), which also requires the court to order restitution "[f]or an offense that is resolved informally

by means of a consent calendar diversion or by another informal method that does not result in a dispositional hearing[.]"

MCR 6.430 governs postjudgment motions to amend restitution. The prosecuting attorney, victim, or defendant may file a postjudgment motion to amend an order of restitution "based upon new or updated information related to the injury, damages, or loss for which the restitution was ordered." MCR 6.430(A).

See the Michigan Judicial Institute's *Crime Victim Rights Benchbook,* Chapter 8, for detailed information about restitution.

1.7 Investigative Stops

The federal and Michigan constitutions grant all persons the right to be secure against unreasonable searches and seizures. US Const, Am IV, and Const 1963, art 1, §11. For a detailed discussion of investigative stops, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 11.

Part A—Civil Infractions

1.8 Courts With Jurisdiction Over Civil Infraction Actions

"Except as otherwise provided in [the Revised Judicature Act,] [a] civil infraction action involving a traffic or parking violation is governed by [the MVC.]" MCL 600.113(2).

MCL 257.741(2) states that the district court and any municipal court "have jurisdiction over civil infraction actions[.]" See also MCL 600.8301(2) ("[t]he district court has jurisdiction over civil infraction actions").

"If the individual cited is a minor, that individual may appear in court or admit responsibility for a civil infraction without the necessity of appointment of a guardian or next friend." MCL 257.741(5). The district court or any municipal court has "jurisdiction over the minor and may proceed in the same manner and in all respects as if that individual were an adult." *Id*.⁵

⁵See Chapter 2 for a more detailed discussion of jurisdiction over civil infractions committed by minors.

1.9 Venue for Civil Infraction Actions

"The place specified in the citation for appearance must be the [district or municipal] court . . . that has territorial jurisdiction of the place where the civil infraction occurred." MCL 257.741(4).

"Venue in the district court is governed by ... MCL 600.8312." MCL 257.741(4). MCL 600.8312(6) provides that "[v]enue in civil infraction actions shall be determined as follows:

(a) In a district of the first class, venue shall be in the county where the civil infraction occurred.

(b) In a district of the second class, venue shall be in the district where the civil infraction occurred.

(c) In a district of the third class, venue shall be in the political subdivision where the civil infraction occurred, except that when the violation is alleged to have taken place within a political subdivision where the court is not required to sit, the action may be heard or an admission entered in any political subdivision within the district where the court is required to sit."

1.10 Time Guidelines for Processing Civil Infraction Actions

"The time specified in a citation for appearance must be within a reasonable time after the citation is issued pursuant to [MCL 257.682] or [MCL 257.742]." MCL 257.741(3). Administrative Order No. 2013-12 provides that "90% of all civil infraction cases, including traffic, nontraffic, and parking cases, should be adjudicated within 35 days from the date of filing and 98% within 84 days." However, "90% of all citations [for juvenile traffic and ordinance violations] should have adjudication and disposition completed within 63 days from the date of first appearance and 98% within 126 days."⁶ Administrative Order No. 2013-12.

1.11 Authority to Conduct Civil Infraction Actions

A. District Court or Municipal Judge

"A formal hearing shall be conducted only by a judge of a court having jurisdiction over civil infraction actions under [MCL

⁶See Chapter 2, for a more detailed discussion of jurisdiction over civil infractions committed by minors.

257.741(2)]."⁷ MCL 257.747(1). A judge of a court having jurisdiction over the action may also conduct informal hearings.⁸ See MCL 257.746(1).

A district court judge must preside over an appeal from an informal hearing conducted by a district court magistrate.⁹ MCL 257.746(5)(b). "The appeal from a judge of the district court shall be heard by a different judge of the district." MCL 257.746(5)(a).

B. District Court Magistrate

"A district court magistrate may exercise the authority conferred by [MCL 600.8512] only to the extent expressly authorized by the chief judge, presiding judge, or only judge of the district court district." MCL 600.8512(4). See also MCR 4.401(B) ("[n]otwithstanding statutory provisions to the contrary, district court magistrates exercise only those duties expressly authorized by the chief judge of the district or division").

"A district court magistrate shall not conduct an informal hearing in a civil infraction action involving a traffic or parking violation governed by [MCL 257.1 to MCL 257.923] until he or she has successfully completed a special training course in traffic law adjudication and sanctions." MCL 600.8512(3).

"An informal hearing shall be conducted by a district court magistrate when authorized by the judge or judges of the district court district or by a judge of a court listed in [MCL 257.741(2)]."¹⁰ MCL 257.746(1). See also MCL 600.8512(1). "A district court magistrate may administer oaths, examine witnesses, and make findings of fact and conclusions of law at an informal hearing." MCL 257.746(1). See also MCL 600.8512(1) (permitting a district court magistrate to "hear, preside over, and conduct" admissions, admissions with explanation, motions to set aside default or withdraw admissions, and informal hearings in civil infractions under the Michigan vehicle code, MCL 257.746).

"Only to the extent expressly authorized by the chief judge, presiding judge, or only judge of the district court district, a district court magistrate may ... [a]ccept an admission of responsibility, decide a motion to set aside a default or withdraw an admission, and order civil sanctions for a civil infraction and order an

⁷ See Section 1.17 for more information on formal hearing requirements.

⁸ See Section 1.16 for more information on informal hearing requirements.

⁹ See Section 1.16(F) for more information on appeals from informal hearings.

¹⁰ See Section 1.16 for more information on informal hearing requirements.

appropriate civil sanction permitted by the statute or ordinance defining the act or omission." MCL 600.8512a(a).

"The judges of the district court shall exercise superintending control over all magistrates within their districts." MCL 600.8541(1). "A district judge may not extend the jurisdiction of a district court magistrate beyond the jurisdiction expressly provided by law." *Id.* See also MCR 4.401(C) ("[a]n action taken by a district court magistrate may be superseded, without formal appeal, by order of a district judge in the district in which the magistrate serves").

1.12 Technical Requirements of Traffic Citations and Parking Violation Notices

"Each citation shall be numbered consecutively, be in a form as determined by the secretary of state, the attorney general, the state court administrator, and the director of the department of state police[.]" MCL 257.727c(1). A citation must consist of the following:

"(a) The original which shall be a complaint or notice to appear by the officer and shall be filed with the court in which the appearance is to be made.

(b) The first copy which shall be retained by the local traffic enforcement agency.

(c) The second copy which shall be delivered to the alleged violator if the violation is a misdemeanor.

(d) The third copy which shall be delivered to the alleged violator if the violation is a civil infraction."¹¹ MCL 257.727c(1).

"The citation serves as a summons to command (a) the initial appearance of the defendant; and (b) a response from the defendant as to his or her responsibility for the alleged violation." MCR 4.101(A)(3).

A. Contents of the Citation¹²

A citation issued pursuant to MCL 257.742 must contain the following:

¹¹ With proper prior approval, "the citation may be appropriately modified as to content or number of copies to accommodate law enforcement and local court procedures and practices. Use of this citation for other than moving violations is optional." MCL 257.727c(2).

¹² With proper prior approval, "the citation may be appropriately modified as to content or number of copies to accommodate law enforcement and local court procedures and practices. Use of this citation for other than moving violations is optional." MCL 257.727c(2).

- The name of the state or political subdivision acting as plaintiff;
- The name and address of the person to whom the citation is issued;
- The civil infraction alleged;¹³
- The place where the person must appear in court;
- The telephone number of the court;
- The time at or by which the appearance must be made. MCL 257.743(1). "The time specified in a citation for appearance must be within a reasonable time after the citation is issued pursuant to [MCL 257.682] or [MCL 257.742]." MCL 257.741(3).

In addition, the issuing officer must sign the citation. MCR 4.101(A)(1). If the citation is prepared electronically and filed with the court as data, including the issuing officer's name satisfies the signature requirement. *Id.* A complaint must be considered signed under oath in certain circumstances. See MCL 257.727c. See Section 1.12(B) for more information.

A citation must also inform the defendant that he or she, at or by the time specified for appearance, may:

- Admit responsibility for the civil infraction in person, by representation, or by mail.
- Admit responsibility for the civil infraction with explanation in person, by representation, or by mail.
- Deny responsibility for the civil infraction by either:
 - Appearing in person for an informal hearing without the opportunity of being represented by an attorney; or
 - Appearing in court for a formal hearing before a judge, with the opportunity of being represented by an attorney. MCL 257.743(2).¹⁴

"The citation shall inform the defendant that if the person desires to admit responsibility 'with explanation' other than by mail or to have an informal hearing or a formal hearing, the person must

¹³ "A violation alleged on a citation may not be amended except by the prosecuting official or a police officer for the plaintiff." MCR 4.101(A)(2).

¹⁴ See Section 1.15 for more information on admitting or denying responsibility.

apply to the court in person, by mail, or by telephone, within the time specified for appearance and obtain a scheduled date and time to appear for a hearing." MCL 257.743(3). "A hearing date may be specified on the citation." *Id*.

"The citation shall contain a notice in boldface type that the failure of a person to appear within the time specified in the citation or at the time scheduled for a hearing or appearance will result in entry of a default judgment against the person and in the immediate suspension of the person's operator's or chauffeur's license." MCL 257.743(4). "Timely application to the court for a hearing or return of the citation with an admission of responsibility and with full payment of applicable civil fines and costs constitute a timely appearance." *Id.*

B. Signed Under Oath¹⁵

A complaint signed by a police officer must be treated as made under oath if the violation alleged in the complaint:

- is a civil infraction or a misdemeanor or ordinance violation for which the maximum permissible penalty does not exceed 93 days in jail or a fine, or both;
- occurred or was committed in the signing officer's presence, or under circumstances permitting the officer's issuance of a citation under MCL 257.625a (warrantless arrests for operating while intoxicated offenses) or MCL 257.728(8) (citations issued after investigation of a traffic accident); and
- if the complaint contains the following statement immediately above the date and signature of the officer:

"'I declare under the penalties of perjury that the statements above are true to the best of my information, knowledge, and belief."" MCL 257.727c(3).

C. Parking Violation Notice

"A parking violation notice may be issued by a police officer, including a limited duty officer, or other personnel duly authorized by the city, village, township, college, or university to issue that parking violation notice under its ordinance." MCL 257.742(7). "If a

¹⁵ The citation must include the issuing officer's signature. See Section 1.12(A) for more information on citation content requirements.

parking violation notice other than a citation is attached to a motor vehicle, and if an admission of responsibility is not made and the civil fine and costs, if any, prescribed by ordinance for the violation are not paid at the parking violations bureau, a citation may be filed with the court described in [MCL 257.741(4)] and a copy of the citation may be served by first-class mail upon the registered owner of the vehicle at the owner's last known address." MCL 257.742(7). "The citation filed with the court pursuant to [MCL 257.742(7)] need not comply in all particulars with [MCL 257.727c and MCL 257.743] but must consist of a sworn complaint containing the allegations stated in the parking violation notice and must fairly inform the defendant how to respond to the citation." MCL 257.742(7).

1.13 Initiating a Civil Infraction Action

"A civil infraction action is commenced upon the issuance and service of a citation as provided in [MCL 257.742]." MCL 257.741(1). See also MCR 4.101(A)(1) ("a civil infraction action may be initiated by a law enforcement officer serving a written citation on the alleged violator and filing the citation in the district court"). The citation serves as the complaint, may be prepared electronically or on paper, and must be signed by the officer in accordance with MCR 1.109(E)(4). MCR 4.101(A)(1). The name of the officer associated with issuance of the citation satisfies the signature requirement if the citation is prepared electronically. *Id*.

"The plaintiff in a civil infraction action must be either this state if the alleged civil infraction is a violation of [the MVC], or a political subdivision if the alleged civil infraction action is a violation of a local ordinance of that subdivision that substantially corresponds to a provision of [the MVC]." MCL 257.741(1).

A. Authority to Issue a Citation¹⁶

"The form of a citation issued under [MCL 257.742(1), MCL 257.742(2), MCL 257.742(3), or MCL 257.742(9)] shall be as prescribed in [MCL 257.727c and MCL 257.743]."¹⁷ MCL 257.742(4).

¹⁶ This subsection does not discuss issuing a citation under MCL 257.742(2), because it is outside the scope of this benchbook.

¹⁷ See Section 1.12(B) for a discussion of the requirements under MCL 257.727c, and Section 1.12(A) on requirements under MCL 257.743.

1. Officer Witnesses Violation

"A police officer who witnesses a civil infraction may stop and temporarily detain the offender for the purpose of issuing a written citation." *People v Chapo*, 283 Mich App 360, 366 (2009), citing MCL 257.742(1), which states:

"If a police officer witnesses an individual violating [the MVC] or a local ordinance substantially corresponding to [the MVC] and that violation is a civil infraction, that police officer may stop the individual, detain the individual temporarily for purposes of making a record of vehicle check, and prepare and subscribe, as soon as possible and as completely as possible, an original and 3 copies of a written citation, which must be a notice to appear in court for 1 or more civil infractions."

"If a police officer of a village, city, township, or county, or a police officer who is an authorized agent of a county road commission, witnesses an individual violating [the MVC] or a local ordinance substantially corresponding to [the MVC] within that village, city, township, or county and that violation is a civil infraction, that police officer may pursue, stop, and detain the individual outside the village, city, township, or county where the violation occurred for the purpose of exercising the authority and performing the duties prescribed in [MCL 257.742] and [MCL 257.749], as applicable." MCL 257.742(1).

2. Officer's Personal Investigation of Accident

"A police officer may issue a citation to an individual who is a driver of a motor vehicle involved in an accident when, based upon personal investigation, the officer has reasonable cause to believe that the individual is responsible for a civil infraction in connection with the accident." MCL 257.742(3). See also *People v Estabrooks*, 175 Mich App 532, 537 (1989) ("[a] police officer may stop and detain a driver involved in a motor vehicle accident for the purpose of issuing a citation for a civil infraction when... based upon the officer's personal investigation, the officer has reasonable cause to believe that the driver is responsible for a civil infraction").

3. Officer's Personal Investigation of Violation Based on Witness Complaint

"A police officer may not stop a driver for a civil infraction solely on the basis of a witness' complaint." *Estabrooks*, 175 Mich App at 537, citing MCL 257.743(3). Rather, the statute requires the officer to investigate and have a reasonable belief that a civil infraction occurred. See *Estabrooks*, 175 Mich App at 538. MCL 257.742(3) states:

"A police officer may issue a citation to an individual who is a driver of a motor vehicle when, based upon personal investigation by the police officer of a complaint by someone who witnessed the individual violating [the MVC] or a local ordinance substantially corresponding to [the MVC] and that violation is a civil infraction, the officer has reasonable cause to believe that the individual is responsible for a civil infraction and if the prosecuting attorney or attorney for the political subdivision approves in writing the issuance of the citation." MCL 257.742(3).

B. Serving and Filing a Citation or Parking Notice Violation

1. General Service and Filing Requirements

"Except as otherwise provided by court rule or statute, a civil infraction action may be initiated by a law enforcement officer serving a written citation on the alleged violator, and filing the citation in the district court." MCR 4.101(A)(1). See also MCL 257.742(5), which states that "[t]he officer shall inform the individual of the alleged civil infraction or infractions and shall deliver the third copy of the citation to the offender."¹⁸ The citation serves as the complaint, may be prepared electronically or on paper, and must be signed by the officer in accordance with MCR 1.109(E)(4). MCR 4.101(A)(1). The name of the officer associated with issuance of the citation satisfies the signature requirement if the citation is prepared electronically. *Id*.

"If the infraction is a parking violation, the action may be initiated by an authorized person placing a citation securely on the vehicle or mailing a citation to the registered owner of the vehicle." MCR 4.101(A)(1)(a). See also MCL 257.742(6).

¹⁸A written citation consists of "an original and 3 copies." MCL 257.742(1).

"If a parking violation notice^[19] other than a citation is attached to a motor vehicle, and if an admission of responsibility is not made and the civil fine and costs, if any, prescribed by ordinance for the violation are not paid at the parking violations bureau, a citation may be filed with the court described in [MCL 257.741(4)] and a copy of the citation may be served by first-class mail upon the registered owner of the vehicle at the owner's last known address." MCL 257.742(7). "The citation filed with the court pursuant to [MCL 257.742(7)] need not comply in all particulars with [MCL 257.727c and MCL 257.743] but must consist of a sworn complaint containing the allegations stated in the parking violation notice and must fairly inform the defendant how to respond to the citation." MCL 257.742(7).

"If the infraction is a municipal civil infraction, the action may be initiated by an authorized local official serving a written citation on the alleged violator." MCR 4.101(A)(1)(b).

2. Depositing Citation With Court

"At or before the completion of his or her tour of duty a police officer to whom a citation book has been issued and who has recorded the occurrence of a vehicle law violation upon a citation shall deliver to his or her police chief or to a person duly authorized by the police chief to receive citations all copies of such citation duly signed." MCL 257.728a(1). "The police chief or a person duly authorized by the police chief shall deposit the original of the citation with the court having jurisdiction over the offense not later than 3 days after the date of the citation, excluding Saturdays, Sundays, and legal holidays." *Id*.²⁰

"The citation shall be considered to have been deposited with the court as required under [MCL 257.728a(1)] if the original of the citation is mailed not later than 2 days after the date of the citation as specified under [MCL 257.728a(2)]." MCL 257.728a(2). "Mailing shall be accomplished by enclosing the original of the citation in a sealed envelope with first class postage fully prepaid, addressed to the court, and depositing the envelope and contents in the United States government mail." *Id*.

¹⁹ Parking violation notice means "a notice, other than a citation, directing a person to appear at a parking violations bureau in the city, village, or township in which, or of the college or university for which, the notice is issued and to pay the fine and costs, if any, prescribed by ordinance for the parking or standing of a motor vehicle in violation of the ordinance." MCL 257.742(9)(a).

²⁰MCL 257.728a does not address the consequences of a violation of the 3-day rule.

C. Sworn Complaint Requirements

"If an officer issues a citation under [MCL 257.742] for a civil infraction or if a citation is issued under [MCL 257.742] for a parking or standing violation, the court may accept an admission with explanation or an admission or denial of responsibility upon the citation without the necessity of a sworn complaint."²¹ MCL 257.744. "If the person denies responsibility for the civil infraction, further proceedings shall not be had until a sworn complaint is filed with the court." *Id.* "[A] sworn traffic citation filed with the district court constitutes a sworn complaint for purposes of MCL 257.744[.]" *People v Ferency*, 133 Mich App 526, 530 (1984).

"A warrant for arrest under [MCL 257.321a] for failure to appear on the civil infraction citation shall not issue until a sworn complaint relative to the civil infraction is filed with the court." MCL 257.744.

1.14 Special Requirements for Nonresidents

"When a nonresident is stopped under [MCL 257.742] for a civil infraction, the police officer making the stop shall issue to that person a citation as provided in [MCL 257.727c and MCL 257.742]." MCL 257.749(1). "The officer shall release the nonresident upon his or her personal recognizance." MCL 257.749(2).

"If a magistrate is available for an immediate appearance, upon demand of the person stopped, the officer immediately shall take the nonresident driver before the magistrate to answer to the civil infraction alleged. If the nonresident defendant requests a formal hearing, the hearing shall be scheduled as provided in [MCL 257.747]." MCL 257.749(3).

"If the person who is released upon his or her personal recognizance as provided in [MCL 257.749(2)] fails to appear as required in the citation or for a scheduled formal hearing, the court having jurisdiction and venue over the civil infraction shall enter a default judgment against the person." MCL 257.749(4).

1.15 Defendant's Options When a Citation Is Issued

"A person to whom a citation is issued under [MCL 257.742] shall appear by or at the time specified in the citation and may respond to the allegations in the citation as provided in [MCL 257.745]." MCL 257.745(1).²² A defendant may appear by mail or in person. See the

²¹ See Section 1.15 for more information on admitting or denying responsibility.

Michigan Judicial Institute's appearance by mail checklist and appearance in person checklist.

A. Admit Responsibility Without Explanation

"If the person wishes to admit responsibility for the civil infraction, the person may do so by appearing in person, by representation, or by mail." MCL 257.745(2). "A defendant may not appear by making a telephone call to the court, but a defendant may telephone the court to obtain a date to appear." MCR 4.101(B)(2). "If appearance is made by representation or mail, the court may accept the admission with the same effect as though the person personally appeared in court." MCL 257.745(2). "Upon acceptance of the admission, the court may order any of the sanctions permitted under [MCL 257.907]." MCL 257.745(2).

"Except as provided in [MCR 4.101(E)(4)], an admission without explanation may be offered to and accepted by

(a) a district judge;

(b) a district court magistrate as authorized by the chief judge, the presiding judge, or the only judge of the district; or

(c) other district court personnel, as authorized by a judge of the district." MCR 4.101(E)(1).

"There is no appeal of right from an admission of responsibility." MCR 4.101(H)(3). "However, within 14 days after the admission, a defendant may file with the district court a written request to withdraw the admission, and must post a bond as provided in [MCR 4.101(H)(1)(a)]." MCR 4.101(H)(3). "If the court grants the request, the case will be scheduled for either a formal hearing or an informal hearing, as ordered by the court." *Id.* "If the court denies the request, the bond may be applied to the fine and costs." *Id.*

B. Admit Responsibility With Explanation

"Depending on the nature of the violation and on the procedure appropriate to the violation, a defendant may appear in person, by videoconferencing technology, by representation, or by mail." MCR 4.101(B)(1). See also MCL 257.745(3). "A defendant may not appear by making a telephone call to the court, but a defendant may

 $^{^{22}}$ "If the violation is a trailway municipal civil infraction, and there has been damage to property or a vehicle has been impounded, the defendant's response must be made at a formal hearing." MCR 4.101(E)(4). See also MCL 600.8717(4).

telephone the court to obtain a date to appear." MCR 4.101(B)(2). "A defendant who obtains a hearing date other than the date specified in the citation, but who does not appear to explain or contest responsibility, is in default, and the procedures established by [MCR 4.101(B)(4)-(6)] apply."²³ MCR 4.101(F)(3). "For any hearing held under [MCR 4.101] and subject to MCR 2.407(B)(5), the use of videoconferencing technology to conduct remote proceedings is presumed." MCR 4.101(F)(4).

"If a person admits responsibility for a civil infraction 'with explanation' under [MCL 257.745(3)], the court shall accept the admission as though the person has admitted responsibility under [MCL 257.745(2)] and may consider the person's explanation by way of mitigating any sanction which the court may order under [MCL 257.907]." MCL 257.745(4). "If appearance is made by representation or mail, the court may accept the admission with the same effect as though the person personally appeared in court, but the court may require the person to provide a further explanation or to appear in court." *Id.*

"Except as provided in [MCR 4.101(E)(4)], an admission with explanation may be written or offered orally to a judge or district court magistrate, as authorized by the district judge." MCR 4.101(E)(2).

"There is no appeal of right from an admission of responsibility." MCR 4.101(H)(3). "However, within 14 days after the admission, a defendant may file with the district court a written request to withdraw the admission, and must post a bond as provided in [MCR 4.101(H)(1)(a)]." MCR 4.101(H)(3). "If the court grants the request, the case will be scheduled for either a formal hearing or an informal hearing, as ordered by the court." *Id.* "If the court denies the request, the bond may be applied to the fine and costs." *Id.*

C. Deny Responsibility

"If the person wishes to deny responsibility for a civil infraction, the person shall do so by appearing for an informal or formal hearing." MCL 257.745(5). "Depending on the nature of the violation and on the procedure appropriate to the violation, a defendant may appear in person, by videoconferencing technology, by representation, or by mail." MCR 4.101(B)(1). See also MCL 257.745(5). "A defendant may not appear by making a telephone call to the court, but a defendant may telephone the court to obtain a date to appear." MCR 4.101(B)(2). "The court shall schedule an informal hearing,"

²³ See Section 1.18 for information on default procedures.

unless the person expressly requests a formal hearing." MCL 257.745(5). "If the hearing date is specified on the citation, the person shall appear on that date for an informal hearing unless the person contacts the court at least 10 days before that date in person, by representation, by mail, or by telephone to request a formal hearing." *Id*.

"A defendant who obtains a hearing date other than the date specified in the citation, but who does not appear to explain or contest responsibility, is in default, and the procedures established by [MCR 4.101(B)(4)-(6)] apply."²⁴ MCR 4.101(F)(3). For any hearing held under [MCR 4.101] and subject to MCR 2.407(B)(5), the use of videoconferencing technology to conduct remote proceedings is presumed." MCR 4.101(F)(4).

"If a hearing is scheduled by telephone, the court shall mail the defendant a confirming notice of that hearing by regular mail to the address appearing on the citation or to an address which may be furnished by the defendant." *Id.* "An informal hearing shall be conducted pursuant to [MCL 257.746²⁵] and a formal hearing shall be conducted pursuant to [MCL 257.747²⁶]." MCL 257.745(5).

"Except as provided in [MCR 4.101(E)(4)], a denial of responsibility must be made by the defendant appearing at a time set either by the citation or as the result of a communication with the court." MCR 4.101(E)(3).

1.16 Informal Hearings

"An informal hearing will be held unless

(a) a party expressly requests a formal hearing, or

(b) the violation is a trailway municipal civil infraction which requires a formal hearing pursuant to MCL 600.8717(4)." MCR 4.101(F)(1). See also MCL 257.745(5) (when the defendant wishes to deny responsibility, "[t]he court shall schedule an informal hearing, unless the person expressly requests a formal hearing").

See also the Michigan Judicial Institute's informal hearing checklist and informal hearing worksheet.

²⁴ See Section 1.18 for information on default procedures.

²⁵ See Section 1.16.

²⁶ See Section 1.17.

A. Conducting the Hearing

"An informal hearing shall be conducted by a district court magistrate when authorized by the judge or judges of the district court district or by a judge of a court listed in [MCL 257.741(2)]." MCL 257.746(1).

"A district court magistrate may administer oaths, examine witnesses, and make findings of fact and conclusions of law at an informal hearing." MCL 257.746(1). "The judge or district court magistrate shall conduct the informal hearing in an informal manner so as to do substantial justice according to the rules of substantive law but shall not be bound by the statutory provisions or rules of practice, procedure, pleading, or evidence, except provisions relating to privileged communications." *Id.*

"There shall not be a jury at an informal hearing." MCL 257.746(1).

"For any hearing held under [MCR 4.101] and subject to MCR 2.407(B)(5),^[27] the use of videoconferencing technology to conduct remote proceedings is presumed." MCR 4.101(F)(4).

"Each municipal judge and each clerk of a court of record shall keep a full record of every case in which an individual is charged with or cited for a violation of [the MVC] or a local ordinance substantially corresponding to [the MVC] regulating the operation of vehicles on highways[.]" MCL 257.732(1). However, "[a] verbatim record of an informal hearing shall not be required." MCL 257.746(1).

B. No Representation at Hearing

"At an informal hearing the person cited may not be represented by an attorney nor may the plaintiff be represented by the prosecuting attorney or attorney for a political subdivision." MCL 257.746(2).

C. Discovery

Discovery is not permitted in a civil infraction action. MCR 2.301(A)(3).

D. Notice and Subpoenas

"Notice of a scheduled informal hearing shall be given to the citing police agency, which agency may subpoena witnesses for the plaintiff." MCL 257.746(3).²⁸ "The defendant may also subpoena

²⁷MCR 2.407 addresses videoconferencing.

witnesses." *Id.* See also MCR 4.101(C)(1), requiring the court to "notify the police officer who issued the citation to appear at the informal hearing." "The attendance of the officer at the hearing may not be waived." MCR 4.101(C)(2). "Except when the court is notified before the commencement of a hearing of an emergency preventing an on-duty officer from appearing, failure of the police officer to appear as required by [MCR 4.101] shall result in a dismissal of the case without prejudice." MCR 4.101(C).

"The provisions of MCR 2.501(C) regarding the length of notice of trial assignment do not apply in civil infraction actions." MCR 4.101(F)(2).

E. Determining Responsibility

The court must find the person responsible for a civil infraction by a preponderance of the evidence. MCL 257.6b(c).

"Preponderance of the evidence means such evidence as, when weighted with that opposed to it, has more convincing force and the greater probability of truth." *People v Cross*, 281 Mich App 737, 740 (2008) (quotation marks omitted).

"If the judge or district court magistrate determines by a preponderance of the evidence that the person cited is responsible for a civil infraction, the judge or magistrate shall enter an order against the person as provided in [MCL 257.907]."²⁹ MCL 257.746(4). "Otherwise, a judgment shall be entered for the defendant, but the defendant shall not be entitled to costs of the action." *Id.* The preponderance of the evidence test, as it relates to civil infraction actions, does not violate due process. *People v Ferency*, 133 Mich App 526, 536 (1984).

See also SCAO Form CIA 02, Judgment.

F. Appeals

"The plaintiff and defendant shall be entitled to appeal an adverse judgment entered at an informal hearing." MCL 257.746(5). "An appeal following an informal hearing is a matter of right, and must be asserted in writing, within 7 days after the decision, on a form to be provided by the court."³⁰ MCR 4.101(H)(2). "The appeal will

²⁸ "Witness fees for a witness on behalf of the plaintiff are payable by the district control unit of the district court for the place where the hearing occurs, by the city or village when the hearing involves an ordinance violation in a district where the district court is not functioning, or by the county when the hearing involves a violation of [the MVC] in a district where the district court is not functioning." MCL 257.746(3).

²⁹ See Section 1.20 for more information on ordering a fine and costs under MCL 257.907.

result in a de novo formal hearing." *Id.* "[A]n appeal shall be de novo in the form of a scheduled formal hearing as follows:

(a) The appeal from a judge of the district court shall be heard by a different judge of the district.

(b) The appeal from a district court magistrate shall be heard by a judge of the district." MCL 257.746(5).

See also MCR 4.401(D) ("[a]ppeals of right may be taken from a decision of the district court magistrate to the district court in the district in which the magistrate serves . . . [and] [t]he action is heard de novo by the district court"); MCL 600.8515. Additionally, "[a]n action taken by a district court magistrate may be superseded, without formal appeal, by order of a district judge in the district in which the magistrate serves." MCR 4.401(C).

"An appeal from a municipal judge shall be a trial de novo in the circuit court." MCL 257.746(5).

Defendant's appeal. "A defendant who appeals must post a bond as provided in [MCR 4.101(H)(1)(a)]." MCR 4.101(H)(2)(a). "If a defendant who has posted a bond defaults by failing to appear at the formal hearing, or if the appeal is dismissed or the judgment is affirmed, the bond may be applied to the fine and costs." *Id*.

Plaintiff's appeal. "A plaintiff's appeal must be asserted by the prosecuting authority of the political unit that is responsible for providing the plaintiff's attorney for the formal hearing." MCR 4.101(H)(2)(b). "A bond is not required." *Id*.

Appeal from an admission of responsibility. "There is no appeal of right from an admission of responsibility. However, within 14 days after the admission, a defendant may file with the district court a written request to withdraw the admission, and must post a bond as provided in [MCR 4.101(H)(1)(a)]. If the court grants the request, the case will be scheduled for either a formal hearing or an informal hearing, as ordered by the court. If the court denies the request, the bond may be applied to the fine and costs." MCR 4.101(H)(3).

³⁰ See SCAO Form CIA 05, *Claim of Appeal of Right/Request to Withdraw Admission and Order, Civil Infraction.* Note that MCR 4.401(D), which generally addresses appeals of district court magistrate decisions, requires a claim of appeal to be written "in substantially the same form provided by MCR 7.104[.]"

1.17 Formal Hearings

District court magistrates do not have the authority to conduct formal hearings. "A formal hearing shall be conducted only by a judge of a court having jurisdiction over civil infraction actions under [MCL 257.741(2)]." MCL 257.747(1).

A formal hearing must be held upon the express request of a party, or if the violation involves a railway municipal civil infraction which requires a formal hearing pursuant to MCL 600.8717(4). MCR 4.101(F)(1). See also MCL 257.745(5) (when the defendant wishes to deny responsibility, "[t]he court shall schedule an informal hearing, unless the person expressly requests a formal hearing").

A. Conducting the Hearing

"A formal hearing shall be conducted only by a judge of a court having jurisdiction over civil infraction actions under [MCL 257.741(2)]." MCL 257.747(1).

"There shall not be a jury trial in a formal hearing." MCL 257.747(4).

"For any hearing held under [MCR 4.101] and subject to MCR 2.407(B)(5),^[31]the use of videoconferencing technology to conduct remote proceedings is presumed." MCR 4.101(F)(4).

"Each municipal judge and each clerk of a court of record shall keep a full record of every case in which an individual is charged with or cited for a violation of [the MVC] or a local ordinance substantially corresponding to [the MVC] regulating the operation of vehicles on highways[.]" MCL 257.732(1).

B. Representation Permitted

"In a formal hearing the person cited may be represented by an attorney, but is not entitled to appointed counsel at public expense." MCL 257.747(2).

C. Discovery

Discovery is not permitted in a civil infraction action. MCR 2.301(A)(3).

³¹MCR 2.407 addresses videoconferencing.

D. Notice and Subpoenas

"Notice of a formal hearing shall be given to the prosecuting attorney or attorney for the political subdivision who represents the plaintiff." MCL 257.747(3). "That attorney shall appear in court for a formal hearing and that attorney shall be responsible for the issuance of a subpoena to each witness for the plaintiff." *Id*.³² "The defendant may also subpoena witnesses." *Id*.

"The provisions of MCR 2.501(C) regarding the length of notice of trial assignment do not apply in civil infraction actions." MCR 4.101(F)(2).

E. Determining Responsibility

The court must find the person responsible for a civil infraction by a preponderance of the evidence. MCL 257.6b(c).

"Preponderance of the evidence means such evidence as, when weighted with that opposed to it, has more convincing force and the greater probability of truth." *People v Cross*, 281 Mich App 737, 740 (2008) (quotation marks omitted).

"If the judge determines by a preponderance of the evidence that the person cited is responsible for a civil infraction, the judge shall enter an order against the person as provided in [MCL 257.907]."³³ MCL 257.747(5). "Otherwise, a judgment shall be entered for the defendant, but the defendant shall not be entitled to costs of the action." *Id.* The "preponderance of the evidence" test, as it relates to civil infraction actions, does not violate due process. *People v Ferency*, 133 Mich App 526, 536 (1984).

See also SCAO Form CIA 02, Judgment.

F. Appeals

"An appeal following a formal hearing is a matter of right." MCR 4.101(H)(1).³⁴

³² "Witness fees for a witness on behalf of the plaintiff are payable by the district control unit of the district court for the place where the hearing occurs, by the city or village when the hearing involves an ordinance violation in a district where the district court is not functioning, or by the county when the hearing involves a violation of this act in a district where the district court is not functioning." MCL 257.747(3).

³³ See Section 1.20 for more information on ordering a fine and costs under MCL 257.907.

³⁴"Except as otherwise provided in [MCR 4.101], the appeal [following a formal hearing] is governed by subchapter 7.100 [of the Michigan Court Rules]." MCR 4.101(H)(1). See the Michigan Judicial Institute's *Appeals & Opinions Benchbook*, Chapter 2, for information on circuit court appeals under subchapter 7.100.

Defendant's appeal. "A defendant who appeals must post with the district court, at the time the appeal is taken, a bond equal to the fine and costs imposed." MCR 4.101(H)(1)(a). "A defendant who has paid the fine and costs is not required to post a bond." *Id*.

"If a defendant who has posted a bond fails to comply with the requirements of MCR 7.104(D), the appeal may be considered abandoned, and the district court may dismiss the appeal on 14 days' notice to the parties pursuant to MCR 7.113." MCR 4.101(H)(1)(b). "The court clerk must promptly notify the circuit court of a dismissal and the circuit court shall dismiss the claim of appeal." *Id.* "If the appeal is dismissed or the judgment is affirmed, the district court may apply the bond to the fine and costs." *Id.*

Plaintiff's appeal. "A plaintiff's appeal must be asserted by the prosecuting authority of the political unit that provided the plaintiff's attorney for the formal hearing." MCR 4.101(H)(1)(c). "A bond is not required." *Id*.

Appeal from an admission of responsibility. "There is no appeal of right from an admission of responsibility. However, within 14 days after the admission, a defendant may file with the district court a written request to withdraw the admission, and must post a bond as provided in [MCR 4.101(H)(1)(a)]. If the court grants the request, the case will be scheduled for either a formal hearing or an informal hearing, as ordered by the court. If the court denies the request, the bond may be applied to the fine and costs." MCR 4.101(H)(3).

1.18 Entering Default Judgment for Failure to Answer a Citation or Appear for a Scheduled Hearing

"Depending on the nature of the violation and on the procedure appropriate to the violation, a defendant may appear in person, by videoconferencing technology, by representation, or by mail." MCR 4.101(B)(1)."A defendant may not appear by making a telephone call to the court, but a defendant may telephone the court to obtain a date to appear." MCR 4.101(B)(2).

"A clerk of the court may enter a default after certifying, on a form to be furnished by the court,^[35] that the defendant has not made a scheduled appearance, or has not answered a citation within the time allowed by statute." MCR 4.101(B)(3). See also MCL 257.6b(d) (*civil infraction determination* includes "a determination that a person is responsible for a civil infraction by ... [a] default judgment, for failing to appear as directed by citation or other notice, at a scheduled appearance under

³⁵ See SCAO Form CIA 07, *Default Judgment, Civil Infraction*.

[MCL 257.745(3)(b) or MCL 257.745(4)], at a scheduled informal hearing under [MCL 257.746], or at a scheduled formal hearing under [MCL 257.747]").

"If a defendant fails to appear or otherwise to respond to any matter pending relative to a civil infraction action, the court:

(a) must enter a default against the defendant;

(b) must make a determination of responsibility, if the complaint is sufficient;

(c) must impose a sanction by entering a default judgment;

(d) must send the defendant a notice of the entry of the default judgment and the sanctions imposed; and

(e) may retain the driver's license of a nonresident^[36] as permitted by statute, if the court has received that license pursuant to statute. The court need not retain the license past its expiration date." MCR 4.101(B)(4). See also MCL 257.748.

Additionally, "[i]f a defendant fails to appear or otherwise to respond to any matter pending relative to a *traffic* civil infraction, the court

(a) must notify the secretary of state of the entry of the default judgment, as required by MCL 257.732,^[37] and

(b) must initiate the procedures required by MCL 257.321a.^[38]" MCR 4.101(B)(5) (emphasis added).

1.19 Setting Aside a Default Judgment

"A defendant may move to set aside a default judgment within 14 days after the court sends notice of the judgment to the defendant." MCR 4.101(D)(1).

"The motion

(a) may be informal,

(b) may be either written or presented to the court in person,

³⁶ See Section 1.14 for more information on special provisions related to nonresidents.

³⁷ See Section 1.41 for information on abstracting requirements under MCL 257.732.

³⁸ See Section 1.45 for information on suspending an individual's license under MCL 257.321a.

(c) must explain the reason for the nonappearance of the defendant,

(d) must state that the defendant wants to offer a defense to or an explanation of the complaint, and

(e) must be accompanied by a cash bond equal to the fine and costs due at the time the motion is filed." MCR 4.101(D)(1).

"For good cause, the court may

(a) set aside the default and direct that a hearing on the complaint take place, or

(b) schedule a hearing on the motion to set aside the default judgment." MCR 4.101(D)(2).

"A defendant who does not file [a motion to set aside a default judgment] on time may use the procedure set forth in MCR 2.603(D)." MCR 4.101(D)(3). "A motion to set aside ... a default judgment, except when grounded on lack of jurisdiction over the defendant, shall be granted only if good cause is shown and a statement of facts showing a meritorious defense, verified in the manner prescribed by MCR 1.109(D)(3), is filed." MCR 2.603(D)(1). "Except as provided in MCR 2.612, if personal service was made on the party against whom the default was taken, the ... default judgment if one has been entered, may be set aside only if the motion is filed ... if a default judgment has been entered, within 21 days after the default judgment was entered." MCR 2.603(D)(2)(b).

1.20 Civil Fines, Costs, and Assessments for Civil Infractions³⁹

"Each district of the district court and each municipal court may establish a schedule of civil fines, costs, and assessments to be imposed for civil infractions that occur within the respective district or city. If a schedule is established, it must be prominently posted and readily available for public inspection. A schedule need not include all violations that are designated by law or ordinance as civil infractions. A schedule may exclude cases on the basis of a defendant's prior record of civil infractions or traffic offenses, or a combination of civil infractions and traffic offenses." MCL 257.907(7).

³⁹ This section discusses the ordering of civil fines, costs, and assessments. See Section 1.21 for information on waiving a fine, costs, or assessment. In addition, see the Michigan Judicial Institute's table summarizing the information discussed in this section.

"The state court administrator shall annually publish and distribute to each district and court a recommended range of civil fines and costs for first-time civil infractions. This recommendation is not binding on the courts that have jurisdiction over civil infractions but is intended to act as a normative guide for judges and district court magistrates and a basis for public evaluation of disparities in the imposition of civil fines and costs throughout this state." MCL 257.907(8).⁴⁰

"Fines, costs, and other financial obligations imposed by the court must be paid at the time of assessment, except when the court allows otherwise, for good cause shown." MCR 1.110. See also MCL 257.907(2) ("[p]ermission may be granted for payment of a civil fine and costs to be made within a specified period of time or in specified installments but, unless permission is included in the order or judgment, the civil fine and costs shall be payable immediately").

A. Civil Fines

In general, if "a person [is] found responsible or responsible 'with explanation'... for a civil infraction under [the MVC] or a local ordinance that substantially corresponds to a provision of [the MVC], the person must be ordered to pay a civil fine of not more than \$100.00." MCL 257.907(2)(a). However, MCL 257.907(2) provides for alternative civil fine amounts for certain enumerated civil infractions.

"A court may not increase a scheduled civil fine because the defendant has requested a hearing." MCR 4.101(G)(1).

1. Specific Fines and Fine Amounts

MCL 257.907(2)⁴¹ provides specific fines and fine amounts for certain MVC violations:

 "If the civil infraction was a moving violation that resulted in an at-fault collision with another vehicle, an individual, or any other object, the civil fine ordered under [MCL 257.907] is increased by \$25.00 but the total civil fine must not be more than \$100.00." MCL 257.907(2)(b).

⁴⁰The trial court fee schedules are available at https://www.courts.michigan.gov/publications/fines,-fees,costs,-and-rates/. The Recommended Range of Fines and Costs for Civil Infractions is available at https:// www.courts.michigan.gov/498aae/siteassets/court-administration/resources/fc_ci.pdf.

⁴¹ This list represents the civil infractions relevant to the scope of this benchbook. It does not include the civil infractions listed for commercial vehicle violations.

- "For a violation of [MCL 257.240, damages or violation of law resulting from use or ownership of vehicle by another after title transfer], the civil fine ordered under [MCL 257.907] is \$15.00." MCL 257.907(2)(c). See also MCL 257.240(3).
- "For a violation of [MCL 257.312a(4)(a), first violation of motorcycle indorsement⁴²], the civil fine ordered under [MCL 257.907] must not be more than \$250.00." MCL 257.907(2)(d).
- For a violation of MCL 257.602b(1), holding or using a mobile electronic device while operating a motor vehicle, the civil fine ordered under MCL 257.907 must be "either of the following:

(A) If the violation does not involve an accident, \$100.00 for a first offense and \$250.00 for a second or subsequent offense.

(B) If the violation involves an accident for which the individual is at fault, \$200.00 for a first offense and \$500.00 for a second or subsequent offense." MCL 257.907(2)(h)(*i*). See also MCL 257.602b(4)(a)-(b); MCL 257.602b(6).

• For a violation of MCL 257.602b(2), holding or using a mobile electronic device, while operating a commercial motor vehicle or school bus, the civil fine ordered under MCL 257.907 must be "either of the following:

(A) If the violation does not involve an accident, \$200.00 for a first offense and \$500.00 for a second or subsequent offense.

(B) If the violation involves an accident for which the individual is at fault, \$400.00 for a first offense and \$1,000.00 for a second or subsequent offense." MCL 257.907(2)(h)(*ii*). See also MCL 257.602b(5)(a)-(b); MCL 257.602b(6).

• "For a violation of [MCL 257.627c, exceeding posted speed limit by 10 miles per hour or more in a work zone], the civil fine ordered under [MCL 257.907] must not be more than \$150.00 for a second violation

⁴² "For a violation that occurs after 1 or more prior judgments, the individual is guilty of a misdemeanor punishable by imprisonment for not more than 1 year or a fine of not more than \$500.00, or both." MCL 257.312a(4)(b).

as described in [MCL 257.627c(2)(b)] and \$300.00 for a third or subsequent violation described in [MCL 257.627c(2)(c)]." MCL 257.907(2)(*i*).

- "For a violation of [MCL 257.674(1)(s), disabled parking violations], or a local ordinance that substantially corresponds to [MCL 257.674(1)(s)], the civil fine ordered under [MCL 257.907] must not be less than \$100.00 or more than \$250.00." MCL 257.907(2)(j).
- For a violation of [MCL 257.676a(3), sale or display for sale of produce or merchandise within highway right-of-way in violation of permit], the civil fine ordered under [MCL 257.907] must not be more than \$10.00." MCL 257.907(2)(k).
- "For a violation of [MCL 257.676c, soliciting business at the scene of an accident or disabled vehicle], the civil fine ordered under [MCL 257.907] is \$1,000.00." MCL 257.907(2)(*I*).
- "For a violation of [MCL 257.682,⁴³ failure to stop for school bus,] or a local ordinance that substantially corresponds to [MCL 257.682], the civil fine ordered under [MCL 257.907] must not be less than \$100.00 or more than \$500.00." MCL 257.907(2)(m).
- "For a violation of [MCL 257.710d, child less than four years of age not properly secured in a child restraint system], the civil fine ordered under [MCL 257.907] must not be more than \$10.00, subject to [MCL 257.907(11)⁴⁴]." MCL 257.907(2)(n).
- "For a violation of [MCL 257.710e, seat belt violations], the civil fine and court costs ordered under [MCL 257.907] must be \$25.00." MCL 257.907(2)(o).

2. Fines Doubled for Moving Violations in Work Zone, Emergency Scene, School Zone, or School Bus Zone

"Notwithstanding any other provision of [the MVC], a person responsible for a moving violation in a work zone, at an emergency scene, or in a school zone during the period

⁴³ "In addition to the civil fine and costs provided for a civil infraction under [MCL 257.907], the judge, district court referee, or district court magistrate may order a person who violates [MCL 257.682] to perform not more than 100 hours of community service at a school." MCL 257.682(4).

⁴⁴ MCL 257.907(11) allows the court to waive the civil fine in certain circumstances. See Section 1.21(B) for more information on waiving a fine for a violation of MCL 257.710d.

Section 1.20

beginning 30 minutes before school in the morning and through 30 minutes after school in the afternoon, or in a school bus zone is subject to a fine that is double the fine otherwise prescribed for that moving violation." MCL 257.601b(1).

3. Minimum Fines For Violating Maximum Speed Limit

MCL 257.629c(1) provides the minimum fine amounts for maximum speed limit violations:

"Notwithstanding [MCL 257.320a and MCL 257.907], a person who is determined responsible or responsible 'with explanation' for a civil infraction for violating the maximum speed limit on a limited access freeway or part of a limited access freeway upon which the maximum speed limit is 55 miles per hour or more shall be ordered by the court to pay a minimum fine... only according to the following schedule, except as otherwise provided in subsections (2) and (3)^[45]:"

Table 1:

Miles per hour over the speed limit	Minimum Fine
1– 5	\$10
6–10	\$20
11–15	\$30
16–25	\$40
26 or over	\$50

⁴⁵MCL 257.629c(2) provides that MCL 257.629c(1) "does not apply to a person operating a vehicle or vehicle combination for which the maximum rate of speed is established pursuant to [MCL 257.627(5)] or [MCL 257.627(7):]" however, effective January 5, 2017, 2016 PA 445 amended MCL 257.627 to delete the content that was in those subsections. The former MCL 257.627(5) and MCL 257.627(7) addressed a person operating a passenger vehicle drawing another vehicle or trailer and a person operating a school bus, respectively. See former MCL 257.627. MCL 257.629c has not been amended to reflect the changes made by 2016 PA 445. And "[f]or a violation of a maximum speed limit on a limited access freeway by a person operating a vehicle or vehicle combination described in [MCL 257.629c(2)], points shall be assessed under [MCL 257.320a] and fines shall be assessed under [MCL 257.907]." MCL 257.629c(3).

4. Distribution of Fines

"Except as provided in [MCL 257.909(2), MCL 257.909(3), and MCL 257.909(6)], a civil fine that is ordered under [MCL 257.907] for a violation of [the MVC] or other state statute must be exclusively applied to the support of public libraries and county law libraries in the same manner as is provided by law for penal fines assessed and collected for violation of a penal law of this state." MCL 257.909(1).⁴⁶

"A civil fine ordered for a violation of [MCL 257.682] that is a camera-based violation as defined in [MCL 257.682] must be paid to the county treasurer or the county treasurer's designee and be distributed by the county treasurer or the county treasurer's designee to the school district that operates the school bus and be used for school transportation safety-related purposes as provided in [MCL 257.682]." MCL 257.909(3). "A civil fine ordered for a violation of a code or ordinance of a local authority that substantially corresponds to [MCL 257.682] that is a camera-based violation as defined in [MCL 257.682] must be paid to the county treasurer or the county treasurer's designee and be distributed by the county treasurer or the county treasurer's designee to the school district that operates the school bus and be used for school transportation safetyrelated purposes as provided in [MCL 257.682]." MCL 257.909(4). "A county treasurer may enter into a contract with and designate a private vendor to process a civil fine described in [MCL 257.909(3)] or [MCL 257.909(4)]." MCL 257.909(5). A private vendor . . . may be . . . contracted by a school district to install, operate, and provide support to a stop-arm camera system on a school bus under...MCL 257.1820." MCL 257.909(5). "A civil fine that is ordered under [MCL 257.907] for a violation of [MCL 257.627c] must be paid to the state transportation department." MCL 257.909(6). "The state transportation department shall deposit any civil fines described in this subsection into the work zone safety fund created in [MCL 257.907a]." MCL 257.909(6).

B. Costs

"Except as otherwise provided, a person found responsible or responsible 'with explanation' for a civil infraction must pay costs as provided in [MCL 257.907(4).]" MCL 257.907(2).

⁴⁶"[MCL 257.909(1)] is intended to maintain a source of revenue for public libraries that previously received penal fines for misdemeanor violations of [the MVC] that are now civil infractions." MCL 257.909(2).

"If a civil fine is ordered under [MCL 257.907(2)], the judge or district court magistrate shall summarily tax and determine the costs of the action, which are not limited to the costs taxable in ordinary civil actions, and may include all expenses, direct and indirect, to which the plaintiff has been put in connection with the civil infraction, up to the entry of judgment." MCL 257.907(4). "Costs must not be ordered in excess of \$100.00." *Id.* "A civil fine ordered under [MCL 257.907(2)] must not be waived unless costs ordered under [MCL 257.907(4)] are waived." MCL 257.907(4).

MCL 257.729 also authorizes the imposition of certain costs:

"In addition to a fine assessed for the . . . civil infraction when . . . determined responsible, the magistrate may also add to any fine and costs levied additional costs incurred in compelling the appearance of the person, which additional costs shall be returned to the general fund of the unit of government incurring the costs."

"Except as otherwise provided by law, costs are payable to the general fund of the plaintiff." MCL 257.907(4).

C. Assessments

"In addition to any civil fines or costs ordered to be paid under [MCL 257.907], the judge or district court magistrate shall order the defendant to pay a justice system assessment of \$40.00 for each civil infraction determination, except for a parking violation or a violation for which the total fine and costs imposed are \$10.00 or less." MCL 257.907(12).

"On payment of the assessment, the clerk of the court shall transmit the assessment collected to the state treasury to be deposited into the justice system fund created in ... MCL 600.181." MCL 257.907(12).

"An assessment levied under [MCL 257.907(12)] is not a civil fine for purposes of [MCL 257.909]." MCL 257.907(12).

1.21 Waiving Civil Fines, Costs, and Assessments

"The court may waive fines, costs and fees, pursuant to statute or court rule, or to correct clerical error." MCR 4.101(G)(4).

"A civil fine ordered under [MCL 257.907(2)] must not be waived unless costs ordered under [MCL 257.907(4)] are waived." MCL 257.907(4).

A. Mandatory Waiver

A court must waive any ordered civil fine, costs, and assessment in the following circumstances:

• Defective safety equipment violation under MCL 257.683.

"If a person has received a civil infraction citation for defective safety equipment on a vehicle under [MCL 257.683], the court shall waive a civil fine, costs, and assessments on receipt of certification by a law enforcement agency that repair of the defective equipment was made before the appearance date on the citation." MCL 257.907(9).

• Failing to carry or display registration certificate under MCL 257.223.

"If a person has received a citation for a violation of [MCL 257.223], the court shall waive any civil fine, costs, and assessment on receipt of certification by a law enforcement agency that the person, before the appearance date on the citation, produced a valid registration certificate that was valid on the date the violation of [MCL 257.223] occurred." MCL 257.907(13).

• Failing to produce certificate of insurance under MCL 257.328.

"If a person has received a citation for a violation of [MCL 257.328(1)] for failing to produce a certificate of insurance under [MCL 257.328(2)], the court...shall waive any fine, costs, and any other fee or assessment otherwise authorized under [the MVC] on receipt of verification by the court that the person, before the appearance date on the citation, produced valid proof of insurance that was in effect when the violation of [MCL 257.328(1)] occurred." MCL 257.907(14).⁴⁷ "Insurance obtained after the violation occurred does not make the person eligible for a waiver under [MCL 257.907(15)]." *Id.*

However, "[i]f, before the appearance date on a citation issued under [MCL 257.328(1)], the defendant submits proof to the court that the motor vehicle had insurance

⁴⁷Additionally, the court may waive the fee of not more than \$25 set out in MCL 257.328(3)(c). MCL 257.907(14).

meeting the requirements of [MCL 500.3101 and MCL 500.3102], at the time the violation of [MCL 257.328(1)] occurred, all of the following apply:

(a) The court shall not assess a fine or costs.

(b) The court shall not forward an abstract of the court record to the secretary of state.

(c) The court may assess a fee of not more than \$25.00 that shall be paid to the court funding unit." MCL 257.328(3).

B. Discretionary Waiver

"Before [April 2, 2025], the court may waive any civil fine, cost, or assessment against an individual who received a civil infraction citation for a violation of [MCL 257.710d (child not properly secured in a child restraint system)] if the individual, before the appearance date on the citation, supplies the court with evidence of acquisition, purchase, or rental of a child seating system meeting the requirements of [MCL 257.710d]." MCL 257.907(11).

"Beginning [April 2, 2025], the court may waive any civil fine, cost, or assessment against an individual who received a civil infraction citation for a violation of [MCL 257.710d] if the individual, before the appearance date on the citation, supplies the court with evidence of acquisition of a child seating system that meets the requirements of [MCL 257.710d] and evidence that the individual has received education from a certified child passenger safety technician." MCL 257.907(11).

1.22 Civil Contempt Proceedings for Nonpayment of Monies Owed

"If a defendant defaults in the payment of a civil fine, costs, or both, or of any installment, as ordered pursuant to [MCL 257.907(2)], the court, upon the motion of the plaintiff or upon its own motion, may require the defendant to show cause why the default should not be treated as in civil contempt and may issue a summons or order to show cause or a bench warrant of arrest for the defendant's appearance." MCL 257.908(1).

"Unless the defendant shows that the default was not attributable to an intentional refusal to obey the order of the court or to a failure on his or her part to make a good faith effort to obtain the funds required for payment, the court shall find that the default constitutes a civil contempt and may order the defendant committed until the civil fine, costs, or both, or a specified part thereof, is paid." MCL 257.908(3). However,

"[t]he court shall not sentence a person to a term of incarceration for nonpayment unless the court has complied with the provisions of MCR 6.425(D)(3)." MCR 3.606(F) (addressing contempts outside the immediate presence of the court). MCR 6.425(D)(3) addresses incarceration for nonpayment, requires an ability to pay determination, provides for payment alternatives, and offers guidance for determining manifest hardship. For a detailed discussion of MCR 6.425(D)(3), see the Michigan Judicial Institute's *Criminal Proceedings Benchbook Vol. 2*, Chapter 8.

"If it appears that the default in the payment of a civil fine or costs does not constitute civil contempt, the court may enter an order allowing the defendant additional time for payment, reducing the amount of payment or of each installment, or revoking the fine or costs or the unpaid portion thereof in whole or in part." MCL 257.908(4).

"The term of imprisonment on civil contempt for nonpayment of a civil fine or costs shall be specified in the order of commitment, and shall not exceed 1 day for each \$10.00 of the fine and costs." MCL 257.908(5). "A person committed for nonpayment of a civil fine or costs shall be given credit toward payment for each day of imprisonment and each day of detention in default of recognizance before judgment at the rate of \$10.00 per day." *Id*.

"A defendant committed to imprisonment for civil contempt for nonpayment of a civil fine or costs shall not be discharged from custody until 1 of the following occurs:

(a) The defendant has been credited with the amount due pursuant to [MCL 257.908(5)].

(b) The amount due has actually been collected through execution of process or otherwise.

(c) The amount due has been satisfied pursuant to a combination of [MCL 257.908(6)(a) and MCL 257.908(6)(b)]." MCL 257.908(6).

"The civil contempt shall be purged upon discharge of the defendant pursuant to [MCL 257.908(6)]." MCL 257.908(7).

1.23 Treatment, Education, and Rehabilitation Programs

"In addition to a civil fine and costs ordered under [MCL 257.907(2)] and [MCL 257.907(4)] and the justice system assessment ordered under [MCL 257.907(12)], the judge or district court magistrate may order the individual to attend and complete a program of treatment, education, or rehabilitation." MCL 257.907(5).

Part B—Misdemeanors

"A misdemeanor case must be conducted in compliance with the constitutional and statutory procedures and safeguards applicable to misdemeanors cognizable by the district court." MCR 6.615(D).

1.24 Courts With Jurisdiction Over Misdemeanor Traffic Offenses

The following courts have jurisdiction over misdemeanor traffic offenses:

- "The district court has jurisdiction of ... [m]isdemeanors punishable by a fine or imprisonment not exceeding 1 year, or both[,]" "[o]rdinance and charter violations punishable by a fine or imprisonment, or both[,]" arraignments, fixing bail, accepting bonds, probable cause conferences in all cases not cognizable by the district court, preliminary examinations in all cases not cognizable by the district court, preliminary examinations in all cases not cognizable by the district court,⁴⁸ and circuit court arraignments in all cases not cognizable by the district court, ⁴⁸ and circuit court. MCL 600.8311(a)-(f). See also MCR 6.008(A) ("The district court has jurisdiction over all misdemeanors and all felonies through the preliminary examination and until the entry of an order to bind the defendant over to the circuit court.").
- "[M]unicipal courts . . . shall have original jurisdiction of all prosecutions and proceedings . . . for all misdemeanors and offenses arising under the laws of [Michigan] and committed within the corporate limits of the cities in which the courts are located, and which are punishable by a fine or imprisonment for not more than 1 year, or both, and may issue all lawful writs and process and do all lawful acts which may be necessary and proper to carry into effect the jurisdiction given by [the Municipal Courts Act]." MCL 730.551.
- The family division of circuit court has jurisdiction over all misdemeanor offenses committed by juveniles under age 18. MCL 712A.2(a)(1). See also MCL 600.1021(1)(e). "A family division of circuit court in any judicial circuit may adopt or institute a juvenile mental health court pursuant to statute or court rules." MCL 600.1099c. A juvenile that is "alleged to have engaged in activity that would constitute a criminal act if committed by an adult" may be admitted into juvenile mental health court. MCL 600.1099f(1).

⁴⁸ "There shall not be a preliminary examination for any misdemeanor to be tried in a district court." MCL 600.8311(e).

- Additionally, "[t]he circuit court or the district court in any judicial circuit or a district court in any judicial district may adopt or institute a mental health court pursuant to statute or court rules." MCL 600.1091(1). Individuals charged with criminal traffic offenses may be admitted into mental health court. See MCL 600.1094(1).
- "As part of a concurrent jurisdiction plan, the circuit court and district court may enter into an agreement for district court probation officers to prepare the presentence investigation report and supervise on probation defendants who either plead guilty to, or are found guilty of, a misdemeanor in circuit court. The case remains under the jurisdiction of the circuit court." MCR 6.008(E).

1.25 Jurisdiction and Duties of District Court Magistrates

MCL 600.8511(b) sets forth the jurisdiction and duty of district court magistrates in regard to violations of the MVC. MCL 600.8511(b) provides:

"A district court magistrate has the ... jurisdiction and dut[y]...[t]o arraign and sentence upon pleas of guilty or nolo contendere for violations of the Michigan vehicle code, ... MCL 257.1 to MCL 257.923, or a local ordinance substantially corresponding to a provision of the Michigan vehicle code, [MCL 257.1 to MCL 257.923], except for violations of [MCL 257.625 and MCL 257.625m], or a local ordinance substantially corresponding to [MCL 257.625 or MCL 257.625m], if authorized by the chief judge of the district court district and if the maximum permissible punishment does not exceed 93 days in jail or a fine, or both." MCL 600.8511(b). "However, the chief judge may authorize the magistrate to arraign defendants and set bond with regard to violations of [MCL 257.625 and MCL 257.625m], or a local ordinance substantially corresponding to [MCL 257.625 or MCL 257.625m]."

Notably, MCL 600.8511(b) "establishes two different grants of authority." *People v VanEss*, _____ Mich App ____, ____ (2024). "The first is to arraign and sentence upon a guilty or nolo contendere plea for violations of the motor vehicle code, except for violations of MCL 257.625 and MCL 257.625m if the penalty does not exceed 93 days in jail." *VanEss*, _____ Mich App at ____. "The second grant of authority is to arraign and set bond for violations of MCL 257.625 and MCL 257.625 and MCL 257.625m if authorized by the chief judge." *VanEss*, _____ Mich App at _____ ("Absent is the authority to sentence upon a plea of guilty or nolo contendere [to violations of MCL 257.625 and MCL 257.625m]."). However, "a district court judge has the express authority

to supersede any action by a district court magistrate, even without a formal appeal." *Id.* at ____ (holding that "the magistrate did not have the authority to take a plea that was binding on the district court judge"), citing MCR 4.401(C).

See MCL 600.8511 for the jurisdiction and duties of district court magistrates in other contexts.

See also the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1,* for additional discussion of the jurisdiction and duties of district court magistrates.

1.26 Initiating a Misdemeanor Action by Issuing a Citation⁴⁹

Under the MVC, a police officer *must* issue a citation⁵⁰ to a person who is arrested without a warrant for "a violation of [the MVC] punishable as a misdemeanor, or an ordinance substantially corresponding to a provision of [the MVC] and punishable as a misdemeanor, under conditions not referred to in [MCL 257.617,⁵¹ MCL 257.619,⁵² or MCL 257.727.⁵³]" MCL 257.728(1). But see MCL 764.9c(1), a provision in the Code of Criminal Procedure authorizing an officer to issue an appearance ticket under the same circumstances in lieu of issuing a formal complaint under MCL 764.13.⁵⁴ "The appearance ticket ..., or other documentation as requested, must be forwarded to the court, appropriate prosecuting authority, or both, for review without delay." MCL 764.9c(1). "Except as provided in [MCL 764.9c(5)], a police officer shall issue to and serve upon a person an appearance ticket... and release the person from custody if the person has been arrested for a misdemeanor or ordinance violation that has a maximum permissible penalty that does not exceed 1 year in jail or a fine, or both, and is not...an operating while intoxicated offense." MCL 764.9c(4). MCL 764.9c "does not create a right to the issuance of an appearance ticket in lieu of an arrest. An arrested person may appeal the legality of his or her

⁴⁹ This section discusses citation requirements. For information on initiating a misdemeanor action by filing an order for an arrest warrant or by issuing a summons, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 3.

⁵⁰ "The citation may be prepared electronically or on paper." MCR 6.615(A)(1)(a).

⁵¹ Leaving the scene of an accident resulting in serious impairment of body function or death. MCL 257.617.

⁵² Failing to give the proper information and aid after an accident. MCL 257.619.

⁵³ Requiring a person who was arrested without a warrant for certain specified violations to be arraigned (if an adult) or taken before the family division of circuit court (if a minor) "without unreasonable delay[.]"

⁵⁴ "The appearance ticket . . . ,or other documentation as requested, must be forwarded to the court, appropriate prosecuting authority, or both, for review without delay." MCL 764.9c(1).

arrest as provided by law. However, an arrested person does not have a claim for damages against a police officer or law enforcement agency because he or she was arrested rather than issued an appearance ticket." MCL 764.9c(8).

However, where no arrest occurs, the MVC provides that "[a] police officer *may* issue a citation to a person who is an operator of a motor vehicle involved in an accident if, based upon personal investigation, the officer has reasonable cause to believe that the person has committed a misdemeanor under [the MVC] in connection with the accident." MCL 257.728(8) (emphasis added). See also MCR 6.615(A)(1)(b). "The citation may serve as a sworn complaint and as a summons to command (a) the initial appearance of the defendant; and (b) for misdemeanor traffic cases, a response from the defendant as to his or her guilt of the violation alleged." MCR 6.615(A)(2). "The citation must be signed by the officer in accordance with MCR 1.109(E)(4); if a citation is prepared electronically and filed with a court as data, the name of the officer that is associated with issuance of the citation satisfies this requirement." MCR 6.615(A)(1)(a).

In lieu of being issued a citation to appear, a person arrested without a warrant for a traffic misdemeanor may demand to be brought to a judge or district court magistrate or, if the person is a juvenile,⁵⁵ to the family division of the circuit court for arraignment as provided in MCL 257.727. MCL 257.728(1). If a nonresident demands an immediate arraignment, and a judge or district court magistrate is not available to conduct the arraignment or if an immediate trial cannot be held, the nonresident may deposit with the officer a guaranteed appearance certificate or a sum of money not to exceed \$100 and be issued a written citation. MCL 257.728(5). However, a nonresident may not be issued a written citation if he or she was arrested for a violation of any offense listed in MCL 257.727(a)-(d). MCL 257.728(5).

A. Citation Requirements

In addition to including a proper signature, MCR 6.615(A)(1)(a), the citation must contain "the name and address of the person, the violation charged, and the time and place when and where the person shall appear in court," MCL 257.728(1) (warrantless arrest for alleged misdemeanor violation). See also MCL 257.728(8) (requiring substantially similar information for traffic accidents allegedly involving a misdemeanor where no arrest is made). The officer must complete an original and three copies of the citation. MCL 257.728(1); MCL 257.728(8). The original must be filed with the court in which the appearance is to be made, the first copy is

⁵⁵ See Chapter 2 for more information on juvenile traffic procedures.

retained by the local traffic enforcement agency, the second copy is delivered to the violator if the violation is a misdemeanor, and the third copy is delivered to the violator if the violation is a civil infraction. MCL 257.727c(1).⁵⁶ See also MCL 257.743, which requires additional information pertaining to an accused's right to admit or deny responsibility for a traffic civil infraction citation.

B. Accepting a Plea Based on a Citation

A judge or district court magistrate (if authorized to do so)⁵⁷ can accept a plea of guilty or not guilty based solely on a citation. MCL 257.728e. However, if the accused pleads not guilty to a misdemeanor, a sworn complaint must be filed with the court before any further proceedings may be conducted. MCL 257.728e. "A warrant for arrest shall not issue for an offense [charged in the citation] until a sworn complaint is filed with the magistrate." MCL 257.728e.

"[N]ot all appearance tickets or citations are considered sworn complaints under the Michigan Vehicle Code or the Code of Criminal Procedure, and not every appearance before the magistrate necessarily is preceded by the issuance of a complaint." City of Plymouth v McIntosh, 291 Mich App 152, 162 (2010). "This procedure, . . . is designed to ensure that, following a plea of not guilty, until the magistrate has in front of him or her either a sworn complaint or a citation that takes the place of a sworn complaint, further proceedings do not occur." Id. Where an appearance ticket is issued for a misdemeanor violation and is in the form of a "Uniform Law Citation" containing the language, "'I declare under the penalties of perjury that the statements above are true to the best of my information, knowledge, and belief," it constitutes a sworn complaint under MCL 257.727c, MCL 764.1e, and MCR 6.615. City of Plymouth, 291 Mich App at 153-154, 154 n 1, 163. A prosecutor is not required to file a second sworn complaint in order to proceed on a not guilty plea. *Id.* at 163.⁵⁸

⁵⁶ With the approval of certain specified officials, the content or number of copies required by MCL 257.727c(1) may be modified "to accommodate law enforcement and local court and procedures and practices." MCL 257.727c(2).

⁵⁷A district court magistrate has the power "[t]o arraign and sentence upon pleas of guilty or nolo contendere for violations of the [MVC] . . . or a local ordinance substantially corresponding to a provision of the [MVC] . . . except for violations of . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, if authorized by the chief judge of the district court district and if the maximum permissible punishment does not exceed 93 days in jail or a fine, or both. However, the chief judge may authorize the magistrate to arraign defendants and set bond with regard to violations of . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m, or a local ordinance substantially corresponding to . . . MCL 257.625 and [MCL] 257.625m." MCL 600.8511(b).

C. Failure to Appear

"Except as provided in MCR 6.615(B), if a defendant fails to appear in court, the court must wait 48 hours, excluding weekends and holidays if the court is closed to the public, before issuing a bench warrant to allow the defendant an opportunity to voluntarily appear before the court." MCR 6.103(A). See also MCL 764.3(1). "If the defendant does not appear within 48 hours, the court must issue a bench warrant unless the court believes there is good reason to instead schedule the case for further hearing." MCR 6.103(A)(3). See also MCL 764.3(1). "The court must not revoke a defendant's release order or forfeit bond during the 48-hour period of delay before a warrant is issued." MCR 6.103(C). See MCL 764.3(2). However, MCR 6.103(A) does not apply if "the defendant previously failed to appear in the case." MCR 6.103(A)(1).

MCR 6.103(A)(2) permits a court to "immediately issue a bench warrant only if the court has a specific articulable reason, stated on the record, to suspect any of the following apply:

(a) the defendant has committed a new crime.

(b) a person or property will be endangered if a bench warrant is not issued.

(c) prosecution witnesses have been summoned and are present for the proceeding.

(d) the proceeding is to impose a sentence for the crime.

(e) there are other compelling circumstances that require the immediate issuance of a bench warrant." See also MCL 764.3(3).

MCR 6.103 "does not abridge a court's authority to issue an order to show cause, instead of a bench warrant, if a defendant fails to appear in court." MCR 6.103(B).

⁵⁸MCL 257.728e applies to misdemeanor citations punishable by imprisonment for not more than 90 days. The *City of Plymouth* case involved a misdemeanor citation that was punishable by imprisonment for not more than 93 days, which was challenged under MCL 764.9g. Similar to MCL 257.728e, MCL 764.9g(1) provides "[i]f the offender pleads not guilty, no further proceedings may be had until a sworn complaint is filed with the magistrate." Although the *City of Plymouth* case involved a different statute, the Court considered the nearly identical language of MCL 257.728e in reaching its conclusion. "[I]n those cases where the prosecution files a sworn complaint before arraignment, MCL 764.9g is satisfied and there is no requirement that a second sworn complaint be filed. The same is true as to . . . MCL 257.728e." *City of Plymouth*, 291 Mich App at 156-157.

MCR 6.615(B) provides that "[i]f a defendant fails to respond to any matter pending relative to a misdemeanor citation issued under MCL 764.9c, the court shall issue an order to show cause." MCR 6.615(B). "The court may immediately issue a bench warrant, rather than an order to show cause, if the court has a specific articulable reason to suspect that any of the following apply and states it on the record:

(a) the defendant has committed a new crime.

(b) the defendant's failure to appear is the result of a willful intent to avoid or delay the adjudication of the case.

(c) another person or property will be endangered if a warrant is not issued." MCR 6.615(B)(1). See also MCL 764.3(4).

"If a defendant fails to appear or otherwise respond to any matter pending relative to a misdemeanor traffic citation, the court must also initiate the procedures required by MCL 257.321a."⁵⁹ MCR 6.615(B)(2).

D. Restrictions on the Issuance of Citations

MCL 257.728(1) prohibits the issuance of citations for the following offenses⁶⁰:

- Leaving the scene of an accident resulting in serious impairment of body function or death. MCL 257.617.
- Failing to give the proper information and aid after an accident. MCL 257.619.
- Committing a moving violation causing death or serious impairment of a body function to another person under MCL 257.601d. MCL 257.727(a).
- Operating a vehicle while intoxicated, visibly impaired, with any bodily alcohol content if under age 21, or while having a controlled substance in his or her body under MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8), or a substantially corresponding ordinance. MCL 257.727(b).

⁵⁹ MCL 257.321a governs the cancellation, suspension, and revocation of a driver's license.

⁶⁰Some of the listed offenses are felonies, not punishable as misdemeanors, or may be punishable as felonies if the person has prior convictions.

- Causing death or serious impairment of a body function by operating a vehicle while intoxicated or visibly impaired, or while having a controlled substance in his or her body, MCL 257.625(4)-257.625(5). MCL 257.727(b).
- Operating a vehicle while intoxicated or visibly impaired, with any bodily alcohol content if under age 21, or while having a controlled substance in his or her body, and having occupants under age 16 in the vehicle, MCL 257.625(7). MCL 257.727(b).
- Reckless driving, MCL 257.626, or a substantially corresponding ordinance, unless the officer deems that issuing a citation and releasing the person will not constitute a public menace. MCL 257.727(c).
- Not having in his or her immediate possession at the time of arrest a valid operator's or chauffeur's license, MCL 257.311, or a receipt for an already surrendered license, MCL 257.311a. However, if the officer can satisfactorily determine the identity of the person and whether the person can be apprehended if he or she fails to appear before the designated magistrate, the officer may issue a citation. MCL 257.727(d).

E. Summons

• As an alternative to filing an order for a warrant after an arrest, the prosecutor may issue a written order for a summons addressed to the accused person under MCL 764.9a(1). See the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Volume 1*, Chapter 3, for more information on summons to appear in misdemeanor cases.

F. Arraignment

MCL 257.727 provides:

"If a person is arrested without a warrant in any of the following cases, the arrested person shall, without unreasonable delay, be arraigned by the magistrate who is nearest or most accessible within the judicial district as provided in . . . MCL 764.13, or, if a minor, taken before the family division of circuit court within the county in which the offense charged is alleged to have been committed:

(a) The person is arrested under [MCL 257.601d].

(b) The person is arrested under [MCL 257.625(1), MCL 257.625(3), MCL 257.625(4), MCL 257.625(5),

MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)], or an ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)].

(c) A person is arrested under [MCL 257.626] or an ordinance substantially corresponding to [MCL 257.626]. If under the existing circumstances it does not appear that releasing the person pending the issuance of a warrant will constitute a public menace, the arresting officer may proceed as provided by [MCL 257.728].

(d) A person arrested does not have in his or her immediate possession a valid operator's or chauffeur's license or the receipt described in [MCL 257.311a]. If the arresting officer otherwise satisfactorily determines the identity of the person and the practicability of subsequent apprehension if the person fails to voluntarily appear before a designated magistrate or the family division of circuit court as directed, the officer may release the person from custody with instructions to appear in court, given in the form of a citation as prescribed by [MCL 257.728]."

1.27 General Criminal Penalty for Misdemeanor Traffic Offenses

A. Generally

"Unless another penalty is provided in [the MVC] or by the laws of [Michigan], a person convicted of a misdemeanor for the violation of [the MVC] shall be punished by a fine of not more than \$100.00, or by imprisonment for not more than 90 days, or both." MCL 257.901(2).

B. Exceptions

"Notwithstanding any other provision of [the MVC], a person responsible for a moving violation in a work zone, at an emergency scene, or in a school zone during the period beginning 30 minutes before school in the morning and through 30 minutes after school in the afternoon, or in a school bus zone is subject to a fine that is double the fine otherwise prescribed for that moving violation." MCL 257.601b(1).

"A person who commits a moving violation in a work zone or a school bus zone for which not fewer than 3 points are assigned under [MCL 257.320a] and as a result causes injury to another person in the work zone or school bus zone is guilty of a misdemeanor punishable by a fine of not more than \$1,000.00 or imprisonment for not more than 1 year, or both." MCL 257.601b(2).

1.28 Rebuttable Presumption Regarding Jail or Probation Sentence

"There is a rebuttable presumption that the court shall sentence an individual convicted of a misdemeanor, other than a serious misdemeanor, with a fine, community service, or other nonjail or nonprobation sentence." MCL 769.5(3). "The court may depart from the presumption under [MCL 769.5(3)] if the court finds reasonable grounds for the departure and states on the record the grounds for the departure." MCL 769.5(4).

"Unlike cases involving felony convictions, there are no sentencing guidelines that a sentencing court must consult when sentencing a person convicted of only a misdemeanor offense." People of the City of Auburn Hills v Mason, ____ Mich App ___, ___ (2024). However, MCL 769.5(3) and MCL 769.5(4) "establish a sentencing framework for misdemeanor convictions that is similar to the framework for felony convictions and the legislative sentencing guidelines." *Mason*, Mich App at ____. Accordingly, "a 'nonjail or nonprobation sentence' imposed on 'an individual convicted of a misdemeanor, other than a serious misdemeanor,' pursuant to MCL 769.5(3) is a presumptively proportionate sentence, as a within-guidelines sentence is for a felony conviction." Mason, ____ Mich App at ____ ("Just like in the context of sentencing guidelines, the district court was obligated to explain why a departure sentence of 93 days in jail was more suitable than a non-jail or non-probation sentence under MCL 769.5(3)."). "Nonetheless, under MCL 769.5(4), a court imposing a sentence for an ordinary misdemeanor conviction remains free to depart from the presumption in MCL 769.5(4) 'if the court finds reasonable grounds for the departure and states on the record the grounds for the departure."" Mason, ____ Mich App at ____ (holding that the "[district] court did not adequately justify the imposed sentence," because it "did not consider the circumstances of the offense and did not explain how its departure sentence was more proportionate than a different sentence would have been").

1.29 Minimum State Costs for Misdemeanor Traffic Offenses

"[W]hen fines and costs are assessed by a judge or district court magistrate, the defendant shall be ordered to pay costs of not less than \$50.00 for each conviction for a misdemeanor or for any ordinance violation." MCL 600.8381(4). See also MCL 769.1j(1)(b) (requiring the same minimum fee for misdemeanor convictions). The costs imposed under MCL 769.1j(1) constitute a tax, and this tax does not violate the separation of powers requirement under Const 1963, art 3, § 2 or the requirement of Const 1963, art 4, § 32 that "[e]very law which imposes, continues or revives a tax shall distinctly state the tax." People v Shenoskey, 320 Mich App 80, 83-84 (2017) (applying the analysis of MCL 769.1k(1)(b)(iii) in People v Cameron, 319 Mich App 215 (2017) to MCL 769.1j(1)(a) because the statutes are "closely related[;]" MCL 769.1j(1)(a) addresses minimum costs in felony cases). Payment of the minimum state cost must be a condition of probation. MCL 771.3(1)(g). "The court shall not sentence a defendant to a term of incarceration, nor revoke probation, for failure to comply with an order to pay money unless the court finds, on the record, that the defendant is able to comply with the order without manifest hardship and that the defendant has not made a good-faith effort to comply with the order." MCR 6.425(D)(3)(a). MCR 6.425(D)(3) also addresses payment alternatives and offers guidance for determining manifest hardship. For a detailed discussion of MCR 6.425(D)(3), see the Michigan Judicial Institute's Criminal Proceedings Benchbook Vol. 2, Chapter 8.

1.30 Costs⁶¹

At the time of sentencing or a delay in sentencing or entry of a deferred judgment of guilt, the court may impose "[a]ny cost authorized by the statute for a violation of which the defendant entered a plea of guilty or nolo contendere or the court determined that the defendant was guilty," or "[u]ntil December 31, 2026, any cost reasonably related to the actual costs incurred by the trial court without separately calculating those costs involved in the particular case[.]" MCL 769.1k(1)(b)(*ii*)-(*iii*).

"MCL 769.1k(1)(b)(*iii*) "is [not] facially unconstitutional." *People v Johnson*, 366 Mich App 688, 705 (2021).⁶² "MCL 769.1k(1)(b)(*iii*) independently authorizes the imposition of costs in addition to those

⁶¹Note that "[t]he court shall not sentence a defendant to a term of incarceration, nor revoke probation, for failure to comply with an order to pay money unless the court finds, on the record, that the defendant is able to comply with the order without manifest hardship and that the defendant has not made a good-faith effort to comply with the order." MCR 6.425(D)(3)(a). MCR 6.425(D)(3) also addresses payment alternatives and offers guidance for determining manifest hardship. For a detailed discussion of MCR 6.425(D)(3), see the Michigan Judicial Institute's *Criminal Proceedings Benchbook Vol. 2*, Chapter 8.

costs authorized by the statute for the sentencing offense," and "[a] trial court possesse[s] the authority under MCL 769.1k, . . . to order defendant to pay court costs." *People v Konopka*, 309 Mich App 345, 350, 358 (2015). "A trial judge does not have unfettered discretion with respect to the amount of costs to impose under [MCL 769.1k(1)(b)(*iii*)] because the costs imposed must be reasonably related to the actual costs incurred by the trial court without separately calculating those costs involved in the particular case." *Johnson*, 366 Mich App at 700 (quotation marks, alteration, and citation omitted). "[A]lthough the costs imposed . . . need not be separately calculated, . . . the trial court [must] . . . establish a factual basis" demonstrating that "the costs imposed [are] reasonably related to the actual costs incurred by the trial court[.]" *Konopka*, 309 Mich App at 359-360.

"MCL 769.1k(1)(b)(iii) does not provide the trial court with the authority to increase the costs imposed on criminal defendants as a means for generating more revenue." Johnson, 366 Mich App at 701 (citation omitted; concluding that although MCL 769.1k(1)(b)(iii) "is a revenuegenerating statute," defendant "failed to show that [the statute] also authorizes courts to administer the revenue so collected" or establish that courts ordering the assessments benefit from a portion of the revenue under circumstances where they do not have control over administration of the revenue). "While . . . use of the funds generated pursuant to MCL 769.1k(1)(b)(*iii*) to finance the operations of the sentencing judge's court, coupled with intense pressure placed on that court by its local funding unit, could create at a minimum an appearance of impropriety, [defendant failed] to establish that no set of circumstances exist under which MCL 769.1k(1)(b)(iii) would be valid. The fact that the statute might operate unconstitutionally under some conceivable set of circumstances is insufficient." Johnson, 366 Mich App at 703 (quotation marks, alterations, and citation omitted).

The imposition of court costs under MCL 769.1k(1)(b)(*iii*) is a tax, rather than a governmental fee, and it must therefore comply with the Distinct-Statement Clause and the separation-of-powers doctrine. *People v Cameron*, 319 Mich App 215, 236 (2017). "[A]lthough it imposes a tax, MCL 769.1k(1)(b)(*iii*) is not unconstitutional[.]" *Cameron*, 319 Mich App at 218.⁶³

MCL 769.34(6) addresses the sentencing guidelines and the duties of the court when sentencing, and it generally authorizes the court to order

⁶²MCL 769.1k does not violate a defendant's due process or equal protection rights, nor does it violate the constitutional prohibition on ex post facto punishments or the principle of separation of powers. *People v Konopka*, 309 Mich App 345, 365, 367-70, 376 (2015).

⁶³For a detailed discussion of the categorization of MCL 769.1k(1)(b)(*iii*) as a tax and of the application of the Distinct-Statement Clause and separation-of-powers, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 8.

court costs ("As part of the sentence, the court may also order the defendant to pay any combination of a fine, costs, or applicable assessments."). However, "MCL 769.34(6) allows courts to impose only those costs or fines that the Legislature has separately authorized by statute[]" and "does not provide courts with the independent authority to impose any fine or cost." *People v Cunningham*, 496 Mich 145, 158 n 11 (2014).⁶⁴ For a complete listing of statutes specifically authorizing the imposition of costs, see the Michigan Judicial Institute's Table of General Costs, Table of Misdemeanor Offenses for Which Costs are Authorized, and Table of Felony Offenses for Which Costs are Authorized.

Further, a defendant may be ordered to pay any additional costs incurred to compel his or her attendance. MCL 769.1k(2). In addition, MCL 769.1k(4) authorizes a court to order that a defendant pay by wage assignment any of the costs authorized in MCL 769.1k(1). A court may provide for the collection of costs imposed under MCL 769.1k at any time. MCL 769.1k(5).

"A defendant must not be imprisoned, jailed, or incarcerated for the nonpayment of costs ordered under [MCL 769.1k] unless the court determines that the defendant has the resources to pay the ordered costs and has not made a good-faith effort to do so." MCL 769.1k(10). See also MCR 6.425(D)(3) (providing similar preconditions to incarceration for nonpayment in felony proceedings). This determination must be made by a judge; district court magistrates are not authorized to make determinations regarding a defendant's ability to pay. See MCL 600.8511, MCL 600.8512; MCL 600.8512a, MCL 600.8513 (setting forth authority of district court magistrates without granting authority for making ability to pay determinations).

In addition to any costs specifically authorized by offense-specific statutes, MCL 257.729 provides that "[i]n addition to a fine assessed for the charge . . . when found guilty . . ., the magistrate may also add to any fine and costs levied additional costs incurred in compelling the appearance of the person, which additional costs shall be returned to the general fund of the unit of government incurring the costs."

1.31 Failure to Comply with Sentence

"If the court finds that the sentenced person has not complied with his or her sentence, including a nonjail or nonprobation sentence, the court may issue an order for the person to show cause why he or she should not be held in contempt of court for not complying with the sentence. If the court finds the person in contempt, it may impose an additional sentence,

⁶⁴For a more in-depth discussion of costs, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 8.

including jail or probation if appropriate." MCL 768.5(5). If the sentenced person is found in contempt "for nonpayment of fines, costs, or other legal financial obligations, the court must find on the record that the person is able to comply with the payments without manifest hardship, and that the person has not made a good-faith effort to do so, before imposing an additional sentence." MCL 769.5(6).

Part C—Felonies

1.32 Courts With Jurisdiction Over Felony Traffic Offenses

The district court has jurisdiction to conduct felony arraignments, probable cause conferences, and preliminary examinations. MCL 600.8311(d)-(f); MCR 6.104; MCR 6.110. The circuit court has jurisdiction to try felony offenses and conduct sentencing proceedings. Const 1963, art 6, §13; MCL 600.8311(f); MCL 762.1. See also MCR 6.008(B) ("The circuit court has jurisdiction over all felonies from the bindover from the district court unless otherwise provided by law. The failure of the court to properly document the bindover decision shall not deprive the circuit court of jurisdiction. A party challenging a bindover decision must do so before any plea of guilty or no contest, or before trial.").

"The circuit court retains jurisdiction over any case in which a plea is entered or a verdict rendered to a charge that would normally be cognizable in the district court." MCR 6.008(C).

"The circuit court shall sentence all defendants bound over to circuit court on a felony that either plead guilty to, or are found guilty of, a misdemeanor." MCR 6.008(D).

Generally, the Family Division of Circuit Court has jurisdiction over criminal cases involving juveniles.⁶⁵ MCL 600.1021(1)(e). "A family division of circuit court in any judicial circuit may adopt or institute a juvenile mental health court pursuant to statute or court rules." MCL 600.1099c. A juvenile that is "alleged to have engaged in activity that would constitute a criminal act if committed by an adult" may be admitted into juvenile mental health court. MCL 600.1099f(1).

Additionally, "[t]he circuit court or the district court in any judicial circuit or a district court in any judicial district may adopt or institute a mental health court pursuant to statute or court rules." MCL 600.1091(1).

⁶⁵ See Chapter 2 for more information on required procedures for juveniles.

Individuals charged with criminal traffic offenses may be admitted into mental health court. See MCL 600.1094(1).

"As part of a concurrent jurisdiction plan, the circuit court and district court may enter into an agreement for district court probation officers to prepare the presentence investigation report and supervise on probation defendants who either plead guilty to, or are found guilty of, a misdemeanor in circuit court. The case remains under the jurisdiction of the circuit court." MCR 6.008(E).

For a general discussion of pretrial proceedings in criminal cases, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*.

1.33 Initiating a Felony Action

"A police officer may arrest an individual without a warrant if a felony has been committed and the officer has probable cause to believe that individual committed a felony." *People v Tierney*, 266 Mich App 687, 705 (2005), quoting *People v Kelly*, 231 Mich App 627, 631 (1998). A criminal defendant may also be arrested on the basis of an arrest warrant. "A court must issue an arrest warrant or a summons . . . if presented with a proper complaint and if the court finds probable cause to believe that the accused committed the alleged offense." MCR 6.102(A). MCR 6.102(C) requires a court to issue a summons rather than a warrant unless:

"(1) the complaint is for an assaultive crime or an offense involving domestic violence, as defined in MCL 764.1a.

(2) there is reason to believe from the complaint that the person against whom the complaint is made will not appear upon a summons.

(3) the issuance of a summons poses a risk to public safety.

(4) the prosecutor has requested an arrest warrant." MCR 6.102(D). See also MCL 764.1a(2).

"A summons must contain the same information as an arrest warrant, except that it should summon the accused to appear before a designated court at a stated time and place." MCR 6.102(C)(1).For more information on the initiation of a felony action, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 3.

1.34 General Criminal Penalties for Felony Traffic Offenses

The MVC and the Michigan Penal Code (MPC) contain general penalty provisions that apply in felony cases where criminal sanctions are not otherwise fixed by statute.

Under the MVC, "[a]ny person who is convicted of a violation of any of the provisions of this act declared to constitute a felony, unless a different penalty is expressly provided herein, shall be punished by imprisonment for not less than 1 year nor more than 5 years, or by a fine of not less than \$500.00 nor more than \$5,000.00, or by both such fine and imprisonment." MCL 257.902.

Under the MPC, if the criminal penalty is not otherwise fixed by statute, a person convicted of a crime declared by the state of Michigan to be a felony "is guilty of a felony punishable by imprisonment for not more than 4 years or a fine of not more than \$5,000.00 dollars, or both." MCL 750.503.

A conflict between the penal provisions of the MVC and the Code of Criminal Procedure must be resolved in favor of the more specific MVC provisions. See *Wayne Co Prosecutor v Wayne Circuit Judge*, 154 Mich App 216, 221-222 (1986).

1.35 Felony Sentencing

"[T]he legislative sentencing guidelines are advisory in every case, regardless of whether the case actually involves judicial fact-finding." *People v Rice*, 318 Mich App 688, 692 (2017); MCL 769.34(2)-(3). See also *People v Steanhouse*, 500 Mich 453, 466 (2017).

When sentencing a defendant, the trial court's objective is to tailor a penalty that is appropriate to the seriousness of the offense and the criminal history of the offender. *People v Rice (On Remand),* 235 Mich App 429, 445 (1999). The "framework" of an appropriate sentence consists of four basic considerations:

- the likelihood or potential that the offender could be reformed;
- the need to protect society;
- the penalty or consequence appropriate to the offender's conduct; and
- the goal of deterring others from similar conduct.

Rice, 235 Mich App at 446, citing *People v Snow*, 386 Mich 586, 592 (1972).

For more information about felony sentencing, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*.

A. Calculating the Minimum Sentence Range

While the legislative sentencing guidelines are advisory, "a sentencing court must determine the applicable guidelines range and take it into account when imposing a sentence." Lockridge, 498 Mich at 365. The recommended minimum sentence range for an offense to which the sentencing guidelines apply is determined by scoring the appropriate offense variables (OVs) and prior record variables (PRVs) for a specific conviction. MCL 777.21. All felony offenses to which the sentencing guidelines apply fall into one of six offense categories called crime groups,⁶⁶ and each crime group is further organized into crime classes⁶⁷ that indicate the severity of the offense. See MCL 777.5 and MCL 777.21(1)(c). An offense's crime class determines which sentencing grid must be used for determination of the offender's recommended minimum sentence range. See, e.g., MCL 777.61 to MCL 777.69. The crime group an offense falls into dictates which OVs must be scored for that offense and how those variables must be scored.⁶⁸ People v Bonilla-Machado, 489 Mich 412, 422 (2011). For cases involving vehicles, OVs 17 (degree of negligence exhibited) and 18 (operator ability affected by alcohol/drugs) contain offense characteristics specific to offenses involving "the operation of a vehicle, vessel, ORV, snowmobile, aircraft, or locomotive." MCL 777.22(1); MCL 777.22(5).⁶⁹

The rule of *Apprendi*, 530 US at 490 ("[o]ther than the fact of a prior conviction, any fact that increases the penalty for a crime beyond the prescribed statutory maximum must be submitted to a jury, and proved beyond a reasonable doubt[]"), does not apply to prior convictions and therefore presumably does not implicate the scoring of *prior record* variables under Michigan's sentencing

⁶⁶There are six crime groups: crimes against a person, crimes against property, crimes involving a controlled substance, crimes against public order, crimes against public trust, and crimes against public safety. MCL 777.5(a)-(f).

⁶⁷An offense's crime class is designated by the letters A through H and M2 (second-degree murder). The crime class determines which sentencing grid applies to the sentencing offense. MCL 777.21(1)(c).

⁶⁸In contrast to OVs, PRVs are scored based on the severity of prior felony convictions, and are scored for every offense regardless of the offense's crime group or class. *People v Peltola*, 489 Mich 174, 187 (2011). See the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 2, for detailed information about PRVs.

⁶⁹MCL 777.22 sets forth a complete list of the OVs that must be scored for each crime group. For a complete discussion of the OVs applicable to a particular crime, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 2.

guidelines. See *Alleyne*, 570 US at 111 n 1 (noting that "[i]n *Almendarez–Torres v United States*, [523 US 224 (1998)], [the United States Supreme Court] recognized a narrow exception to [the] general rule [of *Apprendi*] for the fact of a prior conviction[;]" the *Alleyne* Court declined to revisit *Almendarez-Torres* "[b]ecause the parties [did] not contest that decision's vitality[]"); see also, generally, *Lockridge*, 498 Mich at 370 n 12.

B. Departures

"A court may depart from the appropriate sentence range established under the sentencing guidelines set forth in [MCL 777.1 et seq.] if the departure is reasonable and the court states on the record the reasons for the departure." MCL 769.34(3). "A sentence that departs from the applicable guidelines range will be reviewed by an appellate court for reasonableness[, and] . . . [r]esentencing will be required when a sentence is determined to be unreasonable." People v Lockridge, 498 Mich 358, 392 (2015). "[S]entencing courts must justify the sentence imposed in order to facilitate appellate review." Lockridge, 498 Mich at 392. Appellate courts review the reasonableness of a sentence for an abuse of discretion "informed by the 'principle of proportionality' standard" set forth in *People v* Milbourn, 435 Mich 630, 636 (1990). People v Steanhouse, 500 Mich 453, 476 (2017). The principle of proportionality requires "sentences imposed by the trial court to be proportionate to the seriousness of the circumstances surrounding the offense and the offender." Id. at 474, quoting Milbourn, 435 Mich at 636. See also Steanhouse, 500 Mich at 460, 472, quoting Milbourn, 435 Mich at 661 ("the key test is whether the sentence is proportionate to the seriousness of the matter, not whether it departs from or adheres to the guidelines' recommended range"').

1.36 Minimum State Costs for Felony Traffic Offenses

If a court orders a defendant convicted of a felony to pay any combination of a fine, costs, or assessments, the court must order the defendant to pay minimum state costs of not less than \$68.00. MCL 769.1j(1)(a). The costs imposed under MCL 769.1j(1)(a) constitute a tax, and this tax does not violate the separation of powers requirement under Const 1963, art 3, § 2 or the requirement of Const 1963, art 4, § 32 that "[e]very law which imposes, continues or revives a tax shall distinctly state the tax." *People v Shenoskey*, 320 Mich App 80, 83-84 (2017) (applying the analysis of MCL 769.1k(1)(b)(*iii*) in *People v Cameron*, 319 Mich App 215 (2017) to MCL 769.1j(1)(a) because the statutes are "closely related"). Payment of the state minimum costs must be included as a condition of the defendant's probation. MCL 771.3(1)(g); MCL 769.1j(3). "The court shall not sentence a defendant to a term of incarceration, nor revoke

probation, for failure to comply with an order to pay money unless the court finds, on the record, that the defendant is able to comply with the order without manifest hardship and that the defendant has not made a good-faith effort to comply with the order." MCR 6.425(D)(3)(a). MCR 6.425(D)(3) also addresses payment alternatives and offers guidance for determining manifest hardship. For a detailed discussion of MCR 6.425(D)(3), see the Michigan Judicial Institute's *Criminal Proceedings Benchbook Vol. 2*, Chapter 8.

1.37 Costs⁷⁰

MCL 769.1k(1)(b)(*ii*) provides that, at the time of sentencing or a delay in sentencing or entry of a deferred judgment of guilt, a court may impose "[a]ny cost in addition to the minimum state cost^[71] set forth in [MCL 769.1k(1)(a)[]" on a defendant convicted by plea (guilty or nolo contendere) or found guilty by the court after a hearing or trial. Effective October 17, 2014, 2014 PA 352 amended MCL 769.1k in response to the Michigan Supreme Court's holding in People v Cunningham (Cunningham II), 496 Mich 145 (2014), rev'g 301 Mich App 218 (2013) and overruling People v Sanders (Robert) (After Remand), 298 Mich App 105 (2012), and People v Sanders (Robert), 296 Mich App 710 (2012).⁷² In Cunningham II, the Court held that MCL 769.1k(1)(b) which, at the time, provided for the imposition of "[a]ny cost in addition to the minimum state cost"-did "not provide courts with the independent authority to impose 'any cost[;]" rather, it "provide[d] courts with the authority to impose only those costs that the Legislature has separately authorized by statute." Cunningham II, 496 Mich at 147, 158 (concluding that "[t]he circuit court erred when it relied on [former] MCL 769.1k(1)(b)(ii) as independent authority to impose \$1,000 in court costs[]"). 2014 PA 352, in addition to making other revisions to MCL 769.1k, added MCL 769.1k(1)(b)(iii) to provide for the imposition of "any cost reasonably related to the actual costs incurred by the trial court[.]" The amendments effectuated by 2014 PA 352 "appl[y] to all fines, costs, and assessments ordered or assessed under . . . MCL 769.1k[] before June 18, 2014, and after [October 17, 2014]." 2014 PA 352, enacting section 1 (emphasis supplied).

⁷⁰Note that "[t]he court shall not sentence a defendant to a term of incarceration, nor revoke probation, for failure to comply with an order to pay money unless the court finds, on the record, that the defendant is able to comply with the order without manifest hardship and that the defendant has not made a good-faith effort to comply with the order." MCR 6.425(D)(3)(a). MCR 6.425(D)(3) also addresses payment alternatives and offers guidance for determining manifest hardship. For a detailed discussion of MCR 6.425(D)(3), see the Michigan Judicial Institute's *Criminal Proceedings Benchbook Vol. 2*, Chapter 8.

⁷¹ See the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol 2*, Chapter 9 for a discussion of minimum state cost.

⁷²The amended version of MCL 769.1k does not violate a defendant's due process or equal protection rights, nor does it violate the constitutional prohibition on ex post facto punishments or the principle of separation of powers. *People v Konopka*, 309 Mich App 345, 365, 367-70, 376 (2015).

"MCL 769.1k(1)(b)(*iii*) independently authorizes the imposition of costs in addition to those costs authorized by the statute for the sentencing offense[,]" and "[a] trial court possesses the authority, pursuant to MCL 769.1k, as amended by 2014 PA 352, to order court costs[.]" *People v Konopka*, 309 Mich App 345, 350, 358 (2015). "However, although the costs imposed . . . need not be separately calculated, . . . the trial court [must] . . . establish a factual basis[]" demonstrating that "the court costs imposed [are] 'reasonably related to the actual costs incurred by the trial court[.]"' *Konopka*, 309 Mich App at 359, quoting MCL 769.1k(1)(b)(*iii*).

MCL 769.34(6) addresses the sentencing guidelines and the duties of the court when sentencing, and it generally authorizes the court to order court costs ("As part of the sentence, the court may also order the defendant to pay any combination of a fine, costs, or applicable assessments."). However, "MCL 769.34(6) allows courts to impose only those costs or fines that the Legislature has separately authorized by statute[]" and "does not provide courts with the independent authority to impose any fine or cost." *Cunningham II*, 496 Mich at 158 n 11.⁷³

1.38 Felony in Which a Motor Vehicle Was Used

The MVC defines a *felony in which a motor vehicle was used* as

"a felony during the commission of which the individual operated a motor vehicle and while operating the vehicle presented real or potential harm to individuals or property and 1 or more of the following circumstances existed:

(a) The vehicle was used as an instrument of the felony.

(b) The vehicle was used to transport a victim of the felony.

(c) The vehicle was used to flee the scene of the felony.

(d) The vehicle was necessary for the commission of the felony." MCL 257.732(6). (Note: MCL 257.319(2)(c) and MCL 257.303(7) contain substantially similar definitions.)

The MVC mandates license suspension for one year upon conviction of any offense categorized as a *felony in which a motor vehicle was used*. MCL 257.319(2)(c). The MVC further includes a *felony in which a motor vehicle was used* in the list of specified offenses for which a combination of two or

⁷³For a more in-depth discussion of costs, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 8.

more convictions within 7 years requires license revocation and denial. MCL 257.303(2)(b).

"If a person is charged with a felony in which a motor vehicle was used, other than a felony specified in [MCL 257.732(4) or MCL 257.319], the prosecuting attorney shall include the following statement on the complaint and information filed in district or circuit court:

'You are charged with the commission of a felony in which a motor vehicle was used. If you are convicted and the judge finds that the conviction is for a felony in which a motor vehicle was used, as defined in section 319 of the Michigan vehicle code, 1949 PA 300, MCL 257.319, your driver's license shall be suspended by the secretary of state.'." MCL 257.732(7).

A substantially similar notice is required in juvenile proceedings when a juvenile is accused of an act, the nature of which constitutes a felony in which a motor vehicle was used, other than a felony specified in MCL 257.732(4) or MCL 257.319. MCL 257.732(8).

"If the court determines as part of the sentence or disposition that the felony for which the person was convicted or adjudicated and with respect to which notice was given under [MCL 257.732(7) or MCL 257.732(8)] is a felony in which a motor vehicle was used, the clerk of the court shall forward an abstract of the court record of that conviction to the secretary of state." MCL 257.732(9).

Felonies categorized as a *felony in which a motor vehicle was used* are not separate offenses that are subject to additional criminal penalties; rather, criminal penalties in these cases are imposed according to the applicable penal statute. The *felony in which a motor vehicle was used* categorization is a judicial determination made after conviction, MCL 257.732(9), that results in the imposition of licensing sanctions pursuant to MCL 257.319(2)(c) (suspension of the defendant's driver's license for one year for a first offense), and MCL 257.303(2)(b) (revocation and denial of the defendant's driver's license upon conviction within seven years of two *felonies in which a motor vehicle was used*, or a combination of convictions including one *felony in which a motor vehicle was used* and another enumerated offense).⁷⁴ Further, MCL 257.320a(1)(a) imposes 6 points for a felony resulting from the operation of a motor vehicle.

⁷⁴ Note that revocation and denial under MCL 257.303(2)(b) does not necessarily require a conviction of a *felony in which a motor vehicle was used*; rather, revocation will occur if a defendant is twice convicted within seven years of *any* of the offenses listed in that statute. See Section 1.43 (license denial) and Section 1.44 (license revocation) for more information on licensing sanctions under MCL 257.303.

However, certain traffic offenses that are likely to meet the definition of a *felony in which a motor vehicle was used* will not be subject to the sanctions associated with the designation because the secretary of state has already coded many traffic offenses for specific licensing sanctions.⁷⁵ See the Secretary of State's Offense Code Index for Traffic Violations (setting forth the offense code numbers for particular offenses, including a separate code for a *felony in which a motor vehicle was used*, and designating particularized licensing sanctions for each offense on the basis of its code).

Part D—General Overview of Licensing Sanctions for Civil Infractions, Misdemeanor, and Felony Traffic Offenses

1.39 Authority

"The Secretary of State has the statutory power to review driving records and determine sanctions." *People v Greenlee*, 133 Mich App 734, 737 (1984). "[I]t [is] not within the jurisdiction of the district court to place a civil defendant on probation and restrict his [or her] driving privileges." *Id.*

Accordingly, Part D addresses the secretary of state's authority to impose points, assess driver responsibility fees, impose license suspensions, revocations, and denials, and require vehicle immobilization or forfeiture.

1.40 Multiple Sanctions for the Same Offense

License sanctions imposed by more than one entity for the same offense run concurrently. MCL 257.303a.

⁷⁵For example, the following offenses may be committed in a fashion that satisfies the statutory definition of a *felony in which a motor vehicle was used*, but have separate offense codes and are subject to license revocation and denial, not license suspension: MCL 257.602a (fleeing and eluding); MCL 257.653a (failure to use due care and caution causing injury or death to emergency response personnel); MCL 257.617 (leaving the scene of an accident); MCL 257.601b (moving violation causing death in a work or school bus zone); MCL 257.601c (moving violation causing death of operator of implement of husbandry); MCL 257.904 (operating a motor vehicle while unlicensed); MCL 257.626 (reckless driving causing serious impairment of bodily function or death); and MCL 750.321 (manslaughter). All of these offenses are discussed in Chapter 8. The secretary of state code for a *felony in which a motor vehicle was used* is 1430, the above offenses are assigned different, individual codes that do not include the felony in which a motor vehicle was used code. The code for each offense is set forth in the Offense Code Index for Traffic Violations.

"Except as otherwise provided in [the MVC], the suspension, revocation, denial, disqualification, or cancellation of an operator's license, chauffeur's license, or commercial driver license by another state or the United States shall run concurrently with a suspension, revocation, denial, disqualification, or cancellation of an operator's license, chauffeur's license, or commercial driver license by this state that is imposed for the same offense." MCL 257.303a.

1.41 Abstracts of Convictions

"Each municipal judge and each clerk of a court of record shall keep a full record of every case in which an individual is charged with or cited for a violation of [the MVC] or a local ordinance substantially corresponding to [the MVC] regulating the operation of vehicles on highways[.]" MCL 257.732(1).

"Except as provided in [MCL 257.732(16)], the municipal judge or clerk of the court of record shall prepare and forward to the secretary of state an abstract of the court record as follows:

(a) Not more than 5 days after a conviction, forfeiture of bail, or entry of a civil infraction determination or default judgment upon a charge of or citation for violating or attempting to violate [the MVC] or a local ordinance substantially corresponding to [the MVC] regulating the operation of vehicles on highways.

(b) Immediately for each case charging a violation of [MCL 257.625(1) or MCL 257.625(3)-(8)] or [MCL 257.625m] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8) or MCL 257.625m] in which the charge is dismissed or the defendant is acquitted." MCL 257.732(1)(a)-(b).⁷⁶

"The abstract or report required under [MCL 257.732] must be made upon a form furnished by the secretary of state. An abstract must be certified by signature, stamp, or facsimile signature of the individual required to prepare the abstract as correct. An abstract or report must include all of the following:

(a) The name, address, and date of birth of the individual charged or cited.

⁷⁶ MCL 257.732(1)(c) is not included here as it does not pertain to violations of the MVC.

(b) The number of the individual's operator's . . . license, if any.

(c) The date and nature of the violation.

(d) The type of vehicle driven at the time of the violation[.]

(e) The date of the conviction, finding, forfeiture, judgment, or civil infraction determination.

(f) Whether bail was forfeited.

(g) Any license restriction, suspension, or denial ordered by the court as provided by law.

(h) The vehicle identification number and registration plate number of all vehicles that are ordered immobilized or forfeited.

(i) Other information considered necessary to the secretary of state." MCL 257.732(3).

Certain offenses do not require submission of an abstract. See MCL 257.732(16). "Except for controlled substances offenses described in [MCL 257.732(4)], the court shall not submit, and the secretary of state shall discard and not enter on the master driving record, an abstract for a conviction or civil infraction determination for any of the following violations:

(a) The parking or standing of a vehicle.

(b) A nonmoving violation that is not the basis for the secretary of state's suspension, revocation, or denial of an operator's . . . license.^[77]

(c) A violation of [MCL 257.201 *et seq.* (administration, registration, certificate of title)] that is not the basis for the secretary of state's suspension, revocation, or denial of an operator's . . . license.^[78]

(d) A pedestrian, passenger, or bicycle violation, other than a violation of [MCL 436.1703(1) or MCL 436.1703(2) (minor in possession)], or a local ordinance substantially corresponding to [MCL 436.1703(1) or MCL 436.1703(2)], or

 ⁷⁷ MCL 257.732(18) requires the secretary of state to inform the trial courts "of the nonmoving violations.
 ... that are used by the secretary of state as the basis for the suspension, restriction, revocation, or denial of an operator's . . . license."

⁷⁸ MCL 257.732(18) requires the secretary of state to inform the trial courts "of the . . . violations of [MCL 257.201 *et seq.*,] that are used by the secretary of state as the basis for the suspension, restriction, revocation, or denial of an operator's . . . license."

[MCL 257.624a or MCL 257.624b (transporting or possessing open alcohol)] or a local ordinance substantially corresponding to [MCL 257.624a or MCL 257.624b].

(e) A violation of [MCL 257.710e (safety belt requirements)] or a local ordinance substantially corresponding to [MCL 257.710e].

(f) A violation of [MCL 257.328(1) (proof of insurance)] if, before the appearance date on the citation, the individual submits proof to the court that the motor vehicle had insurance meeting the requirements of . . . MCL 500.3101 and [MCL 500.3102], at the time the citation was issued. Insurance obtained subsequent to the time of the violation does not make the violation an exception under this subsection.

* * *

(h) A violation of [MCL 257.311 (operator's license in immediate possession)] if the individual was driving a noncommercial vehicle and, before the court appearance date or the date fines are to be paid, the individual submits proof to the court that the individual held a valid driver license on the date the citation was issued." MCL 257.732(16).

"Except as provided in [the MVC] and notwithstanding any other provision of law, a court shall not order expunction of any violation reportable to the secretary of state under [MCL 257.732]." MCL 257.732(22).

Upon receipt of an abstract, the secretary of state must determine an individual's eligibility to attend a basic driver improvement course. MCL 257.320d.⁷⁹

1.42 Points

Except as otherwise provided in MCL 257.320a and MCL 257.629c, "[w]ithin 5 days after receipt of a properly prepared abstract from a court of [Michigan] or another state, the secretary of state shall record the date of conviction, civil infraction determination, or probate court disposition, and the number of points for each, based on the [formula set out in MCL 257.320a(1)]." MCL 257.320a(1).⁸⁰

"If more than 1 conviction, civil infraction determination, or probate court disposition results from the same incident, points must be entered

⁷⁹ See Section 1.51 for more information on the basic driver improvement course requirements.

only for the violation that receives the highest number of points under [MCL 257.320a]." MCL 257.320a(5).

"[T]he court shall not submit, and the secretary of state shall discard and not enter on the master driving record, an abstract for a conviction . . . for . . . [a] nonmoving violation that is not the basis for the secretary of state's suspension, revocation, or denial of an operator's . . . license." MCL 257.732(16)(b). Similarly, "[t]he court shall not submit, and the secretary of state shall discard and not enter on the master driving record, an abstract for a conviction . . . for . . . [a] violation of [MCL 257.201 *et seq.*] that is not the basis for the secretary of state's suspension, revocation, or denial of an operator's license." MCL 257.732(16)(c).

Further, points must not be entered for a violation of MCL 257.310e(14); MCL 257.311; MCL 257.602c; MCL 257.625m; MCL 257.658; MCL 257.710d; MCL 257.717; MCL 257.719; MCL 257.719a; or MCL 257.723. MCL 257.320a(2). Points must also not be entered for bond forfeitures, MCL 257.320a(3), overweight loads or defective equipment, MCL 257.320a(4), or when a conviction results solely because of the failure of a Michigan driver to appear in another state to contest a violation committed in another state that would be a civil infraction if committed in Michigan, MCL 257.320a(9).

1.43 License Denial

MCL 257.303(1) lists conditions that require the secretary of state to deny issuance of an operator's license. Although most of the enumerated circumstances are outside the scope of this benchbook, a person's application for a license must be denied if the individual's "license is suspended, revoked, denied, or canceled in any state. If the suspension, revocation, denial, or cancellation is not from the jurisdiction that issued the last license to the individual, the secretary of state may issue a license after the expiration of 5 years from the effective date of the most recent suspension, revocation, denial, or cancellation." MCL 257.303(1)(c). See also MCL 257.303(2) ("[o]n receiving the appropriate records of conviction, the secretary of state shall... deny issuance of an operator's ... license") In addition, certain individuals with certain enumerated convictions, juvenile dispositions, or civil infraction determinations must also be denied an operator's license. See MCL 257.303(1)(h)-(j).

⁸⁰For the exact number of points to be imposed by the secretary of state for a specific traffic offense, see the section detailing that specific traffic offense in Chapters 3 and 4 (civil infractions), Chapters 5 and 6 (misdemeanors), and Chapters 7 and 8 (felonies). In addition, a person eligible to take a basic driver improvement course may be exempt from having points entered against him or her. See Section 1.51 for more information.

Generally,⁸¹ the secretary of state must not issue a license to an individual whose license was denied under MCL 257.303(2) "until all of the following occur, as applicable:

(a) The later of the following

(*i*) The expiration of not less than 1 year after the license was . . . denied.

(*ii*) The expiration of not less than 5 years after the date of a subsequent revocation or denial occurring within 7 years after the date of any prior revocation or denial.

(b) For a denial under [MCL 257.303(2)(a), MCL 257.303(2)(b), MCL 257.303(2)(c), and MCL 257.303(2)(g)], the individual rebuts by clear and convincing evidence the presumption that results from the prima facie evidence that the individual is a habitual offender. The convictions that resulted in the revocation and denial constitute prima facie evidence that the individual is a habitual offender.

(c) The individual meets the requirements of the department." MCL 257.303(4).

"Multiple convictions or civil infraction determinations that result from the same incident must be treated as a single violation for purposes of [license] denial[.]" MCL 257.303(6).

1.44 License Revocation

"On receiving the appropriate records of conviction, the secretary of state shall revoke the operator's . . . license of an individual and deny issuance of an operator's . . . license to an individual" who has a conviction or a combination of convictions for the offenses specified in the statute. MCL 257.303(2).

Generally,⁸² the secretary of state must not issue a license to a person whose license was revoked under the MVC or revoked and denied under MCL 257.303(2) "until all of the following occur, as applicable:

(a) The later of the following

⁸¹ MCL 257.304 provides for the issuance of a restricted license for certain drunk/intoxicated driving offenses. See Section 9.2(H) for more information.

⁸² MCL 257.304 provides for the issuance of a restricted license for certain drunk/intoxicated driving offenses. See Section 9.2(H) for more information.

(*i*) The expiration of not less than 1 year after the license was revoked or denied.

(*ii*) The expiration of not less than 5 years after the date of a subsequent revocation or denial occurring within 7 years after the date of any prior revocation or denial.

(b) For a denial under [MCL 257.303(2)(a), MCL 257.303(2)(b), MCL 257.303(2)(c), and MCL 257.303(2)(g)], the individual rebuts by clear and convincing evidence the presumption that results from the prima facie evidence that the individual is a habitual offender. The convictions that resulted in the revocation and denial constitute prima facie evidence that the individual is a habitual offender.

(c) The individual meets the requirements of the department." MCL 257.303(4).

"Multiple convictions . . . that result from the same incident must be treated as a single violation for purposes of . . . [license] revocation[.]" MCL 257.303(7).

License revocation only applies to misdemeanor and felony traffic offenses and is discussed in conjunction with the applicable offenses.

1.45 License Suspension

A. Mandatory Suspension Under § 319

"The secretary of state shall immediately suspend an individual's license as provided in [MCL 257.319] on receiving a record of the individual's conviction for a crime described in [MCL 257.319], whether the conviction is under a law of [Michigan], a local ordinance that substantially corresponds to a law of [Michigan], a law of another state that substantially corresponds to a law of [Michigan], or, beginning October 31, 2010, a law of the United States that substantially corresponds to a law of [Michigan]." MCL 257.319(1). For information regarding when license suspension applies under MCL 257.319, see the section on sanctions for each offense(s) in Chapters 3 and 4 (civil infractions), Chapters 5 and 6 (misdemeanors), and Chapters 7 and 8 (felonies).

"If the secretary of state receives records of more than 1 conviction of an individual that results from the same incident, a suspension must be imposed only for the violation to which the longest period of suspension applies[.]" MCL 257.319(12). Moreover, the secretary of state may waive any suspension "if the individual submits proof that a court in another state revoked, suspended, or restricted his or her license for a period equal to or greater than the period of a restriction, suspension, or revocation prescribed under [the MVC] for the violation and that the revocation, suspension, or restriction was served for the violation[.]" MCL 257.319(13). The secretary of state may also issue a restricted license under these circumstances. *Id.* However, "[t]he secretary of state shall not issue a restricted license to an individual whose license is suspended under [MCL 257.319] unless a restricted license is authorized under [MCL 257.319] and the individual is otherwise eligible for a license." MCL 257.319(14). For more information on license denial see Section 1.43. See also MCL 257.319(15) and MCL 257.319(16) for the circumstances under which a restricted license may be issued.

B. Mandatory Suspension for Failure to Answer Citation/ Notice to Appear or Failure to Comply with Order/ Judgment

Failure to answer a citation or notice to appear or failure to comply with an order or judgment may result in a mandatory license suspension. See MCL 257.321a(1)-(3); MCR 4.101(G)(2)(b).

1. Notice to Individual of Possible Suspension

"Except as provided in [MCL 257.321a(2)⁸³], 28 days or more after an individual fails to answer a citation, or a notice to appear in court for a violation for which license suspension is allowed under [the MVC] or a local ordinance that substantially corresponds to a violation for which license suspension is allowed under [the MVC], or fails to comply with an order or judgment of the court for a violation for which license suspension is allowed under [the MVC], including, but not limited to, paying all fines, costs, fees, and assessments, the court shall give notice by mail at the last known address of the individual that if the individual fails to appear or fails to comply with the order or judgment within 14 days after the notice is issued, the secretary of state shall suspend the individual's operator's . . . license." MCL 257.321a(1).

Pursuant to MCL 257.321a(2), "[i]f an individual is charged with, or convicted of, a violation of [MCL 257.625, MCL 257.626], any driving violation under [the Michigan Vehicle Code] that causes injury, death, or serious impairment of a body function of another individual, a serious offense involving a motor vehicle, or a local ordinance that

⁸³ MCL 257.321a(2) applies to certain criminal offenses involving alcohol, drugs, and/or intoxicating substances.

substantially corresponds to [MCL 257.625(1)-(3), MCL 257.625(6), MCL 257.625(8), or MCL 257.626], and the individual fails to answer a citation or a notice to appear in court, or a notice to appear for any matter pending, or fails to comply with an order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, the court shall immediately give notice by firstclass mail sent to the individual's last known address that if the individual fails to appear within 7 days after the notice is issued, or fails to comply with the order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, within 14 days after the notice is issued, the secretary of state shall suspend the individual's operator's ... license. If the individual fails to appear within the 7-day period, or fails to comply with the order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, within the 14-day period, the court shall immediately inform the secretary of state who shall immediately suspend the individual's operators . . . license[.]"

2. Notice to Secretary of State to Suspend License

Generally, "[i]f the individual fails to appear or fails to comply with the order or judgment within the 14-day period, the court shall, within 14 days, inform the secretary of state, who shall immediately suspend the license of the individual."84 MCL 257.321a(1). However, if the violation is one listed in MCL 257.321a(2), and "the individual fails to appear within the 7day period, or fails to comply with the order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, within the 14-day period, the court shall immediately inform the secretary of state who immediately shall suspend the individual's operator's . . . license[.]"⁸⁵ Id.

3. Length of Suspension

"A suspension imposed under [MCL 257.321a(1) or MCL 257.321a(2)] remains in effect until both of the following occur:

(a) The secretary of state is notified by each court in which the individual failed to answer a citation or

⁸⁴ "The secretary of state shall immediately notify the individual of the suspension by regular mail at the individual's last known address." MCL 257.321a(1).

⁸⁵ The secretary of state must immediately "notify the individual of the suspension by first-class mail sent to the individual's last known address." MCL 257.321a(2).

notice to appear or failed to pay a fine or cost that the individual has answered that citation or notice to appear or paid that fine or cost.^[86]

(b) The individual has paid to the court a \$45.00 driver license clearance fee for each failure to answer a citation or failure to pay a fine or cost." MCL 257.321a(3).

The distribution of fees collected pursuant to MCL 257.321a(3)(b) is set forth by MCL 257.321a(5).

C. Mandatory Additional Like Periods of Suspension Under § 904

MCL 257.904(10)-(12) require the Secretary of State to impose *additional* periods of license suspension upon conviction or determination of responsibility for enumerated offenses:

"(10) On receiving a record of an individual's conviction or civil infraction determination for the unlawful operation of a motor vehicle or a moving violation reportable under [MCL 257.732] while the individual's operator's or chauffeur's license is suspended or revoked, the secretary of state immediately shall impose an additional like period of suspension or revocation. This subsection applies only if the violation occurs during a suspension of definite length or if the violation occurs before the individual is approved for a license following a revocation.

(11) On receiving a record of an individual's conviction or civil infraction determination for the unlawful operation of a motor vehicle or a moving violation reportable under [MCL 257.732] while the individual's operator's or chauffeur's license is indefinitely suspended or whose application for a license has been denied, the secretary of state immediately shall impose a 30-day period of suspension or denial.

(12) On receiving a record of the conviction, bond forfeiture, or a civil infraction determination of an individual for unlawful operation of a motor vehicle that requires a vehicle group designation while the

⁸⁶ "[T]he court shall give . . . the individual a copy of the information being transmitted to the secretary of state. Upon showing that copy, the individual must not be arrested or issued a citation for driving on a suspended license, on an expired license, or without a license[,] . . . even if the information being sent to the secretary of state has not yet been received or recorded by the department." MCL 257.321a(4).

designation is suspended or revoked under [MCL 257.319b], or while the individual is disqualified from operating a commercial motor vehicle by the United States Secretary of Transportation or under 49 USC 31301 to [49 USC 31317], the secretary of state immediately shall impose an additional like period of suspension or revocation. This subsection applies only if the violation occurs during a suspension of definite length or if the violation occurs before the individual is approved for a license following a revocation."

D. No Appeal to the Secretary of State

"Any period of suspension or restriction required under [MCL 257.319] is not subject to appeal to the secretary of state." MCL 257.319(21).

E. Improper Suspension

"If [a] defendant believe[s] that his license was improperly suspended, his [or her] action should [be] to petition for a hearing in circuit court for an order modifying or setting aside the suspension." *People v Glantz*, 124 Mich App 531, 533 (1983) (holding that "[a] collateral attack on the validity of the suspension of a driver's license in a case where the defendant is charged with driving while his [or her] license is suspended is improper").

1.46 Out-of-State Offenses and Nonresidents

"The secretary of state may suspend or revoke the license issued under [the MVC] upon receiving notice of the conviction of that person in another state of an offense in that state, or the determination of responsibility of that person in an administrative adjudication in another state for a violation in that state which, if committed in [Michigan], would be grounds for the suspension or revocation of the license of an operator[.]" MCL 257.318(1).

"Except as otherwise provided in [the MVC], the suspension, revocation, denial, disqualification, or cancellation of an operator's license . . . by another state or the United States shall run concurrently with a suspension, revocation, denial, disqualification, or cancellation of an operator's license . . . by [Michigan] that is imposed for the same offense." MCL 257.303a.

"The secretary of state may suspend, deny, or revoke the right of a nonresident to operate a motor vehicle in [Michigan] for a cause for which the license of a resident driver may be suspended, denied, or revoked." MCL 257.317(1).

1.47 Vehicle Immobilization

A. Circumstances Under Which Immobilization is Permitted

Under MCL 257.904d(1), vehicle immobilization may be a mandatory or discretionary sanction for a conviction of certain enumerated offenses. Similarly, immobilization is authorized if the person violates any of the enumerated offenses found in MCL 257.904d(2) "during a period of suspension, revocation, or denial[.]" Traffic offenses are discussed throughout the other chapters of this benchbook and will indicate if vehicle immobilization is available or required as a sanction.

"The court may order vehicle immobilization under [MCL 257.904d] under either of the following circumstances:

(a) The defendant is the owner, co-owner, lessee, or colessee of the vehicle operated during the violation.

(b) The owner, co-owner, lessee, or co-lessee knowingly permitted the vehicle to be operated in violation of [MCL 257.625(2) or MCL 257.904(2)] regardless of whether a conviction resulted." MCL 257.904d(4).

When vehicle immobilization applies, "[t]he defendant shall provide to the court the vehicle identification number and registration plate number of the vehicle involved in the violation." MCL 257.904d(3).

B. Exceptions

Certain violations and certain vehicles are exempt from vehicle immobilization. See MCL 257.904d(7), which states that "[MCL 257.904d] does not apply to any of the following:

(a) A suspension, revocation, or denial based on a violation of the support and parenting time enforcement act, . . . [MCL 552.601 to MCL 552.650].

(b) A vehicle that is registered in another state or that is a rental vehicle.

(c) A vehicle owned by the federal government, [Michigan], or a local unit of government of [Michigan].

(d) A vehicle not subject to registration under [MCL 257.216].

(e) Any of the following:

- (*i*) A violation of [MCL 257.201 *et seq*].
- (*ii*) A violation of [MCL 257.501 *et seq*].
- (*iii*) A violation for failure to change address.

(*iv*) A parking violation.

(*v*) A bad check violation.

(*vi*) An equipment violation.

(*vii*) A pedestrian, passenger, or bicycle violation, other than a violation of [MCL 436.1703(1) or MCL 436.1703(2)], or a local ordinance substantially corresponding to [MCL 436.1703(1) or MCL 436.1703(2)], or [MCL 257.624a or MCL 257.624b] or a local ordinance substantially corresponding to [MCL 257.624a or MCL 257.624b].

(*viii*) A violation of a local ordinance substantially corresponding to a violation described in [MCL 257.904d(7)(*i*)-(*vii*)." MCL 257.904d(7).

C. Period of Immobilization

The length of time a vehicle may be immobilized depends on the offense. See MCL 257.904d(1)-(2). For the specific vehicle immobilization periods under MCL 257.904d, see the section on sanctions for each offense(s) in Chapter 5 and Chapter 6 (misdemeanors) and Chapter 7 and Chapter 8 (felonies).

"If a defendant is ordered imprisoned for the violation for which immobilization is ordered, the period of immobilization shall begin at the end of the period of imprisonment." MCL 257.904d(6).

D. Manner of Immobilization

The execution of an immobilization order is governed by MCL 257.904e, which provides:

"(1) A court shall order a vehicle immobilized under [MCL 257.904d] by the use of any available technology approved by the court that locks the ignition, wheels, or steering of the vehicle or otherwise prevents any person from operating the vehicle or that prevents the defendant from operating the vehicle. If a vehicle is immobilized under this section, the court may order the vehicle stored at a location and in a manner considered appropriate by the court. The court may order the person convicted of violating [MCL 257.625] or a suspension, revocation, or denial under [MCL 257.904] to pay the cost of immobilizing and storing the vehicle.

(2) A vehicle subject to immobilization under this section may be sold during the period of immobilization, but shall not be sold to a person who is exempt from paying a use tax under [MCL 205.93(3)], without a court order.

(3) A defendant who is prohibited from operating a motor vehicle by vehicle immobilization shall not purchase, lease, or otherwise obtain a motor vehicle during the immobilization period.

(4) A person shall not remove, tamper with, or bypass or attempt to remove, tamper with, or bypass a device that he or she knows or has reason to know has been installed on a vehicle by court order for vehicle immobilization or operate or attempt to operate a vehicle that he or she knows or has reason to know has been ordered immobilized.

(5) A person who violates this section is guilty of a misdemeanor punishable by imprisonment for not more than 93 days or a fine of not more than \$100.00, or both.

(6) To the extent that a local ordinance regarding the storage or removal of vehicles conflicts with an order of immobilization issued by the court, the local ordinance is preempted.

(7) If a peace officer stops a vehicle that is being operated in violation of an immobilization order, the vehicle shall be impounded pending an order of a court of competent jurisdiction.

(8) The court shall require the defendant or a person who provides immobilization services to the court under this section to certify that a vehicle ordered immobilized by the court is immobilized as required."

E. Immobilization Suspension

"Except as otherwise provided in [MCL 257.904d(11) and MCL 257.904d(13)], an order required to be issued under [MCL 257.904d] shall not be suspended." MCL 257.904d(5).

"Beginning October 31, 2010, if the person obtains a restricted operator's...license from the secretary of state and an ignition interlock device is properly installed in the vehicle, the court shall suspend the immobilization order issued under [MCL 257.904d(1)(c)] for a conviction under [MCL 257.6251(2)]." MCL 257.904d(11).

"Vehicle immobilization under [MCL 257.904d] is subject to [MCL 257.304] if the defendant obtains a restricted license under [MCL 257.304]." MCL 257.904d(13).

F. Immobilization Reinstatement

"Beginning October 31, 2010, the court may reinstate vehicle immobilization issued under [MCL 257.904d(1)(c)] for a conviction under [MCL 257.625l(2)] if an ignition interlock device is tampered with, circumvented, or disabled, or if the person's restricted operator's . . . license is suspended or revoked." MCL 257.904d(12).

1.48 Vehicle Impoundment

Vehicle impoundment is a required sanction when a person is convicted of an offense punishable under MCL 257.904(1)(b) or MCL $257.904(1)(c)^{87}$ or a substantially corresponding local ordinance. MCL 257.904b(1).⁸⁸

⁸⁷Note that MCL 257.904 no longer has subdivision (1)(b) or (1)(c). 1994 PA 450 omitted MCL 257.904(1)(c) and 1998 PA 341 omitted all of the subdivisions from MCL 257.904(1). 1998 PA 341 included MCL 257.904(3)(a)-(b), which contained language very similar to what was formerly in MCL 257.904(1)(b)-(c). The former MCL 257.904(1)(b)-(c) stated that a person who violated MCL 257.904(1) was guilty of a misdemeanor punishable as follows: "(b) For a violation, other than a violation punishable under subdivision a), by imprisonment for not more than 90 days, or by a fine of not more than \$500.00, or both. Unless the vehicle was stolen or used with the permission of a person who did not knowingly permit an unlicensed driver to operate the vehicle, the registration plates of the vehicle shall be confiscated. (c) For a second or subsequent violation punishable under subdivision (b), by imprisonment for not more than 1 year, or a fine of not more than \$1,000.00, or both. Unless the vehicle was stolen, the registration plates of the vehicle shall be confiscated." Currently, MCL 257.904(3)(a)-(b) provide that a person that violates MCL 257.904(1)-(2) is guilty of a misdemeanor as follows: "(a) For a first violation, by imprisonment for not more than 93 days or a fine of not more than \$500.00, or both. Unless the vehicle was stolen or used with the permission of a person who did not knowingly permit an unlicensed driver to operate the vehicle, the registration plates of the vehicle must be canceled by the secretary of state on notification by a peace officer. (b) For a violation that occurs after a prior conviction, by imprisonment for not more than 1 year or a fine of not more than \$1,000.00, or both. Unless the vehicle was stolen, the registration plates of the vehicle shall be canceled by the secretary of state on notification by a peace officer."

Impoundment is also a sanction for certain ignition interlock violations, discussed in Section 9.12(B). See MCL 257.625*l*.

"An order for the impounding of a motor vehicle issued pursuant to [MCL 257.904b] is valid throughout the state. Any peace officer may execute the impoundment order. The order shall include the implied consent of the owner of the vehicle to the storage for insurance coverage purposes." MCL 257.904b(3).

"The owner of a motor vehicle impounded pursuant to [MCL 257.904b] is liable for expenses incurred in the removal and storage of the vehicle whether or not the vehicle is returned to him or her. The vehicle shall be returned to the owner only if the owner pays the expenses for removal and storage. If redemption is not made or the vehicle is not returned as provided in [MCL 257.904b] within 30 days after the time set in the impoundment order for return of the vehicle, the vehicle shall be considered an abandoned vehicle and disposed of as provided in [MCL 257.252a]. MCL 257.904b(4).

"[MCL 257.904b] does not affect the rights of a conditional vendor, chattel mortgagee, or lessor of a motor vehicle registered in the name of another person as owner who becomes subject to this act." MCL 257.904b(5).

1.49 Vehicle Forfeiture

Vehicle forfeiture may be required following a conviction of certain enumerated offenses.⁸⁹ MCL 257.625n(1). Forfeiture of the vehicle used in the offense will occur if the defendant owns the vehicle in whole or in part; if the defendant leases the vehicle, it will be returned to the lessor. *Id.* "The vehicle may be seized under a seizure order issued by the court having jurisdiction upon a showing of probable cause that the vehicle is subject to forfeiture or return to the lessor." MCL 257.625n(2). "The forfeiture of a vehicle is subject to the interest of the holder of a security interest who did not have prior knowledge of or consent to the violation." MCL 257.625n(3).

The prosecuting attorney may file a petition for the forfeiture or return of the vehicle within 14 days of a qualifying conviction. MCL 257.625n(4). Notice must be given to the defendant and his or her attorney, to all owners of the vehicle, and to any person holding a security interest in the

⁸⁸MCL 257.904b(2) is not discussed because it references a subdivision of MCL 257.904 that does not appear to exist in any form in the current version of MCL 257.904.

⁸⁹ Specific traffic offenses are discussed throughout this benchbook and will indicate if forfeiture is an authorized sanction. See Chapters 5 and 6 (misdemeanors), Chapters 7 and 8 (felonies), and Chapter 9 (violations of MCL 257.625).

vehicle that the court may require forfeiture or return of the vehicle. *Id.* The defendant, owner, lessee, or holder of a security interest may file a claim of interest in the vehicle with the court within 14 days after notice by the prosecuting attorney. MCL 257.625n(6). The court must hold a hearing within 21 days after the period for filing claims expires. *Id.* "In considering whether to order forfeiture, the court shall review the defendant's driving record to determine whether the defendant has multiple convictions under [MCL 257.625] or a local ordinance substantially corresponding to [MCL 257.625], or multiple suspensions, restrictions, or denials under [MCL 257.625] or multiple suspensions, restrictions, or denials under [MCL 257.625] or multiple suspensions, restrictions, or denials under [MCL 257.904], or both. If the defendant has multiple convictions under [MCL 257.904], or both, that factor shall weigh heavily in favor of forfeiture." MCL 257.625n(6).

If a vehicle is seized before disposition of the criminal proceedings, the defendant may move the court to require the seizing agency to file a lien against the vehicle and to return the vehicle to the owner or lessee pending disposition of the proceedings. MCL 257.625n(5). The court must hear the motion within seven days. *Id.* "If the defendant establishes at the hearing that he or she holds the legal title to the vehicle or that he or she has a leasehold interest and that it is necessary for him or her or a member of his or her family to use the vehicle pending the outcome of the forfeiture action, the court may order the seizing agency to return the vehicle to the owner or lessee. If the court orders the return of the vehicle to the owner or lessee, the court shall order the defendant to post a bond in an amount equal to the retail value of the vehicle." *Id.*

"The failure of the court or prosecutor to comply with any time limit specified in [MCL 257.625n] does not preclude the court from ordering forfeiture of a vehicle or its return to a lessor, unless the court finds that the owner or claimant suffered substantial prejudice as a result of that failure." MCL 257.625n(11).

"The forfeiture provisions of [MCL 257.625n] do not preclude the prosecuting attorney from pursuing a forfeiture proceeding under any other law of this state or a local ordinance substantially corresponding to [MCL 257.625n]." MCL 257.625n(12).

"A person who knowingly conceals, sells, gives away, or otherwise transfers or disposes of a vehicle with the intent to avoid forfeiture or return of the vehicle to the lessor under [MCL 257.625n] is guilty of a misdemeanor punishable by imprisonment for not more than 1 year or a fine of not more than \$1,000.00, or both." MCL 257.625n(10).

1.50 Registration Denial

Registration denial is required for multiple violations of MCL 257.625, MCL 257.625m, and MCL 257.904. MCL 257.219(1)(d) provides:

"The secretary of state shall refuse issuance of a registration or a transfer of registration [if] . . . (d) [a]t the time of the application, the operator's or chauffeur's license of the owner or co-owner or lessee or co-lessee is suspended, revoked, or denied, except for an applicant who has been issued a license under [MCL 257.304], or the operator has never been licensed by this state for a third or subsequent violation of [MCL 257.625] or [MCL 257.625m], a local ordinance substantially corresponding to [MCL 257.625 or MCL 257.625m], or a law of another state substantially corresponding to [MCL 257.625 or MCL 257.625m], or for a fourth or subsequent suspension or revocation under [MCL 257.904]."

1.51 Basic Driver Improvement Course

"[T]he secretary of state shall not enter the points corresponding to a moving violation committed in [Michigan] by an individual the secretary of state determines to be eligible under [MCL 257.320d] on the individual's driving record or make information concerning that violation available to any insurance company if the individual attends and successfully completes a basic driver improvement course under [MCL 257.320d] and an approved sponsor provides a certificate of successful completion of that course to the secretary of state not more than 60 days after the date on which the secretary of state notified the individual that the individual was eligible to take a basic driver improvement course. [MCL 257.320d(1)] does not apply to an individual who completes a court-ordered basic driver improvement course as described in [MCL 257.320d(17)⁹⁰]." MCL 257.320d(1).

A. Secretary of State Responsibilities

1. Receiving an Abstract

"The secretary of state shall determine if an individual is eligible under [MCL 257.320d(3)] to attend a basic driver

⁹⁰ "An individual who is ordered by a court to complete a basic driver improvement course shall take the basic driver improvement course whether or not the individual is eligible under [MCL 257.320d(3)-(4)]. The secretary of state shall enter the points required under [MCL 257.320a] for an individual who completes a court-ordered basic driver improvement course but is not otherwise eligible under [MCL 257.320d(3)-(4)]." MCL 257.320d(17).

improvement course upon receipt of an abstract of a moving violation." MCL 257.320d(2).

"If the secretary of state determines that an individual is eligible to attend a basic driver improvement course, the secretary of state shall do all of the following:

(a) Notify the individual of the individual's eligibility by first-class mail at the individual's last known address as indicated on the individual's operator's . . . license and inform the individual of the manner and time within which the individual is required to attend and complete a basic driver improvement course.

(b) Provide all eligible participants with information on how to access a list of approved sponsors and basic driver improvement course locations, including the secretary of state's website address and telephone number to call for more information." MCL 257.320d(2).

2. Enter Points

"If an approved sponsor does not provide notice of successful completion of the course by the individual within the time prescribed in [MCL 257.320d(1)], the secretary of state shall enter the points required under [MCL 257.320a]." MCL 257.320d(2)(c). "The secretary of state shall enter the points required under [MCL 257.320a] for an individual who completes a court-ordered basic driver improvement course but is not otherwise eligible under [MCL 257.320d(3)-(4)]." MCL 257.320d(17).

3. Database

"The secretary of state shall maintain a computerized database of the following:

(a) Individuals who have attended a basic driver improvement course.

(b) Individuals who have successfully completed a basic driver improvement course." MCL 257.320d(5).

"The database maintained under [MCL 257.320d(5)] must only be used for determining eligibility under [MCL 257.320d(3) and MCL 257.320d(4)]." MCL 257.320d(6). "The secretary of state shall only make the information contained in the database available to approved sponsors under [MCL 257.320d(10)]." MCL 257.320d(6). "Information in this database concerning an individual must be maintained for the life of that individual." *Id*.

B. Eligibility Requirements

"Except as provided in [MCL $257.320d(17)^{91}$], an individual is ineligible to take a basic driver improvement course if any of the following apply:

(a) The violation occurred while the individual was operating a commercial motor vehicle or was licensed as a commercial driver while operating a noncommercial motor vehicle.

(b) The violation is a criminal offense.

(c) The violation is a violation for which 4 or more points may be assessed under [MCL 257.320a].

(d) The violation is a violation of [MCL 257.626b (careless or negligent operation of a motor vehicle), MCL 257.627(6)⁹² (speeding in construction, survey, or work area), MCL 257.627a (speeding in school zone), or MCL 257.682 (failing to stop for school bus displaying flashing red lights)].

(e) The individual was cited for more than 1 moving violation arising from the same incident.

(f) The individual's license was suspended under [MCL 257.321a(1)] in connection with the violation.

(g) The individual previously successfully completed a basic driver improvement course.

(h) The individual has 3 or more points on the individual's driving record.

⁹¹ "An individual who is ordered by a court to complete a basic driver improvement course shall take the basic driver improvement course whether or not the individual is eligible under [MCL 257.320d(3)-(4)]. The secretary of state shall enter the points required under [MCL 257.320a] for an individual who completes a court-ordered basic driver improvement course but is not otherwise eligible under [MCL 257.320d(3)-(4)]." MCL 257.320d(17).

⁹²MCL 257.320d(3)(d) refers to MCL 257.627(9); however, effective January 5, 2017, MCL 257.627 was amended by 2016 PA 445 and § 627(9) was renumbered to be § 627(6). MCL 257.320d has not been amended to reflect the changes to MCL 257.627.

(*i*) The individual's operator's . . . license is restricted, suspended, or revoked, or the individual was not issued an operator's . . . license." MCL 257.320d(3).

"Except as provided in [MCL 257.320d(17)⁹³], an individual is not eligible to take a basic driver improvement course for a second or subsequent violation an individual receives within the time allowed under [MCL 257.320d(1)]." MCL 257.320d(4).

C. Cost of Course and Allocation of Cost

"An individual shall be charged a fee of not more than \$100.00 by an approved sponsor to participate in a basic driver improvement course and, if applicable, to obtain a certificate in a form as approved by the secretary of state demonstrating that the individual successfully completed the course." MCL 257.320d(7). "An approved sponsor shall remit a portion of the fee, as determined annually by the secretary of state, to cover the costs of implementing and administering this course program." *Id*.

"Fees remitted to the department under [MCL 257.320d(7)] by an approved sponsor must be credited to the basic driver improvement course fund created under [MCL 257.320d(9)]." MCL 257.320d(8).

⁹³"An individual who is ordered by a court to complete a basic driver improvement course shall take the basic driver improvement course whether or not the individual is eligible under [MCL 257.320d(3)-(4)]. The secretary of state shall enter the points required under [MCL 257.320a] for an individual who completes a court-ordered basic driver improvement course but is not otherwise eligible under [MCL 257.320d(3)-(4)]." MCL 257.320d(17).

Chapter 2: Juvenile Traffic Proceedings

Scope Note 2	-2
Jurisdiction Over Juveniles Accused of Committing a Criminal Traffic	2
5	
	-3
Venue and Transfer of Case 2	-3
Court Rules Applicable to Cases Under the Juvenile Code 2	-3
Investigative Stops 2	-4
Taking a Juvenile Into Custody 2	-4
Required Procedures for Traffic-Related Offenses Contained in the	
Michigan Penal Code 2	-5
Special Procedures Applicable in Juvenile Traffic Cases 2	-5
Required Procedures Following a Juvenile's Failure to Respond to a Moto	or
Vehicle Violation 2	-9
Restitution, Crime Victim's Rights Assessment, and Costs 2-7	11
Setting Aside Adjudications	11
Holmes Youthful Trainee Act 2-7	12
	Jurisdiction Over Juveniles Accused of Committing a Criminal Traffic Violation

2.1 Scope Note

This chapter discusses special procedures and rules applicable to juveniles in traffic proceedings. Many procedures and rules apply to both juveniles and adults in the same manner; this chapter addresses only issues specific to juvenile proceedings. See Chapter 1 for more information on generally applicable procedures and rules in traffic cases. See the Michigan Judicial Institute's *Juvenile Justice Benchbook* for a comprehensive discussion of juvenile proceedings.

2.2 Jurisdiction Over Juveniles Accused of Committing a Criminal Traffic Violation

The family division of the circuit court has "[e]xclusive original jurisdiction superior to and regardless of the jurisdiction of another court in proceedings concerning a juvenile under 18 years of age who[,]...[e]xcept as otherwise provided in [MCL 712A.2(a)(1)],... has violated any municipal ordinance or law of the state or of the United States." MCL 712A.2(a)(1). As used in subchapter 3.900 of the Michigan Court Rules (governing practice and procedure in cases filed under the Juvenile Code), a "[d]elinquency proceeding" is "a proceeding concerning an offense by a juvenile, as defined in MCR 3.903(B)(3)." MCR 3.903(A)(5). MCR 3.903(B)(3), in turn, defines "[o]ffense by a juvenile" as "an act that violates a criminal statute, a criminal ordinance, a *traffic law*, or a provision of MCL 712A.2(a) [(governing violations of law or ordinance, civil infractions, and status offenses)] or [MCL 712A.2(d) (governing wayward minors)]." (Emphasis supplied.)

District court or municipal court judges may be assigned to sit as family division judges to hear matters involving juveniles. See MCL 600.1517; MCL 600.225; MCR 8.110(C)(3)(g).

A. Continued Jurisdiction

"If the juvenile attains his or her eighteenth birthday after the filing of the petition, the court's jurisdiction shall continue beyond the juvenile's eighteenth birthday and the court may hear and dispose of the petition under [the Juvenile Code]." MCL 712A.11(4).

B. Age Calculation

The "birthday rule" applies in Michigan for purposes of age calculation. *People v Woolfolk*, 304 Mich App 450, 504 (2014), aff'd 497 Mich 23 (2014). Under the birthday rule, "a person attains a given age on the anniversary date of his or her birth." *Id.* at 464 (quotation and citation omitted).

2.3 Jurisdiction Over Juveniles Accused of Committing a Civil Infraction

MCL 257.741(5) specifically identifies district and municipal courts as the courts with jurisdiction over juvenile traffic civil infraction actions: "If the person cited [for a civil infraction] is a minor, that individual shall be permitted to appear in court or to admit responsibility for a civil infraction without the necessity of appointment of a guardian or next friend. The courts listed in [MCL 257.741(2)¹] shall have jurisdiction over the minor and may proceed in the same manner and in all respects as if that individual were an adult." See also MCL 600.8301(2); MCL 600.8703(2).

See Chapter 1 for more information on civil infraction proceedings.

2.4 Venue and Transfer of Case

Under the Juvenile Code and the Michigan Court Rules, venue is proper in the county where the offense occurred or where the juvenile is physically present. See MCL 712A.2(a); MCL 712A.2(d); MCR 3.926(A).

However, if a juvenile is brought before the family division in a county other than the county in which he or she resides, the court may, before a hearing and with the consent of the court in the juvenile's county of residence, enter an order transferring jurisdiction over the matter to the court of the county of residence.² MCL 712A.2(d); MCR 3.926(B). In addition, a change of venue to a different county may be properly ordered in certain circumstances. MCR 3.926(D).

2.5 Court Rules Applicable to Cases Under the Juvenile Code

MCR 3.901(A) states as follows:

"(1) The rules in . . . subchapter [3.900], in subchapter 1.100, and in subchapter 8.100 govern practice and procedure in the family division of the circuit court in all cases filed under the Juvenile Code.

¹ MCL 257.741(2) states that "[t]he following courts have jurisdiction over civil infraction actions: (a) [t]he district court[,] (b) [a]ny municipal court."

² For a more detailed discussion of venue and transfer in juvenile proceedings, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 2.

(2) Other Michigan Court Rules apply to juvenile cases in the family division of the circuit court only when . . . subchapter [3.900] specifically provides.

(3) The Michigan Rules of Evidence, except with regard to privileges, do not apply to proceedings under . . . subchapter [3.900], except where a rule in . . . subchapter [3.900] specifically so provides. . . . "

As it relates to juvenile traffic offenses, MCR 3.901-MCR 3.950 and MCR 3.991–MCR 3.993 apply. MCR 3.901(B)(1)-(2); MCR 3.903(B)(3).

MCR 3.902(A) states, in part, that these applicable court rules "are to be construed to secure fairness, flexibility, and simplicity. The court shall proceed in a manner that safeguards the rights and proper interests of the parties."

2.6 Investigative Stops

The federal and Michigan constitutions grant all persons the right to be secure against unreasonable searches and seizures. US Const, Am IV, and Const 1963, art 1, §11. The search and seizure protections of the Fourth Amendment to the federal constitution have been extended to minors. *New Jersey v TLO*, 469 US 325, 333 (1985). For a detailed discussion of investigative stops, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 11.

2.7 Taking a Juvenile Into Custody

"Any local police officer, sheriff or deputy sheriff, state police officer, county agent or probation officer of any court of record may, without the order of the court, immediately take into custody any child who is found violating any law or ordinance[.]" MCL 712A.14(1). The officer must "immediately attempt to notify the [juvenile's] parent or parents, guardian, or custodian[,]" and "[u]nless the [juvenile] requires immediate detention as provided for in [the Juvenile Code], the officer shall accept the written promise of the parent or parents, guardian, or custodian, to bring the child to the court at a fixed time[, and t]he [juvenile] shall then be released to the custody of the parent or parents, guardian, or custodian." *Id.* If the juvenile is not released under MCL 712A.14(1), the court must hold a preliminary hearing. See MCL 712A.14(2); MCR 3.935.

However, "if a child less than 18 years of age is arrested, with or without a warrant, the child must be taken immediately before the family division of circuit court of the county where the offense is alleged to have been committed, and the officer making the arrest shall immediately make and file, or cause to be made and filed, a petition against the child as provided in [the Juvenile Code, MCL 712A.1 *et seq.*]" MCL 764.27. See also MCL 257.727.

For more information about detaining a juvenile pending a preliminary hearing or taking a juvenile into custody upon arrest, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 3.

2.8 Required Procedures for Traffic-Related Offenses Contained in the Michigan Penal Code

Unless a case has been designated or waived to a court of general criminal jurisdiction, when a traffic violation contained in the Michigan Penal Code (MPC), MCL 750.1 *et seq.*, is alleged, the court should use the same procedures it would use in any delinquency case. See MCL 712A.2(a)(1) (jurisdiction of the court); MCL 712A.2d(2) (designated cases); MCL 712A.4 (traditional waiver); MCL 764.1f (automatic waiver) MCR 3.903(A)(5) (defining delinquency proceeding); MCR 3.903(B)(3) (defining offense by a juvenile). For a detailed discussion of the required procedures in cases involving juveniles, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*.

2.9 Special Procedures Applicable in Juvenile Traffic Cases

The Juvenile Code provides special procedures that apply when a juvenile is charged with a violation of the MVC or a substantially corresponding local ordinance. MCL 712A.2b states:

"When a juvenile is accused of an act that constitutes a violation of [the MVC], ... or a provision of an ordinance substantially corresponding to any provision of [the MVC], the following procedure applies, any other provision of [the Juvenile Code] notwithstanding:

(a) No petition shall be required, but the court may act upon a copy of the written notice to appear given the accused juvenile as required by . . . [MCL 257.728].^[3]

(b) The juvenile's parent or parents, guardian, or custodian may be required to attend a hearing conducted under this section when notified by the court, without additional service of process or delay.

³ Section 1.26 for a discussion of MCL 257.728.

However, the court may extend the time for that appearance.

(c) If after hearing the case the court finds the accusation to be true, the court may dispose of the case under [MCL 712A.18].^[4]

(d) Within 14 days after entry of a court order of disposition for a juvenile found to be within [the Juvenile Code], the court shall prepare and forward an abstract of the record of the court for the case in accordance with . . . [MCL 257.732].^[5]

(e) This section does not limit the court's discretion to restrict the driving privileges of a juvenile as a term or condition of probation."

Several procedural protections afforded by the Juvenile Code are omitted from the procedures applicable to juveniles charged with violations of the MVC or substantially corresponding ordinances. For example, no provision is made for the appointment of counsel as required by MCL 712A.17c(1)-(3), and formal notice is not required as it is in MCL 712A.12 and MCL 712A.13 and related court rules.

However, the consent and formal calendars, which are procedural mechanisms provided by MCR 3.932(C)-(D), may be used in cases in which the juvenile is charged with a criminal traffic offense. Although some consent calendar procedures are discussed in this chapter as they specifically relate to cases involving MVC violations, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 4, for more information about consent calendar and formal calendar procedures.

A. Commencing Proceedings

"Any request for court action against a juvenile must be by written petition." MCR 3.931(A). However, an officer may issue a citation or appearance ticket to initiate delinquency proceedings in misdemeanor traffic cases. See MCR 3.931(C). Although the citation or appearance ticket must be treated as a petition by the court, "it may not serve as a basis for pretrial detention." MCR 3.931(C)(2). Accordingly, a petition should be filed for any felony traffic cases or if court-ordered pretrial detention is requested. See MCR 3.931(C)(2); MCR 3.933(B). "However, a juvenile may not be detained in a secure facility pending hearing unless the court has considered the results of

⁴ MCL 712A.18 sets forth several possible orders of disposition available to the court. See the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 10, for a discussion of MCL 712A.18.

⁵ See Section 1.41 for more information on abstracting procedures.

a detention screening tool conducted on the juvenile under MCR 3.907." MCR 3.933(B). For more information on a detention screening tool, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 4. The petitioner must complete and file a case inventory of known pending or resolved family division or tribal court cases that involve family member(s) of the person named in the petition. MCR 3.931(A). For more information on petition requirements see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 6.

B. Notice

If the court wants a parent, guardian, or custodian to appear for a hearing in a juvenile's traffic case, that individual must be notified. See MCL 712A.2b(b). However, the procedures outlined in MCL 712A.12 (examination of child; hearing; summons), MCL 712A.13 (summons), MCR 3.920 (service of process), and MCR 3.921 (persons entitled to notice) regarding summonses and notices of hearing do not apply when the juvenile is charged with a violation of the MVC. See MCL 712A.2b(b). Thus, it appears that the court may inform the juvenile's parent, guardian, or custodian of a hearing in any reasonable manner.

Additional notice requirements may exist under the Crime Victim's Rights Act, MCL 780.781 *et seq*.

C. Required Notice When a Juvenile is Charged With a Felony in Which a Motor Vehicle Was Used

MCR 3.931(B)(7) requires a petition to contain the notice provision contained in MCL 257.732(8), if applicable. MCL 257.732(8) provides:

"If a juvenile is accused of an act, the nature of which constitutes a felony in which a motor vehicle was used, other than a felony specified in [MCL 257.732(4)]^[6] or [MCL 257.319],^[7] the prosecuting attorney or family division of circuit court shall include the following statement on the petition filed in the court:

'You are accused of an act the nature of which constitutes a felony in which a motor vehicle was used. If the accusation is found to be true and the

⁶MCL 257.732(4) requires the court clerk to forward an abstract of the court record upon an individual's conviction of any of several enumerated offenses. See Section 1.41 for more information on abstracting procedures.

⁷MCL 257.319 requires the secretary of state to suspend a person's driver's license upon receipt of a record of the person's conviction of any of several enumerated offenses. See Section 1.45 for more information on license suspension.

judge or referee finds that the nature of the act constitutes a felony in which a motor vehicle was used, as defined in ... MCL 257.319, your driver's license shall be suspended by the secretary of state.""

D. Preliminary Inquiries and Preliminary Hearings

1. Preliminary Inquiries

A "[p]reliminary inquiry" is an "informal review by the court to determine appropriate action on a petition." MCR 3.903(A)(23). "When a petition is not accompanied by a request for detention of the juvenile, the court may conduct a preliminary inquiry." MCR 3.932(A). For more information about preliminary inquires, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 6.

2. Preliminary Hearings

If a juvenile has been taken into custody, a preliminary hearing pursuant to MCR 3.935 is required. See MCR 3.933(B); MCR 3.934(A). "However, a juvenile may not be detained in a secure facility pending hearing unless the court has considered the results of a detention screening tool conducted on the juvenile under MCR 3.907." MCR 3.933(B).⁸ At the preliminary hearing, after considering the results of a juvenile's risk screening tool and mental health screening tool, the court must review the petition to determine whether it should be dismissed, referred for alternate services, placed on the consent docket, or continue to a preliminary hearing. MCR 3.935(B)(3). A preliminary hearing in a juvenile delinquency case is functionally similar to an arraignment in a criminal case. In re Wilson, 113 Mich App 113, 122 (1982). A preliminary hearing "must commence no later than 24 hours after the juvenile has been taken into court custody, excluding Sundays and holidays, as defined by MCR 8.110(D)(2), or the juvenile must be released." MCR 3.935(A)(1).

For more information about preliminary hearings, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 6.

⁸For more information on use of a detention screening tool, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 4.

3. Referees

"[T]he court may assign a referee to conduct a preliminary inquiry or to preside at a hearing other than those specified in MCR 3.912(A)^[9] and to make recommended findings and conclusions." MCR 3.913(A)(1). For further discussion of the procedures required at a delinquency trial, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 9.

2.10 Required Procedures Following a Juvenile's Failure to Respond to a Motor Vehicle Violation

MCR 3.931(D) sets out the procedures to follow when a juvenile fails to respond to a motor vehicle violation citation. MCR 3.931(D) provides:

"If the juvenile is a Michigan resident and fails to appear or otherwise to respond to any matter pending relative to a motor vehicle violation, the court

(1) must initiate the procedure required by MCL 257.321a for the failure to answer a citation, and

(2) may issue an order to apprehend the juvenile after a petition is filed with the court." See also MCL 712A.2c.

A. Dispositions¹⁰

When a juvenile is accused of violating the MVC, the hearing procedures in MCL 712A.2b apply, "any other provision of [the Juvenile Code] notwithstanding[.]" MCL 712A.2b. "If after hearing the case the court finds the accusation to be true, the court may dispose of the case under [MCL 712A.18]." MCL 712A.2b(c).

MCL 712A.18(1) provides that if the court finds the juvenile does not come within the jurisdiction of the court, the court must dismiss the petition (or citation). However, except as otherwise provided in MCL 712A.18(8)¹¹ and subject to MCL 712A.18(9),¹² if the court finds that the juvenile is within the jurisdiction of the court, the court must "order the juvenile returned to the juvenile's parent if the return of the

⁹ MCR 3.912(A) provides that "[a] judge must preside at: (1) a jury trial; (2) a waiver proceeding under MCR 3.950; (3) the preliminary examination, trial, and sentencing in a designated case; (4) a proceeding on the issuance, modification or termination of a minor personal protection order."

¹⁰MCR 3.943 governs the required procedures at dispositional hearings. For a detailed discussion of these procedures, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 10.

¹¹ Before entering an order of disposition for certain juvenile offenses, the court must examine the court file and determine whether the juvenile's biometric data and fingerprints have been collected and forwarded as required by MCL 28.243 and the Sex Offenders Registration Act. See MCL 712A.18(10).

juvenile to the juvenile's parent would not cause a substantial risk of harm to the juvenile or society." *Id.* The court may also order any of several types of disposition "appropriate for the welfare of the juvenile and society in view of the facts proven and ascertained[.]" MCL 712A.18(1)(a)-(j); MCL 712A.18(1)(m)-(o). Subject to MCL 712A.18(9), some common dispositional orders imposed in traffic cases include warning the juvenile (MCL 712A.18(1)(a)), in-home probation¹³ (MCL 712A.18(1)(b)), and community service (MCL 712A.18(1)(i)). However, the court may not order the juvenile or the juvenile's parent, guardian, or legal custodian to pay for fees or costs associated with community service or, if another disposition under MCL 712A.18 has been ordered, to pay fines associated with a violation of a municipal ordinance or a state or federal law. MCL 712A.18(1)(i)-(j).

For a discussion of all of the dispositional options open to a court, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 10.

B. Required Reporting to the Secretary of State

"Within 14 days after entry of a court order of disposition for a juvenile found to be within [the Juvenile Code], the court shall prepare and forward an abstract of the record of the court for the case in accordance with . . . [MCL 257.732]." MCL 712A.2b(d). See also MCR 3.943(E)(6) (requiring an abstract after the dispositional hearing in a formal case). MCL 257.732(1) requires "each clerk of a court of record"¹⁴ to keep a "full record" of every case in which an individual is charged with or cited for violating the MVC or a local ordinance substantially corresponding to a provision of the MVC. MCL 257.732(1). For certain traffic-related offenses, the clerk of the court must send an abstract of the court record to the secretary of state. *Id.* For more information on abstracting requirements, see Section 1.41.

¹²Before disposition, a risk and needs assessment must be conducted by a designated individual or agency for each juvenile and the results must be shared with the court and each party to the proceeding and used to inform a dispositional recommendation. See MCL 712A.18(11) and MCR 3.943(E)(2). For more information on a risk and needs assessment, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 4.

¹³ MCL 712A.2b(e) provides that the court retains discretion to restrict the juvenile's driving privileges as a term or condition of probation.

¹⁴ The county clerk is the clerk of the court for the family division and keeps its records and indexes of actions. MCL 600.1007; MCR 8.119.

2.11 Restitution, Crime Victim's Rights Assessment, and Costs

If the juvenile's offense has resulted in financial damages to any victim, then the court must order the juvenile or the juvenile's parent to pay restitution as provided in MCL 712A.30, MCL 712A.31, MCL 780.794, and MCL 780.795. MCL 712A.18(7); MCL 712A.30(2). See the Michigan Judicial Institute's *Crime Victims Rights Benchbook*, Chapter 8, for a discussion of restitution requirements.

When a juvenile offender is found responsible for "an offense committed by [the] juvenile under the jurisdiction of the [family division] under . . . [MCL 712A.2(a)(1)], that if committed by an adult would be a felony, misdemeanor, or ordinance violation, if the juvenile's case is not designated as a case in which the juvenile is to be tried in the same manner as an adult[,]" the court must order the juvenile to pay a crime victim's rights fund assessment. MCL 780.901(f). Each juvenile for whom an order of disposition is entered for a juvenile offense must pay an assessment of \$25.00. MCL 780.905(3). For a complete list of crime victim assessments, visit https://www.courts.michigan.gov/4a3bc4/siteassets/ court-administration/resources/crimevictimassessment.pdf. See the Michigan Judicial Institute's *Crime Victims Rights Benchbook*, for a discussion of the crime victim's rights fund assessment.

The court shall not "order a juvenile or a juvenile's parent, guardian, or legal custodian to pay for the costs of care, services, court-appointed attorney representation, or other costs or assessments related to the juvenile's court proceeding." MCL 712A.18(3). See the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 18, for further discussion.

2.12 Setting Aside Adjudications

"The setting aside of juvenile adjudications is governed by MCL 712A.18e and MCL 712A.18t." MCR 3.925(F)(1). MCL 712A.18e(1) addresses applications to set aside certain juvenile adjudications while MCL 712A.18t addresses automatic set-aside (ie.e. no application) for certain juvenile adjudications.¹⁵ When setting aside a juvenile adjudication, the court must not remove or expunge the adjudication from the juvenile's driving record. MCL 712A.18e(17); MCL 712A.18t(10).

For the offense of unlawfully driving away an automobile (UDAA) or attempted UDAA, MCL 750.413, an adjudication *must* be set aside if the

¹⁵ For more information on setting aside juvenile adjudications, see the Michigan Judicial Institute's *Juvenile Justice Benchbook*, Chapter 21.

applicant "files an application with the court and he or she otherwise meets all the requirements[.]" MCL 712A.18e(10)(a).

2.13 Holmes Youthful Trainee Act

A juvenile charged with a traffic offense cannot be assigned to youthful trainee status under the Holmes Youthful Trainee Act (HYTA), MCL 762.11 *et seq.* MCL 762.11(3)(c); *People v Martinez*, 211 Mich App 147, 149 (1995). For more information about HYTA, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 2*, Chapter 9.

Chapter 3: Civil Infraction Nonmoving Violations

3.1	Introduction and Scope Note	. 3-2
Part	t A—License and Permit Violations	
3.2	Causing or Permitting an Unlicensed Minor to Drive	. 3-2
3.3	Failing to Change Address on Driver's License	. 3-2
3.4	Operating a Vehicle in Violation of Graduated Licensing Requiremer	nts
		3-3
Part	t B—Title, Plate, Registration, and Insurance Violations	
3.5	Failing to Change Address on Registration or Title	. 3-8
3.6	Failing to Assure Title Transfer When Vehicle Is Sold	. 3-8
3.7	Invalid or No Registration Plate	. 3-9
3.8	No Proof of Insurance	3-10
3.9	No Proof of Registration	3-12
3.10	Registration Plate Violations	3-13
Part	t C—Other Nonmoving Violation Civil Infractions Found in the Michigan Vehicle Code	
3.11	Abandoning a Vehicle and Failing to Redeem It Before Disposition	
		3-14
3.12	Child Restraint System and Safety Belt System Violations	3-15
3.13	Equipment Violations	3-19
3.14	Parking, Stopping, or Standing	3-22
3.15	Soliciting Business at the Scene of a Motor Vehicle Accident or Disal Vehicle	
3.16	Transporting or Possessing Alcohol in a Motor Vehicle by a Person L Than 21 Years of Age	

3.1 Introduction and Scope Note

The Michigan Vehicle Code (MVC) contains hundreds of traffic offenses. This chapter includes discussion of the most commonly-cited nonmoving civil infractions. The discussion of each civil infraction includes:

- the name of the offense;
- quotations of the actual statute, or significant parts thereof;
- penalties;
- sanctions; and
- issues of importance regarding that offense.

This chapter does not discuss motor carrier or commercial violations or civil infractions that may be committed only by operation of a motorcycle.

Part A—License and Permit Violations

3.2 Causing or Permitting an Unlicensed Minor to Drive

A. Statutory Authority

"An individual shall not cause or knowingly permit any minor to drive a motor vehicle upon a highway as an operator, unless the minor has first obtained a license to drive a motor vehicle under the provisions of [MCL 257.301 *et seq.*]." MCL 257.325.

B. Penalties

"An individual who violates [MCL 257.325] is responsible for a civil infraction and shall be ordered to pay a civil fine of not more than \$100.00." MCL 257.325.

3.3 Failing to Change Address on Driver's License

A. Statutory Authority

"An operator . . . who changes his or her residence before the expiration of a license granted under this chapter shall immediately

notify the secretary of state of his or her new residence address." MCL 257.315(1).

B. Penalties

The general rules for assessing a civil fine and costs apply to failing to change address on driver's license violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

• Discretionary license suspension or revocation under certain circumstances. See MCL 257.315(3). See Sections 1.45 and 1.44 for more information on license suspension and revocation.

D. Issues

"Under the [MVC], [a] defendant has a duty to show a correct address on his [or her] operator's license." *Hamilton v Gordon*, 135 Mich App 289, 294 (1984). "This duty exists even though the time may not have arrived when the license itself needs to be renewed." *Id*.

3.4 Operating a Vehicle in Violation of Graduated Licensing Requirements

A. Statutory Authority

"Except as otherwise provided in [the MVC], an operator's . . . license issued to a person who is 17 years of age or less shall be in a form as prescribed in [MCL 257.310], and valid only upon the issuance of a graduated driver license." MCL 257.310e(1).

1. Level 1 Graduated License

"Except as otherwise provided in [MCL 257.303¹], a person who is not less than 14 years and 9 months of age may be

issued a level 1 graduated licensing status to operate a motor vehicle if the person has satisfied all of the following conditions:

(a) Passed a vision test and met health standards as prescribed by the secretary of state.

(b) Successfully completed segment 1 of a driver education course as that term is defined in . . . MCL 256.627, including a minimum of 6 hours of on-the-road driving time with the instructor.

(c) Received written approval of a parent or legal guardian." MCL 257.310e(3).

"A person issued a level 1 graduated licensing status may operate a motor vehicle only when accompanied either by a licensed parent or legal guardian or, with the permission of the parent or legal guardian, a licensed driver 21 years of age or older." MCL 257.310e(4).

"Except as otherwise provided in [MCL 257.310e²], a person is restricted to operating a motor vehicle with a level 1 graduated licensing status for not less than 6 months." MCL 257.310e(4).

2. Level 2 Graduated License

"A person may be issued a level 2 graduated licensing status to operate a motor vehicle if the person has satisfied all of the following conditions:

(a) Had a level 1 graduated licensing status for not less than 6 months.

(b) Successfully completed segment 2 of a driver education course as that term is defined in . . . MCL 256.627.

(c) Not incurred a moving violation resulting in a conviction or civil infraction determination or been involved in an accident for which the official police report indicates a moving violation on the part of the person during the 90-day period immediately preceding application.

¹ MCL 257.303 provides limitations on issuing a license.

² See MCL 257.310e(7), providing for an extended graduated licensing period or expanded provisions if certain conditions exist.

(d) Presented a certification by the parent or guardian that the person, accompanied by his or her licensed parent or legal guardian or, with the permission of the parent or legal guardian, any licensed driver 21 years of age or older, has accumulated a total of not less than 50 hours of behind-the-wheel experience including, except as otherwise provided in this subdivision, not less than 10 nighttime hours. The nighttime hours requirement does not apply to a person who has been issued a graduated driver license that permits daylight driving only as provided in [Mich Admin Code, R 257.3].

(e) Successfully completed a secretary of state approved driving skills test. . . This subdivision applies to a person 16 years of age or over only if the person has satisfied subdivisions (a), (b), (c), and (d)." MCL 257.310e(5).

"A person issued a level 2 graduated licensing status under [MCL 257.310e(5)] shall remain at level 2 for not less than 6 months." MCL 257.310e(6). However, this time period may be expanded or extended under MCL 257.310e(7) if certain conditions exist.

"A person issued a level 2 graduated licensing status under [MCL 257.310e(5)] shall not operate a vehicle under the following circumstances:

(a) Between the hours of 10 p.m. and 5 a.m. This subdivision does not apply if either of the following applies:

(*i*) The person is accompanied by a parent or legal guardian or a licensed driver 21 years of age or older designated by the parent or legal guardian.

(*ii*) The person is operating the vehicle in the course of his or her employment or while going to or from employment or while going to or from an authorized activity.

(b) With more than 1 passenger in the vehicle who is less than 21 years of age. This subdivision does not apply if any of the following apply:

(*i*) The person is accompanied by a parent or legal guardian or a licensed driver 21 years of

age or older designated by the parent or legal guardian.

(*ii*) Any additional passengers who are less than 21 years of age are members of his or her immediate family.

(*iii*) The person is operating the vehicle in the course of his or her employment or while going to or from employment or while going to or from an authorized activity." MCL 257.310e(6).

However, these provisions may be expanded or extended under MCL 257.310e(7) if certain conditions exist. See Section 3.3(A)(3) for more information.

3. Extended Graduated Licensing Periods

"The provisions and provisional period described in [MCL 257.310e(4) or MCL 257.310e(6)] shall be expanded or extended, or both, beyond the period described in [MCL 257.310e(4) or MCL 257.310e(6)] if any of the following occur and are recorded on the licensee's driving record during the provisional periods described in [MCL 257.310e(4) or MCL 257.310e(4) or MCL 257.310e(6)] or any additional periods imposed under [MCL 257.310e(7)]:

(a) A moving violation resulting in a conviction, civil infraction determination, or probate court disposition.

(b) An accident for which the official police report indicates a moving violation on the part of the licensee.

(c) A license suspension for a reason other than a mental or physical disability.

(d) A violation of [MCL 257.310e(4) or MCL 257.310e(6)]." MCL 257.310e(7).

"The provisional period described in [MCL 257.310e(4)] shall be extended under [MCL 257.310e(7)] until the licensee completes 90 consecutive days without a moving violation, an accident in which a moving violation resulted, suspension, or provisional period violation listed in [MCL 257.310e(7)], or until age 18, whichever occurs first." MCL 257.310e(8). "The provisional period described in [MCL 257.310e(6)] shall be extended under [MCL 257.310e(7)] until the licensee completes 12 consecutive months without a moving violation, suspension, or restricted period violation listed in [MCL 257.310e(7)] or until age 18, whichever occurs first." MCL 257.310e(8).

In the event that the provisions in MCL 257.310e(4) or MCL 257.310e(6) are expanded or extended as permitted in MCL 257.310e(7), "[n]otice shall be given by first-class mail to the last known address of [the] licensee[.]" MCL 257.310e(10).

4. Level 3 Graduated License

"A person who is not less than 17 years of age may be issued a level 3 graduated licensing status under [MCL 257.310e(9)] if the person has completed 12 consecutive months without a moving violation, an accident in which a moving violation resulted, accident, suspension, or restricted period violation listed in [MCL 257.310e(7)] while the person was issued a level 2 graduated licensing status under [MCL 257.310e(5)]." MCL 257.310e(9).

5. Operating Without a Graduated License in Possession

"A person shall have his or her graduated licensing status in his or her immediate possession at all times when operating a motor vehicle, and shall display the card upon demand of a police officer." MCL 257.310e(14).

B. Penalties

The general rules for assessing a civil fine and costs apply to graduated licensing violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

• Two points for a violation of MCL 257.310e(4) or MCL 257.310e(6) or a substantially corresponding law or

ordinance. MCL 257.320a(1)(t). See Section 1.42 for more information on points.

• No points for a violation of MCL 257.310e(14). MCL 257.320a(2). See Section 1.42 for more information on points.

Part B—Title, Plate, Registration, and Insurance Violations

3.5 Failing to Change Address on Registration or Title

A. Statutory Authority

"If a person, after making application for or obtaining the registration of a vehicle or a certificate of title, moves from the address named in the application as shown upon a registration certificate or certificate of title, the person within 10 days after moving shall notify the secretary of state in writing of the old and new addresses." MCL 257.228(1).

B. Penalties

The general rules for assessing a civil fine and costs apply to failure to change address violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

3.6 Failing to Assure Title Transfer When Vehicle Is Sold

A. Statutory Authority

"The owner of a motor vehicle who has made a bona fide sale by transfer of his or her title or interest and who has delivered possession of the vehicle and the certificate of title to that vehicle properly endorsed to the purchaser or transferee is not liable for any damages or a violation of law that subsequently results from the use or ownership of the vehicle by another, if the owner, other than a licensed dealer, satisfies the conditions prescribed under [MCL 257.240(2)]." MCL 257.240(1).

"The owner of a motor vehicle, other than a licensed dealer, shall satisfy 1 of the following conditions:

(a) Accompany the purchaser of the vehicle to a secretary of state branch office to assure that the title of the vehicle being sold is transferred.

(b) Maintain a record of the sale for not less than 18 months. As used in this subdivision, 'record of the sale' means either a photocopy of the reassigned title or a form or document that includes the name, address, driver license number, and signature of the person to whom the vehicle is sold and the purchase price and date of sale of the vehicle." MCL 257.240(2).

B. Penalties

The general rules for assessing costs apply to violations of MCL 257.240(2). See Section 1.20 for a discussion of the general rules governing the assessment of costs; see also the Michigan Judicial Institute's table for a quick reference guide.

"A person who violates [MCL 257.240(2)] is responsible for a civil infraction and shall be ordered to pay a civil fine of \$15.00." MCL 257.240(3). See also MCL 257.907(2)(c).

C. Issues

"A person who violates [MCL 257.240(2)] is presumed to be the last titled owner and to be liable for towing fees and daily storage fees for an abandoned motor vehicle." MCL 257.240(4).

3.7 Invalid or No Registration Plate

A. Statutory Authority

"Except as otherwise provided in [MCL 257.201–MCL 257.259³], a person shall not operate, nor shall an owner knowingly permit to be operated, upon any highway, a vehicle required to be registered under [the MVC] unless, except as otherwise provided in this subsection, no later than 30 days after the vehicle is registered or the vehicle's registration is renewed, a valid registration plate issued for the vehicle by the department for the current registration year is attached to and displayed on the vehicle as required by [MCL 257.201–MCL 257.259]." MCL 257.255(1). "The 30-day period described in this subsection does not apply to the first registration of

³ See e.g., MCL 257.216.

a vehicle after a transfer of ownership or to a transfer registration under [MCL 257.809]." MCL 257.255(1).

"A registration plate is not required for a wrecked or disabled vehicle, or vehicle destined for repair or junking, that is being transported or drawn on a highway by a wrecker or a registered motor vehicle." MCL 257.255(1).

A noncommercial or commercial vehicle registration "that expires on or after March 1, 2020 but is renewed on or before March 31, 2021" or "that expires after March 31, 2021 but before August 1, 2021, but is renewed within 120 days after the date of the expiration is not in violation of [MCL 257.255]." MCL 257.255(4).

B. Penalties

The general rules for assessing a civil fine and costs apply to invalid or no registration plate violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

3.8 No Proof of Insurance

A. Statutory Authority

"The owner of a motor vehicle who operates or permits the operation of the motor vehicle on the highways of [Michigan] or the operator of the motor vehicle shall produce, under [MCL 257.328(2)], on the request of a police officer, evidence that the motor vehicle is insured under [MCL 500.3101 *et seq.*]." MCL 257.328(1).

"Subject to [MCL 257.907(14)⁴], an owner or operator of a motor vehicle who fails to produce evidence of insurance on request under [MCL 257.328(1)] or who fails to have motor vehicle insurance for the vehicle as required under [MCL 500.3101 *et seq.*], is responsible for a civil infraction." MCL 257.328(1).

Electronic copy of certificate of insurance. "If a an individual displays an electronic copy of his or her certificate of insurance using an electronic device, the police officer shall only view the electronic copy of the certificate of insurance and shall not manipulate the electronic device to view any other information on the electronic device. An individual who displays an electronic copy

⁴ See Section 1.21(A) for more information on MCL 257.907(14).

of his or her certificate of insurance using an electronic device as provided in this subsection is not presumed to have consented to a search of the electronic device. A police officer may require the individual to electronically forward the electronic copy of the certificate of insurance to a specified location provided by the police officer. The police officer may then view the electronic copy of the certificate of insurance in a setting in which it is safe for the officer to verify that the information contained in the electronic copy of the certificate of insurance is valid and accurate. [Michigan], a law enforcement agency, or an employee of [Michigan] or a law enforcement agency is not liable for damage to or loss of an electronic device that occurs as a result of a police officer's viewing an electronic copy of a certificate of insurance in the manner provided in [MCL 257.328], regardless of whether the police officer or the owner or operator of the vehicle was in possession of the electronic device at the time the damage or loss occurred." MCL 257.328(1).

B. Penalties

The general rules for assessing costs apply to no proof of insurance violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

"If, before the appearance date on a citation issued under [MCL 257.328(1)], the defendant submits proof to the court that the motor vehicle had insurance meeting the requirements of [MCL 500.3101 and MCL 500.3102], at the time the violation of [MCL 257.328(1)] occurred, all of the following apply:

(a) The court shall not assess a fine or costs.

(b) The court shall not forward an abstract of the court record to the secretary of state.

(c) The court may assess a fee of not more than \$25.00 that shall be paid to the court funding unit." MCL 257.328(3).

"If a person has received a citation for a violation of [MCL 257.328(1)] for failing to produce a certificate of insurance under [MCL 257.328(2)], the court may waive the fee described in [MCL 257.328(3)(c)] and shall waive any fine, costs, and any other fee or assessment otherwise authorized under [the Michigan Vehicle Code] on receipt of verification by the court that the person, before the appearance date on the citation, produced valid proof of insurance that was in effect when the violation of [MCL 257.328(1)] occurred." MCL 257.907(14). "Insurance obtained after the violation

occurred does not make the person eligible for a waiver under [MCL 257.907(14)]." MCL 257.907(14).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

• Discretionary license surrender/suspension under certain circumstances. See MCL 257.328(4).⁵

3.9 No Proof of Registration

A. Statutory Authority

"A registration certificate shall at all times be carried in the vehicle to which it refers or shall be carried by or electronically accessible to the person driving or in control of the vehicle, who shall display a paper or electronic copy of the registration certificate upon demand of a police officer." MCL 257.223(1).⁶ "A digital photograph of a valid registration certificate satisfies the requirements of [MCL 257.223(1)]." MCL 257.223(3).

Failure to comply with MCL 257.223 is a civil infraction. MCL 257.223(4).

B. Penalties

The general rules for assessing a civil fine and costs apply to no proof of registration violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

⁵MCL 257.328(4) provides: "If an owner or operator of a motor vehicle is determined to be responsible for a violation of [MCL 257.328(1)], the court in which the civil infraction determination is entered may require the individual to surrender his or her operator's or chauffeur's license unless proof that the vehicle has insurance that meets the requirements of . . . MCL 500.3101 and 500.3102, is submitted to the court. If the court requires the license to be surrendered, the court shall order the secretary of state to suspend the individual's license. The court shall immediately destroy the license and shall forward an abstract of the court record to the secretary of state as required by [MCL 257.32]. On receipt of the abstract, the secretary of state shall suspend the individual's license beginning with the date on which the individual is determined to be responsible for the civil infraction for a period of 30 days or until proof of insurance that meets the requirements of . . . MCL 500.3101 and 500.3102, is submitted to the secretary of state, whichever occurs later. An individual who submits proof of insurance to the secretary of state under this subsection shall pay a service fee of \$25.00 to the secretary of state. The individual shall not be required to be examined under [MCL 257.320c] and shall not be required to pay a replacement license fee."

However, "[i]f a person has received a citation for a violation of [MCL 257.223], the court shall waive any civil fine, costs, and assessment on receipt of certification by a law enforcement agency that the person, before the appearance date on the citation, produced a valid registration certificate that was valid on the date the violation of [MCL 257.223] occurred." MCL 257.907(13).

3.10 Registration Plate Violations

A. Statutory Authority

"A registration plate shall at all times be securely fastened in a horizontal position to the vehicle for which the plate is issued so as to prevent the plate from swinging. The plate shall be attached at a height of not less than 12 inches from the ground, measured from the bottom of the plate, in a place and position that is clearly visible. The plate shall be maintained free from foreign materials that obscure or partially obscure the registration information and in a clearly legible condition. The attachment to the rear of a vehicle of a tow ball, bicycle rack, removable hitch, or any other device designed to carry an object on the rear of a vehicle, including the object being carried, does not violate this subsection." MCL 257.225(2).

B. Penalties

"A person who violates [MCL 257.225] is responsible for a civil infraction." MCL 257.225(7).

The general rules for assessing a civil fine and costs apply to failure to change address violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

⁶"If a person displays an electronic copy of his or her registration certificate using an electronic device, the police officer shall only view the electronic copy of the registration certificate and shall not manipulate the electronic device to view any other information on the electronic device. A person who displays an electronic copy of his or her registration certificate using an electronic device. A police officer may require the person to electronically forward the electronic copy of the registration certificate in a setting in which it is safe for the officer to verify that the information contained in the electronic copy of the registration certificate is valid and accurate. This state, a law enforcement agency, or an employee of this state or a law enforcement agency is not liable for damage to or loss of an electronic device that occurs as a result of a police officer's viewing an electronic copy of a registration certificate in the manner provided in [MCL 257.223], regardless of whether the police officer or the owner or operator of the vehicle was in possession of the electronic device at the time the damage or loss occurred." MCL 257.223(2).

C. Issues

Although the Michigan Supreme Court declared a violation of MCL 257.225(2) where the defendant's registration plate was partially obstructed by a bumper-mounted towing ball, effective August 18, 2018, the statute was amended and effectively superseded this holding. See *People v Dunbar*, 449 Mich 60, 73 (2016); 2018 PA 147.

A traffic stop was justified "based on [the officer's] reasonable suspicion that traffic laws were being violated[]" where the defendant's temporary paper license plate "was not in a clearly visible position or in a clearly legible condition." People v Simmons, 316 Mich App 322, 326-327 (2016). The officer testified that "he could not see a plate before pulling over the vehicle[, and h]e could not read the paper in the window when he approached the vehicle from 3 or 4 feet away, and its writing was very 'dim."" Id. at 327. Accordingly, the Court held that the officer "was justified in pulling over the vehicle for a violation of MCL 257.225(2)[.]" Simmons, 316 Mich App at 327 (rejecting the defendant's contention "that the search and seizure became unreasonable when [the officer] asked [him] for his license, registration, and insurance, rather than taking five seconds to examine the paper plate affixed to the rear window of the vehicle and determine its validity[]") (citations omitted). The officer's "actions were reasonably related in scope to the circumstances of the stop[, and e]ven [if the officer had] taken the time to examine the paper plate more closely to determine whether it appeared to be a valid temporary registration plate, the plate would still have been in violation of MCL 257.225(2)." Simmons, 316 Mich App at 327.

Part C—Other Nonmoving Violation Civil Infractions Found in the Michigan Vehicle Code

3.11 Abandoning a Vehicle and Failing to Redeem It Before Disposition⁷

A. Statutory Authority

"A person shall not abandon a vehicle in [Michigan]." MCL 257.252a(1).

⁷See the Michigan Judicial Institute's Abandoned Vehicle Hearing Checklist regarding challenges of towing and storage fees.

"It is presumed that the last titled owner of the vehicle is responsible for abandoning the vehicle unless the person provides a record of the sale as that term is defined in [MCL 257.240]." MCL 257.252a(1).

B. Penalties

The general rules for assessing costs apply to violations of MCL 257.252a(1). See Section 1.20 for a discussion of the general rules governing the assessment of costs; see also the Michigan Judicial Institute's table for a quick reference guide.

3.12 Child Restraint System and Safety Belt System Violations

A. Statutory Authority

1. Child Restraint System Violations

"Except as provided in [MCL 257.710d], or as otherwise provided by law, a rule promulgated under . . . MCL 24.201 to [MCL] 24.328, or federal regulation, each driver transporting a child in a motor vehicle shall properly secure that child in a child restraint system that meets the standards prescribed in 49 CFR 571.213^[8]." MCL 257.710d(1).

"A driver transporting a child as required under [MCL 257.710d(1)] shall position the child in the child restraint system in a rear seat, if the vehicle is equipped with a rear seat." MCL 257.710d(2). "If all available rear seats are occupied by children, then a child may be positioned in the child restraint system in the front seat." *Id*. However, "[a] child in a rear-facing child restraint system may be placed in the front seat only if the front passenger air bag is deactivated." *Id*.

Children who must be seated and positioned in a child restraint system "must be secured in a child restraint system that is appropriate for the child's weight and height and configured according to the child restraint system manufacturer's and vehicle manufacturer's instructions and the standards prescribed in 49 CFR 571.213." MCL 257.710d(2)-(3).

⁸49 CFR 571.213 "specifies requirements for child restraint systems used in motor vehicles" and "applies to passenger cars, multipurpose passenger vehicles, trucks and buses, and to child restraint systems for use in motor vehicles[.]"

- Rear-facing car seat (infant and up to two years of age or until reaching the manufacturer's weight and height limit). A child must be restrained in a rear-facing car seat until the child is at least 2 years old or outgrows "the weight or height limit of the rear-facing child restraint system set by the manufacturer." MCL 257.710d(2)(a)(*î*)-(*ii*).
- Forward-facing car seat (at least two and up to five years of age or until reaching the manufacturer's weight and height limit). Once a child reaches 2 years of age or outgrows the weight and height limit of the rear-facing child restraint system set by the manufacturer, the child must be restrained in a forward-facing car seat with an internal five point harness until the child is at least 5 years old or outgrows "the weight or height limit of the forwardfacing child restraint system set by the manufacturer." MCL 257.710d(2)(b)(*i*)-(*ii*).
- Booster seat with lap-shoulder safety belt (at least five and up to eight years of age or until reaching the height indicated). After a child reaches 5 years of age or reaches the height indicated by statute, the child "must be restrained in a belt-positioning child booster seat secured with a lap-shoulder safety belt until" the child is at least 8 years old or reaches the height of 4 feet 9 inches tall (57 inches). MCL 257.710d(2)(c)(*i*)-(*ii*).
- Seated in a rear seat with adjusted and fastened belt (children who satisfy safety MCL 257.710d(2)(c)(i) or MCL 257.710d(2)(c)(ii) and until thirteen years of age). Children younger than 13 years old must be restrained with a properly adjusted and fastened safety belt meeting the standards of 49 CFR 571.209 in a vehicle's rear seat, unless all rear seats are occupied by other children, or the vehicle does not have rear seating. MCL 257.710d(4). A child described in MCL 257.710d(4) "may be restrained in a belt-positioning child booster seat . . . until the child has reached the weight or height limit of the child booster seat set by the manufacturer." MCL 257.710d(4). Children between the ages of 13 and 16 "must be secured as required in [MCL 257.710e]." MCL 257.710d(5).9

"[MCL 257.710d] does not apply if the motor vehicle being driven is a bus, school bus, taxicab, moped, motorcycle, or

⁹For more information on safety belt violations, see Section 3.12(A)(2).

other motor vehicle not required to be equipped with safety belts under federal law or regulations." MCL 257.710d(6).

"The secretary of state may exempt by rules promulgated under [MCL 24.201–MCL 24.328], a class of children from the requirements of [MCL 257.710d], if the secretary of state determines that the use of the child restraint system required under [MCL 257.710d(1)] is impractical because of physical unfitness, a medical problem, or body size." MCL 257.710d(9). "The secretary of state may specify alternate means of protection for children exempted under [MCL 257.710d(9)]." MCL 257.710d(9).

2. Safety Belt Violations

"Each operator and front seat passenger of a motor vehicle operated on a street or highway in [Michigan] shall wear a properly adjusted and fastened safety belt, except that a child who is less than 13 years of age must be protected as required in [MCL 257.710d]." MCL 257.710e(3).

"Except as otherwise provided in [MCL 257.710d], each operator of a motor vehicle transporting a child 13 years of age or older but less than 16 years of age in a motor vehicle shall secure the child in a properly adjusted and fastened safety belt and seated as required under [MCL 257.710e]." MCL 257.710e(5). "If the motor vehicle is transporting more children than there are safety belts available for use, all safety belts available in the motor vehicle are being utilized in compliance with [MCL 257.710e], and the operator and all front seat passengers comply with [MCL 257.710e(3)], the operator of a motor vehicle transporting a child 13 years of age or older but less than 16 years of age for which there is not an available safety belt is in compliance with [MCL 257.710e(5)] if that child is seated in other than the front seat of the motor vehicle." MCL 257.710e(5). "However, if that motor vehicle is a pickup truck without an extended cab or jump seats, and all safety belts in the front seat are being used, the operator may transport the child in the front seat without a safety belt." Id.

"The operator of a motor vehicle shall wear a lap belt, but is not required to wear a shoulder harness, if the operator is operating the vehicle for the purpose of performing road construction or maintenance in a work zone." MCL 257.710e(6).

"[MCL 257.710e] does not apply to an operator or passenger of any of the following:

(a) A motor vehicle manufactured before January 1, 1965.

(b) A bus.

(c) A motorcycle.

(d) A moped.

(e) A motor vehicle if the operator or passenger possesses a written verification from a physician that the operator or passenger is unable to wear a safety belt for physical or medical reasons.

(f) A motor vehicle that is not required to be equipped with safety belts under federal law.

(g) A commercial or United States Postal Service vehicle that makes frequent stops for the purpose of pickup or delivery of goods or services.

(h) A motor vehicle operated by a rural carrier of the United States Postal Service while serving the carrier's rural postal route." MCL 257.710e(1).

"[MCL 257.710e] does not apply to the passenger of a school bus." MCL 257.710e(2).

B. Penalties

The general rules for assessing a civil fine and costs apply to child restraint system and safety belt violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

1. Special Penalties for Child Restraint System Violations

For a violation of MCL 257.710d, the court may order the person to pay a civil fine and costs. MCL 257.907(2). However, "the civil fine ordered under [MCL 257.907] must not be more than \$10.00, subject to [MCL 257.907(11)]." MCL 257.907(2)(n).

"Before [April 2, 2025], the court may waive any civil fine, cost, or assessment against an individual who received a civil infraction citation for a violation of [MCL 257.710d (child not properly secured in a child restraint system)] if the individual, before the appearance date on the citation, supplies the court with evidence of acquisition, purchase, or rental of a child

seating system meeting the requirements of [MCL 257.710d]." MCL 257.907(11).

"Beginning [April 2, 2025], the court may waive any civil fine, cost, or assessment against an individual who received a civil infraction citation for a violation of [MCL 257.710d if the individual, before the appearance date on the citation, supplies the court with evidence of acquisition of a child seating system that meets the requirements of [MCL 257.710d] and evidence that the individual has received education from a certified child passenger safety technician." MCL 257.907(11).

2. Special Penalties for Safety Belt Violations

For a violation of MCL 257.710e, the court may order the person to pay a civil fine and costs. MCL 257.907(2). However, "the civil fine and court costs ordered under [MCL 257.907(2)] must be \$25.00." MCL 257.907(2)(o).

"Points must not be assessed under [MCL 257.320a] for a violation of [MCL 257.710e]." MCL 257.710e(13). See Section 1.42 for more information on points.

3.13 Equipment Violations

A. Statutory Authority

"Except as otherwise provided in [MCL 257.698 or MCL 257.707d¹⁰], a person who violates a provision of [MCL 257.683-MCL 257.711] with respect to equipment on vehicles is responsible for a civil infraction." MCL 257.683(6).

"A person shall not drive or move or the owner shall not cause or knowingly permit to be driven or moved on a highway a vehicle or combination of vehicles that is in such an unsafe condition as to endanger a person, or that does not contain those parts or is not at all times equipped with lamps and other equipment in proper condition and adjustment as required in [MCL 257.683-MCL 257.711], or that is equipped in a manner in violation of [MCL 257.683-MCL 257.683-MCL 257.683-MCL 257.683(1).

"A person shall not do an act forbidden or fail to perform an act required under [MCL 257.683–MCL 257.711]." MCL 257.683(1).

¹⁰ MCL 257.698 contains a misdemeanor offense for misuse of police or emergency lights, and MCL 257.707d contains noise restrictions, some of which are misdemeanor offenses.

"A police officer on reasonable grounds shown may stop a motor vehicle and inspect the motor vehicle, and if a defect in equipment is found, the officer may issue the driver a citation for a violation of a provision of [MCL 257.683–MCL 257.711]." MCL 257.683(2).

"[MCL 257.683–MCL 257.711] shall not prohibit the use of additional parts and accessories on a vehicle that are not inconsistent with [MCL 257.683–MCL 257.711]." MCL 257.683(4).

"The provisions of [MCL 257.683-MCL 257.711] with respect to equipment on vehicles shall not apply to implements of husbandry, road machinery, road rollers, or farm tractors, except as specifically provided in [MCL 257.683–MCL 257.711]." MCL 257.683(5).

B. Equipment Violations

Common equipment violations include:

- Backing, cowl, fender, or running board lamps—MCL 257.698;
- Brakes—MCL 257.705;
- Brake lights—MCL 257.697;
- Bumpers—MCL 257.710c;
- Device causing emission of flame or smoke—MCL 257.682a;
- Failing to maintain equipment MCL 257.683;
- Head lamps, rear lamps, spot lamps, fog lamps-MCL 257.684, MCL 257.685, MCL 257.686, MCL 257.695, MCL 257.696, MCL 257.700, MCL 257.701, MCL 257.702, MCL 257.704;
- Horn, siren—MCL 257.706;
- Lamp or flag on projecting load MCL 257.693;
- Mirror and window obstruction of view—MCL 257.708 and MCL 257.709;
- Muffler or exhaust system MCL 257.707;
- Parking lights—MCL 257.694;
- Plates (lighting and visibility)-MCL 257.686;

- Reflectors and clearance markers—MCL 257.687, MCL 257.688, MCL 257.689, MCL 257.690, MCL 257.691;
- Safety chains (towing)—MCL 257.721(3);
- Safety glass—MCL 257.711;
- Slow moving vehicles, lights and reflectors—MCL 257.688(1)(g) and MCL 257.703;
- Television or other electronic device that displays video images—MCL 257.708b;
- Tires MCL 257.710;
- Tinted windows—MCL 257.709;
- Trailer, trailer hitch, towing equipment—MCL 257.721;
- Turn signals—MCL 257.697 and 257.697a; and
- Windshield, windows, wipers/washers—MCL 257.708a and MCL 257.709.

C. Penalties

The general rules for assessing a civil fine and costs apply to equipment violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs.

However, "[i]f a person has received a civil infraction citation for defective safety equipment on a vehicle under [MCL 257.683], the court shall waive a civil fine, costs, and assessments on receipt of certification by a law enforcement agency that repair of the defective equipment was made before the appearance date on the citation." MCL 257.907(9).

D. Issues

• Tinted windows, MCL 257.709

"Tinted windows [MCL 257.709] are equipment violations for which no points may be assessed." *People v Pitts*, 222 Mich App 260, 270 (1997). In *Pitts*, 222 Mich App at 271, the Court of Appeals determined that "holding [the] defendant responsible for a moving violation and assessing him two points for having nonreflective window tint upon his windows was improper." Further, "not allowing tinted window film to be applied [to vehicle windows] is rationally related to a legitimate governmental purpose and [a] defendant's right to equal protection of the law is not violated by [MCL 257.709] of the Michigan Vehicle Code." *Pitts,* 222 Mich App at 274-275.

• Tail lamps, MCL 257.686

"The plain language of [MCL 257.686(1)] states that all motor vehicles must be 'equipped with' at least one 'rear lamp."" *People v Williams*, 236 Mich App 610, 613 (1999). MCL 257.686(2) should be read to provide that "a tail lamp must be wired so as to be lighted as specified in order to comply with the Vehicle Code-the implication being that an automobile with a tail lamp not wired so as to be lighted as specified would be in violation of the Vehicle Code." *Williams*, 236 Mich App at 614. Stated another way, "all tail lamps intended for use on an automobile ... have to be operative in order for the automobile to be in compliance with the Vehicle Code." *Id.* at 615. Thus, "a motor vehicle equipped with multiple tail lamps is in violation of [MCL 257.686(2)] if one or more of its tail lamps is inoperative." *Williams*, 236 Mich App at 615.

3.14 Parking, Stopping, or Standing

A. Statutory Authority

1. Parking on Traveled Part of Highway

"Outside of the limits of a city or village, a vehicle shall not be stopped, parked, or left standing, attended or unattended, upon the paved or main traveled part of a highway when it is possible to stop, park, or to leave the vehicle off the paved or main traveled part of the highway." MCL 257.672(1). "Inside or outside of the limits of a city or village, a vehicle shall not be stopped, parked, or left standing, attended or unattended, upon the paved or unpaved part of a limited access highway, except in an emergency or mechanical difficulty." *Id*.

2. Prohibited Parking Areas under MCL 257.674

"A vehicle must not be parked, except if necessary to avoid conflict with other traffic or in compliance with the law or the directions of a police officer or traffic-control device, in any of the following places:

- (a) On a sidewalk.
- (b) In front of a public or private driveway.
- (c) Within an intersection.

(d) Within 15 feet of a fire hydrant.

(e) On a crosswalk.

(f) Within 20 feet of a crosswalk, or if there is not a crosswalk, then within 15 feet of the intersection of property lines at an intersection of highways.

(g) Within 30 feet of the approach to a flashing beacon, stop sign, or traffic-control signal located at the side of a highway.

(h) Between a safety zone and the adjacent curb or within 30 feet of a point on the curb immediately opposite the end of a safety zone, unless a different length is indicated by an official sign or marking.

(i) Within 50 feet of the nearest rail of a railroad crossing.

(j) Within 20 feet of the driveway entrance to a fire station and on the side of a street opposite the entrance to a fire station within 75 feet of the entrance if properly marked by an official sign.

(k) Alongside or opposite a street excavation or obstruction, if the stopping, standing, or parking would obstruct traffic.

(*I*) On the roadway side of a vehicle stopped or parked at the edge or curb of a street.

(m) Upon a bridge or other elevated highway structure or within a highway tunnel.

(n) At a place where an official sign prohibits stopping or parking.

(o) Within 500 feet of an accident at which a police officer is in attendance, if the scene of the accident is outside of a city or village.

(p) In front of a theater.

(q) In a place or in a manner that blocks immediate egress from an emergency exit conspicuously marked as an emergency exit of a building.

(r) In a place or in a manner that blocks or hampers the immediate use of an immediate egress from a fire escape conspicuously marked as a fire escape providing an emergency means of egress from a building.

* * *

(v) Within 500 feet of a fire at which fire apparatus is in attendance, if the scene of the fire is outside a city or village. However, volunteer fire fighters responding to the fire may park within 500 feet of the fire in a manner not to interfere with fire apparatus at the scene. A vehicle parked legally previous to the fire is exempt from [MCL 257.674(1)(v)].

(w) In violation of an official sign restricting the period of time for or manner of parking.

(x) In a space controlled or regulated by a meter on a public highway or in a publicly owned parking area or structure, if the allowable time for parking indicated on the meter has expired, unless the vehicle properly displays 1 or more of the items listed in [MCL 257.675(8)¹¹].

(y) On a street or highway in such a way as to obstruct the delivery of mail to a rural mailbox by a carrier of the United States Postal Service.

(z) In a place or in a manner that blocks the use of an alley.

(aa) In a place or in a manner that blocks access to a space clearly designated as a fire lane.

(bb) On a streetcar track or in a manner that blocks, delays, or otherwise interferes with the movement of a streetcar on a streetcar track." MCL 257.674(1).

3. Clear Vision Areas

"A vehicle shall not be parked in an area purchased, acquired, or used as a clear vision area adjacent to or on a highway right of way." MCL 257.674a(1).

¹¹"To be entitled to free parking in a metered space or in a publicly owned parking structure or area, a vehicle must properly display 1 of the following: (a) [a] windshield placard bearing a free parking sticker issued under [the MVC][;] (b) [a] valid windshield placard issued by another state[;] (c) [a] certificate of identification issued by another state[;] (d) [a] license plate for persons with disabilities issued by another state[;] [or] (e) [a] special registration plate with a tab for persons with disabilities attached issued by another state." MCL 257.675(8).

4. Disabled Person Parking

"A disabled person with a certificate of identification, windshield placard, special registration plates issued under [MCL 257.803d], a special registration plate issued under [MCL 257.803f] that has a tab for persons with disabilities attached, a certificate of identification or windshield placard from another state, or special registration plates from another state issued for persons with disabilities is entitled to courtesy in the parking of a vehicle." MCL 257.675(6). "The courtesy shall relieve the disabled person or the person transporting the disabled person from liability for a violation with respect to parking, other than in violation of [the MVC]." *Id*. ¹²

"A vehicle must not be parked, except if necessary to avoid conflict with other traffic or in compliance with the law or the directions of a police officer or traffic-control device, in any of the following places:

* * *

(s) In a parking space clearly identified by an official sign as being reserved for use by disabled persons that is on public property or private property available for public use, unless the individual is a disabled person as described in [MCL 257.19a] or unless the individual is parking the vehicle for the benefit of a disabled person. In order for the vehicle to be parked in the parking space the vehicle must display 1 of the following:

(*i*) A certificate of identification or windshield placard issued under [MCL 257.675] to a disabled person.

(*ii*) A special registration plate issued under [MCL 257.803d] to a disabled person.

(*iii*) A similar certificate of identification or windshield placard issued by another state to a disabled person.

(*iv*) A similar special registration plate issued by another state to a disabled person.

¹² There are several other disabled person parking offenses designated as misdemeanors. See MCL 257.675(15)-(17).

(*v*) A special registration plate to which a tab for persons with disabilities is attached issued under [the MVC].

(t) In a clearly identified access aisle or access lane immediately adjacent to a space designated for parking by persons with disabilities.

(u) On a street or other area open to the parking of vehicles that results in the vehicle interfering with the use of a curb-cut or ramp by persons with disabilities." MCL 257.674(1).

In addition, a local ordinance may be enacted that "prohibit[s] parking on a street or highway to create a fire lane or to provide for the accommodation of heavy traffic during morning and afternoon rush hours, and the privileges extending to . . . physically disabled persons under [MCL 257.675(6)] do not supersede that ordinance." MCL 257.675(6).

"MCL 257.675(6) precludes liability by . . . a disabled person [who is] cited. . . for violating a local time-restriction parking ordinance not contemplated in MCL 257.675(6) as constituting an exception to the liability exemption for disabled persons." *City of Monroe v Jones*, 259 Mich App 443, 453 (2003).

5. Unattended Vehicle

"A person shall not allow a motor vehicle to stand on a highway unattended without engaging the parking brake or placing the vehicle in park, stopping the motor of the vehicle, and removing and taking possession of the ignition key." MCL 257.676(1). "If the vehicle is standing upon a grade, the front wheels of the vehicle shall be turned to the curb or side of the highway." *Id.* MCL 257.676 "does not apply to a vehicle that is standing in place and is equipped with a remote start feature, if the remote start feature is engaged." *Id.*

6. Parking Within 500 Feet of Fire Apparatus Stopped in Answer to a Fire Alarm

"The driver of a vehicle other than a vehicle on official business shall not . . . park the vehicle within 500 feet where fire apparatus has stopped in answer to a fire alarm."¹³ MCL 257.679(1).

¹³ MCL 257.679(1) also prohibits the following of a fire truck too closely. See Section 4.5 for more information on this civil infraction.

B. Parking, Standing, and Stopping Violations

Common parking violations include:

- Disabled person parking¹⁴ violations—disregarding sign; improper use of handicap identification, plate, or tab; blocking access aisle or curb-cut, MCL 257.674(1)(s)-257.674(1)(u) and MCL 257.675(5) (note that MCL 257.675 now specifically defines disabled person differently than the general definition found in MCL 257.19a);
- Meter violations, MCL 257.674(1)(x);
- Parking in clear vision areas, MCL 257.674a;
- Parking on a highway or limited-access highway, MCL 257.672;
- Prohibited parking areas, MCL 257.674; and
- Unattended vehicle, MCL 257.676(1).

C. Exceptions to Parking, Stopping, and Standing Violations

A vehicle may be parked in a prohibited place "if necessary to avoid conflict with other traffic or in compliance with the law or the directions of a police officer or traffic-control device[.]" MCL 257.674(1).

"The [jury] instruction on sudden emergency is unnecessary when a violation of the no parking statute is alleged, MCL 257.674(n)." *Mason v Wurth*, 181 Mich App 129, 131 (1989). This is because "[t]he statute expressly excepts otherwise prohibited parking when necessary to comply with the law." *Id*. "The exception includes compliance with the law requiring an individual to stop at the scene of an accident and exchange certain information." *Id*.

D. Parking, Standing, and Stopping Violation Proceedings

"[I]n a proceeding for a violation of a local ordinance or state statute relating to a standing or parked vehicle, proof that the particular vehicle described in the citation, complaint, or warrant was parked in violation of the ordinance or state statute, together with proof from the secretary of state that the defendant named in the citation, complaint, or warrant was at the time of the violation the vehicle's registered owner, creates in evidence a presumption that the vehicle's registered owner was the person who parked or placed the

¹⁴ See Section 3.14(A)(4) for a detailed discussion of disabled person parking violations.

vehicle at the point where and at the time that the violation occurred." MCL 257.675a.

E. Liability for Parking, Standing, and Stopping Violations

1. Prima Facie Case of Vehicle Owner Liability

Although a police officer may issue a parking, standing, or stopping violation citation to the operator of the vehicle if he or she is present, MCL 257.675c(4), the registered owner is generally liable for the violation, MCL 257.675c(1). "Instead of requiring the local governmental unit issuing the ticket to identify and pursue the particular driver who violated the parking law, the Legislature has created a rebuttable prima facie case based on vehicle registration." *Ford Motor Credit Co v Detroit*, 254 Mich App 626, 629 (2003). MCL 257.675c(1) states:

"[I]f a vehicle is stopped, standing, or parked in violation of [MCL 257.672, MCL 257.674, MCL 257.674a, MCL 257.675, or MCL 257.676], or other state statute, or a local ordinance prohibiting or restricting the stopping, standing, or parking of a vehicle and the violation is a civil infraction, the person in whose name that vehicle is registered in [Michigan] or another state at the time of the violation is prima facie responsible for that violation and subject to [MCL 257.907]." MCL 257.675c(1).

"The owner of a vehicle cited for a stopping, standing, or parking violation pursuant to [MCL 257.675c(1)] may assert as an affirmative defense that the vehicle in question, at the time of the violation, was in the possession of a person whom the owner had not knowingly permitted to operate the vehicle." MCL 257.675c(2). A vehicle's owner may also rebut a prima facie case established under MCL 257.675c(1) with "evidence that someone else is responsible for the violation." *Ford Motor Co*, 254 Mich App at 630 n 3 (the affirmative defense specified in MCL 257.675c(2) is not the only method of rebutting a prima facie case).

2. Exception: Violations Involving Leased or Rented Vehicles

Where the violation involves a leased or rented vehicle, the owner may not be responsible. MCL 257.675b(1). "The lessee or renter of a motor vehicle and not the leased vehicle owner is liable for a violation of a local ordinance or state statute

relating to a standing or parked vehicle involving the motor vehicle if the leased vehicle owner furnishes proof that the vehicle described in the citation, complaint, warrant, or notice was in the possession of, custody of, or was being operated or used by the lessee or renter of the vehicle at the time of the violation." MCL 257.675b(1).

"If a leased vehicle is leased or rented for 30 days or less, the leased vehicle owner may avoid liability for a violation described in [MCL 257.675b(1)] if the leased vehicle owner provides all of the following information to the clerk of the court or parking violations bureau issuing the violation not later than 30 days after the leased vehicle owner has received notice of the violation:

(a) The lessee's or renter's name, address, and operator's . . . license number.

(b) A copy of the signed rental or lease agreement or an expedited rental agreement without signature as part of a master rental agreement, including proof of the date and time the possession of the vehicle was given to the lessee or renter and the date and time the vehicle was returned to the leased vehicle owner or the leased vehicle owner's authorized agent under the agreement." MCL 257.675b(2).

"If a leased vehicle is leased or rented for 30 days or less, the leased vehicle owner is liable for a violation of a local ordinance or state statute relating to a standing or parked vehicle if 1 or more of the following occur:

(a) The leased vehicle owner does not provide the information described in [MCL 257.675b(2)] within the 30-day period specified in [MCL 257.675b(2)].

(b) The court or parking violations bureau issuing the violation proceeds against the lessee or renter of the vehicle and the lessee or renter of the vehicle is not convicted of or found responsible for the violation." MCL 257.675b(3).

3. Civil Action by Vehicle Owner Against Violator

"The registered owner of a vehicle who is found to be responsible for a civil infraction as the result of [MCL 257.675c(1)] or a leased vehicle owner as defined in [MCL 257.675b] that is found to be responsible for a civil infraction

described in [MCL 257.675b] has the right to recover in a civil action against the person who parked, stopped, or left standing the vehicle in question damages including, but not limited to, the amount of any civil fine or costs, or both, imposed pursuant to [MCL 257.907]." MCL 257.675c(3). "The registered owner of a vehicle or the leased vehicle owner may provide in a written agreement that the person who parked, stopped, or left standing the vehicle in violation of a state statute or local ordinance, when the violation is a civil infraction, shall indemnify the registered owner or the leased vehicle owner for the damages incurred including, but not limited to, any civil fine and costs imposed upon the registered owner for that civil infraction." Id. "With regard to a leased vehicle, [MCL 257.675c(3)] does not apply if the court or parking violations bureau issuing the violation finds that the lessee or renter of the vehicle is not responsible for the violation and it is determined that the lessee or renter did not violate the terms of the rental contract or lease agreement." MCL 275.675c(3).

F. Penalties

"For a violation of [MCL 257.674(1)(s)] or a local ordinance that substantially corresponds to [MCL 257.674(1)(s)], the civil fine ordered under [MCL 257.907(2)] must not be less than \$100.00 or more than \$250.00." MCL 257.907(2)(j).

For all other parking, stopping, or standing civil infraction violations, the general rules for assessing a civil fine and costs apply. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

3.15 Soliciting Business at the Scene of a Motor Vehicle Accident or Disabled Vehicle

A. Statutory Authority

MCL 257.676c provides in relevant part,

"(1) Except as provided in subsection (2), a person shall not travel to the scene of a motor vehicle accident or a disabled vehicle located on public property, property open to the public, or a state trunk line highway and solicit business for a wrecker, recovery, or towing service.... (2) Subsection (1) does not apply if any of the following conditions apply:

(a) A law enforcement agency having jurisdiction over the scene of the accident or disabled vehicle, or an individual involved in that accident or disabled vehicle, requests the owner or operator of a wrecker or towing service to come to the scene.

(b) A wrecker, recovery truck, or tow truck operator, who does not travel to the scene of a motor vehicle accident or disabled vehicle as described in subsection (1) for the purpose of soliciting business for a wrecker, recovery, or towing service, offers assistance to a stranded motorist without creating a nuisance or interfering with management of a motor vehicle accident by law enforcement.

(3) Subject to [MCL 257.252d], the law enforcement agency at the scene shall permit an owner or operator of a motor vehicle to request the towing, wrecker, or recovery service or roadside assistance service of his or her choice unless the vehicle is involved in a suspected criminal activity, fatality, or law enforcement investigation, if the vehicle is being impounded, or if the requested preference wrecker service is unavailable or cannot respond within a timely manner and the vehicle is creating a road or safety hazard as determined by law enforcement at the scene."

B. Penalties

"A person who violates [MCL 257.676c] is responsible for a civil infraction and shall be ordered to pay a civil fine of \$1,000.00." MCL 257.676c(1). See also MCL 257.907(2)(*l*).

Costs must be ordered as provided by MCL 257.907(4). See Section 1.20(B) for a discussion of the general rules governing the assessment of costs; see also the Michigan Judicial Institute's table for a quick reference guide.

3.16 Transporting or Possessing Alcohol in a Motor Vehicle by a Person Less Than 21 Years of Age

"A person who knowingly transfers title to a motor vehicle for the purpose of avoiding [MCL 257.624b] is responsible for a civil infraction

and shall be ordered to pay a civil fine of not more than \$100.00." MCL 257.624b(4).

• Required Notice When Offender is a Juvenile

If the person who allegedly violated MCL 257.624b(1) is less than 18 years old, the law enforcement agency must notify the minor's parent(s), guardian, or custodian if the name of the parent, guardian, or custodian is reasonably ascertainable. MCL 257.624b(5). This notice must be given within 48 hours after the agency determines that the person is less than 18 years old and may be given person, by telephone, or by first-class mail. *Id*.

Chapter 4: Civil Infraction Moving Violations

4.1	Introduction and Scope Note	. 4-2
4.2	Careless or Negligent Driving	. 4-2
4.3	Failing to Remove a Motor Vehicle From Roadway After an Accident	t
		4-4
4.4	Failing to Stop for School Bus	. 4-5
4.5	Following a Fire Truck Too Closely	. 4-8
4.6	Improper Passing of a Stationary Emergency Vehicle	. 4-9
4.7	Improper Use of Emergency Lights	4-11
4.8	Interference With View, Control, or Operation of Vehicle	4-14
4.9	Overtaking and Passing	4-15
4.10	Permitting Minor to Ride in Pickup Truck Bed	4-18
4.11	Railroad Crossings	4-19
4.12	Right-of-Way or Failure to Yield	4-21
4.13	Speed Violations	4-26
4.14	Stop and Go, Signs and Signals	4-42
4.15	Holding or Using a Mobile Electronic Device While Driving	4-47
4.16	Turning and Signaling	4-51
4.17	Using a Cellular Telephone While Operating a Motor Vehicle (Kelsey	/'S
	law)	4-53
4.18	Wrong Side or Wrong Way	4-55

4.1 Introduction and Scope Note

The Michigan Vehicle Code (MVC) contains hundreds of traffic offenses. This chapter includes discussion of the most commonly-cited moving civil infractions. The discussion of each civil infraction includes:

- the name of the offense;
- quotations of the actual statute, or significant parts thereof;
- penalties;
- sanctions; and
- issues of importance regarding that offense.

This chapter does not discuss motor carrier or commercial violations or civil infractions that may be committed only by operation of a motorcycle.

4.2 Careless or Negligent Driving

A. Statutory Authority

"A person who operates a vehicle upon a highway or a frozen public lake, stream, or pond or other place open to the general public including an area designated for the parking of vehicles in a careless or negligent manner likely to endanger any person or property, but without wantonness or recklessness, is responsible for a civil infraction." MCL 257.626b.

B. Penalties

The general rules for assessing a civil fine and costs apply to careless driving. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Three points are assessed for "[c]areless driving in violation of [MCL 257.626b] or a law or ordinance substantially corresponding to [MCL 257.626b;]" MCL 257.320a(1)(m). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

D. Issues

Reckless Driving, MCL 257.626 vs. Careless Driving, MCL 257.626b

The difference between reckless driving (MCL 257.626),¹ a misdemeanor, and careless or negligent driving (MCL 257.626b), a civil infraction, is the degree of negligence. Reckless driving requires driving in "willful or wanton disregard for the safety of persons or property." MCL 257.626(2). Careless driving requires ordinary negligence, which is defined as operating a motor vehicle in a "careless or negligent manner likely to endanger any person or property, but without wantonness or recklessness[.]" MCL 257.626b.

Committee Tip: If the prosecuting attorney, in a plea bargain, decides to reduce the charge from reckless driving to careless or negligent driving, it is necessary to dismiss the misdemeanor charge and to have another citation issued for a civil infraction to which the defendant will then plead responsible.

¹ See Section 6.18.

4.3 Failing to Remove a Motor Vehicle From Roadway After an Accident

A. Statutory Authority

"Unless the operator of a motor vehicle involved in an accident knows or reasonably should know that serious impairment of a bodily function or death has resulted from the accident, the operator or any other occupant of the motor vehicle who possesses a valid driver license shall remove the motor vehicle from the main traveled portion of the roadway into a safe refuge on the shoulder, emergency lane, or median or to a place otherwise removed from the roadway, and, if the motor vehicle is located in a place that would block, delay, or otherwise interfere with the movement of a streetcar on a streetcar track, away from the streetcar track, if both of the following apply:

(a) Moving the motor vehicle may be done safely.

(b) The motor vehicle is capable of being normally and safely operated and can be operated under its own power in its customary manner without further damage or hazard to the traffic elements or to the roadway." MCL 257.618a(1).

B. Penalties

"A person who violates [MCL 257.618a(1)] is responsible for a civil infraction." MCL 257.618a(2).

The general rules for assessing a civil fine and costs apply to failing to remove a vehicle from the roadway after being involved in an accident. See MCL 257.907. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs. See also the Michigan Judicial Institute's table for a quick reference guide.

C. Issues

"The operator or any other person who removes a motor vehicle from the main traveled portion of the roadway as provided in [MCL 257.618a] before the arrival or a police officer is not prima facie at fault regarding the cause of the traffic accident solely by reason of moving the motor vehicle as provided in [MCL 257.618a]." MCL 257.618a(3).

"The decision of the operator or any other person to remove or not to remove a motor vehicle from the main traveled portion of the roadway as provided in [MCL 257.618a] is not admissible in a civil action as evidence that a serious impairment of bodily function has or has not resulted from the accident." MCL 257.618a(4).

"A law enforcement agency or police officer may, without the consent of the owner or operator and with the assistance of the state transportation department, other road agencies, fire department, emergency management, other local public safety agencies, street railway, or towing or recovery companies under the direction of any of those entities remove and dispose of motor vehicles and cargoes of vehicles involved in accidents, including any personal property, from the main traveled portion of a roadway and the right-of-way if the vehicle, cargo, or personal property is blocking the roadway or right-of-way or may otherwise endanger public safety, or from the streetcar track if the motor vehicle, cargo, or personal property is blocking the streetcar track or may delay or interfere with the movement of a streetcar track." MCL 257.618a(5).

"Except as otherwise provided in this subsection, an entity that moves a motor vehicle, cargo, or personal property as described in [MCL 257.618a(5)], and any of the entity's officers, employees, or agents, or anyone acting in good faith under, and within the scope of, the authority conferred under [MCL 257.618a(5)], is not liable for any damages or claims that may arise from the exercise or the failure to exercise any authority granted under [MCL 257.618a(5)]. This subsection does not apply to the transport of a motor vehicle from the scene of an accident, or if the conduct of the individual acting under the authority conferred under [MCL 257.618a(5)] constitutes gross negligence." MCL 257.618a(6).

"The owner or carrier, if any, of a motor vehicle, cargo, or personal property removed or disposed of under subsection (5) shall reimburse the public agency, departments, street railway, and towing companies, if any, for all documented reasonable costs incurred in that removal and disposal." MCL 257.618a(7).

4.4 Failing to Stop for School Bus

A. Statutory Authority

"The operator of a vehicle overtaking or meeting a school bus that has stopped and is displaying 2 alternately flashing red lights located at the same level shall bring the vehicle to a full stop not less than 20 feet from the school bus and shall not proceed until the school bus resumes motion or the visual signs are no longer actuated." MCL 257.682(1). "The operator of a vehicle that fails to stop for a school bus as required by [MCL 257.682(1)], that passes a school bus in violation of [MCL 257.682(1)], or that fails to stop for a school bus in violation of an ordinance that is substantially similar to [MCL 257.682(1)], is responsible for a civil infraction and must be ordered to pay a civil fine of not less than \$100.00 and not more than \$500.00." MCL 257.682(1).

"In a proceeding for a violation of [MCL 257.682(1)], proof that the particular vehicle described in the citation was in violation of [MCL 257.682(1)], together with proof that the defendant named in the citation was, at the time of the violation, the registered owner of the vehicle, constitutes a rebuttable presumption that the registered owner of the vehicle was the driver of the vehicle at the time of the violation." MCL 257.682(3). The presumption may be rebutted at a hearing. MCL 600.8719 (informal hearing); MCL 600.8721 (formal hearing).

Camera-based violation. "Notwithstanding any provision of law to the contrary, if the operator of a vehicle fails to stop for a school bus as required under [MCL 257.682(1)], or passes a school bus in violation of MCL 257.682(1)], or fails to stop for a school bus in violation of an ordinance that is substantially similar to [MCL 257.682(1)], and the school bus is equipped with a stop-arm camera system under [MCL 257.1820], the photograph captured or video recorded by the stop-arm camera system may be used as evidence in a proceeding for a camera-based violation." MCL 257.682(4).

"A school district that uses a stop-arm camera system shall provide a photograph captured or video recorded by a stop-arm camera system for use as evidence in a proceeding for a camera-based violation if requested by an investigating law enforcement agency." MCL 257.682(4). "A photograph or video recorded by a stop-arm camera system is admissible as evidence in a proceeding for a camera-based violation to the extent permitted by the [Michigan Rules of Evidence]." MCL 257.682(4). "However, a photograph captured or video recorded by a stop-arm camera system is not required for the prosecution of a violation of [MCL 257.682(1)]." MCL 257.682(4).

"For a camera-based violation, by not later than 30 days after receiving stop-arm camera system information as described in...MCL 257.1820, a law enforcement agency may review that information to determine if there is sufficient evidence that a violation of [MCL 257.682(1)] occurred and, if there is sufficient evidence that a violation occurred, may issue a citation." MCL 257.682(6). "[I]f a law enforcement agency determines that it has sufficient evidence that a violation of [MCL 257.682(1)] has occurred, the law enforcement agency may initiate an action by mailing via first class mail a citation to the operator of the vehicle involved in the violation." MCL 257.682(7). The mailing must include all of the following information:

- "A copy of the captured photograph or selected images from a recorded video showing the vehicle involved in the violation." MCL 257.682(7)(a).
- "If the violation is based on a recorded video, a method to review the recorded video on a website." MCL 257.682(7)(b).
- "The date, time, and location of the alleged violation." MCL 257.682(7)(c).
- "A statement of the facts inferred from the captured photograph or recorded video." MCL 257.682(7)(d).

B. Exception

"Except where a crosswalk or pedestrian walkway is present, the operator of a vehicle on a highway that has been divided into 2 roadways by leaving a raised intervening space, or by a physical barrier, or clearly indicated dividing sections so constructed as to impede vehicular traffic, is not required to stop upon meeting a school bus that has stopped across the raised intervening space, physical barrier, or dividing section." MCL 257.682(2).

C. Penalties

"For a violation of [MCL 257.682] or a local ordinance that substantially corresponds to [MCL 257.682], the civil fine ordered under [MCL 257.907] must not be less than \$100.00 or more than \$500.00." MCL 257.907(2)(m). "A citation issued under [MCL 257.682(1)] is not a citation for a camera-based violation under [MCL 257.682(4)], and a civil fine for a violation of [MCL 257.682(1)] must be applied as provided in [MCL 257.909(1)]." MCL 257.682(1).

"For a camera-based violation, the operator of a vehicle is responsible for a civil infraction and must be ordered to pay a civil fine of not less than \$100.00 and not more than \$500.00." MCL 257.682(5). "Notwithstanding any provision of law to the contrary, a civil fine for a camera-based violation must be paid to the county treasurer or the county treasurer's designee, who shall distribute the paid civil fines not less than monthly to the school district that operates the school bus." MCL 257.682(8). "A school district that receives money under [MCL 257.682(8)] must use that money for school transportation safety-related purposes." MCL 257.682(8). The general rules for assessing costs apply to violations involving failing to stop for a school bus. See Section 1.20 for a discussion of the general rules governing the assessment of costs; see also the Michigan Judicial Institute's table for a quick reference guide.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*I*). See Section 1.42 for more information on points.
- Three points are assessed for "[d]isobeying a traffic signal" MCL 257.320a(1)(r). See Section 1.42 for more information on points.

4.5 Following a Fire Truck Too Closely

A. Statutory Authority

"The driver of a vehicle other than a vehicle on official business shall not follow any fire apparatus traveling in response to a fire alarm closer than 500 feet or drive[] into . . . within 500 feet where fire apparatus has stopped in answer to a fire alarm." MCL 257.679(1).

B. Penalties

The general rules for assessing a civil fine and costs apply to following a fire truck too closely violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations

published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicles reported under [MCL 257.320a]." MCL 257.320a(1)(u). See Section 1.42 for more information on points.

4.6 Improper Passing of a Stationary Emergency Vehicle

A. Statutory Authority

"(1) Upon approaching and passing a stationary authorized emergency vehicle that is giving a visual signal by means of flashing, rotating, or oscillating red, blue, white, or amber lights as permitted by [MCL 275.698], the driver of an approaching vehicle shall exhibit due care and caution, as required under the following:

(a) On any public roadway with at least 2 adjacent lanes proceeding in the same direction of the stationary authorized emergency vehicle, the driver of the approaching vehicle shall proceed with caution, reduce his or her speed by at least 10 miles per hour below the posted speed limit, and yield the right-of-way by moving into a lane at least 1 moving lane or 2 vehicle widths apart from the stationary authorized emergency vehicle, unless directed otherwise by a police officer. If movement to an adjacent lane or 2 vehicle widths apart is not possible due to weather, road conditions, or the immediate presence of vehicular or pedestrian traffic in parallel moving lanes, the driver of the approaching vehicle proceed required shall as in [MCL 257.653a(1)(b)].

(b) On any public roadway that does not have at least 2 adjacent lanes proceeding in the same direction as the stationary authorized emergency vehicle, or if the movement by the driver of the vehicle into an adjacent lane or 2 vehicle widths apart is not possible as described in [MCL 257.653a(1)(a)], the approaching vehicle shall proceed with due care and caution and

reduce his or her speed by at least 10 miles per hour below the posted speed limit, or as directed by a police officer.

* * *

(5) The operator of a vehicle upon a highway that has been divided into 2 roadways by leaving an intervening space, or by a physical barrier or clearly indicated dividing sections so constructed as to impede vehicular traffic, is not required to proceed with caution, reduce his or her speed, or yield the right-of-way for an authorized emergency vehicle that is stopped across the dividing space, barrier, or section." MCL 257.653a(1); MCL 257.653a(5).

B. Penalties

A violation of MCL 257.653a(1) that does not result in injury or death to emergency response personnel is a civil infraction. MCL 257.653a(2). The violator must be ordered to pay a civil fine of \$400.00. *Id*.

Note: Different penalties apply if the violation of MCL 257.653a(1) causes injury or death to a police officer, firefighter, or other emergency response personnel in the immediate area of the stationary authorized emergency vehicle. MCL 257.653a(3); MCL 257.653a(4). See Section 8.3 for a discussion of these felony offenses.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(o); MCL 257.732(1)(a). See Section 1.42 for more information on points.

4.7 Improper Use of Emergency Lights

A. Statutory Authority

"The use or possession of flashing, oscillating, or rotating lights of any color is prohibited except as otherwise provided by law or under the following circumstances:

(a) A police vehicle must be equipped with flashing, rotating, or oscillating red or blue lights, for use in the performance of police duties.

(b) A fire vehicle or ambulance available for public use or for use of the United States, this state, or any unit of this state, whether publicly or privately owned, must be equipped with flashing, rotating, or oscillating red lights and used as required for safety.

(c) An authorized emergency vehicle may be equipped with flashing, rotating, or oscillating red lights for use when responding to an emergency call if when in use the flashing, rotating, or oscillating red lights are clearly visible in a 360-degree arc from a distance of 500 feet.

(d) Flashing, rotating, or oscillating amber or green lights, placed in a position as to be visible throughout an arc of 360 degrees, must be used by a state, county, or municipal vehicle engaged in the removal of ice, snow, or other material from the highway and in other operations designed to control ice and snow, or engaged in other non-winter operations. This subdivision does not prohibit the use of a flashing, rotating, or oscillating green light by a fire service.

(e) A vehicle used for the cleanup of spills or a necessary emergency response action taken under state or federal law or a vehicle operated by an employee of the department of natural resources or the department of environment, Great Lakes, and energy that responds to a spill, emergency response action, complaint, or compliance activity may be equipped with flashing, rotating, or oscillating amber or green lights. The lights described in this subdivision must not be activated unless the vehicle is at the scene of a spill, emergency response action, complaint, or compliance activity. This subdivision does not prohibit the use of a flashing, rotating, or oscillating green light by a fire service.

(f) A vehicle to perform public utility service, a vehicle owned or leased by and licensed as a business for use in the collection and hauling of refuse, an automobile service car or wrecker, a vehicle of a peace officer, a vehicle operated by a rural letter carrier or a person under contract to deliver newspapers or other publications by motor route, a vehicle utilized for snow or ice removal under [MCL 257.682c], a private security guard vehicle as authorized in [MCL 257.698(7)], a motor vehicle while engaged in escorting or transporting an oversize load that has been issued a permit by the state transportation department or a local authority with respect to highways under its jurisdiction, a vehicle owned by the National Guard or a United States military vehicle while traveling under the appropriate recognized military authority, a motor vehicle while towing an implement of husbandry, or an implement of husbandry may be equipped with flashing, rotating, or oscillating amber lights. However, a wrecker may be equipped with flashing, rotating, or oscillating red lights that must be activated only when the wrecker is engaged in removing or assisting a vehicle at the scene of a traffic accident or disablement. The flashing, rotating, or oscillating amber lights must not be activated except when the warning produced by the lights is required for public safety. A vehicle engaged in authorized highway repair or maintenance may be equipped with flashing, rotating, or oscillating amber or green lights. This subdivision does not prohibit the operator of a vehicle utilized for snow or ice removal under [MCL 257.682c] that is equipped with flashing, rotating, or oscillating amber lights from activating the flashing, rotating, or oscillating amber lights when that vehicle is traveling between locations at which it is being utilized for snow or ice removal.

(g) A vehicle engaged in leading or escorting a funeral procession or any vehicle that is part of a funeral procession may be equipped with flashing, rotating, or oscillating purple or amber lights that must not be activated except during a funeral procession.

(h) An authorized emergency vehicle may display flashing, rotating, or oscillating white lights in conjunction with an authorized emergency light as prescribed in [MCL 257.698].

(i) A private motor vehicle of a physician responding to an emergency call may be equipped with and the physician may use flashing, rotating, or oscillating red lights mounted on the roof section of the vehicle either as a permanent installation or by means of magnets or suction cups and clearly visible in a 360-degree arc from a distance of 500 feet when in use. The physician shall first obtain written authorization from the county sheriff.

(j) A public transit vehicle may be equipped with a flashing, oscillating, or rotating light mounted on the roof of the vehicle approximately 6 feet from the rear of the vehicle that displays a white light to the front, side, and rear of the vehicle, which light may be actuated by the driver for use only in inclement weather such as fog, rain, or snow, when boarding or discharging passengers, from 1/2 hour before sunset until 1/2 hour after sunrise, or when conditions hinder the visibility of the **public transit vehicle**....

(k) A person engaged in the manufacture, sale, or repair of flashing, rotating, or oscillating lights governed by [MCL 257.698(5)] may possess the lights for the purpose of employment, but shall not activate the lights upon the highway unless authorized to do so under [MCL 257.698(6)].

(*I*) A vehicle used as part of a neighborhood watch program may be equipped with flashing, rotating, or oscillating amber lights, if the vehicle is clearly identified as a neighborhood watch vehicle and the neighborhood watch program is working in cooperation with local law enforcement. The lights described in [MCL 257.698(5)(*I*)] must not be activated when the vehicle is not being used to perform neighborhood watch program duties." MCL 257.698(5).

B. Penalties

"A person who operates a vehicle in violation of [MCL 257.698] is responsible for a civil infraction and shall be ordered to pay a civil fine of not more than \$100.00." MCL 257.698(9).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of

State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

4.8 Interference With View, Control, or Operation of Vehicle

A. Statutory Authority

"A person shall not drive a vehicle when it is loaded, or when there are in the front seat a number of persons, as to obstruct the view of the driver to the front or sides of the vehicle or as to interfere with the driver's control over the driving mechanism of the vehicle." MCL 257.677(1).

"A passenger in a vehicle or streetcar shall not ride in a position as to interfere with the driver's or operator's view ahead or to the sides, or to interfere with the driver's or operator's control over the driving mechanism of the vehicle or streetcar." MCL 257.677(2).

B. Penalties

The general rules for assessing a civil fine and costs apply to interference with view, control, or operation of vehicle violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*I*). See Section 1.42 for more information on points.

• Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicles reported under [MCL 257.320a]." MCL 257.320a(1)(u). See Section 1.42 for more information on points.

4.9 Overtaking and Passing

A. Statutory Authority

1. Overtaking and Passing Vehicles

"The following rules shall govern the overtaking and passing of vehicles proceeding in the same direction, subject to [MCL 257.637–MCL 257.643a]:

(a) The driver of a vehicle overtaking another vehicle proceeding in the same direction shall pass at a safe distance to the left of that vehicle, and when safely clear of the overtaken vehicle shall take up a position as near the right-hand edge of the main traveled portion of the highway as is practicable.

(b) Except when overtaking and passing on the right is permitted, the driver of an overtaken vehicle shall give way to the right in favor of the overtaking vehicle and shall not increase the speed of his or her vehicle until completely passed by the overtaking vehicle." MCL 257.636(1).

"A person who violates [MCL 257.636] is responsible for a civil infraction." MCL 257.636(4).

"The driver of a vehicle may overtake and pass upon the right of another vehicle only if 1 or more of the following conditions exist:

(a) When the vehicle overtaken is making or about to make a left turn.

(b) Upon a street or highway with unobstructed pavement not occupied by parked vehicles of sufficient width for 2 or more lines of moving vehicles in each direction and when the vehicles are moving in substantially continuous lanes of traffic.

(c) Upon a 1-way street, or upon a roadway on which traffic is restricted to 1 direction of

Section 4.9

movement, where the roadway is free from obstructions and of sufficient width for 2 or more lines of moving vehicles and when the vehicles are moving in substantially continuous lanes of traffic." MCL 257.637(1).

"The driver of a vehicle may overtake and pass another vehicle upon the right only under conditions permitting the overtaking and passing in safety." MCL 257.637(2). "The driver of a vehicle shall not overtake and pass another vehicle upon the right by driving off the pavement or main-traveled portion of the roadway." *Id*.

"A person who violates [MCL 257.637] is responsible for a civil infraction." MCL 257.637(4).

"The operator of a vehicle shall not follow a snowplow more closely than 200 feet." MCL 257.643b(1). However, MCL 257.643b(1) "does not apply when the operator of a vehicle is legally overtaking or passing a snowplow." MCL 257.643b(2). "The operator of a vehicle that is not a snowplow who approaches a snowplow from the rear that is stopped at or in an intersection shall stop not less than 20 feet from the snowplow." MCL 257.643b(3). "A person that violates [MCL 257.643b] is responsible for a civil infraction." MCL 257.643b(4).

2. Overtaking and Passing Bicycles

"The driver of a motor vehicle overtaking a bicycle proceeding in the same direction shall pass at a safe distance of at least 3 feet to the left of that bicycle or, if it is impracticable to pass the bicycle at a distance of 3 feet to the left, at a safe distance to the left of that bicycle at a safe speed, and when safely clear of the overtaken bicycle shall take up a position as near the righthand edge of the main traveled portion of the highway as practicable." MCL 257.636(2). "Notwithstanding [MCL 257.640], if it is safe to do so, the driver of a vehicle overtaking a bicycle proceeding in the same direction may overtake and pass the bicycle in a no-passing zone." MCL 257.636(3).

"A person who violates [MCL 257.636] is responsible for a civil infraction." MCL 257.636(4).

MCL 257.637(3) similarly provides: "The driver of a vehicle overtaking a bicycle proceeding in the same direction shall, when otherwise permitted by [MCL 257.637], pass at a distance of 3 feet to the right of that bicycle or, if it is impracticable to

pass the bicycle at a distance of 3 feet to the right, at a safe distance to the right of that bicycle at a safe speed."

"A person who violates [MCL 257.637] is responsible for a civil infraction." MCL 257.637(4).

B. Overtaking and Passing Violations:

Common overtaking and passing violations include:

- Disobeying no-passing zones, MCL 257.640;
- Failing to give way when overtaken, MCL 257.636(1)(b);
- Following vehicle too closely, MCL 257.643;
- Following snowplow too closely, MCL 257.643b;
- Improper lane use (multiple lane highway), MCL 257.642;
- Improper passing on hill or curve, MCL 257.639(1)(a);
- Improper passing on right, MCL 257.637;
- Improper passing within 100 feet of bridge, viaduct, or tunnel with obstructed view, MCL 257.639(1)(b)

C. Penalties

The general rules for assessing a civil fine and costs apply to overtaking and passing violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

1. Improper Passing

• Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle,

an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.

• Three points are assessed for improper passing. MCL 257.320a(1)(r). See Section 1.42 for more information on points.

2. Following Too Closely, Tailgating, Improper Lane Use

• Two points are assessed for following too closely, tailgating, and improper lane use. See MCL 257.320a(1)(u) (applicable to "[a]ll other moving violations pertaining to the operation of motor vehicles reported under [MCL 257.320a]"). See Section 1.42 for more information on points.

4.10 Permitting Minor to Ride in Pickup Truck Bed

A. Statutory Authority

"Except as provided in [MCL 257.682b], an operator shall not permit a person less than 18 years of age to ride in the open bed of a pickup truck on a highway, road, or street in a city, village, or township at a speed greater than 15 miles per hour." MCL 257.682b(1).

B. Exceptions

"[MCL 257.682b(1)] does not apply to the operator of any of the following:

(a) A motor vehicle operated as part of a parade pursuant to a permit issued by the governmental unit with jurisdiction over the highway or street.

(b) A military motor vehicle.

(c) An authorized emergency vehicle.

(d) A motor vehicle controlled or operated by an employer or an employee of a farm operation, construction business, or similar enterprise during the course of work activities.

(e) A motor vehicle used to transport a search and rescue team to and from the site of an emergency." MCL 257.682b(2).

C. Penalties

The general rules for assessing a civil fine and costs apply to permitting a minor to ride in a pickup truck bed. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*I*). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

4.11 Railroad Crossings

A. Statutory Authority

"When a person driving a vehicle approaches a railroad grade crossing under any of the following circumstances, the driver shall stop the vehicle not more than 50 feet but not less than 15 feet from the nearest rail of the railroad, and shall not proceed until the driver can do so safely:

(a) A clearly visible electric or mechanical signal device gives warning of the immediate approach of a railroad train or other on-track equipment.

(b) A crossing gate is lowered or a flagman gives or continues to give a signal of the approach or passage of a railroad train or other on-track equipment.

(c) A railroad train or other on-track equipment approaching within approximately 1,500 feet of the

highway crossing gives a signal audible from that distance, and the railroad train or on-track equipment by reason of its speed or nearness to the crossing is an immediate hazard.

(d) An approaching railroad train or other on-track equipment is plainly visible and is in hazardous proximity to the crossing." MCL 257.667(1).²

"A person shall not drive a vehicle through, around, or under a crossing gate or barrier at a railroad crossing while the gate or barrier is closed or is being opened or closed or against the direction of a police officer." MCL 257.667(2).

Certain grade crossings may be designated as "stop" crossings and "yield" crossings; if so designated, appropriate signs are to be erected to notify drivers. A driver's duties depend on the designation. MCL 257.668.

Stop crossings—"[T]he driver of a vehicle shall stop not more than 50 feet but not less than 15 feet from the railway tracks. The driver shall then traverse the crossing when it may be done in safety." MCL 257.668(1).

Yield crossings—"Drivers of vehicles approaching a yield sign at the grade crossing of a railway shall maintain a reasonable speed based upon existing conditions and shall yield the right-of-way." MCL 257.668(2).

A person who violates either MCL 257.667 or MCL 257.668 is responsible for a civil infraction. MCL 257.667(3); MCL 257.668(4).

B. Penalties

The general rules for assessing a civil fine and costs apply to railroad crossing violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations

 $^{^2}$ MCL 257.1857 addresses railroad crossing requirement for school buses. However, that provision is outside the scope of this benchbook.

published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.20 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Three points are assessed for "[d]isobeying a traffic signal or stop sign" MCL 257.320a(1)(r). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). Section 1.42 for more information on points.

4.12 Right-of-Way or Failure to Yield

A. Statutory Authority

MCL 257.649 and MCL 257.649a contain several provisions for yielding the right-of-way to another vehicle. However, "[t]he driver of a vehicle traveling at an unlawful speed forfeits a right of way that the driver might otherwise have under [MCL 257.649]." MCL 257.649(7). A person who violates MCL 257.649 or MCL 257.649a is responsible for a civil infraction. See MCL 257.649(10)and MCL 257.649a(2).

1. Entering Intersection from Different Highways

"The driver of a vehicle approaching an intersection shall yield the right of way to a vehicle that has entered the intersection from a different highway." MCL 257.649(1).

"When 2 vehicles enter an intersection from different highways at approximately the same time, the driver of the vehicle on the left shall yield the right of way to the vehicle on the right." MCL 257.649(2).

"The right of way rules in [MCL 257.649(1) and MCL 257.649(2)] are modified at through highways and otherwise as provided in [MCL 257.649(4) and MCL 257.649–MCL 257.655]." MCL 257.649(3).

2. Approaching an Intersection Controlled by Certain Traffic Control Signals

"The driver of a vehicle approaching an intersection that is controlled by a traffic control signal shall do all of the following, if the signal facing the driver exhibits no colored lights or colored lighted arrows, exhibits a combination of colored lights or colored lighted arrows that fails to clearly indicate the assignment of the right of way, or the signals are otherwise malfunctioning:

> (a) Stop at a clearly marked stop line, or if there is no clearly marked stop line, stop before entering the crosswalk on the near side of the intersection, or, if there is no crosswalk, stop before entering the intersection.

> (b) Yield the right of way to all vehicles in the intersection or approaching on an intersecting road, if those vehicles will constitute an immediate hazard during the time the driver is moving across or within the intersection.

(c) Exercise ordinary care while proceeding through the intersection." MCL 257.649(4).

"[MCL 257.649(4)] does not apply to either of the following:

(a) An intersection that is controlled by a traffic control signal that is flashing yellow unless certain events occur, including, but not limited to, activation by an emergency vehicle.

(b) A traffic control signal that is located in a school zone and is flashing yellow only during prescribed periods of time." MCL 257.649(5)(a)-(b).

3. Approaching a Yield Sign

"The driver of a vehicle approaching a yield sign, in obedience to the sign, shall slow down to a speed reasonable for the existing conditions and shall yield the right of way to a vehicle in the intersection or approaching on another highway so closely as to constitute an immediate hazard during the time the driver would be moving across or within the intersection." MCL 257.649(6). "However, if required for safety to stop, the driver shall stop before entering the crosswalk on the near side of the intersection or, if there is not a crosswalk, at a clearly marked stop line; but if there is not a crosswalk or a clearly marked stop line, then at the point nearest the intersecting roadway where the driver has a view of approaching traffic on the intersecting roadway." *Id.*

4. Traveling at an Unlawful Speed

"The driver of a vehicle traveling at an unlawful speed forfeits a right of way that the driver might otherwise have under [MCL 257.649]." MCL 257.649(7).

5. Approaching a Stop Sign³

"Except when directed to proceed by a police officer, the driver of a vehicle approaching a stop intersection indicated by a stop sign shall stop before entering the crosswalk on the near side of the intersection, or if there is not a crosswalk shall stop at a clearly marked stop line; or if there is not a crosswalk or a clearly marked stop line, then at the point nearest the intersecting roadway where the driver has a view of approaching traffic on the intersecting roadway." MCL 257.649(8). "After having stopped, the driver shall yield the right of way to a vehicle that has entered the intersection from another highway or that is approaching so closely on the highway as to constitute an immediate hazard during the time when the driver would be moving across or within the intersection." *Id.*

6. Merging

"When a vehicle approaches the intersection of a highway from an intersecting highway or street that is intended to be, and is constructed as, a merging highway or street, and is plainly marked at the intersection with appropriate merge signs, the vehicle shall yield right of way to a vehicle so close as to constitute an immediate hazard on the highway about to be entered and shall adjust its speed so as to enable it to merge safely with the through traffic." MCL 257.649(9).

7. Turning Left Across Lane of Traffic

"The driver of a vehicle at a location other than an intersection intending to turn to the left across a lane of traffic traveling in

³ For more information on stop and go, signs and signal violations, see Section 4.14.

the opposite direction shall yield the right-of-way to all vehicles approaching from the opposite direction that are so close to the driver as to constitute an immediate hazard." MCL 257.649a(1).

B. Right-of-Way or Failure to Yield Violations

Common right-of-way or failure to yield violations include:

- Failing to obey stop, yield, or merge signs on through highways, MCL 257.671(3);
- Failing to stop at stop sign, MCL 257.649(8);
- Failing to yield at yield sign, MCL 257.649(6);
- Failing to yield from alley, private road, or driveway, MCL 257.652(1);
- Failing to yield to emergency vehicles, MCL 257.653⁴;
- Failing to yield to funeral procession, MCL 257.654;
- Failing to yield to oncoming traffic when merging onto highway, MCL 257.649(9);
- Failing to yield to pedestrians or bicyclists, MCL 257.612;
- Failing to yield to vehicle on the right at an uncontrolled intersection, MCL 257.649(2);
- Failing to yield to vehicle that has already entered an intersection, MCL 257.649(1); and
- Turning left at intersection into oncoming traffic, MCL 257.650.

C. Penalties

The general rules for assessing a civil fine and costs apply to rightof-way and failure to yield violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

⁴MCL 257.653 (failure to yield to emergency vehicles) is a civil infraction which is similar to MCL 257.653a (duty to exhibit due care and caution to stationary emergency vehicle giving visual signal), a misdemeanor. See Section 6.10 for more information on MCL 257.653a.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Three points are assessed for "[d]isobeying a . . . stop sign" MCL 257.320a(1)(r). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

E. Issues

"[T]he standard of care to which the . . . driver with the right of way . . . must adhere is a standard of reasonable care or due care under the circumstances." *Placek v Sterling Hts*, 405 Mich 638, 669 (1979).

• Failing to yield to emergency vehicles, MCL 257.653

"[A] driver traveling on a through street, as against an emergency vehicle, has a right to cross the intersection unless, by the reasonable exercise of the senses of sight and hearing, he or she should have noticed or heard warning to the contrary." *Placek*, 405 Mich at 671-672.

• Turning left at intersection into oncoming traffic, MCL 257.650

"[T]he law [does not] require [a driver], after entering [an] intersection under a favorable green light, to stop and wait for a change in the traffic light before completing [a] left turn." *Neander v Clampett*, 344 Mich 292, 295 (1955).

4.13 Speed Violations

"An individual operating a vehicle on a highway shall operate that vehicle at a careful and prudent speed not greater than nor less than is reasonable and proper, having due regard to the traffic, surface, and width of the highway and of any other condition existing at the time. An individual shall not operate a vehicle on a highway at a speed greater than that which will permit a stop within the assured, clear distance ahead." MCL 257.627(1).

"A violation of [MCL 257.627(1)] shall be known and may be referred to as a violation of the basic speed law or 'VBSL'." *Id.*

"An individual who violates a speed limit established under [MCL 257.627] is responsible for a civil infraction." MCL 257.627(16).

MCL 257.627(1) identifies two concepts: *careful and prudent speed* and *assured, clear distance ahead.* Underlying the concept of *careful and prudent speed* is the premise of ordinary care, e.g., the rate of speed that the average person would conclude to be proper, considering all conditions. Some conditions to consider include:

- weather (rain, wind, snow, etc.);
- time of day (day or night);
- road surface (rough, wet, icy, etc.);
- sight limitations (hills, curves, parked cars, etc.);
- traffic volume (pedestrians, other types of vehicles); and
- vehicle type (stopping distance or braking capacity).

The concept of *assured clear distance ahead* is typically applied to accident cases because the collision itself is evidence of the inability to stop within an assured, clear distance ahead. The ability to stop as a measurement of speed is contingent on several factors, including:

- driver's perception and reaction time;
- road surface conditions; and
- the vehicle's braking capacity.

A. Statutory Authority: Lawful Highway Speeds

"Except as provided in [MCL 257.627(1)], it is lawful for the operator of a vehicle to operate that vehicle on a highway at a speed not exceeding the following:

(a) Fifteen miles per hour on a highway segment within the boundaries of a mobile home park

(b) Twenty-five miles per hour on a highway segment within a business district.

(c) Twenty-five miles per hour on a highway segment within the boundaries of a public park. A local authority may decrease the speed limit to not less than 15 miles per hour in a public park under its jurisdiction.

(d) Twenty-five miles per hour on a highway segment within the boundaries of a residential subdivision, including a condominium subdivision, consisting of a system of interconnected highways with no through highways and a limited number of dedicated highways that serve as entrances to and exits from the subdivision.

(e) Twenty-five miles per hour on a highway segment that is part of the local street system as designated by a local jurisdiction and approved by the state transportation commission under . . . MCL 247.651 to [MCL] 247.675, and that is within land that is zoned for residential use by the governing body of an incorporated city or village under the Michigan zoning enabling act, . . . MCL 125.3101 to [MCL] 125.3702, unless another speed is fixed and posted.

(f) Twenty-five miles per hour on a highway segment with 60 or more vehicular access points within 1/2 mile.

(g) Thirty miles per hour on a highway segment with not less than 50 vehicular access points but no more than 59 vehicular access points within 1/2 mile.

(h) Thirty-five miles per hour on a highway segment with not less than 45 vehicular access points but no more than 49 vehicular access points within 1/2 mile.

(i) Forty miles per hour on a highway segment with not less than 40 vehicular access points but no more than 44 vehicular access points within 1/2 mile. (j) Forty-five miles per hour on a highway segment with not less than 30 vehicular access points but no more than 39 vehicular access points within 1/2 mile." MCL 257.627(2).

"All of the following apply to the speed limits described in [MCL 257.627(2)]:

(a) A highway segment adjacent to or lying between 2 or more areas described in [MCL 257.627(2)(a), MCL 257.627(2)(b), MCL 257.627(2)(c), or MCL 257.627(2)(d)] is not considered to be within the boundaries of those areas.

(b) A highway segment of more than 1/2 mile in length with a consistent density of vehicular access points equal to the number of vehicular access points described in [MCL 257.627(2)(f), MCL 257.627(2)(g), MCL 257.627(2)(h), MCL 257.627(2)(i), or MCL 257.627(2)(j)] must be posted at the speed limit specified in the adjoining segment. A separate determination must be made for each adjoining highway segment where vehicular access point density is different.

(c) A speed limit may be posted on highways less than 1/2 mile in length by prorating in 1/10 mile segments the vehicular access point density described in [MCL 257.627(2)(f), MCL 257.627(2)(g), MCL 257.627(2)(h), MCL 257.627(2)(i), or MCL 257.627(2)(j)]." MCL 257.627(5).

1. Special Rules for Certain Trucks

"An individual operating a truck with a gross weight of 10,000 pounds or more, a truck-tractor, a truck-tractor with a semitrailer or trailer, or a combination of these vehicles shall not exceed a speed of 35 miles per hour during the period when reduced loadings are being enforced in accordance with this chapter." MCL 257.627(3).

2. Special Rules Where Posted Speed Limit is Greater than 65 miles per hour

"Where the posted speed limit is greater than 65 miles per hour, an individual operating a school bus, a truck with a gross weight of 10,000 pounds or more, a truck-tractor, or a trucktractor with a semi-trailer or trailer or a combination of these vehicles shall not exceed a speed of 65 miles per hour on a limited access freeway or a state trunk line highway." MCL 257.627(4).

3. Work Zones

"An individual operating a vehicle on a highway, when entering and passing through a work zone described in [MCL 257.79d(a)] where a normal lane or part of the lane of traffic has been closed due to highway construction, maintenance, or surveying activities, shall not exceed a speed of 45 miles per hour unless a different speed limit is determined for that work zone by the state transportation department, a county road commission, or a local authority, based on accepted engineering practice." MCL 257.627(6). "The state transportation department, a county road commission, or a local authority shall post speed limit signs in each work zone described in [MCL 257.79d(a)] that indicate the speed limit in that work zone and shall identify that work zone with any other traffic control devices necessary to conform to the Michigan manual on uniform traffic control devices." MCL 257.627(6). "An individual operating a vehicle shall not exceed a speed limit established under [MCL 257.627] or a speed limit established under [MCL 257.628⁵]."MCL 257.627(6).

Automated speed enforcement systems. "Beginning [April 2, 2026], the state transportation department may install and use or authorize the installation and use of an automated speed enforcement system in a work zone that is not separated from traffic by concrete barriers on a street or highway under the jurisdiction of the state transportation department." MCL 257.627c(1). "A sign must be placed 1 mile before the start of a work zone where an automated speed enforcement system is installed and used under [MCL 257.627c] indicating that the work zone is monitored by an automated speed enforcement system." MCL 257.627c(1)(a). "A digital display must be placed on or near the automated speed enforcement system that shows the speed of an approaching vehicle as measured by the automated speed enforcement system." MCL 257.627c(1)(b). "Prioritization must be given to work zones that meet the following factors:

(*i*) The work zone is active and located on a freeway with a speed limit of 45 miles per hour or more.

⁵ MCL 257.628 authorizes certain governmental entities to establish reasonable and safe maximum or minimum speed limits on certain roads and highways.

(*ii*) Workers are exposed or there are traffic hazards, including, but not limited to, lane shifts, lane splits, lane width reductions, closed shoulders, and rough pavement.

(*iii*) The work zone will be active for 30 days or more.

(*iv*) There are no significant obstructions to line of sight for the automated speed enforcement system." MCL 257.627c(1)(c).

"Automated speed enforcement systems must be used in an equitable manner to avoid causing a disparate impact on specific communities." MCL 257.627c(1)(d). "Not more than 4 automated speed enforcement systems may be installed and used at the same time within the jurisdiction of a state transportation department region office." MCL 257.627c(1)(e). "The state transportation department may install and use an automated speed enforcement system only by contracting with a third-party vendor." MCL 257.627c(1)(f). "An automated speed enforcement system must be activated and detecting vehicle speed for the purpose of enforcing [MCL 257.627c] only while workers are present in the work zone." MCL 257.627c(1)(g).

Work zone speed limit violation(s). "Except for an individual operating a police vehicle, a fire department or fire patrol vehicle, or a public or private ambulance as provided in [MCL 257.632], if an individual violates an applicable speed limit described in [MCL 257.627] by exceeding a posted speed limit by 10 miles per hour or more in a work zone while workers are present, on the basis of a recorded image produced by an automated speed enforcement system, all of the following apply:

(a) An individual must be issued a written warning using a form that is created by the automated speed enforcement unit under [MCL 257.627c(7)] for either of the following violations as described in [MCL 257.627c(2)]:

(*i*) A first violation.

(*ii*) A violation that occurs more than 3 years after that individual's most recent violation.

(b) For a second violation as described in [MCL 257.627c(2)] that occurs less than 3 years after a written warning is issued under [MCL

257.627c(2)(a)], the individual is responsible for a civil infraction and must be ordered to pay a civil fine of not more than \$150.00.

(c) For a third or subsequent violation as described in [MCL 257.627c(2)] that occurs less than 3 years after a second or subsequent violation, the individual is responsible for a civil infraction and must be ordered to pay a civil fine of not more than \$300.00." MCL 257.627c(2).

Recorded image and data. "A sworn statement of an automated speed enforcement system operator, based on inspecting a recorded image produced by an automated speed enforcement system, is prima facie evidence of the facts contained in the recorded image." MCL 257.627c(3). "A recorded image indicating a violation must be available for inspection in any proceeding to adjudicate the responsibility for a violation of [MCL 257.627c]." MCL 257.627c(3). "A recorded image indicating a violation must be destroyed as soon as the period for contesting the violation has lapsed, including any period for appeals, or as soon as the individual pays the civil fine in full, whichever occurs first." *Id*.

"[A] recorded image and any other data collected by an automated speed enforcement system . . . may be used only for the purpose of adjudicating a violation of [MCL 257.627c]." MCL 257.627c(4)(a). "Except to the extent necessary to adjudicate a violation of [MCL 257.627c], the image and data are confidential and exempt from disclosure under the freedom of information act . . . MCL 15.231 to 15.246." MCL 257.627c(4)(b). "The image and data must not be shared with or sold to any private or public third party not involved with installing and using the automated speed enforcement system." MCL 257.627c(4)(c).

Rebuttable presumption. "In a proceeding for a violation of [MCL 257.627c], prima facie evidence that the vehicle described in the citation issued was operated in violation of [MCL 257.627c], together with proof that the individual who was issued the citation was at the time of the violation the registered owner of the vehicle, creates a rebuttable presumption that the registered owner of the vehicle was the individual who committed the violation." MCL 257.627c(5). "The presumption is rebutted if the registered owner of the vehicle files an affidavit by regular mail with the clerk of the court stating that the registered owner was not the operator of the vehicle at the time of the alleged violation or testifies in open court under oath that the registered owner was not the

operator of the vehicle at the time of the alleged violation." *Id.* "The presumption also is rebutted if a certified copy of a police report, showing that the vehicle had been reported to the police as stolen before the time of the alleged violation of this section, is presented before the appearance date established on the citation." *Id.* "For purposes of [MCL 257.627c(5)], the owner of a leased or rented vehicle shall provide the name and address of the individual to whom the vehicle was leased or rented at the time of the violation." MCL 257.627c(5).

Citation. Notwithstanding [MCL 257.742], a citation for a violation of [MCL 257.627c] may be executed by the state transportation department or a person designated by the state transportation department by mailing by first-class mail a copy to the address of the registered owner of the vehicle as shown on the records of the secretary of state." MCL 257.627c(6). "The state transportation department shall also file the citation with the court having jurisdiction over the offense." *Id.* "If the summoned individual fails to appear on the date of return set out in the citation mailed under [MCL 257.627c(6)], the clerk of the court shall enter a default after certifying, on a form furnished by the court, that the defendant has not made a scheduled appearance or has not answered the citation within the time provided by law." *Id.*

4. School Zones

"A school zone speed limit on a highway segment in a school zone, which, except as otherwise provided in this subsection, shall be in force not more than 30 minutes before the first regularly scheduled school session, rounded to the nearest multiple of 5 minutes, until school commences and from dismissal until not more than 30 minutes after the last regularly scheduled school session, rounded to the nearest multiple of 5 minutes, may be decreased by not more than 20 miles per hour less than the speed limit normally posted but shall be not less than 25 miles per hour. A school superintendent may begin the 30-minute period before the first regularly scheduled school session described in this subsection at a time that is less than 30 minutes before the first regularly scheduled school session and that extends beyond the time school commences, may begin the 30-minute period after dismissal at a time other than dismissal, and, if a school has an off-campus lunch period, may designate the period provided for off-campus lunch as a period during which the school zone speed limit described in this subsection applies." MCL 257.627a(2).

5. State Trunkline Highways Within Political Subdivisions That Prohibit Operation of Nonemergency Motor Vehicles

"Notwithstanding [MCL 257.627] and [MCL 257.628], an individual who operates a vehicle, a bicycle, or any other device on a state trunkline highway within a political subdivision [that prohibits the operation of nonemergency motor vehicles in the political subdivision by ordinance, regulation, or resolution] shall not exceed either of the following speed limits:

(a) Except as otherwise provided in [MCL 257.628b(2)(a)] and [MCL 257.628b(2)(b)], 15 miles per hour. [MCL 257.628b(2)(a)] does not apply to a snowmobile.

(b) Within a business district of a political subdivision [that prohibits the operation of nonemergency motor vehicles in the political subdivision by ordinance, regulation, or resolution], 10 miles per hour." MCL 257.628b(2).

"An individual who violates [MCL 257.628b] is responsible for a civil infraction." MCL 257.628b(3).

B. Statutory Authority: Maximum and Minimum Speed Limits and General Speed Limit

"If the county road commission, the township board, and the department of state police unanimously determine upon the basis of an engineering and traffic investigation that the speed of vehicular traffic on a county highway is greater or less than is reasonable or safe under the conditions found to exist upon any part of the highway, then acting unanimously they may establish a reasonable and safe maximum or minimum speed limit on that county highway that is effective at the times determined when appropriate signs giving notice of the speed limit are erected on the highway." MCL 257.628(1).

"Subject to [MCL 257.627(17)⁶], the speed limit on all trunk line highways and all county highways upon which a speed limit is not otherwise fixed under [the MVC] is 55 miles per hour, which shall be known as the 'general speed limit'." MCL 257.627(9).

"Subject to [MCL 257.627(17)], the maximum speed limit on all limited access freeways upon which a speed limit is not otherwise fixed under [the MVC] is 70 miles per hour, which shall be known as the 'limited access freeway general speed limit'." MCL 257.627(8).

"The minimum speed limit on all limited access freeways upon which a minimum speed limit is not otherwise fixed under [the MVC] is 55 miles per hour." MCL 257.627(8).

"Except as otherwise provided in this subsection, the speed limit on all county highways with a gravel or unimproved surface upon which a speed limit is not otherwise fixed under [the MVC] is 55 miles per hour, which shall be known as the 'general gravel road speed limit'." MCL 257.627(10).

C. Statutory Authority: Modified Speed Limits

"The governor may reduce the maximum speed limit on a street, highway, or freeway pursuant to an executive order issued during a state of energy emergency as provided by law." MCL 257.629b(1).

Township boards may petition for changes in the speed limit. MCL 257.628(1)-(2).

MCL 257.627(7) requires the state transportation department, a county road commission, or a local authority to decrease the speed limit in a hospital highway zone by up to 10 miles per hour upon the request of a hospital located in the zone. The speed limit may be decreased by more than 10 miles per hour if supported by an engineering safety study. *Id.*

MCL 257.627(10) permits the decrease of the general gravel road speed limit⁷ under certain circumstances that differ depending on the population size of the municipality.

⁶MCL 257.627(17) provides: "No later than January 5, 2018, the state transportation department and the department of state police shall increase the speed limits on at least 600 miles of limited access freeway to 75 miles per hour if an engineering and safety study and the eighty-fifth percentile speed of free-flowing traffic under ideal conditions of that section contain findings that the speed limit may be raised to that speed, and the department shall increase the speed limit of 900 miles of trunk line highway to 65 miles per hour if an engineering and safety study and the eighty-fifth percentile speed of free-flowing traffic under ideal conditions of that section contain findings that the speed limit may be raised to that speed ideal conditions of that section contain findings that the speed limit may be raised to that speed."

⁷See Section 4.13(B) for discussion of the general gravel road speed limit.

D. Special Citation Requirements

In addition to the requirements discussed in Section 1.12, "[i]n every charge of a violation of a speed limit in [Chapter VI of the MVC] the complaint or citation and the summons or notice to appear shall specify the speed at which the respondent is alleged to have driven and the speed limit applicable at the location." MCL 257.633(1).

E. Evidence in a Speed Case

Evidence in a speed case may be presented by testimony of the defendant, complaining police officer, or a witness, or by physical evidence. In general, the court must determine the admissibility of evidence based on principles of relevancy. See MRE 401–MRE 403. However, a district court magistrate conducting an informal hearing in a civil infraction case is not bound by the rules of evidence. See MCL 257.746(1).

"[A]dmissions made by a driver to a police officer are admissible in any court proceeding." *People v Chandler*, 75 Mich App 585, 590 (1976).

1. Speed Evidence Based on Witness Observation

"One need not qualify as an expert in order to testify as to matters one learns through ordinary observation, such as the rate of speed at which a vehicle is going, provided a witness is fully interrogated as to the knowledge upon which his judgment is based, so that a jury can determine what weight should be given to his [or her] statements." *Hicks v Bacon*, 26 Mich App 487, 493 (1970). "With respect to testimony regarding the speed of a vehicle, the admissibility of this testimony is not contingent upon specific times and distances, and the testimony should be admitted where the jury is made aware of the witness' opportunity to observe." *Cole v Eckstein*, 202 Mich App 111, 114 (1993).

2. Speed Evidence Based on Radar, Vascar, or Laser Devices

• Radar

Radar (Radio Detection and Ranging) detects the presence of a moving object and determines its speed. According to *People v Ferency*, 133 Mich App 526, 542-544 (1984),"a speeding case involving 'moving' radar, the following seven guidelines must be met in order to allow into evidence speed readings from a radar speedmeter [thus satisfying due process requirements]:

1. The officer operating the device has adequate training and experience in its operation.

2. That the radar device was in proper working condition and properly installed in the patrol vehicle at the time of the issuance of the citation.

3. That the device was used in an area where road conditions are such that there is a minimum possibility of distortion.

4. That the input speed of the patrol vehicle was verified. This also means that the speedometer of the patrol vehicle was independently calibrated.

5. That the speedmeter be retested at the end of the shift in the same manner that it was tested prior to the shift and that the speedmeter be serviced by the manufacturer or other professional as recommended.^[8]

6. That the radar operator be able to establish that the target vehicle was within the operational area of the beam at the time the reading was displayed.

7. That the particular unit has been certified for use by an agency with some demonstrable expertise in the area."

The Michigan Commission on Law Enforcement Standards (MCOLES) has adopted Michigan Speed Measurement Standards applicable to individuals certified as speed measurement operators on or after January 1, 2010. The standards also contain specifications for speedmeter calibration. See http://www.michigan.gov/documents/mcoles/Web_Language_304054_7.pdf.

The Michigan Speed Measurement Task Force set out *Guidelines for the Adjudication of Radar Speeding Cases* in 2000 to "provide recommendations to members of the Michigan judicial system regarding the adjudication of radar speeding cases." See <u>http://www.michigan.gov/documents/ADJUDICATIONOFRADARSPEEDING_CASES_11138_7.pdf</u>. The recommendations are as follows:

⁸This "requirement... does not mandate any specific actions." *City of Adrian v Strawcutter*, 259 Mich App 142, 144 (2003). "The requirement . . . only mandates service *as recommended*." *Id*. "This does not preclude the possibility that *no* service may be recommended." *Id*. at 144-145 (officer complied with relevant servicing requirements under [*People v Ferency*, 133 Mich App 526 (1984)] where the Michigan Speed Measurement Task Force did not recommend any servicing for the speedmeter unit, and no servicing was performed).

"1. The Michigan Speed Measurement Task Force recommends that the guidelines listed in [*People v Ferency*, 133 Mich App 526 (1984)] be considered valid for both stationary-mode and moving-mode radar citations.

2. The Michigan Speed Measurement Task Force recommends that speed-measuring radar evidence be admissible in court only if the radar device used was <u>certified</u>, as determined by the Michigan Speed Measurement Task Force.

3. The Michigan Speed Measurement Task Force recommends that it is <u>not necessary</u> to have radar devices periodically recertified because a properly trained radar operator will be able to determine when a specific device is malfunctioning.

4. The Michigan Speed Measurement Task Force recommends that speed measuring radar device evidence be admissible in court only if the radar operator was <u>certified</u> [by] the Michigan Commission On Law Enforcement Standards at the time the radar speed reading was made.

5. Only if the radar device and radar operator were each properly certified should issues related to this particular case be addressed in order to determine if the specific facts warrant that the defendant be held responsible. Specific points that should be covered, once the certification issues have been dispensed with, include:

a. Was the radar device in proper working order? And when was this verification done?

b. Was the patrol vehicle's speedometer independently calibrated? And, if so, when was it last calibrated?

c. What mode of operation was used (e.g., stationary or moving)?

d. Was the radar device being used in an area where road conditions or environmental conditions might have led to spurious display readings?

e. What was the nature of the roadway (i.e., type of roadway, general visibility, terrain,

visual obstructions, and volume of traffic flow)?

f. What was the target-tracking history (i.e., visual observations of the target, operational area of the radar beam, characteristics of the Doppler-audio signal, display readings, and correlation between the patrol speed display window reading and the reading from the patrol vehicle's speedometer—the latter only being needed during moving-mode operation). *Guidelines for the Adjudication of Radar Speeding Cases, supra.*

In summary, the Michigan Speed Measurement Task Force recommends that the defendant be held responsible for the speeding infraction if the following three conditions are met: first, the radar device was certified as determined by the Michigan Speed Measurement Task Force; second, the radar operator was certified by the Michigan Commission On Law Enforcement Standards; and, third, the preponderance of the forensic evidence related to this specific case indicates that the speeding infraction did occur as stated by the radar operator."

• Laser (Lidar)

Laser emits an invisible infrared light beam that measures both speed and distance.

"Although the Michigan Court of Appeals ruling in [*People v Ferency*, 133 Mich App 526 (1984)] deals with the adjudication of a case involving traffic radar, the Michigan Speed Measurement Task Force is of the opinion that the ruling can be applied to cases involving laser speed measurement devices." See https://www.michigan.gov/msp/0,4643,7-123-72297_30536_53738-16134--,00.html. "The interim guidelines for adjudicating speeding cases involving laser speed measurement devices that have been developed by the Michigan Speed Measurement Task Force reflect this opinion:

1. The officer operating the laser speed measurement device must have adequate training and experience.

2. The particular laser device must have been certified for use in Michigan by the Michigan Speed Measurement Task Force.

3. The laser device must be verified in the same manner at the beginning and end of the shift to ensure that it is in proper working condition, and the device must be serviced by the manufacturer or other professional as recommended.

4. The officer using the laser device must be able to testify that a down-the-road speed reading was obtained at a distance that was within the operational range of the device.

5. The target vehicle must be properly identified.

6. The laser device must be in proper working condition at the time the speed measurement reading is obtained. Additionally, across-the-road laser devices must be properly positioned and aligned." *Id.*

• VASCAR

VASCAR (Visual Average Speed Computer and Recorder) is a computer device that allows a police officer to enter a precisely measured distance and the time it took the target vehicle to travel that same distance. The computer then calculates the average speed of the target vehicle.

A proper foundation must be laid for the admission of evidence from a VASCAR unit. People v Blondia, 69 Mich App 554, 559 (1976). In *Blondia*, 69 Mich App at 558, "[t]he trooper testified that he calibrated the VASCAR unit before each shift on a premeasured half-mile course . . . [and] that he had not participated in the measuring of the course but that other troopers had used it and had marked it and he thought it was accurate." The trooper "also used a stop watch to test the VASCAR unit but he did not have anything to offer on the accuracy of the stop watch except that it was not a certified stop watch as prescribed in the manufacturer's instructions on the use of the VASCAR unit." Id. The Court of Appeals stated that "[i]f the outcome of th[e] appeal depended on the VASCAR speed evidence alone, [it] would hold that there was an insufficient showing of the accuracy of the unit because a proper foundation was not laid." Id. at 559.

F. Penalties

The general rules for assessing a civil fine and costs apply to speed violations. See Section 1.20 for a discussion of the general rules

governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

"Notwithstanding [MCL 257.320a and MCL 257.907], a person who is determined responsible or responsible 'with explanation' for a civil infraction for violating the maximum speed limit on a limited access freeway or part of a limited access freeway upon which the maximum speed limit is 55 miles per hour or more shall be ordered by the court to pay a minimum fine . . . only according to the . . . schedule [established in MCL 257.629c], except as otherwise provided in [MCL 257.629c(2) and MCL 257.629c(3).]" MCL 257.629c(1). The schedule is as follows:

- 1-5 mph over speed limit—\$10
- 6-10 mph over speed limit—\$20
- 11-15 mph over speed limit—\$30
- 16-25 mph over speed limit—\$40
- 26+ mph over speed limit—\$50

"[MCL 257.629c(1)] does not apply to a person operating a vehicle or a vehicle combination for which the maximum rate of speed is established pursuant to [MCL 257.627(5)-(7)]." MCL 257.629c(2).⁹

"For a violation of a maximum speed limit on a limited access freeway by a person operating a vehicle or vehicle combination described in [MCL 257.629c(2)], . . . fines shall be assessed under [MCL 257.907]." MCL 257.629c(3).

G. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

"Within 5 days after receipt of a properly prepared abstract from a court of [Michigan] or another state, the secretary of state shall record the . . . civil infraction determination . . . and the number of

⁹Effective January 5, 2017, 2016 PA 445 amended MCL 257.627 to delete the content of former § 627(5) and former § 627(7). The content in § 627(6) was re-numbered as § 627(3) and amended. MCL 257.629c has not been amended to reflect the changes to MCL 257.627.

points for each, based on the following formula, except as otherwise provided in [MCL 257.320a and MCL 257.629c]:

* * *

(g) A violation of MCL 257.627(6)] pertaining to speed in a work zone described in [MCL 257.627] by exceeding the lawful maximum by more than 15 miles per hour5 points

(h) A violation of any law or ordinance pertaining to speed by exceeding the lawful maximum by more than 15 miles per hour 4 points

* * *

(k) A violation of [MCL 257.627(6)] pertaining to speed in a work zone described in that section by exceeding the lawful maximum by more than 10 but not more than 15 miles per hour 4 points

(*l*) Beginning October 31, 2010, a moving violation resulting in an at-fault collision with another vehicle, aan individual, or any other object 4 points

(m) Careless driving in violation of [MCL 257.626b] or a law substantially corresponding to [MCL 257.626b] 3 points

(n) A violation of any law or ordinance pertaining to speed by exceeding the lawful maximum by more than 10 miles per hour but not more than 15 miles per hour 3 points

(o) A violation of [MCL 257.653a(2)] 2 points

(p) A violation of any law or ordinance pertaining to speed by exceeding the lawful maximum by more than 5 miles per hour but not more than 10 miles per hour 2 points

(q) A violation of any law or ordinance pertaining to speed by exceeding the lawful maximum by more than 1 mile per hour but not more than 5 miles per hour 1 point

* * *

(u) All other moving violations pertaining to the operation of motor vehicles reported under [MCL 257.320a] 2 points

(w) A violation of [MCL 257.627(6)] pertaining to speed in a work zone described in that section by exceeding the lawful maximum by 10 miles per hour or less . . . 3 points[.]" MCL 257.320a(1). See Section 1.42 for more information on points.

"Notwithstanding [MCL 257.320a and MCL 257.907], a person who is determined responsible or responsible 'with explanation' for a civil infraction for violating the maximum speed limit on a limited access freeway or part of a limited access freeway upon which the maximum speed limit is 55 miles per hour or more . . . shall have points entered on his or her driving record by the secretary of state only according to the . . . schedule [established in MCL 257.629c], except as otherwise provided in [MCL 257.629c(2) and MCL 257.629c(3).]" MCL 257.629c(1). The schedule is as follows:

- 1-5 mph over speed limit—0 points
- 6-10 mph over speed limit-1 point
- 11-15 mph over speed limit-2 points
- 16-25 mph over speed limit—3 points
- 26+ mph over speed limit 4 points

"[MCL 257.629c(1)] does not apply to a person operating a vehicle or vehicle combination for which the maximum rate of speed is established pursuant to [MCL 257.627(5)–MCL 257.627(7)]." MCL 257.629c(2).¹⁰

"For a violation of a maximum speed limit on a limited access freeway by a person operating a vehicle or vehicle combination described in [MCL 257.629c(2)], points shall be assessed under [MCL 257.320a.]" MCL 257.629c(3).

4.14 Stop and Go, Signs and Signals

"The state transportation department and department of state police shall adopt a manual^[11] and specifications for a uniform system of traffic control devices consistent with the provisions of [MCL 257.1 *et seq.*,] for use upon highways within [Michigan]." MCL 257.608. "The manual shall

¹⁰Effective January 5, 2017, 2016 PA 445 amended MCL 257.627 to delete the content of former § 627(5) and former § 627(7). The content in § 627(6) was re-numbered as § 627(3) and amended. MCL 257.629c has not been amended to reflect the changes to MCL 257.627.

¹¹ "Definitions and meanings found in the manual adopted under this section are supplemental to the definitions in chapter I [of the MVC]. However, if a definition or meaning found in the manual adopted under this section conflicts with a definition in chapter I, the definition in chapter I prevails." MCL 257.608.

correlate with and so far as possible conform to the federal manual then current as approved by the United States Department of Transportation, Federal Highway Administration, and may be revised whenever necessary to carry out the provisions of [the MVC]." *Id.* "It is the policy of [Michigan] to achieve, insofar as is practicable, uniformity in the design, shape, and color scheme of traffic signs, signals and guide posts erected and maintained upon the streets and highways within [Michigan] with other states." *Id.*

See the Michigan Manual of Uniform Traffic Control Devices for more information.

A. Statutory Authority: Stop Signs

"Except when directed to proceed by a police officer, the driver of a vehicle approaching a stop intersection indicated by a stop sign shall stop before entering the crosswalk on the near side of the intersection, or if there is not a crosswalk shall stop at a clearly marked stop line; or if there is not a crosswalk or a clearly marked stop line, then at the point nearest the intersecting roadway where the driver has a view of approaching traffic on the intersecting roadway." MCL 257.649(8). "After having stopped, the driver shall yield the right of way to a vehicle that has entered the intersection from another highway or that is approaching so closely on the highway as to constitute an immediate hazard during the time when the driver would be moving across or within the intersection." *Id*.

"[I]t is a general rule that a person entering an intersection must exercise that degree of care and caution that an ordinarily prudent and careful driver would exercise under the same or similar circumstances." *Stabler v Copeland*, 304 Mich 1, 7 (1942).

"A stop sign on the highway is a warning of possible danger." *Shoniker v English*, 254 Mich 76, 80 (1931). However, it is also "a direction, not merely a caution, to drivers entering a through street, to stop." *Rife v Colestock*, 297 Mich 194, 197 (1941). "It imposes the duty not only to stop the vehicle but also to make reasonable observation of the intersecting highway at a place where the view is clear and to keep the car under such control as will enable the driver to stop at once, if observation discloses approaching vehicles." *Shoniker*, 254 Mich at 80-81.

"Where . . . a stop sign is placed a considerable distance from the stop intersection, it is generally recognized that the sign serves only to notify motorists of the approaching highway intersection." *People v McIntosh*, 23 Mich App 412, 415 (1970). "It does not signify the exact spot at which vehicles are required to stop." *Id.* "Placement of

the sign some distance from the intersection of a servient and dominant highway gives the motorist ample time to slow down and stop before entering the intersection." *Id.* at 416. "It provides a necessary warning of impending danger." *Id.*

B. Statutory Authority: Traffic Lights or Signals

"When traffic is controlled by traffic control signals, not fewer than 1 signal shall be located over the traveled portion of the roadway so as to give vehicle operators a clear indication of the right-of-way assignment from their normal positions approaching the intersection." MCL 257.612(1). "The vehicle signals shall exhibit different colored lights successively, 1 at a time, or with arrows." *Id*.

1. Solid Green

"If the signal exhibits a green indication, vehicular traffic facing the signal may proceed straight through or turn right or left unless a sign at that place prohibits either turn." MCL 257.612(1)(a). "Vehicular traffic, including vehicles turning right or left, shall yield the right-of-way to other vehicles and to pedestrians and bicyclists lawfully within the intersection or an adjacent crosswalk at the time the signal is exhibited." *Id*.

2. Solid Yellow

"If the signal exhibits a steady yellow indication, vehicular traffic facing the signal shall stop before entering the nearest crosswalk at the intersection or at a limit line when marked, but if the stop cannot be made in safety, a vehicle may be driven cautiously through the intersection." MCL 257.612(1)(b).

3. Solid Red

"If the signal exhibits a steady red indication, the following apply:

(*i*) Vehicular traffic facing a steady red signal alone shall stop before entering the crosswalk on the near side of the intersection or at a limit line when marked or, if there is no crosswalk or limit line, before entering the intersection and shall remain standing until a green indication is shown, except as provided in [MCL 257.612(1)(c)(*ii*)].

(*ii*) Vehicular traffic facing a steady red signal, after stopping before entering the crosswalk on the near

side of the intersection or at a limit line when marked or, if there is no crosswalk or limit line, before entering the intersection, may make a right turn from a 1-way or 2-way street into a 2-way street or into a 1-way street carrying traffic in the direction of the right turn or may make a left turn from a 1-way or 2-way street into a 1-way roadway carrying traffic in the direction of the left turn, unless prohibited by sign, signal, marking, light, or other traffic control device. The vehicular traffic shall yield the right of way to pedestrians and bicyclists lawfully within an adjacent crosswalk to other traffic lawfully and using the intersection." MCL 257.612(1)(c).

4. Green Arrows

"If the signal exhibits a steady green arrow indication, vehicular traffic facing the green arrow signal, shown alone or in combination with another indication, may cautiously enter the intersection only to make the movement indicated by the arrow or other movement permitted by other indications shown at the same time." MCL 257.612(1)(d). "The vehicular traffic shall yield the right-of-way to pedestrians or bicyclists lawfully within an adjacent crosswalk and to other traffic lawfully using the intersection." *Id*.

5. Red and Yellow Arrows

"Red arrow and yellow arrow indications have the same meaning as the corresponding circular indications, except that they apply only to vehicle operators intending to make the movement indicated by the arrow." MCL 257.612(1).

6. Flashing Red (Stop Signal)

"When a red lens is illuminated by rapid intermittent flashes, drivers of vehicles shall stop before entering the nearest crosswalk at an intersection or at a limit line when marked and the right to proceed shall be subject to the rules applicable after making a stop at a stop sign." MCL 257.614(1)(a).

7. Flashing Yellow (Caution Signal)

"When a yellow lens is illuminated with rapid intermittent flashes, drivers of vehicles may proceed through the intersection or past the signal only with caution." MCL 257.614(1)(b).

C. Stop and Go, Sign and Signal Violations

Stop and go, sign and signal violations include:

- Avoiding traffic control device, MCL 257.611;
- Disregarding stop sign, MCL 257.649;
- Disregarding flashing red or flashing yellow signal, MCL 257.614;
- Disregarding yellow signal, MCL 257.612;
- Disregarding stop and go light, MCL 257.612; and
- Right turn on red light without stopping, MCL 257.612.

D. Penalties

The general rules for assessing a civil fine and costs apply to stop and go, and sign and signal violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

E. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Three points are assessed for "[d]isobeying a traffic signal or stop sign" MCL 257.320a(1)(r). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

4.15 Holding or Using a Mobile Electronic Device While Driving

A. Statutory Authority

Under MCL 257.602b(1), "an individual shall not hold or use a mobile electronic device while operating a motor vehicle" except as otherwise provided in MCL 257.602b.

MCL 257.602b(2) provides that "an individual shall not hold or use a mobile electronic device while operating a commercial motor vehicle or a school bus" except as otherwise provided in MCL 257.602b.

"[MCL 257.602b] supersedes all local ordinances regulating the use of a mobile electronic device while operating a motor vehicle in motion on a highway or street, except that a unit of local government may adopt an ordinance or enforce an existing ordinance substantially corresponding to [MCL 257.602b]." MCL 257.602b(7).

B. Exceptions

"[MCL 257.602b(1) and (2)] do not apply to any of the following situations:

(a) The use of a mobile electronic device by a law enforcement officer, firefighter, emergency medical technician, paramedic, operator of an authorized emergency vehicle, or similarly engaged paid or volunteer public safety first responder during the performance of that individual's official duties, or a public utility employee or contractor acting within the scope of that individual's employment when responding to a public utility emergency.

(b) The use of a mobile electronic device for emergency purposes, including calling or texting a 9-1-1 system, or making an emergency call to a law enforcement agency, health care provider, fire department, or other emergency services agency or entity to report to appropriate authorities any of the following:

(i) A fire, traffic accident, serious road hazard, or medical or hazardous materials emergency.

(ii) An operator of another motor vehicle who is driving in a reckless or otherwise unsafe manner or who appears to be driving under the influence of alcohol or drugs.

(iii) A crime being committed.

(c) The use of a global positioning or navigation feature of a mobile electronic device if information is not entered by hand into the global positioning or navigation system feature of the mobile electronic device.

(d) The use of a mobile electronic device in a voiceoperated or hands-free mode if the operator of the motor vehicle does not use the operator's hands to operate the device, except for either of the following:

(i) Using a single button press, tap, or swipe to activate or deactivate a feature or function of the mobile electronic device or to select a telephone number or name.

(ii) Using a mobile electronic device that is integrated into a motor vehicle and utilizes the user interfaces that are permanently installed into the motor vehicle.

(e) The use of a mobile electronic device used for the sole purpose of continuously recording or broadcasting video inside or outside of a motor vehicle.

(f) The use of a mobile electronic device that is placed in a mount and used in any manner as described in subdivisions (a) to (e)." MCL 257.602b(3).

"[MCL 257.602b] does not apply to any of the following:

(a) A level 3, 4, or 5 automated driving system, as described in 'J3016: Taxonomy and Definitions for Terms Related to Driving Automation Systems for On-Road Motor Vehicles', April 2021 edition, published by SAE International, or an automated vehicle equipped with an automated driving system as described in this subdivision, during testing or operation with the automated driving system engaged.

(b) Viewing or using in a hands-free manner a device mounted in a vehicle for displaying information related to testing or operating an automated driving system or automated technology. (c) Operating, or programming the operation of, an automated motor vehicle while testing or operating the automated motor vehicle without a human operator while the automated driving system is engaged." MCL 257.602b(10).

C. Penalties

The general rules for assessing costs apply to holding or using a mobile electronic device while driving. See MCL 257.907(2)(h). See Section 1.20 for a discussion of the general rules governing the assessment of costs; see also the Michigan Judicial Institute's table for a quick reference guide.

1. Holding or Using a Mobile Electronic Device While Driving

For a first violation of MCL 257.602b(1), holding or using a mobile electronic device while operating a motor vehicle, that does not involve an accident, an individual is responsible for a civil infraction and must be ordered to pay a "\$100.00 civil fine or perform 16 hours of community service, or both." MCL 257.602b(4)(a); MCL 257.907(2)(h)(*i*)(A) (providing for a civil fine of the same amount and making no reference to community service). For a second or subsequent offense that does not involve an accident, an individual is responsible for a civil infraction and must be ordered to pay "a \$250.00 civil fine or perform 24 hours of community service, or both." MCL 257.602b(4)(b); MCL 257.907(2)(h)(*i*)(A) (providing for a civil fine of the same amount and making no reference to community service).

However, if the violation of MCL 257.602b(1) involves an accident for which the individual is at fault, the individual must be ordered to pay a civil fine "double the amount that would be ordered under [MCL 257.602b(4)], as applicable." MCL 257.602b(6); MCL 257.907(2)(h)(*i*)(B) (providing for civil fines of "\$200.00 for a first offense and \$500.00 for a second or subsequent offense" and making no reference to fault).

2. Holding or Using a Mobile Electronic Device While Driving a Commercial Motor Vehicle or School Bus

For a first violation of MCL 257.602b(2), holding or using a mobile electronic device while operating a commercial motor vehicle or school bus, that does not involve an accident, an individual is responsible for a civil infraction and must be ordered to "pay a \$200.00 civil fine or perform 32 hours of

community service, or both." MCL 257.602b(5)(a); MCL 257.907(2)(h)(ii)(A) (providing for a civil fine of the same amount and making no reference to community service). For a second or subsequent offense that does not involve an accident, an individual is responsible for a civil infraction and must be ordered to pay "a \$500.00 civil fine or perform 48 hours of community service, or both." MCL 257.602b(5)(b); MCL 257.907(2)(h)(*ii*)(A) (providing for a civil fine of the same amount and making no reference to community service). However, if the violation of MCL 257.602b(2) involves an accident for which the individual is at fault, the individual must be ordered to pay a civil fine of "double the amount that would be ordered under [MCL 257.602b(5)], as applicable." MCL 257.602b(6); MCL 257.907(2)(h)(*ii*)(B) (providing for civil fines of "\$400.00 for a first offense and \$1,000.00 for a second or subsequent offense" and making no reference to fault).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*I*). See Section 1.42 for more information on points.
- Two points are assessed for "a third or subsequent violation of [MCL 257.602b]. MCL 257.320a(1)(x). See Section 1.42 for more information on points.
- One point is assessed for "a second violation of [MCL 257.602b]." MCL 257.320a(1)(y). See Section 1.42 for more information on points.
- "Points shall not be entered for a violation of [MCL 257.602b(1)]." MCL 257.320a(2). See Section 1.42 for more information on points.

4.16 Turning and Signaling

A. Statutory Authority: Turning at Intersection

1. Right turn

"Both the approach for a right turn and a right turn shall be made as close as practicable to the right-hand curb or edge of the roadway." MCL 257.647(1)(a).

However, where local authorities have placed pavement markers, signs, or signals "direct[ing] that a different course from that specified in [MCL 257.647] be traveled by vehicles turning at an intersection[,]" the "driver of a vehicle shall not turn a vehicle at an intersection other than as directed and required by those markers, signs, or signals." MCL 257.647(1)(e).

2. Left turn

"Approach for a left turn shall be made in that portion of the right half of the roadway nearest the center line in a manner as not to interfere with the progress of any streetcar, and after entering the intersection the left turn shall be made so as to leave the intersection to the right of the center line of the roadway being entered." MCL 257.647(1)(b).

"Approach for a left turn from a 2-way roadway into a 1-way roadway shall be made in that portion of the right half of the roadway nearest the center line and clear of existing car tracks in use, and by passing to the right of the center line where it enters the intersection." MCL 257.647(1)(c).

"Approach for a left turn from a 1-way roadway into a 2-way roadway shall be made as close as practicable to the left curb or edge of the roadway and by passing to the right of the center line of the roadway being entered." MCL 257.647(1)(c).

"Where both streets or roadways are 1-way, both the approach for a left turn and a left turn shall be made as close as practicable to the left-hand curb or edge of the roadway." MCL 257.647(1)(d).

However, where local authorities have placed pavement markers, signs, or signals "direct[ing] that a different course from that specified in [MCL 257.647] be traveled by vehicles turning at an intersection[,]" the "driver of a vehicle shall not turn a vehicle at an intersection other than as directed and required by those markers, signs, or signals." MCL 257.647(1)(e).

B. Statutory Authority: Signals for Stopping or Turning

"The operator of a vehicle . . . upon a highway, before stopping or turning from a direct line, shall first determine that the stopping or turning can be made in safety and shall give a signal as required in [MCL 257.648]." MCL 257.648(1).

"Except as provided in [MCL 257.648(5)¹²], a signal required under [MCL 257.648] shall be given either by means of the hand and arm in the manner specified in [MCL 257.648], or by a mechanical or electrical signal device that conveys an intelligible signal or warning to other highway traffic." MCL 257.648(2).

"When a person is operating a vehicle and signal is given by means of the hand and arm, the operator shall signal as follows:

(a) For a left turn, the operator shall extend his or her left hand and arm horizontally.

(b) For a right turn, the operator shall extend his or her left hand and arm upward.

(c) To stop or decrease speed, the operator shall extend his or her left hand and arm downward." MCL 257.648(3).

C. Turning and Signaling Violations

Turning and signaling violations include:

- Failing to signal or improper signal, MCL 257.648;
- Improper or prohibited right or left turn, MCL 257.647 and MCL 257.648; and
- Improper turn from wrong lane, MCL 257.647.

D. Penalties

The general rules for assessing a civil fine and costs apply to turning and signal violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

¹²MCL 257.648(5) concerns commercial motor vehicles.

E. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

F. Issues

• Using turn signal on highway, MCL 257.648

"MCL 257.648 requires drivers to use a turn signal when changing lanes on a highway[.]" *People v Hrlic*, 277 Mich App 260, 266 (2007).

4.17 Using a Cellular Telephone While Operating a Motor Vehicle (Kelsey's law)¹³

A. Statutory Authority

"Except as provided in [MCL 257.602c], and in addition to the requirements of [MCL 257.602b], an individual issued a level 1 or level 2 graduated license under [MCL 257.310e¹⁴] shall not use a cellular telephone while operating a motor vehicle upon a highway or street." MCL 257.602c(1).

"[MCL 257.602c] supersedes all local ordinances regulating the use of a cellular telephone by an individual issued a level 1 or level 2 graduated license while operating a motor vehicle in motion on a highway or street, except that a unit of local government may adopt an ordinance or enforce an existing ordinance substantially corresponding to [MCL 257.602c]." MCL 257.602c(4).

¹³MCL 257.602c "may be known as 'Kelsey's law.'" MCL 257.602c(6).

¹⁴ See Section 3.4 for more information on graduated licenses.

B. Exceptions

"[MCL 257.602c(1)] does not apply to an individual who is using a cellular telephone to do any of the following:

(a) Report a traffic accident, medical emergency, or serious road hazard.

(b) Report a situation in which the individual believes the individual's personal safety is in jeopardy.

(c) Report or avert the perpetration or potential perpetration of a criminal act against the individual or another individual." MCL 257.602c(2).

C. Penalties

The general rules for assessing a civil fine and costs apply to using a cellular telephone while operating a motor vehicle. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- "Points shall not be entered for a violation of [MCL 257.602c]." MCL 257.320a(2). See Section 1.42 for more information on points.

4.18 Wrong Side or Wrong Way

A. Statutory Authority: "Keep to the Right" Rule

"Upon each roadway of sufficient width, the driver of a vehicle shall drive the vehicle upon the right half of the roadway, except as follows:

(a) When overtaking and passing another vehicle proceeding in the same direction under the rules governing that movement.

(b) When the right half of a roadway is closed to traffic while under construction or repair or when an obstruction exists making it necessary to drive to the left of the center of the highway. A driver who is driving on the left half of a roadway under [MCL 257.634(1)(b)] shall yield the right-of-way to an oncoming vehicle traveling in the proper direction upon the unobstructed portion of the roadway.

(c) When a vehicle operated by a state agency or local authority or an agent of a state agency or local authority is engaged in work on the roadway.

(d) Upon a roadway divided into 3 marked lanes for traffic under the rules applicable on the roadway." MCL 257.634(1).

"Upon a roadway having 2 or more lanes for travel in 1 direction, the driver of a vehicle shall drive the vehicle in the extreme righthand lane available for travel except as otherwise provided in [MCL 257.634]." MCL 257.634(2). "However, the driver of a vehicle may drive the vehicle in any lane lawfully available to traffic moving in the same direction of travel when the lanes are occupied by a streetcar or vehicles moving in substantially continuous lanes of traffic and in any left-hand lane lawfully available to traffic moving in the same direction of travel for a reasonable distance before making a left turn or to avoid blocking, delaying, or otherwise interfering with the movement of a streetcar on a streetcar track." *Id*.

"On a roadway that is divided into 4 or more lanes and provides for 2-way movement of traffic, a vehicle must be operated within the extreme right-hand lane except if overtaking and passing, and must not cross the center line of the roadway except if making a left turn." MCL 257.642(1)(b).

B. Exceptions to the "Keep to the Right" Rule

"While [a] defendant driver may have been on that portion of the road normally used by traffic in the opposite direction, he [or she] was justifiably there, [if he or she was] directed to use it because of construction and resurfacing operations." *Smith v Whitehead*, 342 Mich 542, 546 (1955). "Under such circumstances he was not violating the statutes or rules of the road." *Id*.

A driver may drive on the wrong side of the road, around parked cars, provided he or she exercises reasonable care in doing so. *Rosen v Beh*, 272 Mich 487, 492 (1935).

"[T]he evidence that [the] defendant, while recognizing that the highway was icy, hit a patch of ice that caused a skid was sufficient evidence of a sudden emergency that has been recognized by Michigan courts as an excuse for violation of the statutes requiring drivers to keep to the right." *Young v Flood*, 182 Mich App 538, 544 (1990).

A driver is not liable for driving on the wrong side of the highway if he or she faints or becomes unconscious immediately before an accident, such that the passing of his or her vehicle to the wrong side of the highway is not a voluntary act. *Soule v Grimshaw*, 266 Mich 117, 119 (1934). However, if a driver continues to drive while feeling dizzy or sleepy, he or she may be considered negligent. *Id*. at 120.

C. Statutory Authority: One-Way and Two-Way Traffic

"Upon a roadway designated and signposted for 1-way traffic a vehicle shall be driven only in the direction designated." MCL 257.641(2).

"Drivers of vehicles proceeding in opposite directions shall pass each other to the right, each giving to the other not less than 1/2 of the main traveled portion of the roadway as nearly as possible." MCL 257.635(1).

D. Wrong Side or Wrong Way Violations

Wrong side or wrong way violations include:

- Driving on the wrong side of divided highway, MCL 257.644;
- Driving on wrong side of undivided highway, MCL 257.642(1);

- Driving the wrong way on a one-way road, MCL 257.641;
- Entering freeway improperly, MCL 257.645;
- Failing to keep to the right, MCL 257.634; and
- Failing to keep to the right half of the roadway when passing vehicle going in opposite direction, MCL 257.635.

E. Penalties

The general rules for assessing a civil fine and costs apply to wrong side or wrong way violations. See Section 1.20 for a discussion of the general rules governing the assessment of a civil fine and costs; see also the Michigan Judicial Institute's table for a quick reference guide.

F. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points are assessed for "a moving violation resulting in an at-fault collision with another vehicle, an individual, or any other object." MCL 257.320a(1)(*l*). See Section 1.42 for more information on points.
- Two points are assessed for "[a]ll other moving violations pertaining to the operation of motor vehicle reports under [MCL 257.320a.]" MCL 257.320a(1)(u). See Section 1.42 for more information on points.

Chapter 5: Misdemeanor Nonmoving Violations

	5.1	Introduction and Scope Note 5-3				
Part A—Failing to Report or Leaving the Scene of an Accident						
	5.2	Failing to Give Information and Aid at the Scene of an Accident 5-3				
	5.3	Failing to Report Accident Involving Death, Personal Injury, or Property Damage of \$1,000 or More				
	5.4	Garage or Repair Shop Failing to Report Evidence of Accident or Bullet 5-6				
	5.5	Leaving the Scene of an Accident Resulting in Damage to Fixtures That Are Upon or Adjacent to a Highway				
	5.6	Leaving the Scene of or Failing to Report an Accident Resulting in Personal Injury				
	5.7	Leaving the Scene of or Failing to Report an Accident Resulting in Vehicle Damage Only				
	5.8	Leaving the Scene of or Failing to Report an Accident With an Attended or Unattended Vehicle				
	Part B—License and Permit Violations					
	5.9	Allowing Another to Operate in Violation of the Michigan Vehicle Code 5-10				
	5.10	Allowing Another Person to Operate a Motor Vehicle With a Suspended/ Revoked License				
	5.11	Possessing a Reproduced, Altered, Counterfeit, Forged, or Duplicate License				
	5.12	Reporting a False Address Change to the Secretary of State 5-13				
	5.13	Reproducing, Altering, Counterfeiting, Forging, or Duplicating a License, or Using Such License, With Intent to Commit a Misdemeanor 5-14				
	5.14	Unlawful Use or Display of License 5-14				
Part C—Title, Plate, Registration, and Insurance Violations						
	5.15	Allowing Another to Drive or Move an Unregistered Vehicle 5-16				

5.16	Failing to Apply for Registration and Certificate of Title	5-17
5.17	Failing to Transfer Title	5-18
5.18	Forging Proof of Insurance	5-19
5.19	Producing False Evidence of Motor Vehicle Insurance	5-19
5.20	Reproducing, Altering, Counterfeiting, Forging, or Duplicating Certific of Title, or Using Such Certificate of Title, With Intent to Commit a Cr	rime
5.21	Temporary Registration Violations	5-20
5.22	Unlawful Lending or Use of Title, Registration Certificate, Plate, or Pe	
Part	D—Other Nonmoving Violation Misdemeanors Found in the M	IVC
5.23	Failing to Answer Citation, Appear in Court, or Comply With an Order	r or
	Judgment	5-22
5.24	Failing to Disclose Odometer Mileage	5-23
5.25	Falsifying or Improperly Disposing of a Citation	5-24
5.26	Improper Use of Disabled Person Identification	5-24
5.27	Traffic Control Device Violations	5-26
5.28	Transporting or Possessing Alcohol in a Motor Vehicle by a Person Le Than 21 Years of Age	ess 5-27
5.29	Minor Purchasing, Consuming, or Possessing Alcohol, or Having Any Bodily Alcohol Content/Furnishing Fraudulent Identification to a Min or Minor Using Fraudulent Identification to Purchase Alcoholic Liquo	r
5.30	Transporting or Possessing Open Alcohol in a Vehicle	

5.1 Introduction and Scope Note

The principal source of Michigan traffic law is the Michigan Vehicle Code (MVC). The MVC clearly distinguishes the misdemeanor traffic offense from the felony and civil infraction. "It is a misdemeanor for a person to violate [the MVC], unless that violation is by [the MVC] or other law of this state declared to be a felony or a civil infraction." MCL 257.901(1).¹

Part A—Failing to Report or Leaving the Scene of an Accident

5.2 Failing to Give Information and Aid at the Scene of an Accident²

A. Statutory Authority

"The driver of a vehicle who knows or who has reason to believe that he or she has been involved in an accident with an individual or with another vehicle that is operated or attended by another individual shall do all of the following:

(a) Give his or her name and address, and the registration number of the vehicle he or she is operating, including the name and address of the owner, to a police officer, the individual struck, or the driver or occupants of the vehicle with which he or she has collided.

(b) Exhibit his or her operator's . . . license to a police officer, individual struck, or the driver or occupants of the vehicle with which he or she has collided.

(c) Render to any individual injured in the accident reasonable assistance in securing medical aid or arrange for or provide transportation to any injured individual." MCL 257.619.

¹A detailed table including traffic misdemeanors and related sanctions is available at http://www.michigan.gov/documents/OffenseCode_73877_7.pdf.

 $^{^2}$ This offense is often tied to the felony hit-and-run offense found in MCL 257.617. See Section 8.4 for more information on MCL 257.617.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Issues

• Purpose of MCL 257.619

MCL 257.619 "reflect[s] the legislative intent to curb hit-and-run accidents and encourage drivers involved in accidents to assume responsibility for identifying themselves and offering assistance, thus promoting public safety." *People v Sartor*, 235 Mich App 614, 623 (1999).

• Self-Incrimination

"MCL 257.619 requires the driver involved in an accident to give his name, address, registration number of the vehicle driven, name and address of the vehicle's owner, and to display his driver's license, and further requires him [or her] to render injured persons reasonable assistance in securing medical aid or transportation." *People v Goodin*, 257 Mich App 425, 429 (2003). "[T]he disclosure[] of one's name, address, vehicle-registration number, and driver's license required by . . . MCL 257.619 [is] neutral and do[es] not implicate a driver in criminal conduct." *Goodin*, 257 Mich App at 430. "Thus, the disclosure[] mandated under . . . MCL 257.619 do[es] not create a substantial risk of self-incrimination." *Goodin*, 257 Mich App at 431.

• Constitutionality of MCL 257.619

"MCL 257.619 is not unconstitutionally vague on its face" because "[t]he standard [for what constitutes reasonable assistance] clearly anticipates a determination of reasonableness under the circumstances of a particular case." *People v Noble*, 238 Mich App 647, 655 (1999).

5.3 Failing to Report Accident Involving Death, Personal Injury, or Property Damage of \$1,000 or More

A. Statutory Authority

"The driver of a motor vehicle involved in an accident that injures or kills any person, or that damages property to an apparent extent totaling \$1,000.00 or more, shall immediately report that accident at the nearest or most convenient police station, or to the nearest or most convenient police officer." MCL 257.622.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Issues

• Purpose of MCL 257.622

"The purpose of the accident-report statute [MCL 257.622] is to apprise the police that an accident has occurred and provide them with statistical information concerning the number and causes of accidents." *People v Schmidt*, 196 Mich App 104, 107 (1992).

Property Damage Threshold

To satisfy the \$1,000³ property damage threshold under MCL 257.622, "the prosecutor ha[s] the burden of introducing some evidence of value, either the opinion of a qualified expert or the testimony of a person with knowledge, under the applicable rules of evidence." *Schmidt*, 196 Mich App at 107.

³ The *Schmidt* case was decided when the threshold was \$200.

5.4 Garage or Repair Shop Failing to Report Evidence of Accident or Bullet

A. Statutory Authority

"The person in charge of any garage or repair shop to which is brought any motor vehicle which shows evidence of having been involved in an accident or having been struck by any bullet shall report the same to the nearest police station or sheriff's office immediately after such motor vehicle is received, giving the engine number, registration number and the name and address of the owner, and/or operator of such vehicle." MCL 257.623.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

5.5 Leaving the Scene of an Accident Resulting in Damage to Fixtures That Are Upon or Adjacent to a Highway

A. Statutory Authority

"The driver of any vehicle involved in an accident resulting only in damage to fixtures legally upon or adjacent to a highway shall take reasonable steps to locate and notify the owner or person in charge of such property of such accident and of his name and address and of the registration number of the vehicle he is driving and shall upon request exhibit his operator's . . . license and, if such owner cannot be found, shall forthwith report such accident to the nearest or most convenient police officer." MCL 257.621(a).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

5.6 Leaving the Scene of or Failing to Report an Accident Resulting in Personal Injury

A. Statutory Authority

"The driver of a vehicle who knows or who has reason to believe that he has been involved in an accident upon public or private property that is open to travel by the public shall immediately stop his or her vehicle at the scene of the accident and shall remain there until the requirements of [MCL 257.619] are fulfilled or immediately report the accident to the nearest or most convenient police agency or officer to fulfill the requirements of [MCL 257.619(a) and MCL 257.619(b)] if there is a reasonable and honest belief that remaining at the scene will result in further harm. The stop shall be made without obstructing traffic more than is necessary." MCL 257.617a(1).

B. Relevant Jury Instruction

• M Crim JI 15.14 addresses the elements of leaving the scene of an accident causing personal injury.

C. Penalties

For an accident that results in injury to any individual,⁴ MCL 257.617a(2) provides for:

- imprisonment for not more than one year; or
- fine of not more than \$1,000; or
- both.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Six points. See MCL 257.320a(1)(d); MCL 257.732(1)(a). See Section 1.42 for more information on points.

⁴ If the accident does not result in injury to any individual, MCL 257.901(2) applies.

• License suspension for 90 days. See MCL 257.319(3)(a). See Section 1.45 for more information on license suspension.

5.7 Leaving the Scene of or Failing to Report an Accident Resulting in Vehicle Damage Only

A. Statutory Authority

"The driver of a vehicle who knows or who has reason to believe that he has been involved in an accident upon public or private property that is open to travel by the public shall immediately stop his or her vehicle at the scene of the accident and shall remain there until the requirements of [MCL 257.619] are fulfilled or immediately report the accident to the nearest or most convenient police agency or officer to fulfill the requirements of [MCL 257.619(a) and MCL 257.619(b)] if there is a reasonable and honest belief that remaining at the scene will result in further harm. The stop shall be made without obstructing more traffic than is necessary." MCL 257.618.

B. Relevant Jury Instruction

• M Crim JI 15.14 addresses the elements of leaving the scene of an accident causing damage to a vehicle.

C. Penalties

For an accident that results in "damage to a vehicle operated by or attended by any individual,"⁵ MCL 257.618(2) provides for the following penalties:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and

⁵ If the accident does not result in damage to a vehicle, MCL 257.901(2) applies.

applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Six points. See MCL 257.320a(1)(d); MCL 257.732(1)(a). See Section 1.42 for more information on points.

5.8 Leaving the Scene of or Failing to Report an Accident With an Attended or Unattended Vehicle

A. Statutory Authority

"The driver of any vehicle which collides upon either public or private property with any vehicle which is attended or unattended shall immediately stop and shall then and there either locate and notify the operator or owner of such vehicle of the name and address of the driver and owner of the vehicle striking the vehicle or, if such owner cannot be located, shall forthwith report it to the nearest or most convenient police officer." MCL 257.620.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Six points. See MCL 257.320a(1)(d); MCL 257.732(1)(a). See Section 1.42 for more information on points.

Part B—License and Permit Violations

5.9 Allowing Another to Operate in Violation of the Michigan Vehicle Code

A. Statutory Authority

"No person shall knowingly authorize or permit a motor vehicle owned by him or under his control to be driven by any person in violation of the provisions of [the MVC]." MCL 257.326.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Issues

Allowing a person without a driver's license to "take possession and control" of a motor vehicle does not violate MCL 257.326; however, granting a person without a driver's license permission to drive a motor vehicle implicates MCL 257.326 and may subject the person granting permission to prosecution under MCL 257.326. *Monaco v Home-Owners Ins Co*, 317 Mich App 738, 750 (2016) (discussing the authorization to drive a vehicle in relation to unlawful taking for no-fault insurance purposes).

5.10 Allowing Another Person to Operate a Motor Vehicle With a Suspended/Revoked License⁶

A. Statutory Authority

"A person shall not knowingly permit a motor vehicle owned by the person to be operated on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within [Michigan] by an individual whose license or registration certificate is suspended or revoked, whose application for license has been denied, or who has never applied for a license, except as permitted under [the MVC]." MCL 257.904(2).

⁶The felony offenses contemplated by MCL 257.904 are discussed in Chapter 7.

B. Relevant Jury Instruction

• M Crim JI 15.23 addresses the elements of permitting another person to drive a motor vehicle with a suspended or revoked license.

C. Penalties

MCL 257.904(3) provides for the following penalties:

- First Violation:
 - imprisonment for not more than 93 days; or
 - fine of not more than \$500; or
 - both.

• Violation Occurring After Prior Conviction:

- imprisonment for not more than one year; or
- fine of not more than \$1,000; or
- both.

"A prior conviction under [MCL 257.904] must be established at or before sentencing by 1 or more of the following:

- (a) A copy of a judgment of conviction.
- (b) An abstract of conviction.
- (c) A transcript of a prior trial, plea, or sentencing.
- (d) A copy of a court register of action.
- (e) A copy of the defendant's driving record.
- (f) Information contained in a presentence report.
- (g) An admission by the defendant." MCL 257.904(9).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

- Vehicle immobilization authorized. See MCL 257.904d(4)(b). See Section 1.47 for more information on vehicle immobilization.
- Registration plate cancellation required.⁷ See MCL 257.904(3).

E. Issues

• Scope of MCL 257.904.

MCL 257.904 "does not apply to an individual who operates a vehicle solely for the purpose of protecting human life or property if the life or property is endangered and summoning prompt aid is essential." MCL 257.904(15).

"For purposes of [MCL 257.904], an individual who never applied for a license includes an individual who applied for a license, was denied, and never applied again." MCL 257.904(19).

5.11 Possessing a Reproduced, Altered, Counterfeit, Forged, or Duplicate License

A. Statutory Authority

"[A] person who is in possession of a reproduced, altered, counterfeited, forged, or duplicated license photograph, negative of the photograph, image, license, or electronic data contained on a license or part of a license is guilty of a misdemeanor[.]" MCL 257.310(10).⁸

B. Penalties

MCL 257.310(10) provides the following penalties:

- imprisonment for not more than one year; or
- fine of not more than \$2,000; or

⁷ Upon a first violation of MCL 257.904(1) or MCL 257.904(2), "[u]nless the vehicle was stolen or used with the permission of a person who did not knowingly permit an unlicensed driver to operate the vehicle, the registration plates of the vehicle must be canceled by the secretary of state on notification by a peace officer." MCL 257.904(3)(a). For any violation of MCL 257.904(1) or MCL 257.904(2) that occurs after a prior conviction, the registration plates of the vehicle must be canceled unless the vehicle was stolen. MCL 257.904(3)(b).

 $^{^{8}}$ [MCL 257.310(10)] do[es] not apply to a person who is in possession of 1 or more photocopies, reproductions, or duplications of a license to document the identity of the licensee for a legitimate business purpose." MCL 257.310(16).

both.

5.12 Reporting a False Address Change to the Secretary of State

A. Statutory Authority

"A person shall not knowingly report a change of address to the secretary of state for himself or herself that is not his or her residence address." MCL 257.315(4).

"A person shall not knowingly report a change of address to the secretary of state for another person without the consent of the other person." MCL 257.315(4).

B. Penalties

MCL 257.315 provides for the following penalties:

- First Offense:
 - imprisonment for not more than 93 days; or
 - fine of \$1,000; or
 - both. MCL 257.315(4).
- Second or Subsequent Offense:
 - imprisonment for not more than 93 days; or
 - fine of \$5,000; or
 - both. MCL 257.315(5).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• License suspension for six months authorized for first offense; license revocation required for second or subsequent offense. See MCL 257.315(4); MCL

257.315(5). See Section 1.45 for more information on license suspension and see Section 1.44 for more information on license revocation.

5.13 Reproducing, Altering, Counterfeiting, Forging, or Duplicating a License, or Using Such License, With Intent to Commit a Misdemeanor

A. Statutory Authority

"[A] person who intentionally reproduces, alters, counterfeits, forges, or duplicates a license photograph, the negative of the photograph, image, license, or electronic data contained on a license or a part of a license or who uses a license, image, or photograph that has been reproduced, altered, counterfeited, forged, or duplicated" with the intent "to commit or aid in the commission of an offense that is a misdemeanor punishable by imprisonment for less than 6 months, the person committing the reproduction, alteration, counterfeiting, forging, duplication, or use is guilty of a misdemeanor[.]" MCL 257.310(7)(c).

B. Penalties

MCL 257.310(7)(c) provides the following penalties:

- imprisonment for not more than one year; or
- fine of not more than \$2,000; or
- both.

5.14 Unlawful Use or Display of License

A. Statutory Authority

"A person shall not do any of the following:

(a) Display, or cause or permit to be displayed, or have in possession an operator's...license knowing the operator's...license to be fictitious or to have been canceled, revoked, suspended, or altered.

(b) Lend to or knowingly permit use of, by one not entitled to its use, the operator's \ldots license issued to the person lending or permitting the use of the operator's \ldots license.

(c) Display or to represent as one's own any operator's...license not issued to the person displaying the operator's...license.

(d) Fail or refuse to surrender to the department upon demand, any operator's...license which has been suspended, canceled, or revoked as provided by law.

(e) Use a false or fictitious name or give a false or fictitious address in an application for an operator's ... license, or any renewal or duplicate of an operator's ... license, or knowingly make a false statement or knowingly conceal a material fact or otherwise commit a fraud in making an application.

(f) Alter or otherwise cause to be altered any operator's...license so as to knowingly make a false statement or knowingly conceal a material fact in order to misrepresent as one's own the operator's...license.

(g) Use or have in possession in committing a crime an operator's... license that has been altered or that is used to knowingly make a false statement or to knowingly conceal a material fact in order to misrepresent as one's own the operator's... license.

(h) Furnish to a peace officer false, forged, fictitious, or misleading verbal or written information identifying the person as another person, if the person is detained for a violation of [the MVC] or of a local ordinance substantially corresponding to a provision of [the MVC] \dots ^{"9} MCL 257.324(1).

"An operator's . . . license issued to a person under [MCL 257.301 *et seq.*] upon an application that is untrue, or that contains false statements as to any material matters, or that was obtained by fraud in the testing for or issuance of the license, is void from the date of issuance." MCL 257.324(2). "The operator . . . who was issued the license is considered unlicensed and the license issued shall be returned upon request or order of the department." *Id*.

B. Penalties

MCL 257.901(2) provides for:

• imprisonment for not more than 90 days; or

 $^{^{9}}$ MCL 257.324(1)(i)-(j) were omitted from this quotation because they contain prohibitions outside the scope of this benchbook.

- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• License denial required for unlicensed violation; license suspension for 90 days required for first offense; license suspension for one year required for offense committed with prior conviction within seven years. See MCL 257.303(1)(i); MCL 257.319(5). See Section 1.43 for more information on license denial and Section 1.45 for more information on license suspension.

Part C—Title, Plate, Registration, and Insurance Violations

5.15 Allowing Another to Drive or Move an Unregistered Vehicle

A. Statutory Authority

"Every motor vehicle ... when driven or moved on a street or highway, is subject to the registration and certificate of title provisions of [the MVC] except [as provided in MCL 257.216(1)(a)-(o)[.]" MCL 257.216(1). "Except as otherwise provided in [MCL 257.216(2)], a registration issued to a motor vehicle ... when driven or moved on a street or highway, that expires on or after March 1, 2020 is valid until March 31, 2021.... If the registration issued to a motor vehicle ... when driven or moved on a street or highway, that is used for commercial purposes expires on or after March 1, 2020, the registration is valid until March 31, 2021." MCL 257.216(2). "Except as otherwise provided in [MCL 257.216(2)], a registration issued to a motor vehicle ... when driven or moved on a street or highway, that expires after March 31, 2021 but before August 1, 2021, is valid until 120 days after the date of the expiration. ... If the registration issued to a motor vehicle . . . when driven or moved on a street or highway, that is used for commercial purposes expires after March 31, 2021 but before August 1, 2021, the registration is valid until 120 days after the date of the expiration." *Id*.

"It is a misdemeanor for . . . an owner knowingly to permit to be driven or moved upon any highway any vehicle of a type required to be registered hereunder which is not registered or for which a certificate of title has not been applied for or for which the appropriate fee has not been paid when and as required hereunder[.]" MCL 257.215.

B. Penalties

MCL 257.901(2) provides the following penalties:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

5.16 Failing to Apply for Registration and Certificate of Title

A. Statutory Authority

"An owner of a vehicle that is subject to registration under [the MVC] shall apply to the secretary of state, on an appropriate form furnished by the secretary of state, for the registration of the vehicle and issuance of a certificate of title for the vehicle." MCL 257.217(1). "The application must be accompanied by the required fee." *Id.* "An application for a certificate of title must bear the signature or verification and certification of the owner." *Id.*

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• License suspension for 90 days required for first offense; license suspension for one year required for second offense within seven years. See MCL 257.319(5). See Section 1.45 for more information on license suspension.

5.17 Failing to Transfer Title

A. Statutory Authority

"The purchaser or transferee, unless the person is a licensed dealer, shall present or cause to be presented the certificate of title and registration certificate if plates are being transferred to another vehicle, assigned as provided in [the Michigan Vehicle Code], to the secretary of state accompanied by the fees as provided by law, whereupon a new certificate of title and registration certificate shall be issued to the assignee." MCL 257.234(1). "The certificate of title shall be mailed or delivered to the owner or another person the owner may direct in a separate instrument in a form the secretary of state shall prescribe." *Id*.

"Unless the transfer is made and the fee paid within 15 days, the vehicle is considered to be without registration, the secretary of state may repossess the license plates, and transfer of the vehicle ownership may be effected and a valid registration acquired thereafter only upon payment of a transfer fee of \$15.00 in addition to the fee provided for in [MCL 257.806¹⁰]." MCL 257.234(3).

B. Penalties

MCL 257.901(2) provides for:

¹⁰"Until October 1, 2027, an applicant for a certificate of title required by [the Michigan Vehicle Code] or an applicant for a duplicate of a certificate of title shall accompany the application with a fee of \$10.00." MCL 257.806(1). "The secretary of state shall collect a \$3.00 service fee . . . for each title issued[.]" *Id.* "In addition to paying the fees required by [MCL 257.806(1)], until December 31, 2027, each person that applies for a certificate of title . . . under [the Michigan Vehicle Code] shall pay a tire disposal surcharge of \$1.50 for each certificate of title or duplicate of a certificate of title that person receives." MCL 257.806(3).

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

5.18 Forging Proof of Insurance

A. Statutory Authority

"Any person who forges or, or without authority, signs any evidence of ability to respond in damages as required by the secretary of state under [MCL 257.501 *et seq.*] . . . is guilty of a misdemeanor[.]" MCL 257.905(1).

B. Penalties

MCL 257.905(1) provides for:

- fine of not less than \$100 or more than \$1,000; or
- imprisonment for not more than 90 days; or
- both.

5.19 Producing False Evidence of Motor Vehicle Insurance

A. Statutory Authority

"An owner or operator of a motor vehicle who knowingly produces false evidence under [MCL 257.328] is guilty of a misdemeanor[.]" MCL 257.328(6).

B. Penalties

MCL 257.328(6) provides for:

- imprisonment for not more than one year; or
- fine of not more than \$1,000; or
- both.

5.20 Reproducing, Altering, Counterfeiting, Forging, or Duplicating Certificate of Title, or Using Such Certificate of Title, With Intent to Commit a Crime

A. Statutory Authority

"A person that intentionally reproduces, alters, counterfeits, forges, or duplicates a certificate of title or a document releasing a security interest or that uses a reproduced, altered, counterfeited, forged, or duplicated certificate of title or document releasing a security interest must be punished [as set out in MCL 257.222(6).]" MCL 257.222(6).

B. Penalties

MCL 257.222(6) provides the following penalties, which depend upon the length of imprisonment for the intended crime:

- If the intent of the act was to commit or aid in committing a crime punishable by imprisonment for one year or more:
 - imprisonment for a period equal to that which could be imposed for the commission of the crime; and
 - fine of not more than \$10,000.
- If the intent of the act was to commit or aid in committing a crime punishable by imprisonment for not more than one year:
 - imprisonment for not more than one year; or
 - fine of not more than \$1,000; or
 - both.

5.21 Temporary Registration Violations

A. Statutory Authority

"A temporary registration may be issued to an owner of a vehicle." MCL 257.226b(1). "The registration shall be valid for either 30 days or 60 days from the date of issue, at the discretion of the owner, and shall be in a form as determined by the secretary of state." *Id*.

"A vehicle which has a temporary registration shall not be used for the transportation of passengers for hire or for the transportation of goods, wares, or merchandise or draw other vehicles transporting goods, wares, or merchandise." MCL 257.226b(2).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

5.22 Unlawful Lending or Use of Title, Registration Certificate, Plate, or Permit

A. Statutory Authority

"A person shall not lend to another person, or knowingly permit the use of, any certificate of title, registration certificate, registration plate, special plate, or permit issued to him or her if the person receiving or using the certificate of title, registration certificate, registration plate, special plate, or permit would not be entitled to the use thereof." MCL 257.256(1).

"A person shall not carry or display upon a vehicle any registration certificate or registration plate not issued for the vehicle or not otherwise lawfully used under [the MVC]." MCL 257.256(1).

B. Penalties

MCL 257.256(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

Part D—Other Nonmoving Violation Misdemeanors Found in the MVC

5.23 Failing to Answer Citation, Appear in Court, or Comply With an Order or Judgment

A. Statutory Authority

"Except as provided [MCL 257.321a(2)], 28 days or more after an individual fails to answer a citation, or a notice to appear in court for a violation for which license suspension is allowed under [the MVC] or a local ordinance that substantially corresponds to a violation for which license suspension is allowed under [the MVC], or fails to comply with an order or judgment of the court for a violation for which license suspension is allowed under [the MVC], including, but not limited to, paying all fines, costs, fees, and assessments, the court shall give notice by mail at the last known address of the individual that if the individual fails to appear or fails to comply with the order or judgment within 14 days after the notice is issued, the secretary of state shall suspend the individual's operator's...license. If the individual fails to appear or fails to comply with the order or judgment within the 14-day period, the court shall, within 14 days, inform the secretary of state, who shall immediately suspend the license of the individual. The secretary of state shall immediately notify the individual of the suspension by regular mail at the individual's last known address." MCL 257.321a(1).

"If an individual is charged with, or convicted of, a violation of [MCL 257.625, MCL 257.626], any driving violation under [the MVC] that causes injury, death, or serious impairment of a body function of another individual, a serious offense involving a motor vehicle, or a local ordinance that substantially corresponds to MCL 257.625(1)-(3), MCL 257.625(6), MCL 257.625(8) or MCL 257.626], and the individual fails to answer a citation or a notice to appear in court, or a notice to appear for any matter pending, or fails to comply with an order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, the court shall immediately give notice by first-class mail sent to the individual's last known address that if the individual fails to appear within 7 days after the notice is issued, or fails to comply with the order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, within 14 days after the notice is issued, the secretary of state shall suspend the individual's operator's or chauffeur's license. If the individual fails to appear within the 7-day period, or fails to comply with the order or judgment of the court, including, but not limited to, paying all fines, costs, and crime victim rights assessments, within the 14-day period, the court shall immediately inform the secretary of state who shall immediately suspend the individual's operator's or chauffeur's license and notify

the individual of the suspension by first-class mail sent to the individual's last known address." MCL 257.321a(2).

B. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• License suspension required under certain circumstances. See MCL 257.321a–MCL 257.321b. See Section 1.45 for more information on license suspension.

5.24 Failing to Disclose Odometer Mileage¹¹

A. Statutory Authority

"Except as otherwise provided in [MCL 257.233a(17)], if the owner of a registered motor vehicle transfers the owner's title or interest in that vehicle, the transferor shall present to the transferee before delivery of the vehicle, written disclosure of odometer mileage by means of the certificate of title or a written statement signed by the transferor" MCL 257.233a(1).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Issues

• Voidable Transaction

¹¹Odometer tampering is a separate felony offense. See MCL 257.233a(6)-(7). For more information on that crime, see Section 7.7.

"[F]ailure to comply with the odometer statute requirements merely renders the transaction voidable by the purchaser." *Whitcraft v Wolfe*, 148 Mich App 40, 54 (1985).

• Civil Liability

"A person who, with intent to defraud, violates any requirement under [MCL 257.233a(1)] . . . is liable in an amount equal to 3 times the amount of actual damages sustained or \$1,500.00 whichever is greater, and in the case of a successful recovery of damages, the costs of the action together with reasonable attorney fees." MCL 257.233a(15).

5.25 Falsifying or Improperly Disposing of a Citation

A. Statutory Authority

It is a misdemeanor to "knowingly falsif[y] a citation or copies thereof or a record of the issuance of same, or dispose[] of such citation, copy or record, in a manner other than as required in [the Michigan Vehicle Code], or attempt[] so to falsify or dispose, or attempt[] to incite or procure another so to falsify or dispose[.]" MCL 257.728d.

B. Penalties

MCL 257.728d provides for:

- imprisonment not to exceed one year; or
- fine of not more than \$500; or
- both.

5.26 Improper Use of Disabled Person Identification

A. Statutory Authority

 "A person who intentionally makes a false statement of material fact or commits or attempts to commit a deception or fraud on a medical statement attesting to a disability, submitted in support of an application for a windshield placard, free parking sticker, special registration plate, or tab for persons with disabilities under [MCL 257.675], [MCL 257.803d], or [MCL 257.803f], is guilty of a misdemeanor[.]" MCL 257.675(14). • "A person who commits or attempts to commit a deception or fraud by 1 or more of the following methods is guilty of a misdemeanor[:]

(a) Using a windshield placard or free parking sticker issued under [MCL 257.675] or by another state to provide transportation to a disabled person, if the person is not providing transportation to a disabled person.

(b) Altering, modifying, or selling a windshield placard or free parking sticker issued under [MCL 257.675] or by another state.

(c) Copying or forging a windshield placard or free parking sticker described in [MCL 257.675] or selling a copied or forged placard or sticker described in [MCL 257.675]....

(d) Using a copied or forged windshield placard or free parking sticker described in [MCL 257.675].

(e) Making a false statement of material fact to obtain or assist an individual in obtaining a placard or sticker described in [MCL 257.675], a special registration plate under [MCL 257.803d], or a tab for persons with disabilities under [MCL 257.803f].

(f) Knowingly using or displaying a placard or sticker described in [MCL 257.675] that has been canceled by the secretary of state." MCL 257.675(15).

B. Penalties

MCL 257.675(14) and MCL 257.675(15) provide for:

- imprisonment for not more than 30 days; or
- fine of not more than \$500; or
- both.

However, "[i]n the case of a violation of [MCL 257.675(15)(c)], the fine described in [MCL 257.675(15)] shall be not less than \$250.00." MCL 257.675(15)(c).

C. Issues

• Confiscating Windshield Placard/Parking Sticker by Court

"Upon conviction of an offense involving a violation of the special privileges conferred upon a holder of a windshield placard or free parking sticker, a magistrate or judge trying the case, as a part of any penalty imposed, may confiscate the windshield placard or free parking sticker and return the confiscated item or items to the secretary of state together with a certified copy of the sentence imposed." MCL 257.675(13).

"Upon receipt of a windshield placard or free parking sticker from a judge or magistrate, the secretary of state shall cancel and destroy the placard or sticker, and the disabled person to whom it was issued shall not receive another placard or sticker until he or she submits a completed application and presents a current medical statement attesting to his or her condition." MCL 257.675(13).

• Confiscating Windshield Placard/Parking Sticker by Law Enforcement Officer

"A law enforcement officer who observes a misuse of a windshield placard or free parking sticker may immediately confiscate the placard or sticker and forward it with a copy of his or her report to the secretary of state." MCL 257.675(13).

5.27 Traffic Control Device Violations

A. Statutory Authority

"A person who willfully and maliciously damages, destroys, injures, defaces, dismantles, tampers with, or removes a traffic control device is guilty of a crime[.]" MCL 750.377d(1). Violation of MCL 750.377d is a misdemeanor. MCL 750.377d(1)(a)-(c).

MCL 750.377d "does not prohibit the person from being charged with, convicted of, or sentenced for any other violation of law arising out of the same transaction as the violation of [MCL 750.377d] in addition to being charged with, convicted of, or sentenced for the violation of [MCL 750.377d]." MCL 750.377d(2).

B. Penalties

MCL 750.377d provides for:

• First Violation:

- imprisonment for not more than 93 days; or
- fine of not more than \$500; or

- both. MCL 750.377d(1)(a).
- Second Violation:
 - imprisonment for not more than 180 days; or
 - fine of not more than \$1,000; or
 - both. MCL 750.377d(1)(b).
- Third of Subsequent Violation:
 - imprisonment for not more than one year; or
 - fine of not more than \$10,000; or
 - both. MCL 750.377d(1)(c).

5.28 Transporting or Possessing Alcohol in a Motor Vehicle by a Person Less Than 21 Years of Age

A. Statutory Authority

"An individual less than 21 years of age shall not knowingly transport or possess alcoholic liquor in a motor vehicle as an operator or occupant unless the individual is employed by a licensee under [MCL 436.1101 to MCL 436.2303], a common carrier designated by the liquor control commission under [MCL 436.1101 to MCL 436.2303], the liquor control commission, or an agent of the liquor control commission and is transporting or having the alcoholic liquor in a motor vehicle under the individual's control during regular working hours and in the course of the individual's employment." MCL 257.624b(1).

"[MCL 257.624b] does not prevent an individual less than 21 years of age from knowingly transporting alcoholic liquor in a motor vehicle if an individual at least 21 years of age is present inside the motor vehicle." MCL 257.624b(1).¹²

B. Penalties

MCL 257.901(2) provides for:

¹²MCL 436.1703(1) provides that it is a state civil infraction or a misdemeanor, depending on whether there is a prior judgment, for a minor to "purchase or attempt to purchase alcoholic liquor, consume or attempt to consume alcoholic liquor, possess or attempt to possess alcoholic liquor, or have any bodily alcohol content, except as provided in [MCL 436.1703]." For a discussion of this offense, see Section 5.29(A).

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

"As part of the sentence, the individual may be ordered to perform community service and undergo substance abuse screening and assessment at his or her own expense as described in [MCL 436.1703(1)]." MCL 257.624b(1).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Two points. See MCL 257.320a(1)(s); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- License denial mandatory for unlicensed violation; license suspension for 90 days required for prior conviction; license suspension for one year required for two or more prior convictions. See MCL 257.303(1)(j); MCL 257.319(7). A restricted license may be issued. See MCL 257.319(7). See Section 1.45 for more information on license suspension and Section 1.43 for more information on license denial.
- Vehicle impoundment authorized. See MCL 257.624b(2)-(3). See Section 9.12(B) for more information on impoundment.

D. Issues

Related Civil Infraction. See Section 3.16 for information on the related civil infraction found in MCL 257.625b(4).

5.29 Minor Purchasing, Consuming, or Possessing Alcohol, or Having Any Bodily Alcohol Content/ Furnishing Fraudulent Identification to a Minor, or

Minor Using Fraudulent Identification to Purchase Alcoholic Liquor

A. Statutory Authority

"A minor shall not purchase or attempt to purchase alcoholic liquor, consume or attempt to consume alcoholic liquor, possess or attempt to possess alcoholic liquor, or have any bodily alcohol content, except as provided in [MCL 436.1703]." MCL 436.1703(1).¹³

"An individual who furnishes fraudulent identification to a minor or, notwithstanding [MCL 436.1703(1)], a minor who uses fraudulent identification to purchase alcoholic liquor, is guilty of a misdemeanor[.]" MCL 436.1703(2).

B. Penalties

A minor who violates MCL 436.1703(1) is responsible for a state civil infraction or guilty of a misdemeanor as follows:

- First Violation:
 - "[T]he minor is responsible for a state civil infraction and shall be fined not more than \$100.00." MCL 436.1703(1)(a).
 - Additionally, the "court may order a minor under [MCL 436.1703(1)(a)] to participate in substance use disorder services as defined in . . . MCL 333.6230, and designated by the administrator of the office of substance abuse services, and may order the minor to perform community service and to undergo substance abuse screening and assessment at his or her own expense as described in [MCL 436.1703(5)]." MCL 436.1703(1)(a).
 - "A minor may be found responsible or admit responsibility only once under [MCL 436.1703(1)(a)]." MCL 436.1703(1)(a).
- Violation After One Prior Judgment:
 - The minor is guilty of a misdemeanor punishable by:

¹³ A juvenile who pleads guilty to or offers a plea of admission for violating MCL 436.1703(1) and who has not previously been convicted of or received a juvenile adjudication for violating MCL 436.1703(1), may have proceedings against him or her deferred. See MCL 436.1703(3). See Section 5.29(C) for more information on deferring proceedings under MCL 436.1703(3).

- imprisonment for not more than 30 days if the court finds that the minor violated an order of probation, failed to successfully complete any treatment, screening, or community service ordered by the court or failed to pay any fine for that conviction or juvenile adjudication; or
- by a fine of not more than \$200; or
- both. MCL 436.1703(1)(b).
- Additionally, the "court may order a minor under [MCL 436.1703(1)(a)] to participate in substance use disorder services as defined in . . . MCL 333.6230, and designated by the administrator of the office of substance abuse services, and may order the minor to perform community service and to undergo substance abuse screening and assessment at his or her own expense as described in [MCL 436.1703(5)]." MCL 436.1703(1)(b).

• Violation After Two or More Prior Judgments:

- The minor is guilty of a misdemeanor punishable by:
 - imprisonment for not more than 60 days, if the court finds that the minor violated an order of probation, failed to successfully complete any treatment, screening, or community service ordered by the court, or failed to pay any fine for that conviction or juvenile adjudication; or
 - by a fine of not more than \$500; or
 - both. MCL 436.1703(1)(c).
- Additionally, the "court may order a minor under [MCL 436.1703(1)(a)] to participate in substance use disorder services as defined in . . . MCL 333.6230, and designated by the administrator of the office of substance abuse services, and may order the minor to perform community service and to undergo substance abuse screening and assessment at his or her own expense as described in [MCL 436.1703(5)]." MCL 436.1703(1)(c).

A violation of MCL 436.1703(2) is a misdemeanor punishable by:

- imprisonment for not more than 93 days; or
- fine of not more than \$100; or

• both. MCL 436.1703(2).

C. Issues

• Affirmative Defense

"In a prosecution for the violation of [MCL 436.1703(1)] concerning a minor having any bodily alcohol content, it is an affirmative defense that the minor consumed the alcoholic liquor in a venue or location where that consumption is legal." MCL 436.1703(16).

• Custody

An officer or an inspector of the liquor control commission who witnesses a violation of MCL 436.1703 "may stop and detain a person and obtain satisfactory identification, seize illegally possessed alcoholic liquor, and issue an appearance ticket under...MCL 764.9c."¹⁴ MCL 436.1705.

• Deferred Proceedings

MCL 436.1703(3) allows the court to defer proceedings and impose probation, without a judgment of guilt or adjudication of responsibility, for individuals who consent to the deferral and plead guilty to a misdemeanor violation of MCL 436.1703(1)(b) (criminal proceedings), or offer a plea of admission (delinquency proceedings) for a misdemeanor violation of MCL 436.1703(1)(b). "If an individual fulfills the terms and conditions of probation, the court shall discharge the individual and dismiss the proceedings." MCL 436.1703(3). "An individual may obtain only 1 discharge and dismissal under [MCL 436.1703(3)]." *Id.* MCL 436.1703(3) further provides:

"The court shall maintain a nonpublic record of the matter while proceedings are deferred and the individual is on probation and if there is a discharge and dismissal under [MCL 436.1703(3)]. The secretary of state shall retain a nonpublic record of a plea and of the discharge and dismissal under [MCL 436.1703(3)]. These records shall be furnished to any of the following:

(a) To a court, prosecutor, or police agency on request for the purpose of determining if an individual has already utilized [MCL 436.1703(3)].

(b) To the department of corrections, a prosecutor, or a law enforcement agency, on the department's,

¹⁴ See Section 2.9(A) for a discussion of appearance tickets.

a prosecutor's, or a law enforcement agency's request, subject to all of the following conditions:

(*i*) At the time of the request, the individual is an employee of the department of corrections, the prosecutor, or the law enforcement agency, or an applicant for employment with the department of corrections, the prosecutor, or the law enforcement agency.

(*ii*) The record is used by the department of corrections, the prosecutor, or the law enforcement agency only to determine whether an employee has violated his or her conditions of employment or whether an applicant meets criteria for employment."

However, even in cases of deferral, MCL 257.732(5) requires that "[t]he clerk of the court shall . . . forward an abstract of the court record to the secretary of state if an individual has pled guilty to, or offered a plea of admission in a juvenile proceeding for, a violation of . . . MCL 436.1703, or a local ordinance substantially corresponding to [MCL 436.1703], and has had further proceedings deferred under that section. If the individual is sentenced to a term of probation and terms and conditions of probation are fulfilled and the court discharges the individual and dismisses the proceedings, the court shall also report the dismissal to the secretary of state."

• Preliminary Chemical Breath Analysis

"A peace officer who has reasonable cause to believe a minor has consumed alcoholic liquor or has any bodily alcohol content may request that individual to submit to a preliminary chemical breath analysis." MCL 436.1703(6). "If a minor does not consent to a preliminary chemical breath analysis, the analysis shall not be administered without a court order, but a peace officer may seek to obtain a court order." *Id.* "The results of a preliminary chemical breath analysis or other acceptable blood alcohol test are admissible in a state civil infraction proceeding or criminal prosecution to determine if the minor has consumed or possessed alcoholic liquor or had any bodily alcohol content." *Id.*

• Required Notice When Offender is a Juvenile

If the person who allegedly violated MCL 436.1703(1) is less than 18 years old and unemancipated, the law enforcement agency must notify the minor's parent(s), guardian, or custodian if the name of the parent, guardian, or custodian is reasonably ascertainable. MCL 436.1703(7). This notice must be given within 48 hours after the

agency determines that the person is less than 18 years old and umenacipated, and may be by any means reasonably calculated to give prompt actual notice of the offense, including notice in person, by telephone, or by first-class mail. *Id.* If the minor is less than 17 years old and incarcerated for a violation of MCL 436.1703(1), the minor's parents or legal guardian must be notified immediately. MCL 436.1703(7).

• Statutory Exemptions

MCL 436.1703 includes several provisions setting forth individuals who are not considered to be in violation of MCL 436.1703(1):

- "A minor who has consumed alcoholic liquor and who voluntarily presents himself or herself to a health facility or agency^[15] for treatment or for observation including, but not limited to, medical examination and treatment for any condition arising from a violation of sections . . . MCL 750.520b to [MCL 750.520g], committed against a minor." MCL 436.1703(9)(a).
- "A minor who accompanies an individual who . . . [h]as consumed alcoholic liquor[and who] [v]oluntarily presents himself or herself to a health facility or agency^[16] for treatment or for observation including, but not limited to, medical examination and treatment for any condition arising from a violation of . . . MCL 750.520b to [MCL 750.520g], committed against a minor." MCL 436.1703(9)(b).
- "A minor who initiates contact with a peace officer or emergency medical services personnel for the purpose of obtaining medical assistance for a legitimate health care concern." MCL 436.1703(9)(c).
- "The consumption of alcoholic liquor by a minor who is enrolled in a course offered by an accredited postsecondary educational institution in an academic building of the institution under the supervision of a faculty member is not prohibited by this act if the purpose of the consumption is solely educational and is a requirement of the course." MCL 436.1703(12).

¹⁵ If the minor is under 18 years old and unemancipated, the health facility or agency must notify the minor's parent(s), guardian, or custodian as to the nature of the treatment or observation if the name of a parent, guardian, or custodian is reasonably ascertainable. MCL 436.1703(10).

¹⁶ If the minor is under 18 years old and unemancipated, the health facility or agency must notify the minor's parent(s), guardian, or custodian as to the nature of the treatment or observation if the name of a parent, guardian, or custodian is reasonably ascertainable. MCL 436.1703(10).

- "The consumption by a minor of sacramental wine in connection with religious services at a church, synagogue, or temple is not prohibited by this act." MCL 436.1703(13).
- Minors involved in undercover operations. MCL 436.1703(14).

Further, MCL 436.1703 "does not prohibit a minor from possessing alcoholic liquor during regular working hours and in the course of his or her employment if employed by a person licensed by this act, by the commission, or by an agent of the commission, if the alcoholic liquor is not possessed for his or her personal consumption." MCL 436.1703(8).

5.30 Transporting or Possessing Open Alcohol in a Vehicle

A. Statutory Authority

"[A] person who is an operator or occupant shall not transport or possess alcoholic liquor in a container that is open or uncapped or upon which the seal is broken within the passenger area of a vehicle upon a highway, or within the passenger area of a moving vehicle in any place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, in [Michigan]." MCL 257.624a(1).

However, "[a] person may transport or possess alcoholic liquor in a container that is open or uncapped or upon which the seal is broken within the passenger area of a vehicle upon a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles in [Michigan], if the vehicle does not have a trunk or compartment separate from the passenger area, and the container is in a locked glove compartment, behind the last upright seat, or in an area not normally occupied by the operator or a passenger." MCL 257.624a(2).

Further, "[e]xcept as otherwise provided in MCL 257.624a, unless prohibited by local ordinance, [MCL 257.624a(1)-(2)] do not apply to a passenger in a commercial quadricycle. A passenger in a commercial quadricycle shall not transport or possess alcoholic liquor other than beer, wine, spirits, or a mixed spirits drink." MCL 257.624a(5).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

"As part of the sentence, the person may be ordered to perform community service and undergo substance abuse screening and assessment at his or her own expense as described in [MCL 436.1703(1)]." MCL 257.624a(3).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Two points. See MCL 257.320a(1)(s); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- License denial mandatory for unlicensed violation; license suspension for 90 days required for prior conviction; license suspension for one year required for two or more prior convictions. See MCL 257.303(1)(j); MCL 257.319(7). A restricted license may be issued. See MCL 257.319(7). See Section 1.43 for more information on license denial and Section 1.45 for more information on license suspension.

D. Issues

• Guilty/Nolo Contendere Pleas

"A court shall not accept a plea of guilty or nolo contendere for a violation of [MCL 257.624a] from a person charged solely with a violation of [MCL 257.625(6)¹⁷]." MCL 257.624a(3).

• Definition of Vehicle

A personal electric scooter qualifies as a vehicle under MCL 257.624a because the scooter is "a device upon which a person [is] transported upon a highway." *People v Lyon*, 310 Mich App 515, 516-517, 520

¹⁷MCL 257.625(6) prohibits a person who is less than 21 years of age from operating a vehicle with "any bodily alcohol content."

(2015) (the defendant was driving the scooter "along the paved portion of the 'curb lane'' and "weaving into the traffic lane" on a public highway while "holding an open can of beer[]"). The Court noted that "the definition of 'vehicle,' the term actually used in MCL 257.624a . . . , is much more inclusive than the definition of 'motor vehicle[.]" *Lyon*, 310 Mich App at 519. "[T]he MVC govern[s a] defendant's conduct when he use[s a] scooter as a vehicle upon a highway[]" and thereby undertakes "the duties of a vehicle driver, which include refraining from driving . . . with an open container." *Id.* at 521.

Chapter 6: Misdemeanor Moving Violations

6.1	Introduction and Scope Note	. 6-2
6.2	Disobeying the Direction of a Police Officer Who Is Regulating Traffic	C
		6-2
6.3	Drag Racing	. 6-3
6.4	Driving in Violation of a Restricted License	. 6-4
6.5	Driving or Moving an Unregistered Vehicle	. 6-5
6.6	Driving With an Invalid License	. 6-6
6.7	Failing to Stop for School Crossing Guard	. 6-7
6.8	Failing to Yield to Handicapped Individual	. 6-8
6.9	Improper Operation of a Golf Cart	. 6-8
6.10	Improper Use of Emergency Vehicle	6-11
6.11	Moving Violation Causing Injury to Person in Work Zone or School B	us
	Zone	
6.12	Moving Violation Causing Injury to Person Operating Farm Equipment	
6.13	Moving Violation Causing Serious Impairment of a Body Function	
1 1 4		
6.14	Moving Violation Causing Death	
6.15	Operating Motor Vehicle While License Is Suspended or Revoked	
6.16	Operating a Motor Vehicle While Unlicensed	
6.17		
6.18	Operating a Motor Vehicle Without a License in Possession Reckless Driving	
6.19	Reckless Driving Causing Miscarriage, Stillbirth, or Death	
6.20	Unlawful Use of an Automobile Without Intent to Steal (Joyriding)	
		0-20

6.1 Introduction and Scope Note

The principal source of Michigan traffic law is the Michigan Vehicle Code (MVC). The MVC clearly distinguishes the misdemeanor traffic offense from the felony and civil infraction. "It is a misdemeanor for a person to violate [the MVC], unless that violation is by [the MVC] or other law of this state declared to be a felony or a civil infraction." MCL 257.901(1).¹

6.2 Disobeying the Direction of a Police Officer Who Is Regulating Traffic²

A. Statutory Authority

"A person shall not refuse to comply with a lawful order or direction of a police officer when that officer, for public interest and safety, is guiding, directing, controlling, or regulating traffic on the highways of [Michigan]." MCL 257.602.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

¹The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

² Failure to obey some directions by a police officer may constitute a felony. See e.g., MCL 257.602a. For a discussion of felonies found in the MVC, see Chapters 7 and 8.

6.3 Drag Racing

A. Statutory Authority

"A person shall not do either of the following:

(a) Operate a vehicle upon a highway, or any other place open to the general public, including an area designated for the parking of motor vehicles, in a speed or acceleration contest or drag race or to make a speed record.

(b) Assist in a violation of [MCL 257.626a(1)(a)]. MCL 257.626a(1).

"[MCL 257.626a(1)] do[es] not apply within the Silver Lake State Park scramble area." MCL 257.626a(3).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Four points. See MCL 257.320a(1)(j); MCL 257.732(1)(a). See Section 1.42 for more information on points.

D. Issues

• Prima Facie Evidence of *Drag Racing*

"The operation of 2 or more vehicles either at speeds in excess of the prima facie lawfully established speed or rapidly accelerating from a common starting point to a speed in excess of the prima facie lawful speed is prima facie evidence of a drag race." MCL 257.626a(2).

"[MCL 257.626a(2)] do[es] not apply within the Silver Lake State Park scramble area." MCL 257.626a(3).

6.4 Driving in Violation of a Restricted License

A. Statutory Authority

"Upon proper showing of extenuating circumstances and special reasons, or need by an applicant who meets the age qualifications and when accompanied by the fee as provided in [the MVC], the secretary of state may recommend a restricted operator's . . . license containing conditions and restrictions applicable to the licensee, the type of special mechanical control devices required in a motor vehicle operated by the licensee, and the area, time, or other condition that the secretary of state considers necessary to assure the safe operation of a vehicle by the licensee and under which the licensee may operate a motor vehicle." MCL 257.312(1).

"A license issued to a person who is at least 14 years of age and under 16 years of age shall contain only the conditions determining the hours during which the licensee may drive a motor vehicle and the purpose for which it is to be driven." MCL 257.312(1). "A license issued to a minor who is at least 14 years of age and under 16 years of age shall be revoked by the secretary of state on the written request of a parent, guardian, or person standing in loco parentis." MCL 257.312(1).

"A person who violates a restriction imposed in a restricted license issued to that person is guilty of a misdemeanor." MCL 257.312(4). "[MCL 257.312(4)] does not apply to a person who is at least 14 years of age and under 16 years of age." *Id*.

"If a motor vehicle is being driven by a person who is at least 14 years of age and under 16 years of age, and that person is accompanied by a parent, guardian, or person standing in loco parentis, the conditions, limitations, and restrictions set forth in [MCL 257.312] do not apply." MCL 257.312(5).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- License suspension or revocation authorized. See MCL 257.312(3). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.

6.5 Driving or Moving an Unregistered Vehicle

A. Statutory Authority

"Every motor vehicle . . . when driven or moved on a street or highway, is subject to the registration and certificate of title provisions of [the MVC] except [as provided in MCL 257.216(1)(a)-(o)[.]" MCL 257.216(1). "Except as otherwise provided in [MCL 257.216(2)], a registration issued to a motor vehicle . . . when driven or moved on a street or highway, that expires on or after March 1, 2020 is valid until March 31, 2021. . . . If the registration issued to a motor vehicle . . . when driven or moved on a street or highway, that is used for commercial purposes expires on or after March 1, 2020, the registration is valid until March 31, 2021." MCL 257.216(2). "Except as otherwise provided in [MCL 257.216(2)], a registration issued to a motor vehicle . . . when driven or moved on a street or highway, that expires after March 31, 2021 but before August 1, 2021, is valid until 120 days after the date of the expiration. . . . If the registration issued to a motor vehicle . . . when driven or moved on a street or highway, that is used for commercial purposes expires after March 31, 2021 but before August 1, 2021, the registration is valid until 120 days after the date of the expiration." Id.

"It is a misdemeanor for any person to drive or move . . . upon any highway any vehicle of a type required to be registered hereunder which is not registered or for which a certificate of title has not been applied for or for which the appropriate fee has not been paid when and as required hereunder[.]" MCL 257.215.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

6.6 Driving With an Invalid License

A. Statutory Authority

"Except as provided in this act, an individual shall not drive a motor vehicle on a highway in [Michigan] unless that individual has a valid operator's...license with the appropriate group designation and indorsements for the type or class of vehicle being driven[.]" MCL 257.301(1).

"An individual shall not have more than 1 valid driver license." MCL 257.301(3). Thus, "[a]n individual shall not receive a license to operate a motor vehicle until that individual surrenders to the secretary of state all valid licenses to operate a motor vehicle issued to that individual by [Michigan] or any state or certifies that he or she does not possess a valid license." MCL 257.301(2).

"An operator's . . . license that expires on or after March 1, 2020 is valid until March 31, 2021." MCL 257.301(6). "An operator's . . . license that expires after March 31, 2021 but before August 1, 2021 is valid until 120 days after the date of the expiration." *Id*.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of

State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

6.7 Failing to Stop for School Crossing Guard

A. Statutory Authority

"A driver of a motor vehicle who fails to stop when a school crossing guard is in a school crossing and is holding a stop sign in an upright position visible to approaching vehicular traffic is guilty of a misdemeanor." MCL 257.613d(1).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Three points. See MCL 257.320a(1)(r); MCL 257.732(1)(a). See Section 1.42 for more information on points.

D. Issues

"In a proceeding for a violation of [MCL 257.613d], proof that the particular vehicle described in the citation, complaint, or warrant was used in the violation, together with proof that the defendant named in the citation complaint or warrant was the registered owner of the vehicle at the time of the violation, constitutes in evidence a

presumption that the registered owner of the vehicle was the driver of the vehicle at the time of the violation." MCL 257.613d(2).

6.8 Failing to Yield to Handicapped Individual

A. Statutory Authority

"A vehicle operator who approaches a person using a wheelchair or a device to aid the person to walk at a crosswalk or any other pedestrian crossing shall take necessary precautions to avoid accident or injury to the person using the wheelchair or device. A person who violates this subsection is guilty of a misdemeanor." MCL 257.612(4).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

6.9 Improper Operation of a Golf Cart

A. Statutory Authority

Where operation of a golf cart on the streets of a village, city, or township is allowed under MCL 257.657a(1), the following apply:

"(5) A person shall not operate a golf cart on any street unless he or she is at least 16 years old and is licensed to operate a motor vehicle. (6) The operator of a golf cart shall comply with the signal requirements of [MCL 257.648³] that apply to the operation of a vehicle.

(7) A person operating a golf cart upon a roadway shall ride as near to the right side of the roadway as practicable, exercising due care when passing a standing vehicle or a vehicle proceeding in the same direction.

(8) Except as otherwise provided in [MCL 257.657a(9)],⁴ a person shall not operate a golf cart on a state trunk line highway. This subsection does not prohibit a person from crossing a state trunk line highway when operating a golf cart on a street of a village, city, or township, using the most direct line of crossing.

* * *

(12) Where a usable and designated path for golf carts is provided adjacent to a highway or street, a person operating a golf cart may, by local ordinance, be required to use that path.

(13) A person operating a golf cart shall not pass between lines of traffic, but may pass on the left of traffic moving in his or her direction in the case of a 2-way street or on the left or right of traffic in the case of a 1-way street, in an unoccupied lane.

(14) A golf cart shall not be operated on a sidewalk constructed for the use of pedestrians.

(15) A golf cart shall be operated at a speed not to exceed 15 miles per hour and shall not be operated on a state trunk line highway or a highway or street with a speed limit of more than 30 miles per hour except to cross that state trunk line highway or highway or street. A village, city, or township may, by resolution, designate roads or classifications of roads for use by golf carts under this subsection.

(16) A golf cart shall not be operated on a state trunk line highway or the streets of a city, village, or township during the time period from 1/2 hour before sunset to 1/2 hour after sunrise."

* * *

³For more information on MCL 257.648, see Section 4.16.

⁴MCL 257.657a(9) authorizes local units of government to adopt an ordinance allowing the operation of golf carts on a state trunk line highway if the state transportation has granted a request for such use.

(18) A person operating a golf cart on a state trunk line highway shall ride as near to the right side of the roadway as practicable." MCL 257.657a(5)-(8), MCL 257.657a(12)-(16), and MCL 257.657a(18).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Issues

Golf cart operators and passengers are not required to wear a crash helmet. MCL 257.657a(17).

MCL 257.657a "does not apply to a police officer in the performance of his or her official duties." MCL 257.657a(19).

Golf carts operated under MCL 257.657a are not required to be registered under the MVC for purposes of MCL 500.3101. MCL 257.657a(20). However, a local unit of government may require registration. See generally MCL 257.657a(21), which states:

"A village, city, or township may require a golf cart registered within its jurisdiction to meet any or all of the following vehicle safety requirements of a low-speed vehicle for approval under [MCL 257.657a]:

(a) At least 2 headlamps that comply with [MCL 257.685].

(b) At least 1 tail lamp that complies with [MCL 257.686].

(c) At least 1 stop lamp and 1 lamp or mechanical signal device that comply with [MCL 257.697 and MCL 257.697b].

(d) At least 1 red reflector on each side of the golf cart as far to the rear as practicable and 1 red reflector on the rear of the golf cart as required for low-speed vehicles by 49 CFR 571.500.

(e) One exterior mirror mounted on the driver's side of the golf cart and either 1 exterior mirror

mounted on the passenger side of the golf cart or 1 interior mirror as required for low-speed vehicles by 49 CFR 571.500.

(f) Brakes and a parking brake that comply with [MCL 257.704].

(g) A horn that complies with [MCL 257.706].

(h) A windshield that complies with [MCL 257.708a].

(i) A manufacturer's identification number permanently affixed to the frame of the golf cart.

(j) Safety belts that comply with [MCL 257.710a] and that are used as required by [MCL 257.710e].

(k) The crash helmet requirements applicable to low-speed vehicles under [MCL 257.658b]." MCL 257.657a(21).

6.10 Improper Use of Emergency Vehicle

A. Statutory Authority

"The driver of an authorized emergency vehicle when responding to an emergency call, but not while returning from an emergency call, or when pursuing or apprehending a person who has violated or is violating the law or is charged with or suspected of violating the law may exercise the privileges set forth in [MCL 257.603], subject to the conditions of [MCL 257.603]." MCL 257.603(2).

"The driver of an authorized emergency vehicle may do any of the following:

(a) Park or stand, irrespective of [the MVC].

(b) Proceed past a red or stop signal or stop sign, but only after slowing down as may be necessary for safe operation.

(c) Exceed the prima facie speed limits so long as he or she does not endanger life or property.^[5]

⁵ See also MCL 257.632.

(d) Disregard regulations governing direction of movement or turning in a specified direction." MCL 257.603(3).

"The exemptions granted in [MCL 257.603] to an authorized emergency vehicle apply only when the driver of the vehicle while in motion sounds an audible signal by bell, siren, air horn, or exhaust whistle as may be reasonably necessary, except as provided in [MCL $257.603(5)^{\circ}$], and when the vehicle is equipped with at least 1 lighted lamp displaying a flashing, oscillating, or rotating red or blue light visible under normal atmospheric conditions from a distance of 500 feet in a 360 degree arc unless it is not advisable to equip a police vehicle operating as an authorized emergency vehicle with a flashing, oscillating or rotating light visible in a 360 degree arc. In those cases, a police vehicle shall display a flashing, oscillating, or rotating red or blue light visible under normal atmospheric conditions from a distance of 500 feet to the front of the vehicle. Only police vehicles that are publicly owned shall be equipped with a flashing, oscillating, or rotating blue light that when activated is visible under normal atmospheric conditions from a distance of 500 feet in a 360 degree arc." MCL 257.603(4).

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

⁶"A police vehicle shall retain the exemptions granted in [MCL 257.603] to an authorized emergency vehicle without sounding an audible signal if the police vehicle is engaged in an emergency run in which silence is required." MCL 257.603(5).

D. Issues

• Operating with Due Regard for Others' Safety

"While authorized emergency vehicles are excused by statute from obeying most of the rules of the road, MCL 257.603, MCL 257.632, they must be driven with due regard for the safety of others." *Terry v Detroit*, 226 Mich App 418, 429 (1997). See also *Flanagin v Kalkaska Co Rd Comm*, 319 Mich App 633, 637-638 (2017) (holding that "a plow truck operator is not necessarily committing a moving violation by driving across the centerline while plowing the road[,]" but that the requirement to drive "with due regard for the safety of others" when a vehicle is authorized to disobey the rules of the road applies to a plow truck in the same way it applies to emergency vehicles) (quotation marks and citation omitted). MCL 257.603 does not establish immunity from suit or "an excuse to be negligent[;]" but rather, it "merely recognize[s] that drivers, under the covered circumstances, are not violating these particular provisions of the [MVC]." *Flanagin*, 319 Mich App at 639.

6.11 Moving Violation Causing Injury to Person in Work Zone or School Bus Zone

A. Statutory Authority

"A person who commits a moving violation in a work zone or a school bus zone for which not fewer than 3 points are assigned under [MCL 257.320a] and as a result causes injury to another person in the work zone or school bus zone is guilty of a misdemeanor[.]" MCL 257.601b(2).

B. Penalties

MCL 257.601b(2) provides for:

- imprisonment for not more than one year; or
- fine of not more than \$1,000; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of

State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(b); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- Mandatory 90 day license suspension; license revocation or denial authorized under certain circumstances. See MCL 257.303(2)(b); MCL 257.319(3)(b). See Section 1.45 for more information on license suspension, Section 1.44 for more information on license revocation, and Section 1.43 for more information on license denial.

6.12 Moving Violation Causing Injury to Person Operating Farm Equipment

A. Statutory Authority

"A person who commits a moving violation that has criminal penalties and as a result causes injury to a person operating an implement of husbandry on a highway in compliance with [the MVC] is guilty of a misdemeanor[.]" MCL 257.601c(1).

B. Penalties

MCL 257.601c(1) provides for:

- imprisonment for not more than one year; or
- fine of not more than \$1,000; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Six points. See MCL 257.320a(1)(b); MCL 257.732(1)(a). See Section 1.42 for more information on points.

 Mandatory 90 day license suspension; license revocation or denial authorized under certain circumstances. See MCL 257.303(2)(b); MCL 257.319(3)(b). See Section 1.45 for more information on license suspension, Section 1.44 for more information on license revocation, and Section 1.43 for more information on license denial.

6.13 Moving Violation Causing Serious Impairment of a Body Function

A. Statutory Authority

"A person who commits a moving violation while operating a vehicle upon a highway or other place open to the general public, including, but not limited to, an area designated for the parking of motor vehicles, that causes serious impairment of a body function to another person is guilty of a misdemeanor[.]" MCL 257.601d(2).

MCL 257.601d "does not prohibit the person from being charged with, convicted of, or punished for any other violation of law." MCL 257.601d(3).

B. Relevant Jury Instruction

• M Crim JI 15.19 addresses the elements of a moving violation causing serious impairment of a body function.⁷

C. Penalties

MCL 257.601d(2) provides for:

- imprisonment for not more than 93 days; or
- fine of not more than \$500; or
- both.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and

⁷The instruction should only be used for act committed on or after October 31, 2010.

applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(b); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- Mandatory one year license suspension; license denial authorized under certain circumstances. See MCL 257.303; MCL 257.319(2)(e). See Section 1.45 for more information on license suspension, and Section 1.43 for more information on license denial.

E. Issues

• Elements

MCL 257.601d(2) "plainly requires a causal link between the moving violation and the injury, not simply a causal link between the operation of the vehicle and the injury." People v Czuprynski, 325 Mich App 449, 461 (2018) ("a moving violation together with the operation of the motor vehicle must cause the serious impairment of a body function"). M Crim JI 15.19 erroneously "tells the jury that operation of the vehicle *alone* must be the proximate cause [of the injury]." Czuprynski, 325 Mich App at 462. M Crim JI 15.19 "misapplies proximate causation to [MCL 257.601d(2)]" because it "relieve[s] the prosecution of proving that the moving violation caused the accident and instead require[s] only that 'the operation of the motor vehicle' caused the accident." Czuprynski, 325 Mich App at 461, 462 (M Crim JI 15.19 includes an incorrect statement of law, and its use by the trial court coupled with "strongly controverted evidence as to the cause of the accident" was not harmless error because the error "almost certainly was outcome determinative").

"MCL 257.601d imposes strict liability upon a motorist who commits a moving violation causing serious impairment of a body function to another person[.]" *People v Pace*, 311 Mich App 1, 12 (2015). "[T]he prosecution is required to prove solely (1) the commission of a moving violation; (2) another person suffered a serious impairment of a body function; and (3) a causal link between the bodily injury and the moving violation, i.e., factual and proximate causation. The prosecution is not required to also prove that [the] defendant operated his [or her] vehicle in a negligent manner[.]" *Id.* See also *Czuprynski*, 325 Mich App at 462, 468, 469.

A general verdict as to the moving violation, without unanimity, is sufficient to support a jury's conclusion that a violation of MCL 257.601d(2) occurred. *Czuprynski*, 325 Mich App at 467 (the prosecutor may present alternative theories regarding the type of moving violation).

• Due Process

"[I]mposing strict liability for the offense of committing a moving violation causing serious impairment of a body function[, MCL 257.601d,] does not offend due process." *Pace*, 311 Mich App at 12 (noting that "the offense is a misdemeanor [and despite the severe harm that such an offense inflicts upon the victim,]...[t]he penalty is ... relatively small[,]" and that "because the crime is a misdemeanor only, it is far less likely to 'besmirch' the defendant[]") (citations omitted).

6.14 Moving Violation Causing Death

A. Statutory Authority

"A person who commits a moving violation while operating a vehicle upon a highway or other place open to the general public, including, but not limited to, an area designated for the parking of motor vehicles, is guilty of a misdemeanor . . . if the moving violation was the proximate cause of the death of another person." MCL 257.601d(1).

MCL 257.601d "does not prohibit the person from being charged with, convicted of, or punished for any other violation of law." MCL 257.601d(3).

B. Relevant Jury Instruction

• M Crim JI 15.18 addresses the elements of a moving violation causing death.⁸

C. Penalties

MCL 257.601d(1) provides for:

- imprisonment for not more than 1 year; or
- fine of not more than \$2,000; or
- both.

⁸The instruction should be used only for acts committed on or after October 31, 2010.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(b); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- Mandatory one year license suspension; license denial authorized under certain circumstances. See MCL 257.303; MCL 257.319(2)(e). See Section 1.45 for more information on license suspension and Section 1.43 for more information on license denial.

6.15 Operating Motor Vehicle While License Is Suspended or Revoked⁹

A. Statutory Authority

"An individual whose operator's . . . license or registration certificate has been suspended or revoked, whose application for license has been denied, or who has never applied for a license, shall not operate a motor vehicle on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of motor vehicles, within [Michigan]." MCL 257.904(1).

B. Relevant Jury Instruction

• M Crim JI 15.20 addresses the elements of driving while license is suspended or revoked.

C. Penalties

MCL 257.904(3) provides for:

- First Violation:
 - imprisonment for not more than 93 days; or

⁹The felony offenses contemplated by MCL 257.904 are discussed in Chapter Chapter 8.

- fine of not more than \$500; or
- both.
- Violation Occurring After Prior Conviction:
 - imprisonment for not more than one year; or
 - fine of not more than \$1,000; or
 - both.

"A prior conviction under [MCL 257.904] must be established at or before sentencing by 1 or more of the following:

- (a) A copy of a judgment of conviction.
- (b) An abstract of conviction.
- (c) A transcript of a prior trial, plea, or sentencing.
- (d) A copy of a court register of action.
- (e) A copy of the defendant's driving record.
- (f) Information contained in a presentence report.
- (g) An admission by the defendant." MCL 257.904(9).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points (at-fault collision) or two points (any other moving violation). See MCL 257.320a(1)(*l*); MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- License suspension and/or revocation required, depending on the circumstances. See MCL 257.904(10)-(12). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
- Registration plate cancellation required.¹⁰ See MCL 257.904(3).

- Vehicle impoundment required. See MCL 257.904b(1).¹¹ See Section 1.48 and Section 9.12(B) for more information on vehicle impoundment.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues

• Scope of MCL 257.904

MCL 257.904 "does not apply to an individual who operates a vehicle solely for the purpose of protecting human life or property if the life or property is endangered and summoning prompt aid is essential." MCL 257.904(15).

"For purposes of [MCL 257.904], a person who never applied for a license includes a person who applied for a license, was denied, and never applied again." MCL 257.904(19).

• Individuals With Expired License

"The plain language of [MCL 257.904(1)] applies to persons who never apply for a license or who obtain one but subsequently have the license suspended or revoked because of improper driving." *People v Acosta-Baustista*, 296 Mich App 404, 408 (2012). "In the first instance, the person has never been adjudged fit to drive." *Id*. "In the second instance, the person has specifically been adjudged unfit to drive." *Id*. MCL 257.904(1) does not apply to "a person driving on a valid but recently expired license[.]" *Acosta-Bautista*, 296 Mich App at 409. "[A] person with a valid license who has simply let it lapse is a person adjudged fit to drive who has merely failed to keep up the related paperwork." *Id*. at 408.

• Inference that Operator is the Registered Owner

A traffic stop conducted "after running a vehicle's license plate and learning that the registered owner has a revoked driver's license," is reasonable "when the officer lacks information negating an inference that the owner is the driver of the vehicle[.]" *Kansas v Glover*, ____ US

¹⁰ Upon a first violation of MCL 257.904(1) or MCL 257.904(2), "[u]nless the vehicle was stolen or used with the permission of a person who did not knowingly permit an unlicensed driver to operate the vehicle, the registration plates of the vehicle must be canceled by the secretary of state upon notification by a peace officer." MCL 257.904(3)(a). For any violation of MCL 257.904(1) or MCL 257.904(2) that occurs after a prior conviction, the registration plates of the vehicle must be canceled unless the vehicle was stolen. MCL 257.904(3)(b).

¹¹Note that MCL 257.904b references subdivisions of MCL 257.904 that are not longer current. For further explanation, see Section 1.48.

_____, ____ (2020). "The fact that the registered owner of a vehicle is not always the driver of the vehicle does not negate the reasonableness of [the officer's] inference." *Id.* at _____. However, "the presence of additional facts might dispel reasonable suspicion," such as "if an officer knows that the registered owner of the vehicle is in his midsixties but observes that the driver is in her mid-twenties, then the totality of the circumstances would not raise a suspicion that the particular individual being stopped is engaged in wrongdoing." *Id.* at _____ (quotation marks and citation omitted). In *Glover*, the officer "drew the commonsense inference that [defendant] was likely the driver of the vehicle, which provided more than reasonable suspicion to initiate the stop," where he "knew that the registered owner of the truck had a revoked license and that the model of the truck matched the observed vehicle," and he "possessed no exculpatory information[.]" *Id.* at _____.

• Immigration Status

"[T]he plain language of [MCL 257.904] and [the Convention on the Regulation of Inter-American Automotive Traffic 1943] support a finding that the reasoning underlying MCL 257.904(1) . . . is not affected by the motor vehicle operator's immigration status and will remain the same regardless of whether the motor vehicle operator is driving pursuant to a license from Michigan, a foreign country that is a signatory to the convention, or one of the other 49 states." *Acosta-Baustista*, 296 Mich App at 411.

6.16 Operating a Motor Vehicle While Unlicensed

A. Statutory Authority

"Any person, not exempt from license under [the MVC], who operates a motor vehicle on the highways of [Michigan] and who is unable to show that he or she has been issued a license to operate a motor vehicle by any state or foreign country valid within the last 3 years before operating the vehicle, is guilty of a misdemeanor[.]" MCL 257.904a.

B. Penalties

MCL 257.904a provides for:

- First Offense:
 - imprisonment for not more than 90 days; or
 - fine of \$50 to \$100; or

- both.
- Second Offense:
 - imprisonment for not more than 90 days; or
 - fine of \$100; or
 - both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.

6.17 Operating a Motor Vehicle Without a License in Possession

A. Statutory Authority

"The licensee shall have his or her operator's . . . license, or the receipt described in [MCL 257.311a¹²], in his or her immediate possession at all times when operating a motor vehicle, and shall display the same upon demand of any police officer, who shall identify himself or herself as such." MCL 257.311.

B. Penalties

MCL 257.901(2) provides for:

- imprisonment for not more than 90 days; or
- fine of not more than \$100; or

 $^{^{12}}$ "If the court requires a person who is accused of a misdemeanor or ordinance violation to surrender his or her operator's . . . license pursuant to [MCL 765.6 or MCL 780.64], and if the license is not expired, suspended, revoked, or canceled, the court shall issue to the licensee a receipt for the license." MCL 257.311a. "The receipt shall have the effect of granting driving privileges identical to the operator's . . . license surrendered to the court[.]" *Id*.

• both.

"If a person has received a citation for a violation of [MCL 257.311], the court shall waive any fine and costs, upon receipt of certification by a law enforcement agency that the person, before the appearance date on the citation, has produced his or her operator's . . . license and that the license was valid on the date the violation of [MCL 257.311] occurred." MCL 257.901a.

C. Issues

• Establishing Police Officer's Identity

"Under MCL 257.311, . . . not only can identification of [a police] officer occur by observation of the officer's uniform and fully marked vehicle, in the case of a non-uniformed police officer, identification can occur by use of a badge or oral or other identification." *People v McKinley*, 255 Mich App 20, 30 (2003). In *McKinley*, 255 Mich App at 29, the police officer "had activated the emergency equipment in his fully marked sheriff's department vehicle and approached [the] defendant while he was fully uniformed as a . . . sheriff's deputy." The Court of Appeals held that "such indicia are sufficient to satisfy the statutory requirement [in MCL 257.311] that an officer identify himself as an officer when requesting a driver's license from a person who was operating a motor vehicle." *McKinley*, 255 Mich App at 29.

6.18 Reckless Driving

A. Statutory Authority

"A person who violates [MCL 257.626] is guilty of reckless driving as provided in [MCL 257.626]." MCL 257.626(1).

"Except as otherwise provided in [MCL 257.626], a person who operates a vehicle upon a highway or a frozen public lake, stream, or pond or other place open to the general public, including, but not limited to, an area designated for the parking of motor vehicles, in willful or wanton disregard for the safety of persons or property is guilty of a misdemeanor[.]" MCL 257.626(2).

B. Relevant Jury Instruction

• M Crim JI 15.15 addresses the elements of reckless driving.

C. Penalties

MCL 257.626(2) provides for:

- imprisonment for not more than 93 days; or
- fine of not more than \$500; or
- both.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(e); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- Mandatory 90 day license suspension; license revocation/denial required under certain circumstances. See MCL 257.303(2)(a); MCL 257.319(3)(b). See Section 1.45 for more information on license suspension, Section 1.44 for more information on license revocation, and Section 1.43 for more information on license denial.

E. Issues

• Willful or Wanton Disregard

"Willful or wanton disregard means more than simple carelessness but does not require proof of an intent to cause harm. It means knowingly disregarding the possible risks to the safety of people or property." *People v Carll*, 322 Mich App 690, 695 (2018) (quotation marks omitted). There was "[e]xtensive evidence" showing the defendant "drove in a manner that willfully or wantonly disregarded a high risk of serious injury to the people in his vehicle and other vehicles" where "there was evidence that defendant purposefully drove through a stop sign at high speed without any attempt to brake and that he might even have accelerated into the intersection." *Id.* at 696-697.

6.19 Reckless Driving Causing Miscarriage, Stillbirth, or Death

A. Statutory Authority

"If a person operates a motor vehicle in a careless or reckless manner, but not willfully or wantonly, that is the proximate cause of an accident involving a pregnant individual and the accident results in a miscarriage or stillbirth by that individual or death to the embryo or fetus, the person is guilty of a misdemeanor[.]"¹³ MCL 750.90e.

B. Penalties

MCL 750.90e provides for:

- imprisonment for not more than 2 years; or
- fine of not more than \$2,000; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(a); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- License revocation/denial required under certain circumstances. See MCL 257.303(2)(b)(*iii*); MCL 257.303(2)(e). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

¹³MCL 750.8 provides that "[w]hen any act or omission, not a felony, is punishable according to law, by a fine, penalty or forfeiture, and imprisonment, or by such fine, penalty or forfeiture, or imprisonment, in the discretion of the court, such act or omission shall be deemed a misdemeanor." Accordingly, although MCL 750.90e is specifically designated as a "misdemeanor," it is treated as a felony for purposes of licensing sanctions under the Michigan Vehicle Code.

6.20 Unlawful Use of an Automobile Without Intent to Steal (Joyriding)

A. Statutory Authority

"Any person who takes or uses without authority any motor vehicle without intent to steal the same, or who is a party to such unauthorized taking or using, is guilty of a misdemeanor[.]" MCL 750.414.

"[MCL 750.414] does not apply to any person or persons employed by the owner of said motor vehicle or anyone else, who, by the nature of his or her employment, has the charge of or the authority to drive said motor vehicle if said motor vehicle is driven or used without the owner's knowledge or consent." MCL 750.414.

B. Relevant Jury Instructions

- M Crim JI 24.2 addresses the elements of use of an automobile without authority and without intent to steal.
- M Crim JI 24.4 addresses the distinction between joyriding and the felony offense of unlawfully driving away an automobile, which is addressed in Section 8.12.

C. Penalties

MCL 750.414 provides for:

- imprisonment for not more than 2 years; or
- fine of not more than \$1,500.

However, if it is a first offense, "the court may reduce to punishment to imprisonment for not more than 3 months or a fine of not more than \$500.00." MCL 750.414.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Two points. See MCL 257.320a(1)(u); MCL 257.732(1)(a). See Section 1.42 for more information on points.
- Mandatory 90 day license suspension for no prior conviction; mandatory one year license suspension for one or more prior convictions. See MCL 257.319(6). See Section 1.45 for more information on license suspension.

E. Issues

• General Intent Crime

"'To be convicted of [MCL 750.414], a defendant must have intended to take or use the vehicle, knowing that he [or she] had no authority to do so." *Landon v Titan Ins Co*, 251 Mich App 633, 644 (2002), quoting *People v Laur*, 128 Mich App 453, 455 (1983). "Because no intent is required beyond that of doing the act itself, this is a general intent crime." *Landon*, 251 Mich App at 644.

• Distinguished From Crime of Unlawfully Driving Away an Automobile

The difference between the felony offense of unlawfully driving away an automobile (UDAA), MCL 750.413, and the misdemeanor offense of using a motor vehicle without authority but without intent to steal, MCL 750.414, is that "'UDAA (joyriding) requires the defendant to *take possession* of the motor vehicle without the owner's permission, while the misdemeanor offense of unlawful use of a motor vehicle is committed when an individual, who has been given lawful possession^[14] of a motor vehicle, *uses* it beyond the authority which has been granted to him by the owner." *Landon*, 251 Mich App at 644-645, quoting *People v Hayward*, 127 Mich App 50, 61 (1983) (emphasis in original). See also M Crim JI 24.4 for more information on the distinction between unlawfully driving away an automobile and unlawful use of an automobile without intent to steal (joyriding).

Rental Agreement

The evidence was "sufficient to demonstrate that plaintiff [an unlicensed driver] was in possession of the car... through voluntary action, and thus he 'took' it," in violation of MCL 750.414 where plaintiff's wife had rented the car from a dealership and "the rental agreement prohibited an unlicensed person from driving it," thus demonstrating "plaintiff's use and operation of the car was without

¹⁴"'Lawful possession' is not an element of the offense of unlawful use of an automobile." *People v Crosby*, 82 Mich App 1, 3 (1978). "It is just that 'unlawful possession' does not have to be shown to support a conviction for unlawful use of an automobile." *Id*.

authority of . . . the owner[.]" *Ahmed v Tokio Marine America Ins Co*, 337 Mich App 1, 13-14 (2021).

Chapter 7: Felony Nonmoving Violations

7.1	Introduction and Scope Note 7-2
7.2	Allowing A Person Who is Unlicensed or Whose License is Suspended or
	Revoked to Operate a Motor Vehicle Causing Serious Impairment of a
	Body Function or Death 7-2
7.3	Altering, Forging, or Falsifying Documents 7-4
7.4	Buying, Selling, or Using a Portable Signal Preemption Device
7.5	False Certification
7.6	False Statement in Application for Certificate of Title or in Assignment of
	Title; Possession of Stolen Vehicle 7-9
7.7	Odometer Tampering 7-11
7.8	Police Officer Knowingly Making Materially False Statement in Citation as
	Perjury 7-12
7.9	Possessing, Selling, or Offering for Sale a Stolen, False, or Counterfeit
	Certificate of Insurance

7.1 Introduction and Scope Note

This chapter contains an overview of nonmoving felony traffic offenses found in the Michigan Vehicle Code (MVC). Each section discusses the statutory authority for the offense, criminal penalties, and any applicable licensing sanctions.

7.2 Allowing A Person Who is Unlicensed or Whose License is Suspended or Revoked to Operate a Motor Vehicle¹ Causing Serious Impairment of a Body Function or Death

A. Statutory Authority

"A person shall not knowingly permit a motor vehicle owned by the person to be operated on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within this state, by an individual whose license or registration certificate is suspended or revoked, whose application for license has been denied, or who has never been licensed except as permitted by this act." MCL 257.904(7).

B. Relevant Jury Instructions

• M Crim JI 15.24 and M Crim JI 15.25 address the elements of permitting another person to drive a motor vehicle with a suspended or revoked license, causing serious impairment of a body function or death.

C. Criminal Penalties

If the operation of the motor vehicle results in serious impairment of a body function, the crime of knowingly allowing an individual who is unlicensed or whose license is suspended or revoked to operate a motor vehicle is punishable by:

- imprisonment for not more than 2 years,
- a fine of not less than \$1,000.00 or more than \$5,000.00,
- or both. MCL 257.904(7).

¹The moving felony offenses contemplated by MCL 257.904 are discussed in Chapter 8.

If the operation of the motor vehicle results in the death of another an individual, the crime of knowingly allowing an individual whose license is suspended or revoked to operate a motor vehicle is punishable by:

- imprisonment for not more than 5 years,
- a fine of not less than \$1,000.00 or more than \$5,000.00,
- or both. MCL 257.904(7).

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• It is unclear whether 6 points may be assessed for this violation. See MCL 257.320a(1)(a), which requires imposition of six points where "[m]anslaughter, negligent homicide, or a felony result[s] from the operation of a motor vehicle[.]" See Section 1.42 for more information on points.

E. Issues

MCL 257.904 "does not apply to an individual who operates a vehicle solely for the purpose of protecting human life or property if the life or property is endangered and summoning prompt aid is essential." MCL 257.904(15).

For purposes of MCL 257.904, "an individual who never applied for a license includes an individual who applied for a license, was denied, and never applied again." MCL 257.904(19).

F. Related Misdemeanor

It is a misdemeanor to "knowingly permit a motor vehicle owned by the person to be operated on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within this state by an individual whose license or registration certificate is suspended or revoked, whose application for license has been denied, or who has never applied for a license, except as permitted under [the MVC]." MCL 257.904(2)-(3). See Section 5.10 for a discussion of this misdemeanor offense.

7.3 Altering, Forging, or Falsifying Documents

A. Statutory Authority

"A person who commits any of the following acts is guilty of a felony:

(a) Alters with fraudulent intent any certificate of title, registration certificate, or registration plate issued by the department.

(b) Forges or counterfeits a certificate of title, registration certificate, or registration plate purporting to have been issued by the department.

(c) Alters or falsifies with fraudulent intent or forges an assignment upon a certificate of title.

(d) Holds or uses a certificate of title, registration certificate, or registration plate knowing that it has been altered, forged, or falsified.

(e) Knowingly possesses, sells, or offers for sale a stolen, false, or counterfeit certificate of title, registration certificate, registration plate, registration decal, or registration tab.

(f) Fraudulently indicates on a certificate of title that there is no security interest on record for the vehicle.

(g) Forges or counterfeits a letter from the holder of a security interest in a vehicle stating that the security interest has been released." MCL 257.257(1).

B. Criminal Penalties

For a **first conviction** of this offense, the following penalties apply pursuant to MCL 257.902:

- imprisonment for not less than one year or more than five years; or
- fine of not less than \$500 or more than \$5,000; or
- both.

For a **second conviction** of this offense, the following penalties apply pursuant to MCL 257.257(2):

- imprisonment for not less than two or more than seven years; or
- fine of not less than \$1,500 or more than \$7,000; or
- both.

For a **third or subsequent conviction** of this offense, the following penalties apply pursuant to MCL 257.257(3):

- imprisonment for not less than five years or more than 15 years; or
- fine of not less than \$5,000 or more than \$15,000; or
- both.

"A person who is convicted of a violation of subsection (1)(f) or (g), in addition to any other penalty, shall pay restitution to the holder of a security interest in the vehicle in the amount of the outstanding lien on the vehicle." MCL 257.257(4).

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• Suspension of the defendant's license is mandatory for a period of one year. MCL 257.319(2)(a). See Section 1.45 for more information on license suspension.

7.4 Buying, Selling, or Using a Portable Signal Preemption Device

A. Statutory Authority

"(1) Except as provided in subsection[] (3) . . . a person shall not do any of the following:

* * *

(b) Use a portable signal preemption device.

(c) Sell a portable signal preemption device to a person other than a person described in subsection (3).

(d) Purchase a portable signal preemption device for use other than a duty as described in subsections (3) and (4).

* * *

(3) This section does not apply to any of the following:

(a) A law enforcement agency in the course of providing law enforcement services.

(b) A fire station or a firefighter in the course of providing fire prevention or fire extinguishing services.

(c) An emergency medical service or ambulance in the course of providing emergency medical transportation or ambulance services.

(d) An operator, passenger, or owner of an authorized emergency vehicle in the course of his or her emergency duties." MCL 257.616a.

B. Criminal Penalties

Unless the violation results in a traffic accident, serious impairment of a body function, or death, improper use of a portable signal preemption device is punishable by:

- imprisonment for not more than two years; or
- fine of not more than \$10,000; or
- both. MCL 257.616a(2)(b).

The improper sale or purchase of a portable signal preemption device is punishable by:

- imprisonment for not more than two years; or
- fine of not more than \$10,000; or
- both. MCL 257.616a(2)(f).

When the improper use of a portable signal preemption device results in a traffic accident, the penalties are as follows:

- imprisonment for not more than five years; or
- fine of not more than \$15,000; or
- both. MCL 257.616a(2)(c).

When the improper use of a portable signal preemption device results in serious impairment of a body function, the penalties are increased as follows:

- imprisonment for not more than 10 years; or
- fine of not more than \$20,000; or
- both. MCL 257.616a(2)(d).

When the improper use of a portable signal preemption device results in the death of another person, the penalties are increased as follows:

- imprisonment for not more than 15 years; or
- fine of not more than \$25,000; or
- both. MCL 257.616a(2)(e).

C. Sanctions

Generally, buying, selling, or using a portable signal preemption device in violation of MCL 257.616a is not a reportable offense under the secretary of state's guidelines; thus, there are no licensing sanctions associated with it. However, licensing sanctions will apply if the manner in which the crime is committed satisfies the definition of a *felony in which a motor vehicle was used*. MCL 257.732(6); MCL 257.319(2)(c); MCL 257.303(7).

The MVC mandates license suspension for one year upon conviction of any offense categorized as a *felony in which a motor vehicle was used.* MCL 257.319(2)(c). The MVC further includes a *felony in which a motor vehicle was used* in the list of specified offenses for which a combination of two or more convictions within 7 years requires license revocation and denial. MCL 257.303(2)(b). Further, MCL 257.320a(1)(a) imposes 6 points for a felony resulting from the operation of a motor vehicle.

D. Related Misdemeanor

Possession of a signal preemption device by anyone other than one of the authorized emergency response services or a delivery service in the course of shipping or delivering the device is a misdemeanor. MCL 257.616a(1)(a).

7.5 False Certification

A. Statutory Authority

"A person who makes a false certification to a matter or thing required by the terms of this act to be certified, including but not limited to an application for any type of driver license, dealer license, vehicle certificate of title, vehicle registration, vehicle inspection, self-insurance, personal information, or commercial driver training school, is guilty of a felony" MCL 257.903(1).

B. Criminal Penalties

For a **first conviction** of this offense, the following penalties apply pursuant to MCL 257.902:

- imprisonment for not less than one year or more than five years; or
- fine of not less than \$500 or more than \$5,000; or
- both.

For a **second conviction** of this offense, the following penalties apply pursuant to MCL 257.903(2):

- imprisonment for not less than two years or more than seven years; or
- fine of not less than \$1,500 or more than \$7,000; or
- both.

For a **third or subsequent conviction** of this offense, the following penalties apply pursuant to MCL 257.903(3):

- imprisonment for not less than five years or more than 15 years; or
- fine of not less than \$5,000 or more than \$15,000; or
- both.

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• License suspension for 90 days if the person has no prior conviction within seven years; if the person has one or more prior convictions within seven years, the suspension period is one year. MCL 257.319(5). See Section 1.45 for more information on license suspension.

7.6 False Statement in Application for Certificate of Title or in Assignment of Title; Possession of Stolen Vehicle

A. Statutory Authority

"Any person who shall knowingly make any false statement of a material fact, either in his or her application for the certificate of title required by this act, or in any assignment of that title, or who, with intent to procure or pass title to a motor vehicle which he or she knows or has reason to believe has been stolen, shall receive or transfer possession of the same from or to another, or who shall have in his or her possession any vehicle which he or she knows or has reason to believe has been stolen, and the or she knows or has reason to believe has been stolen, and who is not an officer of the law engaged at the time in the performance of his or her duty as such officer, is guilty of a felony[.]" MCL 257.254.

B. Relevant Jury Instructions

- M Crim JI 24.6 addresses the elements of possession of a stolen automobile with intent to pass title.
- M Crim JI 24.7 addresses the elements of the offense of false statement about title to a motor vehicle.

C. Criminal Penalties

MCL 257.254 provides the following criminal penalties for this offense:

• imprisonment for not more than ten years; or

- fine of not more than \$5,000; or
- both.

These criminal sanctions are not exclusive of any other penalties prescribed by any law for larceny or the unauthorized taking of a vehicle. MCL 257.254.

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. See Section 1.41 for information on abstracting procedures.

• License suspension for 90 days if the person has no prior conviction within seven years; if the person has one or more prior convictions within seven years, the suspension period is one year. MCL 257.319(5). See Section 1.45 for more information on license suspension.

E. Issues

• Required elements

In view of the original title of this chapter, "'The Motor Vehicle Title Act', [²] and its stated purpose—'to protect titles of motor vehicles'— all parts of the act must be read to be germane only to conduct affecting titles or their fraudulent transfer." *People v Morton*, 384 Mich 38, 40 (1970). See also Const 1963, art 4, § 24.

Conviction for possession of a stolen vehicle is unauthorized in the absence of a showing of possession with knowledge that the vehicle was stolen coupled with intent to fraudulently transfer title. *Morton*, 384 Mich at 40. A defendant must transfer or intend to transfer the right of ownership of a stolen vehicle; transfer or intent to transfer the vehicle's certificate of title is not an element of the offense. *People v Ross*, 204 Mich App 310, 312-313 (1994); *People v Harbour*, 76 Mich App 552, 558-559 (1977).

Specific intent to fraudulently pass title is not an element of making a false application for a certificate of title; intent "can be inferred from the other necessary elements." *People v Jensen*, 162 Mich App 171, 181 (1987). This criminal offense is distinguishable from MCL 257.222(6), a misdemeanor offense that involves reproducing, altering, counterfeiting, forging, or duplicating certificate of title. *Jensen*, 162 Mich App at 184-185.

²Now titled the Michigan Vehicle Code, see 1949 PA 300; MCL 257.1 et seq.

• Meaning of motor vehicle

To constitute a "stolen *motor vehicle*" within the meaning of MCL 257.254, "the vehicle does not have to be immediately operable to be a motor vehicle under the normal definition of a motor vehicle [(see MCL 257.33)], [but] it must be constituted in such a way that it contains the major essential parts of a self-propelled vehicle. Therefore, the obvious meaning of a *stolen* motor vehicle is a vehicle the essential major elements of which have been stolen." *People v Boscaglia*, 419 Mich 556, 564 (1984) (rejecting the prosecution's argument that a truck with a stolen cab was a stolen motor vehicle for purposes of MCL 257.254, noting that "[t]he truck certainly is a vehicle which includes a stolen part, but it is not a stolen motor vehicle").

• The "false statement of a material fact" requirement

Whether a false statement was of *material* fact is an issue for the jury to decide. See *United States v Gaudin*, 515 US 506, 511 (1995) (holding that the defendant had a right to have a jury decide the question of materiality when it was an element of the charged offense).

7.7 Odometer Tampering

A. Statutory Authority

"(6) A person shall not alter, set back, or disconnect an odometer; cause or allow an odometer to be altered, set back, or disconnected; or advertise for sale, sell, use, install, or cause or allow to be installed a device which causes an odometer to register other than the actual mileage driven. This subsection does not prohibit the service, repair, or replacement of an odometer if the mileage indicated on the odometer remains the same as before the service, repair, or replacement. If the odometer is incapable of registering the same mileage as before the service, repair, or replacement, the odometer must be adjusted to read zero and a notice in writing must be attached to the left door frame of the vehicle by the owner or the owner's agent specifying the mileage prior to service, repair, or replacement of the odometer and the date on which it was serviced, repaired or replaced. A person shall not remove, deface, or alter any notice affixed to a motor vehicle under this subsection.

(7) A person who violates subsection (6) is guilty of a felony." MCL 257.233a(6)-(7).

B. Criminal Penalties

MCL 257.902 provides the following penalties:

- imprisonment for not less than one year or more than five years; or
- fine of not less than \$500 or more than \$5,000; or
- both.

C. Issues

• Applicability

The odometer statute also applies to a new or used vehicle dealer, a lessor of a leased vehicle, and an auction dealer or vehicle salvage pool operator. See MCL 257.233a(11)-(13).

• Required intent

"[T]he odometer statute in Michigan does not require the intent to defraud The main purpose behind the odometer statute is to protect a buyer from being defrauded by a seller who fraudulently turns back the odometer." *People v Houseman*, 128 Mich App 17, 22 (1983) (footnote and citations omitted).

• Civil liability

MCL 257.233a(15) governs the civil liability of a person who, with intent to defraud, violates MCL 257.233a(6). The person is liable in an amount equal to three times the amount of actual damages sustained or \$1,500, whichever is greater, plus costs and reasonable attorney fees in the case of a successful recovery of damages. MCL 257.233a(15).

7.8 Police Officer Knowingly Making Materially False Statement in Citation as Perjury

A. Statutory Authority

"A police officer who, knowing the statement is false, makes a materially false statement in a citation issued under [MCL 257.742³] is guilty of perjury, a felony[.]" MCL 257.744a.

³ Citations issued under MCL 257.742 are for civil infractions only. Citations for misdemeanors are issued under MCL 257.728.

B. Criminal Penalties

MCL 257.744a provides for the following penalties:

- imprisonment for not more than 15 years; and
- contempt of court.

C. Issues

The materiality of the defendant's false statement is an issue for the jury to decide. See *United States v Gaudin*, 515 US 506, 511 (1995).

7.9 Possessing, Selling, or Offering for Sale a Stolen, False, or Counterfeit Certificate of Insurance

A. Statutory Authority

"A person who knowingly possesses, sells, or offers for sale a stolen, false, or counterfeit certificate of insurance is guilty of a felony." MCL 257.329(1).

B. Criminal Penalties

For a **first conviction** of this offense, the following penalties apply pursuant to MCL 257.902:

- imprisonment for not less than one year or more than five years; or
- fine of not less than \$500 or more than \$5,000; or
- both.

For a **second conviction** of this offense, the following penalties apply pursuant to MCL 257.329(2):

- imprisonment for not less than two or more than seven years; or
- fine of not less than \$1,500 or more than \$7,000; or
- both.

For a **third or subsequent conviction** of this offense, the following penalties apply pursuant to MCL 257.329(3):

• imprisonment for not less than five years or more than 15 years; or

- fine of not less than \$5,000 or more than \$15,000; or
- both.

Chapter 8: Felony Moving Violations

8.1	Introduction and Scope Note 8-2	
Part A—MVC Offenses		
8.2	Failure to Stop at Signal of Police Officer ("Fleeing and Eluding") 8-2	
8.3	Failure to Use Due Care and Caution Causing Injury or Death to	
	Emergency Response Personnel 8-7	
8.4	Leaving the Scene of an Accident Resulting in Serious Impairment of a	
	Body Function or Death 8-9	
8.5	Moving Violation Causing Death in Work Zone or School Bus Zone	
8.6	Moving Violation Causing Death of Operator of Implement of Husbandry	
8.7	Operating a Motor Vehicle While Unlicensed or With Suspended/	
	Revoked License	
8.8	Reckless Driving Causing Death	
8.9	Reckless Driving Causing Serious Impairment of a Body Function	
Part B—MPC Offenses		
8.10	Malicious Destruction of Trees, Shrubs, Plants, or Soil with a Motor	
	Vehicle	
8.11	Manslaughter with a Motor Vehicle	
8.12	Unlawfully Driving Away an Automobile	

8.1 Introduction and Scope Note

This chapter contains an overview of moving felony traffic offenses found in the Michigan Vehicle Code (MVC) and the Michigan Penal Code (MPC). This chapter does not discuss violations of MCL 257.625, which are the subject of Chapter 9.

Part A—MVC Offenses

8.2 Failure to Stop at Signal of Police Officer ("Fleeing and Eluding")

A substantially similar statute appears in both the MVC and the Michigan Penal Code. MCL 257.602a and MCL 750.479a. Differences in the two statutes are noted below.

MCL 257.602a applies only to the operation of vehicles on the highways. See MCL 257.601, which states that provisions of the MVC apply "exclusively to the operation of vehicles upon highways except where a different place is specifically referred to"

A. Statutory Authority

"(1) A driver of a motor vehicle who is given by hand, voice, emergency light, or siren a visual or audible signal by a police or conservation officer, acting in the lawful performance of his or her duty, directing the driver to bring his or her motor vehicle to a stop shall not willfully fail to obey that direction by increasing the speed of the motor vehicle, extinguishing the lights of the motor vehicle, or otherwise attempting to flee or elude the officer. This subsection does not apply unless the police or conservation officer giving the signal is in uniform and the officer's vehicle is identified as an official police or department of natural resources vehicle."

(2) Except as provided in subsection (3), (4), or (5), an individual who violates subsection (1) is guilty of fourth-degree fleeing and eluding, a felony[,]....

(3) Except as provided in subsection (4) or (5), an individual who violates subsection (1) is guilty of third-degree fleeing and eluding, a felony[,]... if 1 or more of the following circumstances apply:

(a) The violation results in a collision or accident.

(b) A portion of the violation occurred in an area where the speed limit is 35 miles an hour or less, whether that speed limit is posted or imposed as a matter of law.

(c) The individual has a prior conviction for fourthdegree fleeing and eluding, attempted fourth-degree fleeing and eluding, or fleeing and eluding under a current or former law of this state prohibiting substantially similar conduct.

(4) Except as provided in subsection (5), an individual who violates subsection (1) is guilty of second-degree fleeing and eluding, a felony[,]... if 1 or more of the following circumstances apply:

(a) The violation results in serious injury^[1] to an individual.

(b) The individual has 1 or more prior convictions for first-, second-, or third-degree fleeing and eluding, attempted first-, second-, or third-degree fleeing and eluding, or fleeing and eluding under a current or former law of this state prohibiting substantially similar conduct.

(c) The individual has any combination of 2 or more prior convictions for fourth-degree fleeing and eluding, attempted fourth-degree fleeing and eluding, or fleeing and eluding under a current or former law of this state prohibiting substantially similar conduct.

(5) If the violation results in the death of another individual, an individual who violates subsection (1) is guilty of first-degree fleeing and eluding, a felony[.]" MCL 257.602a(1)-(5). See also MCL 750.479a(1)-(5).²

B. Relevant Jury Instructions

- M Crim JI 13.6d addresses the elements of fourth-degree fleeing and eluding.
- M Crim JI 13.6c addresses the elements of third-degree fleeing and eluding.

¹ MCL 750.479a(4)(a) requires the violation to result in serious impairment of a body function.

²Effective November 1, 2012, 2012 PA 60 amended MCL 750.479a to include vessels as well as motor vehicles in the fleeing and eluding statute. See the Michigan Judicial Institute's *Recreational Vehicles Benchbook* for information on fleeing and eluding as it relates to vessels.

- M Crim JI 13.6b addresses the elements of second-degree fleeing and eluding.
- M Crim JI 13.6a addresses the elements of first-degree fleeing and eluding.

C. Criminal Penalties

For **fourth-degree fleeing and eluding**, the penalties are as follows:

- imprisonment for not more than two years; or
- fine of not more than \$500;³ or
- both. MCL 257.602a(2).

For **third-degree fleeing and eluding**, the penalties are as follows:

- imprisonment for not more than five years; or
- fine of not more than \$1,000;⁴ or
- both. MCL 257.602a(3).

For **second-degree fleeing and eluding**, the penalties are as follows:

- imprisonment for not more than 10 years; or
- fine of not more than \$5,000;⁵ or
- both. MCL 257.602a(4).

For **first-degree fleeing and eluding**, the penalties are as follows:

- imprisonment for not more than 15 years; or
- fine of not more than \$10,000;⁶ or
- both. MCL 257.602a(5).

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this

³ MCL 750.479a(2) provides for a fine of not more than \$2,000.

⁴ MCL 750.479a(3) provides for a fine of not more than \$5,000.

⁵ MCL 750.479a(4) provides for a fine of not more than \$10,000.

⁶ MCL 750.479a(5) provides for a fine of not more than \$15,000.

offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(f). See Section 1.42 for more information on points.
- Following convictions of third- or fourth-degree fleeing and eluding, possible denial of license. See MCL 257.303(1)(c) (requiring license denial if a person's license is suspended, revoked, denied or canceled in any state). Following convictions of second- or first-degree fleeing and eluding, denial of license for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.43 for more information on license denial.
- Following convictions of second- or first-degree fleeing and eluding, license revocation for at least one year. MCL 257.303(2)(d); MCL 257.303(2)(f); MCL 257.303(4)(a)(*i*)-(*ii*); MCL 750.479a(7)(a). See Section 1.44 for more information on license revocation.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for fleeing and eluding and any of the motor vehicle felonies listed in MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- Following convictions of fourth- or third-degree fleeing and eluding, license suspension for a period of one year. MCL 257.319(2)(d); MCL 750.479a(6)(a). See Section 1.45 for more information on license suspension.

E. Issues

• Aiding and abetting

A passenger may be convicted of fleeing and eluding under an aiding and abetting theory. *People v Branch*, 202 Mich App 550, 551–552 (1993). In *Branch*, during a high-speed chase, the defendant threw full beer cans at a police car giving chase and instructed the driver. *Id.* at 551. The Court of Appeals held that the aiding and abetting statute, MCL 767.39, may be applied to passengers, and that the jury was properly instructed that a defendant must have intentionally assisted the driver to commit fleeing and eluding. *Branch*, 202 Mich App at 552.

• Causation

The statutory language, "results in," that appears in MCL 257.602a(5), requires only that a defendant's conduct was a factual cause of death; MCL 257.602a(5) does not require that a defendant's conduct be the proximate cause of death. *People v Wood*, 276 Mich App 669, 676 (2007). The Court explained that "[f]actual causation means just that: was defendant's criminal conduct a factual cause of the officer's death; or, but for defendant's fleeing and eluding, would the officer's death have occurred?" *Id.* In *Wood*, "[b]ecause the officer's death would not have occurred absent [the] defendant's fleeing and eluding, i.e., the police officer would not have lost control of his vehicle during the pursuit of the fleeing defendant, factual causation exist[ed]." *Id.*

• Conduct arising out of the same transaction

A person may be convicted under either MCL 257.602a(2)-(5) or MCL 750.479a(2)-(5) but not both, for conduct arising out of the same transaction. MCL 257.602a(6); MCL 750.479a(8). A person may be charged with and convicted of MCL 257.602a(5) or MCL 750.479a(5), for each death arising out of the same criminal transaction, and a court may impose consecutive sentences upon conviction. MCL 769.36(1)(a)-(b).

• Nature of the defendant's fleeing or eluding

The fleeing and eluding statutes are not limited to prohibiting only high-speed or long-distance "police chases." The Court of Appeals found sufficient evidence to bind over the defendant for trial where, after the police officer signaled for the defendant to stop, the defendant sped up slightly, made two turns, stopped the car, and attempted to flee on foot. *People v Grayer*, 235 Mich App 737, 742-744 (1999).⁷ A defendant's intent to flee or elude a police officer may be inferred from his or her acceleration, turning off the vehicle's headlights, or other similar actions after the officer signals the defendant to stop. *Id.* at 743-744. See also *People v Grayer (After Remand)*, 252 Mich App 349, 355–356 (2002) (holding that the evidence in this case was sufficient to support the defendant's conviction).

• Required evidence of the officer's signal to stop

Whether sufficient evidence exists to bind over a defendant for fleeing and eluding depends on "the type of signal given and the context in which it occurs[.]" *People v Green*, 260 Mich App 710, 718 (2004).⁸ In *Green*, the Michigan Court of Appeals explained that the

⁷Grayer interpreted MCL 750.479a; however, this statute is substantially similar to MCL 257.602a.

⁸*Green* interpreted MCL 750.479a; however, this statute is substantially similar to MCL 257.602a.

plain language of the fleeing and eluding statute requires a driver to stop when given a visual or audible signal by a police officer. *Id.* at 717. The officer's signal may be given by hand, voice, emergency light, or siren, but the Court emphasized that MCL 750.479a "does not require that this signal to the driver of a motor vehicle be given from within the officer's officially identified police vehicle." *Green*, 260 Mich App at 717. The Court further explained that the "fair and natural import" of the statutory language indicates that if the signal to stop is given by an officer away from that officer's vehicle, the statute requires that the officer be in uniform. *Id.* at 718. Similarly, "if the signal occurs by emergency light or siren, that signal must come from an officially identified police vehicle in order to hold a driver accountable for the offense of fleeing and eluding." *Id.*

F. Related Misdemeanor

Refusing to comply with a lawful order or direction of a police officer under MCL 257.602 is a separate misdemeanor offense. See MCL 257.901(1).

8.3 Failure to Use Due Care and Caution Causing Injury or Death to Emergency Response Personnel

A. Statutory Authority

"(1) Upon approaching and passing a stationary authorized emergency vehicle that is giving a visual signal by means of flashing, rotating, or oscillating red, blue, white, or amber lights as permitted by [MCL 257.698], the driver of an approaching vehicle shall exhibit due care and caution, as required under the following:

(a) On any public roadway with at least 2 adjacent lanes proceeding in the same direction of the stationary authorized emergency vehicle, the driver of the approaching vehicle shall proceed with caution, reduce his or her speed by at least 10 miles per hour below the posted speed limit, and yield the right-of-way by moving into a lane at least 1 moving lane or 2 vehicle widths apart from the stationary authorized emergency vehicle, unless directed otherwise by a police officer. If movement to an adjacent lane or 2 vehicle widths apart is not possible due to weather, road conditions, or the immediate presence of vehicular or pedestrian traffic in parallel moving lanes, the driver of the approaching vehicle shall proceed as required in subdivision (b). (b) On any public roadway that does not have at least 2 adjacent lanes proceeding in the same direction as the stationary authorized emergency vehicle, or if the movement by the driver of the vehicle into an adjacent lane or 2 vehicle widths apart is not possible as described in subdivision (a), the approaching vehicle shall proceed with due care and caution and reduce his or her speed by at least 10 miles per hour below the posted speed limit, or as directed by a police officer.

* * *

(3) A person who violates this section and causes injury to a police officer, firefighter, or other emergency response personnel in the immediate area of the stationary authorized emergency vehicle is guilty of a felony

(4) A person who violates this section and causes death to a police officer, firefighter, or other emergency response personnel in the immediate area of the stationary authorized emergency vehicle is guilty of a felony

(5) The operator of a vehicle upon a highway that has been divided into 2 roadways by leaving an intervening space, or by a physical barrier or clearly indicated dividing sections so constructed as to impede vehicular traffic, is not required to proceed with caution, reduce his or her speed, or yield the right-of-way for an authorized emergency vehicle that is stopped across the dividing space, barrier, or section. MCL 257.653a(1); MCL 257.653a(3)-(5).

B. Criminal Penalties

For causing injury to emergency response personnel, the penalties are as follows:

- imprisonment for not more than two years; or
- fine of not more than \$1,000; or
- both. MCL 257.653a(3).

For causing death to emergency response personnel, the penalties are as follows:

- imprisonment for not more than 15 years; or
- fine of not more than \$7,500; or
- both. MCL 257.653a(4).

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(b). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.653a(3)-(4) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

D. Related Civil Infraction

A violation of MCL 257.653a that does not result in injury or death to emergency response personnel is a civil infraction. MCL 257.653a(2).⁹

8.4 Leaving the Scene of an Accident Resulting in Serious Impairment of a Body Function or Death

A. Statutory Authority

"(1) The driver of a vehicle who knows or who has reason to believe that he or she has been involved in an accident upon public or private property that is open to travel by the public shall immediately stop his or her vehicle at the scene of the accident and shall remain there until the requirements of [MCL 257.619 (providing required information and rendering reasonable assistance)] are fulfilled or immediately report the accident to the nearest or most convenient police agency or officer to fulfill the requirements of MCL 257.619(a)-(b)] if there is a reasonable and

⁹ See Section 4.6 for more information on this offense.

honest belief that remaining at the scene will result in further harm. The stop shall be made without obstructing traffic more than is necessary.

(2) Except as provided in subsection (3), if the individual violates subsection (1) and the accident results in serious impairment of a body function or death, the individual is guilty of a felony

(3) If the individual violates subsection (1) following an accident caused by that individual and the accident results in the death of another individual, the individual is guilty of a felony" MCL 257.617.

"MCL 257.617(1) and MCL 257.617(3) establish crimes distinct from each other, each with its own elements, and [MCL 257.617(3)] is not a mere 'penalty enhancement' or 'penalty provision." *People v Dumback*, 330 Mich App 631, 642 (2019). See Section 8.4(E) for information on death as an element of the offense.

MCL 257.619¹⁰ requires:

"The driver of a vehicle who knows or who has reason to believe that he or she has been involved in an accident with an individual or with another vehicle that is operated or attended by another individual shall do all of the following:

(a) Give his or her name and address, and the registration number of the vehicle he or she is operating, including the name and address of the owner, to a police officer, the individual struck, or the driver or occupants of the vehicle with which he or she has collided.

(b) Exhibit his or her operator's or chauffeur's license to a police officer, individual struck, or the driver or occupants of the vehicle with which he or she has collided.

(c) Render to any individual injured in the accident reasonable assistance in securing medical aid or arrange for or provide transportation to any injured individual."

¹⁰ Violation of MCL 257.619 constitutes a misdemeanor offense. See MCL 257.901. See Section 5.2 for more information on this offense.

B. Relevant Jury Instructions

- M Crim JI 15.14 addresses the elements of leaving the scene of an accident resulting in serious impairment of a body function.
- M Crim JI 15.14a addresses the elements of leaving the scene of an accident causing death.

"M Crim JI 15.14a directs trial courts to inform the jury that it must find beyond a reasonable doubt that the defendant both caused the accident and that the accident resulted in death." *People v Dumback*, 330 Mich App 631, 647 (2019). See Section 8.4(E) for information on death as an element of the offense.

C. Criminal Penalties

If a defendant driver violates MCL 257.617(1) and the accident results in serious impairment of a body function or death, MCL 257.617(2) provides for the following penalties:

- imprisonment for not more than 5 years; or
- fine of not more than \$5,000; or
- both.

If, after an accident caused by the defendant, the defendant driver violates MCL 257.617(1) and the accident results in another individual's death, MCL 257.617(3) provides for the following penalties:

- imprisonment for not more than 15 years; or
- fine of not more than \$10,000; or
- both.

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• Six points. MCL 257.320a(1)(d). See Section 1.42 for more information on points.

- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.617 and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

E. Issues

• Aiding and abetting

The Michigan Court of Appeals has recognized that "a passenger may be charged with aiding and abetting a driving offense." People v Branch, 202 Mich App 550, 552 (1993), citing People v Hoaglin, 262 Mich 162, 172-173 (1933). In Branch, the defendant was charged with aiding and abetting the principal in fleeing and eluding. *Branch*, 202 Mich App at 551. In *Hoaglin*, the defendant was charged with aiding and abetting the principal in failing to render assistance after an accident. Hoaglin, 262 Mich at 170. Hoaglin was decided on the basis of the former versions of MCL 257.617 and MCL 257.619. Further, the defendant in *Hoaglin* argued that because the principal was only found guilty of a misdemeanor, he could not be convicted of aiding and abetting. Hoaglin, 262 Mich at 172. The Hoaglin Court rejected that contention, and upheld the defendant's conviction. Id. at 172-173. Thus, while not directly on point, Hoaglin suggests that it is proper to convict a defendant of aiding and abetting in violation of MCL 257.617 and MCL 257.619.

• Causation

MCL 257.617(3) "requires the prosecution to establish that the accident was 'caused' by the accused[.]" *People v Feezel,* 486 Mich 184, 194 (2010). In order to show causation, the prosecution must prove both factual and proximate causation. *Id.* "Factual causation exists if a finder of fact determines that 'but for' defendant's conduct the result would not have occurred. . . . Proximate causation 'is a legal construct designed to prevent criminal liability from attaching when the result of the defendant's conduct is viewed as too remote or unnatural." *Id.* at 194-195, quoting *People v Schaefer,* 473 Mich 418, 436 (2005). Proximate cause is not established if there is an intervening cause that supersedes the defendant's conduct and the victim's injury was broken[.]" *Feezel,* 486 Mich at 195, quoting *Schaefer,* 473 Mich at 436-437. "Whether an intervening cause

supersedes a defendant's conduct is a question of reasonable foreseeability." Feezel, 486 Mich at 195. A victim's ordinary negligence is not a superseding cause that severs proximate causation because ordinary negligence is reasonably foreseeable. Id. However, a victim's gross negligence or intentional misconduct is not reasonably foreseeable and is sufficient to sever proximate causation. Id. The fact that an accident victim was driving with a blood alcohol concentration of 0.054 grams per 100 milliliters of blood and had tetrahydrocannabinol (THC) in his blood stream was not an intervening cause that superseded the defendant's conduct such that the causal link between the defendant's conduct and the victim's injury was broken where "[t]here was no evidence that [the victim] was not properly driving within his marked lane, or that [the victim's] vehicle would not have safely passed [the] defendant if [the] defendant had not crossed the center line in front of [the victim], presenting a serious and unexpected hazard." People v Bergman, 312 Mich App 471, 486 (2015). "Thus, there was no evidence that [the victim] did anything that contributed to the accident in a way that would establish that he was negligent or grossly negligent and by his conduct was an intervening cause of the accident." Id. at 486 (holding that the trial court properly excluded evidence of the victim's intoxication during the defendant's trial for second-degree murder, OUIL causing death, and operating a vehicle with a suspended license causing death because the evidence was not relevant to causation under the facts of the case).

• Consecutive sentencing

A person may be charged with and convicted of MCL 257.617 for each death arising out of the same criminal transaction, and a court may impose consecutive sentences upon conviction. MCL 769.36(1)(a).

• Death as an element

"To support conviction under [MCL 257.617(3)], the prosecution must establish that the driver of a vehicle was involved in an accident, knew or had reason to know that the accident occurred, and did not stop or report," as is required to support a conviction under MCL 257.617(1), "but the prosecution must also prove that the driver caused the accident and that another person died as a result of the accident." *People v Dumback*, 330 Mich App 631, 642 (2019). "These additional requirements are elements of the offense because they increase the prescribed range of penalties to which a criminal defendant is exposed," and the "additional requirements must be presented to and found by the jury." *Id.* (alteration, quotation marks, and citation omitted).

• Intent to injure and knowledge not required

Intent to injure is not a necessary element of failing to stop and identify at the scene of a personal injury accident. *People v Strickland (Ronald)*, 79 Mich App 454, 456 (1977). Moreover, the driver need not know or have reason to believe that an accident resulted in physical injury, death, or property damage. See 2005 PA 3 (amending MCL 257.617, MCL 257.617a, MCL 257.618, and MCL 257.619 to remove the requirement that a driver know or have reason to believe that an accident resulted in physical injury, death, or property damage).

• Meaning of *accident*

Where the term *acciden*t appears in criminal statutes that forbid leaving the scene of a personal injury accident, it includes intentional conduct; the cause of the accident is not a concern. *People v Martinson*, 161 Mich App 55, 57 (1987).

• Right against self-incrimination

A defendant's Fifth Amendment right against self-incrimination is not implicated by requiring the defendant to comply with a statutory mandate to stop and disclose neutral information at the scene of a serious accident. *People v Goodin*, 257 Mich App 425, 432 (2003). The Michigan Court of Appeals held that the disclosures required of drivers involved in serious accidents do not create a significant risk of self-incrimination. *Id.* at 431. According to the *Goodin* Court:

"[T]he disclosures of one's name, address, vehicleregistration number, and driver's license required by MCL 257.617 and MCL 257.619 are neutral and do not implicate a driver in criminal conduct. Moreover, MCL 257.617 is not directed at a 'highly selective group' or a group 'inherently suspect of criminal activities,' but rather is aimed at any driver involved in an accident that results in serious personal injuries or death." *Goodin*, 257 Mich App at 430, citing *California v Byers*, 402 US 424 (1971).

• Single-vehicle accidents

The requirements of MCL 257.617 apply to single-vehicle accidents. *People v Noble*, 238 Mich App 647, 659 (1999) (stating that "[b]ecause the statutory language does not specifically limit the provisions of [MCL 257.617] to accidents involving two vehicles or a vehicle and a pedestrian, we conclude that the Legislature did not intend to so limit those provisions.")

• Statutory meaning of involved in

To be *involved in* an accident means to be implicated in an accident or connected with an accident in a logical or substantial manner, and the defendant need not have caused the accident in order to have been *involved in* the accident. *People v Oliver*, 242 Mich App 92, 97-98 (2000).

F. Related Misdemeanors

Leaving the scene of an accident resulting in personal injury, but not serious or aggravated personal injury, is a separate misdemeanor offense. MCL 257.617a.¹¹

Leaving the scene of an accident resulting in damage to an attended vehicle is also a separate misdemeanor offense. MCL 257.618.¹²

8.5 Moving Violation Causing Death in Work Zone or School Bus Zone

A. Statutory Authority

"A person who commits a moving violation in a work zone or school bus zone for which not fewer than 3 points are assigned under [MCL 257.320a] and as a result causes death to another person in the work zone or school bus zone is guilty of a felony" MCL 257.601b(3).

B. Criminal Penalties

MCL 257.601b(3) provides for the following penalties:

- imprisonment for not more than 15 years; or
- fine of not more than \$7,500; or
- both.

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by

¹¹ See Section 5.6 for more information on this offense.

¹² See Section 5.6 for more information on this offense.

the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(b). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.601b(3) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

D. Related Misdemeanor

Committing a moving violation for which not fewer than three points are assigned under MCL 257.320a and which causes injury to another person in a work zone or a school bus zone is a misdemeanor. MCL 257.601b(2).¹³

8.6 Moving Violation Causing Death of Operator of Implement of Husbandry

A. Statutory Authority

"A person who commits a moving violation that has criminal penalties and as a result causes death to a person operating an implement of husbandry on a highway in compliance with this act is guilty of a felony punishable by imprisonment for not more than 15 years or a fine of not more than \$7,500.00, or both." MCL 257.601c(2).

B. Criminal Penalties

MCL 257.601c(2) provides for the following penalties:

• imprisonment for not more than 15 years; or

¹³ See Section 6.11 for more information on this offense.

- fine of not more than \$7,500; or
- both.

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(b). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.601c(2) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

D. Related Misdemeanor

Committing a moving violation that has criminal penalties and causes injury to a person operating an implement of husbandry on a highway is a misdemeanor. MCL 257.601c(1).¹⁴

8.7 Operating a Motor Vehicle While Unlicensed or With Suspended/Revoked License¹⁵

A. Statutory Authority

"(1) An individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked, whose application for license has been denied, or who has never applied for a license, shall not operate a motor vehicle on a highway or other

¹⁴ See Section 6.12 for more information on this offense.

¹⁵The nonmoving felony offense contemplated by MCL 257.904 is discussed in Section 7.2.

place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of motor vehicles, within this state.

* * *

(4) An individual who operates a motor vehicle in violation of [MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle during the period of suspension or revocation and who, by operation of that motor vehicle, causes the death of another individual is guilty of a felony[.]...[MCL 257.904(4)] does not apply to an individual whose operator's or chauffeur's license was suspended because that individual failed to answer a citation or comply with an order or judgment pursuant to [MCL 257.321a].

(5) An individual who operates a motor vehicle in violation of [MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle during the period of suspension or revocation and who, by operation of that motor vehicle, causes the serious impairment of a body function of another individual is guilty of a felony[.]...[MCL 257.904(4)] does not apply to an individual whose operator's or chauffeur's license was suspended because that individual failed to answer a citation or comply with an order or judgment pursuant to [MCL 257.321a]." MCL 257.904(1); MCL 257.904(4)-(5).

B. Criminal Penalties

If the operation of the motor vehicle results in serious impairment of a body function, the crime is punishable by:

- imprisonment for not more than 5 years,
- a fine of not less than \$1,000.00 or more than \$5,000.00,
- or both. MCL 257.904(5).

If the operation of the motor vehicle results in the death of another individual, the crime is punishable by:

- imprisonment for not more than 15 years,
- a fine of not less than \$2,500.00 or more than \$10,000.00,
- or both. MCL 257.904(4).

C. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- While MCL 257.320a does not specifically mention MCL 257.904 in its list of offenses requiring an assessment of points, this offense will necessarily be committed in a fashion that constitutes "a felony resulting from the operation of a motor vehicle," for which six points are assessed. MCL 257.320a(1)(a). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.904(4)-(5) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

D. Issues

• Applicability of MCL 257.904

MCL 257.904 "does not apply to an individual who operates a vehicle solely for the purpose of protecting human life or property if the life or property is endangered and summoning prompt aid is essential." MCL 257.904(15).

"For purposes of [MCL 257.904], an individual who never applied for a license includes an individual who applied for a license, was denied, and never applied again." MCL 257.904(19). See also *People v Acosta-Baustista*, 296 Mich App 404, 408 (2012) (holding that MCL 257.904(1) "applies to persons who never apply for a license or who obtain one but subsequently have the license suspended or revoked because of improper driving"). MCL 257.904(1) does not apply to "a person driving on a valid but recently expired license[.]" *Acosta-Baustista,* 296 Mich App at 409. "[A] person with a valid license who has simply let it lapse is a person adjudged fit to drive who has merely failed to keep up the related paperwork." *Id.* at 408.

MCL 257.904 does not prohibit attempts. *People v Burton*, 252 Mich App 130, 138 (2002). In Burton, the Court noted that if "the Legislature intended to include the attempt to commit DWLS within the conduct proscribed by the version of MCL 257.904 applicable to the case at hand, it could have written subsection 1 of the statute to 'A person read, whose operator's . . . license . . . has been suspended . . . and notified . . . of who has been that suspension ... shall not operate [or attempt to operate] a motor vehicle upon a highway or other place open to the general public' That it chose not to do so will be considered to be a purposeful, not an inadvertent, act."¹⁶ Burton, 252 Mich App at 138.

• Causation

The causation element of MCL 257.625(4), a provision that is worded similarly to MCL 257.904(4)-(5), requires only that a defendant's operation of a motor vehicle—not a defendant's operation of a vehicle as affected by the defendant's state of intoxication—be a factual and proximate cause of the harm resulting from the statutory violation. *People v Schaefer*, 473 Mich 418, 446 (2005). The language of MCL 257.625(4) is almost identical¹⁷ to the language of MCL 257.904(4)-(5). Both statutes require a finding that the operator "by operation of that motor vehicle, causes the death [or 'the serious impairment of a body function[,]' in the case of MCL 257.904(5),] of another person" in order for the operator to be guilty of the specified felony. MCL 257.625(4); MCL 257.904(4)-(5). Thus, it seems clear that the prosecution need not demonstrate any causal link between a defendant's revoked or suspended license and the subsequent

¹⁶For more information on attempts, see Section 1.3.

¹⁷MCL 257.625(4) provides in pertinent part: "A person, whether licensed or not, who operates a motor vehicle in violation of subsection (1), (3), or (8) and by the operation of that motor vehicle causes the death of another person is guilty of a crime[.]" MCL 257.904(4) provides in pertinent part: "An individual who operates a motor vehicle in violation of [MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle during the period of suspension or revocation and who, by operation of that motor vehicle, causes the death of another individual is guilty of a felony[.]" MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate a motor vehicle in violation of [MCL 257.904(1)] or a nindividual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle in violation of [MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle in violation of [MCL 257.904(1)] or an individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked by another state who operates a motor vehicle during the period of suspension or revocation and who, by operation of that motor vehicle, causes the serious impairment of a body function of another individual is guilty of a felony[.]"

serious impairment of a body function or death in order to prove a violation of MCL 257.904(4) or (5).

The Schaefer Court explained:

"[MCL 257.625(4)] plainly requires that the victim's death be caused by the defendant's *operation* of the vehicle, not the defendant's *intoxicated* operation. Thus, the manner in which the defendant's intoxication affected his or her operation of the vehicle is unrelated to the causation element of the crime. The defendant's status as 'intoxicated' is a separate element of the offense used to identify the class of persons subject to liability under §625(4)." *Schaefer*, 473 Mich at 433.

• Double Jeopardy

A defendant's multiple convictions of second-degree murder, OUIL causing death, and driving with a suspended license causing death do not violate the double jeopardy clauses of the United States and Michigan constitutions. *People v Bergman*, 312 Mich App 471, 492 (2015). "The statutes governing second-degree murder and driving with a suspended license causing death enforce distinct societal norms, and their respective elements of malice and lack of a valid operator's license are distinctive to each[;] . . . [s]imilarly, the OUIL and suspended license statutes enforce distinct societal norms, and their respective elements of intoxication while driving and lack of a valid operator's license are distinctive to each." *Id.* at 492 (citations omitted).

• Immigration Status of the Defendant

"[T]he plain language of [MCL 257.904] and [the Convention on the Regulation of Inter-American Automotive Traffic 1943] support a finding that the reasoning underlying MCL 257.904(1)... is not affected by the motor vehicle operator's immigration status and will remain the same regardless of whether the motor vehicle operator is driving pursuant to a license from Michigan, a foreign country that is a signatory to the convention, or one of the other 49 states." *Acosta-Baustista*, 296 Mich App at 411.

• Proof of Notice

Admission of a secretary of state certificate of mailing to prove that the defendant was notified of the revocation or suspension of his or her license does not violate the defendant's constitutional right of confrontation because the certificate of mailing is not testimonial in nature. *People v Nunley*, 491 Mich 686, 689 (2012). The certificate of mailing is not testimonial because it "is necessarily generated before the commission of any crime, [and thus,] is a function of the legislatively authorized administrative role of the [secretary of state] independent from any investigatory or prosecutorial purpose." *Id.* at 689-690. Thus, a "certificate of mailing may be admitted into evidence absent accompanying witness testimony without violating the Confrontation Clause." *Id.* at 690.

E. Related Misdemeanor

"An individual whose operator's or chauffeur's license or registration certificate has been suspended or revoked, whose application for license has been denied, or who has never applied for a license, shall not operate a motor vehicle on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of motor vehicles, within [Michigan]." MCL 257.904(1). Violation of MCL 257.904(1), without causing serious impairment of a body function or death, is a misdemeanor. MCL 257.904(3).¹⁸

8.8 Reckless Driving Causing Death

A. Statutory Authority

"(1) A person who violates this section is guilty of reckless driving punishable as provided in this section.

(2) Except as otherwise provided in this section, a person who operates a vehicle upon a highway or a frozen public lake, stream, or pond or other place open to the general public, including, but not limited to, an area designated for the parking of motor vehicles, in willful or wanton disregard for the safety of persons or property is guilty of a misdemeanor^[19]....

* * *

(4) Beginning October 31, 2010, a person who operates a vehicle in violation of [MCL 257.626](2) and by the operation of that vehicle causes the death of another person is guilty of a felony[.]" MCL 257.626.

¹⁸See Section 6.15 for more information on this misdemeanor offense.

¹⁹ See Section 6.18 for a discussion of this misdemeanor offense.

B. Relevant Jury Instruction

• M Crim JI 15.16 addresses the elements of reckless driving causing death.²⁰

C. Criminal Penalties

For a violation of reckless driving causing death, the following penalties apply:

- imprisonment for not more than 15 years; or
- fine of not less than \$2,500 or more than \$10,000; or
- both. MCL 257.626(4).

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. See MCL 257.320a(1)(e). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- Possible vehicle immobilization for not more than 180 days. MCL 257.904d(1)(a). Vehicle immobilization must be ordered if the vehicle is not ordered forfeited. MCL 257.626(4). See Section 1.47 for more information on vehicle immobilization.
- Possible forfeiture of the vehicle. MCL 257.625n(1); MCL 257.626(4). See Section 1.49 for more information on forfeiture.

²⁰The instruction should be used only for acts committed on or after October 31, 2010.

E. Issues

• Jury Cannot be Instructed on Moving Violation Causing Death

MCL 257.626(5) provides that "[i]n a prosecution under [MCL 257.626(4)], the jury shall not be instructed regarding the crime of moving violation causing death." While "MCL 768.32(1) sets forth the general rule that a defendant is entitled to have the jury instructed on necessarily included lesser offenses[,] MCL 257.626(5) . . . sets forth a clear exception to this general rule[.]" *People v Jones*, 497 Mich 155, 168 (2014).²¹ The Court found that "MCL 257.626(5) is not a matter of practice and procedure, and, consequently [is not a] violation of separation of powers[.]" *Jones*, 497 Mich at 169. (concluding that creation of a substantive exception to the rule generally allowing instruction on necessarily included lesser offenses was a permissible exercise of legislative authority).

• Failing to maintain a vehicle and then operating the poorly maintained vehicle

"MCL 257.626 criminalizes the manner or nature of operation rather than the decision to operate following poor maintenance." People v Otto, ____ Mich App ____, ___ (2023). "[F]ailing to maintain a vehicle and then operating the poorly maintained vehicle-cannot support conviction under MCL 257.626(4)." Otto, ____ Mich App at ____. "[T]he second element-reckless driving-requires the manner of operation to be reckless rather than the *decision* to operate a vehicle that is negligently or carelessly maintained." Id. at ____. "[T]he third element-operation causing death-requires operation to be the factual and proximate cause of the victim's death." Id. at ____. However, "careless maintenance [cannot] satisfy the second and third elements: reckless operation, and operation causing death." Id. at ____. Indeed, "the prosecution was required to prove that [defendant's] operation, rather than [defendant's] poor maintenance must have been the cause of the accident." Id. at ____. In Otto, "an intervening event, a sudden mechanical failure, superseded [defendant's] conduct, such that the causal link between [defendant's] driving and the victim's death was broken." *Id.* at _____. The Otto Court held that "the prosecution failed to allege or prove that [defendant's] operation, not [defendant's] maintenance or an unexpected brake failure, was the proximate cause of the victim's death[.]" Id. at ____. Although "[t]here [was] no dispute that

²¹ The *Jones* Court "[a]ssum[ed], based on the record concession, that moving violation causing death . . . constitutes a necessarily included lesser offense of reckless driving causing death," and noted that even if it is a cognate lesser offense, the result would be the same because "MCL 768.32(1) does not permit cognate lesser offense instructions." *Jones*, 497 Mich at 161-162 n 16, 167-168. For more information on jury instructions, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 12.

[defendant's] operation was the factual cause," "the prosecution could not establish that [defendant's] operation was the proximate cause because it was attenuated from the harm by the superseding brake failure, severing the causal chain." *Id.* at ____.

• Willful and Wanton Disregard

"Willful or wanton disregard means more than simple carelessness but does not require proof of an intent to cause harm. It means knowingly disregarding the possible risks to the safety of people or property." *People v Carll*, 322 Mich App 690, 695 (2018) (quotation marks omitted). There was "[e]xtensive evidence" proving the defendant "drove in a manner that willfully or wantonly disregarded a high risk of serious injury to the people in his vehicle and other vehicles" where "there was evidence that defendant purposefully drove through a stop sign at high speed without any attempt to brake and that he might even have accelerated into the intersection." *Id.* at 696-697, 698.

8.9 Reckless Driving Causing Serious Impairment of a Body Function

A. Statutory Authority

"(1) A person who violates this section is guilty of reckless driving punishable as provided in this section.

(2) Except as otherwise provided in this section, a person who operates a vehicle upon a highway or a frozen public lake, stream, or pond or other place open to the general public, including, but not limited to, an area designated for the parking of motor vehicles, in willful or wanton disregard for the safety of persons or property is guilty of a misdemeanor^[22]....

(3) Beginning October 31, 2010, a person who operates a vehicle in violation of [MCL 257.626](2) and by the operation of that vehicle causes serious impairment of a body function to another person is guilty of a felony[.]" MCL 257.626(1)-(3).

B. Relevant Jury Instruction

• M Crim JI 15.17 addresses the elements of reckless driving causing serious impairment of a body function.²³

²² See Section 6.18 for a discussion of this misdemeanor offense.

²³The instruction should be used only for acts committed on or after October 31, 2010.

C. Criminal Penalties

For a violation of reckless driving causing serious impairment of a body function, the following penalties apply:

- imprisonment for not more than 5 years; or
- fine of not less than \$1,000 or more than \$5,000; or
- both. MCL 257.626(3).

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(e). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(d); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- Possible vehicle immobilization for not more than 180 days. MCL 257.904d(1)(a). Vehicle immobilization must be ordered if the vehicle is not ordered forfeited. MCL 257.626(3). See Section 1.47 for more information on vehicle immobilization.
- Possible forfeiture of the vehicle. MCL 257.625n(1); MCL 257.626(3). See Section 1.49 for more information on forfeiture.

E. Issues

• Willful and Wanton Disregard

"Willful or wanton disregard means more than simple carelessness but does not require proof of an intent to cause harm. It means knowingly disregarding the possible risks to the safety of people or property." *People v Carll*, 322 Mich App 690, 695 (2018) (quotation marks omitted). There was "[e]xtensive evidence" proving the defendant "drove in a manner that willfully or wantonly disregarded a high risk of serious injury to the people in his vehicle and other vehicles" where "there was evidence that defendant purposefully drove through a stop sign at high speed without any attempt to brake and that he might even have accelerated into the intersection." *Id.* at 696-697, 698.

Part B—MPC Offenses

8.10 Malicious Destruction of Trees, Shrubs, Plants, or Soil with a Motor Vehicle

A. Statutory Authority

"A person who willfully and maliciously, or wantonly and without cause, cuts down, destroys, or injures any tree, shrub, grass, turf, plants, crops, or soil of another that is standing, growing, or located on the land of another is guilty of a crime as follows:

* * *

(c) If any of the following apply, the person is guilty of a felony[:]

(*i*) The value of the trees, shrubs, grass, turf, plants, crops, or soil cut down, destroyed, or injured is \$1,000.00 or more but less than \$20,000.00.

(*ii*) The person violates [MCL 750.382(1)(b)(*i*) (value of property is \$200 or more but less than \$1,000)] and has 1 or more prior convictions for committing or attempting to commit an offense under [MCL 750.382]. For purposes of this subparagraph, however, a prior conviction does not include a conviction for a violation or attempted violation of [MCL 750.382(1)(a) (value of property is less than \$200) or MCL 750.382(1)(b)(*ii*) (violation of MCL 750.382(1)(a) with one prior conviction)].

(d) If any of the following apply, the person is guilty of a felony[:]

(*i*) The value of the trees, shrubs, grass, turf, plants, crops, or soil cut down, destroyed, or injured is \$20,000.00 or more.

(*ii*) The person violates [MCL 750.382(1)(c)(*i*) (value of property is \$1,000 or more but less than \$20,000)] and has 2 or more prior convictions for committing or attempting to commit an offense under [MCL 750.382]. For purposes of this subparagraph, however, a prior conviction does not include a conviction for a violation or attempted violation of [MCL 750.382(1)(a) (value of property is less than \$200) or MCL 750.382(1)(b)(*ii*) (violation of MCL 750.382(1)(a) with one prior conviction)]." MCL 750.382(1).

B. Criminal Penalties

A violation of MCL 750.382(1)(c) is a felony punishable as follows:

- imprisonment for not more than 5 years;
- fine of not more than \$10,000 or three times the value of the trees, shrubs, grass, turf, plants, crops, or soil, whichever is greater; or
- both.

A violation of MCL 750.382(1)(d) is a felony punishable as follows:

- imprisonment for not more than 10 years;
- fine of not more than \$15,000 or three times the value of the trees, shrubs, grass, turf, plants, crops, or soil, whichever is greater; or
- both.

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• While points are not assessed on the basis of this offense alone, it is likely the offense will be committed in a fashion that constitutes "a felony resulting from the operation of a motor vehicle," for which six points are assessed. MCL 257.320a(1)(a). See Section 1.42 for more information on points.

- Possible denial of license. See MCL 257.303(1)(c) (requiring license denial if a person's license is suspended, revoked, denied or canceled in any state). See Section 1.43 for more information on license denial.
- License suspension for 90 days. MCL 257.319(3)(c). See Section 1.45 for more information on license suspension.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.904(4)-(5) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

8.11 Manslaughter with a Motor Vehicle

A. Statutory Authority

"Manslaughter—Any person who shall commit the crime of manslaughter shall be guilty of a felony[.]" MCL 750.321.

B. Relevant Jury Instructions

- M Crim JI 16.8 addresses the elements of voluntary manslaughter.
- M Crim JI 16.10 addresses the elements of involuntary manslaughter.

C. Criminal Penalties

A violation of MCL 750.321 is a felony punishable by:

- imprisonment for not more than 15 years;
- fine of not more than \$7,500; or
- both. MCL 750.321.

"A person may be charged with and convicted of [MCL 750.321, where death results from the operation of a vehicle,] for each death arising out of the same criminal transaction, and the court may order the terms of imprisonment to be served consecutively to each other[.]" MCL 769.36(1)(b).

D. Sanctions

Only applicable sanctions are discussed; accordingly, if a particular sanction is omitted from this section it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

- Six points. MCL 257.320a(1)(a). See Section 1.42 for more information on points.
- License revocation and denial for at least one year. MCL 257.303(2)(e); MCL 257.303(4)(a)(*i*)-(*ii*). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for manslaughter and any of the motor vehicle felonies listed in MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

E. Issues

"The manslaughter statute, MCL 750.321, encompasses two types of common-law manslaughter: voluntary and involuntary." *In re Nale Estate*, 290 Mich App 704, 708 (2010). "Although the punishment for manslaughter is defined by statute, the common law defines the elements of voluntary and involuntary manslaughter." *Id*.

- "[V]oluntary manslaughter [is] an intentional killing committed under the influence of passion or hot blood produced by adequate provocation and before a reasonable time has passed for the blood to cool." *In re Nale Estate*, 290 Mich App at 708, citing *People v Mendoza*, 468 Mich 527, 534-535 (2003). Accordingly, voluntary manslaughter is "specifically designated [as] an 'intentional' killing." *In re Nale Estate*, 290 Mich App at 709.
- "[I]nvoluntary manslaughter [is] the killing of another without malice and unintentionally, but in doing some unlawful act not amounting to a felony nor naturally tending to cause death or great bodily harm, or in negligently doing some act lawful in itself, or by the negligent omission to perform a legal duty." *In re Nale Estate*, 290 Mich App at 709 (citations and quotations omitted).

A defendant may be convicted of involuntary manslaughter, MCL 750.321, if his or her conduct was "a" cause of death (not "the" cause of death). *People v Tims*, 449 Mich 83, 96 (1995).

8.12 Unlawfully Driving Away an Automobile

A. Statutory Authority

"Taking possession of and driving away a motor vehicle—Any person who shall, wilfully and without authority, take possession of and drive or take away, and any person who shall assist in or be a party to such taking possession, driving or taking away of any motor vehicle, belonging to another, shall be guilty of a felony[.]" MCL 750.413.

B. Relevant Jury Instructions

- M Crim JI 24.1 addresses the elements of unlawfully driving away an automobile.
- M Crim JI 24.4 addresses the distinction between unlawfully driving away an automobile and the misdemeanor offense of joyriding, which is addressed in Section 6.20.

C. Criminal Penalties

A person guilty of unlawfully driving away an automobile is punishable as follows:

• imprisonment for not more than 5 years. MCL 750.413.

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing certain snowmobile offenses and applicable sanctions. See Section 1.41 for information on abstracting procedures.

• While MCL 257.320a does not specifically mention MCL 750.413 in its list of offenses requiring an assessment of points, it is likely the offense will be committed in a fashion that constitutes "a felony resulting from the operation of a motor vehicle," for which six points are assessed. MCL

257.320a(1)(a). See Section 1.42 for more information on points.

- Possible denial of license. See MCL 257.303(1)(c) (requiring license denial if a person's license is suspended, revoked, denied or canceled in any state). See Section 1.43 for more information on license denial.
- License suspension for a period of one year. MCL 257.319(2)(b). See Section 1.45 for more information on license suspension.
- License revocation and denial also occur when a defendant has any combination of two or more convictions within seven years for violation of MCL 257.904(4)-(5) and any of the motor vehicle felonies listed at MCL 257.303(2)(b). See Section 1.44 for more information on license revocation and Section 1.43 for more information on license denial.

E. Issues

• Related Crime of Unlawful Use of an Automobile Without Intent to Steal (Joyriding)

The difference between the felony offense of unlawfully driving away an automobile (UDAA), MCL 750.413, and the misdemeanor offense of using a motor vehicle without authority but without intent to steal, MCL 750.414, is that "'UDAA (joyriding) requires the defendant to take possession of the motor vehicle without the owner's permission, while the misdemeanor offense of unlawful use of a motor vehicle is committed when an individual, who has been given lawful possession[²⁴] of a motor vehicle, uses it beyond the authority which has been granted to him [or her] by the owner." *Landon v Titan Ins Co*, 251 Mich App 633, 644-645 (2002), quoting *People v Hayward*, 127 Mich App 50, 61 (1983) (emphasis added in original). See also M Crim JI 24.4 for more information on the distinction between unlawfully driving away an automobile and unlawful use of an automobile without intent to steal (joyriding).

• UDAA Elements

A defendant must take possession of the motor vehicle without the owner's permission in order to be guilty of UDAA. *Landon*, 251 Mich App at 644. However, [t]he unlawful taking and driving away of a motor vehicle does not require an intent to steal, that is, to permanently deprive the owner of his [or her] property." *Id.* at 640.

²⁴"Lawful possession' is not an element of the offense of unlawful use of an automobile." *People v Crosby*,
82 Mich App 1, 3 (1978). "It is just that 'unlawful possession' does not have to be shown to support a conviction for unlawful use of an automobile." *Id*.

"While the offense requires the specific intent to take possession of the vehicle unlawfully, it does not require an intent to steal the vehicle; the offense punishes conduct that does not rise to the level of larceny because an intent to permanently deprive the owner of the property is lacking." *Id.* A completed larceny is not an element of UDAA. *People v Sherman*, 497 Mich 1025, 1025 (2015). See also *People v Cain*, 495 Mich 874, 874 (2013) (holding that "[a] completed larceny is not an element of UDAA because the offense does not require felonious intent, only movement of the vehicle without the owner's consent[]") (internal quotations omitted). "UDAA merely requires driving or taking away a motor vehicle without the owner's consent." *Id.*

Chapter 9: Section 625 Offenses

9.1	Scope Note	. 9-2
9.2	Special Procedures for Section 625 Offenses	. 9-2
9.3	DWI/Sobriety Court and the Specialty Court Interlock Program	9-13
9.4	Operating While Intoxicated (OWI)—Section 625(1)	9-15
9.5	Permitting or Authorizing Certain Individuals to Drive—Section 625((2)
		9-23
9.6	Operating While Visibly Impaired (OWVI)—Section 625(3)	9-24
9.7	OWI or OWVI Causing Death—Section 625(4)	9-28
9.8	OWI or OWVI Causing Serious Impairment of a Body Function—Sect	ion
	625(5)	9-33
9.9	Zero Tolerance Violations—Section 625(6)	9-35
9.10	Child Endangerment—Section 625(7)	9-38
9.11	Operating with the Presence of Certain Controlled Substances in Bo	dy—
	Section 625(8)	9-41
9.12	Ignition Interlock Device	9-46
9.13	Preliminary Chemical Breath Analysis (PBT) & Chemical Tests of Blog	od,
	Urine, or Breath	9-48
9.14	Roadside Drug Testing	9-61
9 .15	Testimony Regarding Field Sobriety Tests	9-63
9.16	Drug Recognition Evaluator or Expert Testimony	9-63
9.17	Obtaining or Transferring a Vehicle to Circumvent Vehicle or License	Ģ
	Sanctions	9-64

9.1 Scope Note

This chapter outlines the various criminal offenses listed in MCL 257.625 of the Michigan Vehicle Code (MVC), as well as other related offenses. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions.

9.2 Special Procedures for Section 625 Offenses

A. Warrantless Arrest

"A peace officer may arrest a person without a warrant under either of the following circumstances:

(a) The peace officer has reasonable cause to believe the person was, at the time of an accident in [Michigan], the operator of a vehicle involved in the accident and was operating the vehicle in violation of [MCL 257.625] or a local ordinance substantially corresponding to [MCL 257.625].

(b) The person is found in the driver's seat of a vehicle parked or stopped on a highway or street within [Michigan] if any part of the vehicle intrudes into the roadway and the peace officer has reasonable cause to believe the person was operating the vehicle in violation of [MCL 257.625] or a local ordinance substantially corresponding to [MCL 257.625]." MCL 257.625a(1).

In lieu of filing a complaint as required by MCL 764.13, a police officer may issue an appearance ticket to a person who is arrested without a warrant "for a misdemeanor or ordinance violation[.]" MCL 764.9c(1). "The appearance ticket . . . , or other documentation as requested, must be forwarded to the court, appropriate prosecuting authority, or both, for review without delay." *Id*.

"Except as provided in [MCL 764.9c(5)], a police officer shall issue to and serve upon a person an appearance ticket . . . and release the person from custody if the person has been arrested for a misdemeanor or ordinance violation that has a maximum permissible penalty that does not exceed 1 year in jail or a fine, or both, and is not . . . an operating while intoxicated offense." MCL 764.9c(4). MCL 764.9c "does not create a right to the issuance of an appearance ticket in lieu of an arrest. An arrested person may appeal the legality of his or her arrest as provided by law. However, an arrested person does not have a claim for damages against a police officer or law enforcement agency because he or she was arrested rather than issued an appearance ticket." MCL 764.9c(8).

B. Arraignment

"A person arrested for a misdemeanor violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)] or [MCL 257.625m] must be arraigned on the citation, complaint, or warrant not more than 14 days after the arrest for the violation or, if an arrest warrant is issued or reissued, not more than 14 days after the issued or reissued arrest warrant is served, whichever is later." MCL 257.625b(1). "The court shall not dismiss a case or impose any other sanction for a failure to comply with this time limit." *Id.* "The time limit does not apply to a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] punishable as a felony or a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] joined with a felony charge." MCL 257.625b(1).

C. Pretrial Conference

"The court shall schedule a pretrial conference between the prosecuting attorney, the defendant, and the defendant's attorney in each case in which the defendant is charged with a misdemeanor violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)] or [MCL 257.625(8)]

"The pretrial conference must be held not more than 35 days after the person's arrest for the violation or, if an arrest warrant is issued or reissued, not more than 35 days after the issued or reissued arrest warrant is served, whichever is later." MCL 257.625b(2).

"If the court has only 1 judge who sits in more than 1 location in that district, the pretrial conference must be held not more than 42 days after the person's arrest for the violation or, if an arrest warrant is issued or reissued, not more than 42 days after the date the issued or reissued arrest warrant is served, whichever is later." MCL 257.625b(2).

"The court shall not dismiss a case or impose any other sanction for a failure to comply with the applicable time limit." MCL 257.625b(2).

"The 35- and 42-day limits do not apply to a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] punishable as a felony or a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), MCL 257.625(8)] or [MCL 257.625m] joined with a felony charge."¹ MCL 257.625b(2).

"The court shall order the defendant to attend the pretrial conference and may accept a plea by the defendant at the conclusion of the pretrial conference." MCL 257.625b(2).

"The court may adjourn the pretrial conference upon the motion of a party for good cause shown." MCL 257.625b(2). However, "[n]ot more than 1 adjournment shall be granted to a party, and the length of an adjournment must not exceed 14 days." MCL 257.625b(2).

D. Plea Provision in Section 625(16)

"If a person is charged with a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(4), MCL 257.625(5), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m], the court shall not permit the defendant to enter a plea of guilty or nolo contendere to a charge of violating [MCL 257.625(6)²] in exchange for dismissal of the original charge. This subsection does not prohibit the court from dismissing the charge on the prosecuting attorney's motion." MCL 257.625(16). See also *People v Bulger*, 291 Mich App 1, 6 (2010), which states that "the Legislature differentiated the zero-tolerance provision from the crimes of operating while intoxicated or operating while visibly impaired by prohibiting a plea of guilty or nolo contendere to a zero-tolerance charge in exchange for dropping either of the more serious charges."

E. Advice to the Accused Before Accepting Plea

"Before accepting a plea of guilty or nolo contendere under [MCL 257.625] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(2), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)], the court shall advise the accused of the maximum possible term of imprisonment and the maximum possible fine that may be imposed for the violation and shall advise the defendant that the maximum possible license sanctions that may be imposed

¹ See the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapters 7 and 9, for information on felony pretrial conferences.

² MCL 257.625(6) is a "zero-tolerance provision," which prohibits a person who is under 21 years of age from operating a motor vehicle with any bodily alcohol content. See Section 9.9 for more information on this offense.

will be based upon the master driving record maintained by the secretary of state under [MCL 257.204a]." MCL 257.625b(4).

F. Final Adjudication

"Except for delay attributable to the unavailability of the defendant, a witness, or material evidence or due to an interlocutory appeal or exceptional circumstances, but not a delay caused by docket congestion, the court shall finally adjudicate, by a plea of guilty or nolo contendere, entry of a verdict, or other final disposition, a case in which the defendant is charged with a misdemeanor violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625(1), MCL 257.625(3), MCL 257.625(1), MCL 257.625(8)] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)] or [MCL 257.625(1), within 77 days after the person is arrested for the violation or, if an arrest warrant is issued or reissued, not more than 77 days after the date the issued or reissued arrest warrant is served, whichever is later." MCL 257.625b(3). "The court shall not dismiss a case or impose any other sanction for a failure to comply with this time limit." *Id*.

"The 77-day time limit does not apply to a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] punishable as a felony or a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or [MCL 257.625m] joined with a felony charge." MCL 257.625b(3).

G. Order Screening and Assessment for Rehabilitative Services Before Imposing Sentence

"Before imposing sentence for a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(4), MCL 257.625(5), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)], the court shall order the person to undergo screening and assessment by a person or agency designated by the office of substance abuse services to determine whether the person is likely to benefit from rehabilitative services, including alcohol or drug education and alcohol or drug treatment programs." MCL 257.625b(5).

"Except as otherwise provided in [MCL 257.625b(5)], the court may order the person to participate in and successfully complete 1 or more appropriate rehabilitative programs as part of the sentence." MCL 257.625b(5).

"If the person was convicted under [MCL 257.625(1)(c)] or has 1 or more prior convictions, the court shall order the person to participate in and successfully complete 1 or more appropriate rehabilitative programs as part of the sentence, including, but not limited to, an alcohol treatment program or a self-help program for a period of not less than 1 year." MCL 257.625b(5).

"The treatment plan must be devised from an assessment performed by an appropriately licensed alcohol assessor and approved by the court." MCL 257.625b(5).

"If the person has 2 or more prior convictions, the court shall order the person to undergo an assessment that uses a standardized evidence-based instrument performed by a provider or other licensed or certified substance use disorder professional to determine whether he or she has a diagnosis for alcohol dependence and would likely benefit from a United States Food and Drug Administration approved medication-assisted treatment that is indicated for the treatment of alcohol dependence, as specified in the most recent Diagnostic and Statistical Manual of Mental Disorders published by the American Psychiatric Association." MCL 257.625b(5).

"A person may request an independent assessment[.]" MCL 257.625b(5). The independent assessment must satisfy the assessment requirements set forth in MCL 257.625b(5). *Id.* "A court shall grant a request for an independent assessment and shall consider the results of the independent assessment along with the assessment required under [MCL 257.625b(5)] when determining if the court will refer the person to a rehabilitative program that offers 1 or more forms of United States Food and Drug Administration-approved medications for the treatment of alcohol dependence." *Id.*

"Only a provider may recommend that a person take medicationassisted treatment. A person always maintains the right to refuse ingestion or injection of medication. Only a provider may determine the type, dosage, and duration of the medication-assisted treatment. If the person refuses to take the medication-assisted treatment, the court shall not hold that person in contempt." *Id*.

"If no other identified funding source is available, the person shall pay for the costs of the screening, assessment, or assessments, as applicable, and rehabilitative services ordered under [MCL 257.625b(5)]." MCL 257.625b(5).

"[MCL 257.625b(5)] does not require the person to successfully complete an ordered rehabilitative program before driving a vehicle with an ignition interlock device on a restricted license." MCL 257.625b(5).

H. Issuance of a Restricted License

Unless a person is ineligible under MCL 257.304(3), the secretary of state must issue a person a restricted license if the person's license "was suspended or restricted under [MCL 257.319] or revoked or denied under [MCL 257.303] based on either of the following:

(a) Two or more convictions for violating [MCL 257.625(1) or MCL 257.625(3)] or a local ordinance of this state that substantially corresponds to [MCL 257.625(1) or MCL 257.625(3)].

(b) One conviction for violating [MCL 257.625(1) or MCL 257.625(3)] or a local ordinance of this state that substantially corresponds to [MCL 257.625(1) or MCL 257.625(3)], preceded by 1 or more convictions for violating a local ordinance or law of another state that substantially corresponds to [MCL 257.625(1) or MCL 257.625(3), or MCL 257.625(6)], or a law of the United States that substantially corresponds to [MCL 257.625(1), MCL 257.625(3), or MCL 257.625(3), or MCL 257.625(3), or MCL 257.625(6)]." MCL 257.625(1), MCL 257.625(3), or MCL 257.625(6)]."

1. Requirements for Issuance

A restricted license under MCL 257.304(1) must not be issued until:

- the individual's license has been suspended or revoked for 45 days, MCL 257.303(2), and
- the judge assigned to a specialty court certifies to the secretary of state both of the following:
 - the individual has been admitted into a specialty court interlock program,³ and
 - an approved and certified ignition interlock device has been properly installed on each motor vehicle owned, operated, or both, by the individual. MCL 257.304(2)(a)-(b).

Under MCL 257.304(3), a restricted license must not be issued under MCL 257.304(1) if the individual is ineligible for a license under the MVC, unless the individual's ineligibility is based on 1 or more of the following:

³See Section 9.3 for discussion of specialty courts.

- License denied under MCL 257.303(1)(h) or MCL 257.303(1)(j);
- License revoked and denied under MCL 257.303(2)(c)(*i*) or MCL 257.303(2)(c)(*iii*);
- License revoked and denied under MCL 257.303(2)(g)(*i*) or MCL 257.303(2)(g)(*iii*);
- License suspended under MCL 257.319(4)-(7), MCL 257.319(8)(a)-(e), or MCL 257.319(9);
- Under investigation or reexamination by the secretary of state under MCL 257.320(1)(d);
- License suspended for violating MCL 257.321a(1) or MCL 257.321a(2);
- Issued a court-ordered restricted license under MCL 257.323c;
- License suspended or denied under MCL 257.625f;
- An additional like period of suspension or revocation was issued under MCL 257.904(10);
- License suspended under MCL 324.82105a(2);
- License suspended or revoked under MCL 500.3177;
- License suspended under MCL 257.1110.

2. Privileges of a Restricted License

An individual issued a restricted license under MCL 257.304 may operate only the vehicle equipped with a proper ignition interlock device, take any driving test required by the secretary of state, and drive to and from any combination of the following places:

- in the course of the individual's employment/ occupation as long as no commercial driver license is required;
- the individual's residence;
- the individual's workplace;
- an alcohol, drug, or mental health education and treatment program as ordered by the court;

- Alcoholics Anonymous, Narcotics Anonymous, or other court-ordered self-help programs;
- court hearings and probation appointments;
- court-ordered community service;
- an educational institution at which the person is enrolled as a student;
- a place of regularly occurring medical treatment for a serious condition or medical emergency for the person or a member of the person's household or immediate family;
- alcohol or drug testing as ordered by the court;
- an ignition interlock service provider as required;
- the judge has discretion to permit a minor's custodian to drive to a day care facility or educational institution (if the child is going to class or is a participant in certain school-sanctioned activities) at which the custodian's minor child is enrolled. MCL 257.304(4).

"While driving with a restricted license, the individual shall carry proof of the individual's destination and the hours of any employment, class, or other reason for traveling and shall display that proof on a peace officer's request." MCL 257.304(5).

3. Removing Restriction From License

"Except as otherwise provided in [MCL 257.304], a restricted license under [MCL 257.304(1)] is effective until a hearing officer orders an unrestricted license under [MCL 257.322]. Subject to [MCL 257.304(7)], the hearing officer shall not order an unrestricted license until the later of the following events occur:

(a) The court notifies the secretary of state that the individual has successfully completed the specialty court program.

(b) The minimum period of license sanction that would have been imposed under [MCL 257.303 or MCL 257.319] but for this section has been completed.

(c) The individual demonstrates that the individual has operated with an ignition interlock device for not less than 1 year.

(d) The individual satisfies the requirements of [MCL 257.303] and R 257.313 of the Michigan Administrative Code." MCL 257.304(6).

"A hearing officer shall not issue an unrestricted license for at least 1 year if either of the following applies:

(a) The hearing officer determines that the individual consumed any alcohol during the period that the individual's license was restricted under [MCL 257.304], as determined by breath, blood, urine, or transdermal testing unless a second test, administered within 5 minutes after administering the first test, showed an absence of alcohol.

(b) The hearing officer determines that the individual consumed or otherwise used any controlled substance during the period that the individual's license was restricted under [MCL 257.304], except as lawfully prescribed." MCL 257.304(7).

"In determining whether to order an unrestricted license under [MCL 257.304(6)], the successful completion of the specialty court program and a certificate from the specialty court judge must be considered positive evidence of the petitioner's abstinence while the petitioner participated in the specialty court program. This subsection does not apply to a determination made under [MCL 257.304(7)]." MCL 257.304(8). *Certificate* "includes, but is not limited to, a statement that the participant has maintained a period of abstinence from alcohol for not less than 6 months at the time the participant completed the specialty court program." *Id.*

4. Violating the Terms of the Restricted License Before Completion of Specialty Court Program

The secretary of state must summarily impose the following sanctions if it receives notice from the specialty court that the person (1) has been removed from the specialty court program; (2) operated a motor vehicle that was not equipped with an ignition interlock device; (3) tampered with, circumvented, or removed a court-ordered ignition interlock device without

court approval; or (4) has been charged with a new violation of MCL 257.625:

- License suspension for the full length of time provided in MCL 257.319(8), except that a restricted license cannot be issued under MCL 257.319(8). This sanction only applies if the underlying conviction(s) would have been punishable under MCL 257.319(8) if MCL 257.304 did not apply.
- License revocation and denial for the full length of time provided in MCL 257.303. The minimum length of license revocation and denial must be imposed as if MCL 257.304 did not apply. This sanction only applies if the underlying conviction(s) would have been punishable under MCL 257.303 if MCL 257.304 did not apply. MCL 257.304(9). See also MCL 600.1084(7) (setting forth circumstances under which a specialty court judge must notify the secretary of state).⁴

5. Violating the Terms of the Restricted License After Completion of the Specialty Court Program

"After the individual completes the specialty court program, the following apply:

(a) The secretary of state shall postpone considering the issuance of an unrestricted license under [MCL 257.322] for a period of 3 months for each act that would be a minor violation if the individual's license had been issued under [MCL 257.322(6)]. As used in this subdivision, 'minor violation' means that term as defined in R 257.301a of the Michigan Administrative Code.

(b) The restricted license issued under this section must be suspended or revoked or denied as provided in [MCL 257.304(9)], unless set aside under [MCL 257.322(5)], if any of the following events occur:

(*i*) The individual operates a motor vehicle without an ignition interlock device that meets the criteria under [MCL 257.304(2)(b)].

(*ii*) The individual removes, or causes to be removed, an ignition interlock device from a

⁴See Section 9.3 for discussion of specialty courts.

vehicle the individual owns or operates unless the secretary of state has authorized its removal under [MCL 257.322a].

(*iii*) The individual commits any other act that would be a major violation if the individual's license had been issued under [MCL 257.322(6)]. As used in this subparagraph, 'major violation' means that term as defined in R 257.301a of the Michigan Administrative Code.

(*iv*) The individual is arrested for a violation of any of the following:

(A) [MCL 257.625].

(B) A local ordinance of this state or another state that substantially corresponds to [MCL 257.625].

(C) A law of the United States that substantially corresponds to [MCL 257.625.]

(c) If the individual is convicted of or found responsible for any offense that requires the suspension, revocation, denial, or cancellation of the individual's operator's or chauffeur's license, the restricted license issued under this section must be suspended until the requisite period of license suspension, revocation, denial, or cancellation, as appropriate, has elapsed.

(d) If the individual has failed to pay any courtordered fines or costs that resulted from the operation of a vehicle, the restricted license issued under this section must be suspended pending payment of those fines and costs." MCL 257.304(10).

6. Vehicle Immobilization/Forfeiture⁵ and Restricted Licenses

"The vehicle of an individual admitted to the specialty court interlock program whose vehicle would otherwise be subject to immobilization or forfeiture under this act is exempt from both

⁵ For more general information on vehicle immobilization, see Section 1.47; for more general information on forfeiture, see Section 1.49.

immobilization and forfeiture under [MCL 257.625n] and [MCL 257.904d] if both of the following apply:

(a) The individual is a specialty court interlock program participant in good standing or the individual successfully satisfactorily completes the specialty court interlock program.

(b) The individual does not subsequently violate a law of this state for which vehicle immobilization or forfeiture is a sanction."MCL 257.304(11).

I. Appeals to Circuit Court⁶

"If the judgment and sentence are appealed to circuit court, the court may ex parte order the secretary of state to stay the suspension, revocation, or restricted license issued by the secretary of state pending the outcome of the appeal." MCL 257.625b(6).

9.3 DWI/Sobriety Court and the Specialty Court Interlock Program

MCL 600.1084 governs the DWI/sobriety court and the specialty court interlock program. See MCL 600.1084(1). DWI/sobriety courts must be certified by the State Court Administrative Office (SCAO) before performing any of the functions of a DWI/sobriety court. MCL 600.1084(3). See the Michigan Judicial Institute's *Controlled Substances Benchbook*, Chapter 10, for more information about certification. See also SCAO's problem-solving courts website, which contains detailed information about the certification process.

In cooperation with SCAO, all DWI/sobriety courts that participate in the program must provide the legislature, the secretary or state, and the Michigan Supreme Court with documentation regarding the participants' compliance with court-ordered conditions. MCL 600.1084(5).

A. Guiding Principles

"All DWI/sobriety courts shall comply with the 10 guiding principles of DWI courts as promulgated by the National Center for DWI Courts." MCL 600.1084(2).

The 10 guiding principles are:

⁶ For more information on circuit court appeals, see the Michigan Judicial Institute's *Appeals & Opinions Benchbook*, Chapter 2.

- determine the population;
- perform a clinical assessment;
- develop the treatment plan;
- supervise the offender;
- forge agency, organization, and community partnerships;
- take a judicial leadership role;
- develop case management strategies;
- address transportation issues;
- evaluate the program; and
- ensure a sustainable program.⁷

B. Requirements for Placement

"In order to be considered for placement in the program, an individual must have been convicted of either of the following:

(a) Two or more convictions for violating [MCL $257.625(1)^8$] or [MCL $257.625(3)^9$], or a local ordinance of this state substantially corresponding to [MCL 257.625(1)] or [MCL 257.625(3).]

(b) One conviction for violating [MCL 257.625(1)] or [MCL 257.625(3),] . . . or a local ordinance of this state substantially corresponding to [MCL 257.625(1)], or [MCL 257.625(3),] . . . preceded by 1 or more convictions for violating a local ordinance or law of another state substantially corresponding to [MCL 257.625(1), MCL 257.625(3), or MCL 257.625(6),¹⁰] . . . or a law of the United States substantially corresponding to [MCL 257.625(1), MCL 257.625(1), MCL 257.625(1), MCL 257.625(3), or MCL 257.625(3), or MCL 257.625(6),¹⁰] . . . or a law of the United States substantially corresponding to [MCL 257.625(1), MCL 257.625(1), MCL 257.625(3), or MCL 257.625(3), or MCL 257.625(6).]" MCL 257.625(1), MCL 257.625(3), or MCL 257.625(6).]

⁷Source: the National Center for DWI Courts.

⁸ Operating motor vehicle while intoxicated.

⁹ Operating motor vehicle while visibly impaired.

¹⁰ Person under 21 years of age operating motor vehicle with any bodily alcohol content.

C. Restricted Licensing for Program Participants

"Before the secretary of state issues a restricted license to a program participant under . . . MCL 257.304, the specialty court judge shall certify to the secretary of state that the individual seeking the restricted license has been admitted into the program and that an interlock device has been installed on each motor vehicle owned or operated, or both, by the individual." MCL 600.1084(6).

"If any of the following occur, the specialty court judge shall immediately inform the secretary of state of that occurrence:

(a) The court orders that a program participant be removed from the specialty court program before he or she successfully completes it.

(b) The court becomes aware that a program participant operates a motor vehicle that is not equipped with an interlock device or that a program participant tampers with, circumvents, or removes a court-ordered interlock device without prior court approval.

(c) A program participant is charged with a new violation of . . . MCL 257.625." MCL 600.1084(7).

"The receipt of notification by the secretary of state under subsection (7) must result in summary revocation or suspension of the restricted license under . . . MCL 257.304." MCL 600.1084(8).

9.4 Operating While Intoxicated (OWI)—Section 625(1)

A. Statutory Authority

"A person, whether licensed or not, shall not operate a vehicle on a highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within [Michigan] if the person is operating while intoxicated." MCL 257.625(1).

B. Relevant Jury Instructions

- M Crim JI 15.1 addresses the different types of operating while intoxicated offenses that a jury may consider.
- M Crim JI 15.1a addresses the elements of operating with high bodily alcohol content under MCL 257.625(1)(c).

- M Crim JI 15.2 addresses the common elements of operating while intoxicated under MCL 257.625(1) and operating while visibly impaired under MCL 257.625(3), discussed in Section 9.6.
- M Crim JI 15.3 addresses the elements of operating while intoxicated under MCL 257.625(1).
- M Crim JI 15.5 addresses factors for the jury to consider when the defendant is charged with operating while intoxicated and/or operating while visibly impaired, which is discussed in Section 9.6.
- M Crim JI 15.6 addresses possible verdicts where operating with a high bodily alcohol content under MCL 257.625(1)(c) is *not* charged.
- M Crim JI 15.6a addresses possible verdicts where operating with a high bodily alcohol content under MCL 257.625(1)(c) *is* charged.
- M Crim JI 15.7 provides a verdict form for use where operating with a high bodily alcohol content under MCL 257.625(1)(c) is *not* charged.
- M Crim JI 15.7a provides a verdict form for use where operating with a hight bodily alcohol content under MCL 257.625(1)(c) *is* charged.

C. Penalties

"If a person is convicted of violating [MCL 257.625(1)], all of the following apply:

(a) Except as otherwise provided in [MCL 257.625(9)(b) and MCL $257.625(9)(c)^{11}$], the person is guilty of a misdemeanor punishable by 1 or more of the following:

(*i*) Community service for not more than 360 hours.

(*ii*) Imprisonment for not more than 93 days, or, if the person is convicted of violating [MCL 257.625(1)(c)], imprisonment for not more than 180 days.

(*iii*) A fine of not less than \$100.00 or more than \$500.00, or, if the person is guilty of violating [MCL

¹¹Stating that violations of MCL 257.625(1) constitute a felony when certain circumstances involving prior convictions exist.

257.625(1)(c)], a fine of not less than \$200.00 or more than \$700.00.

(b) If the violation occurs within 7 years of 1 prior conviction, the person must be sentenced to pay a fine of not less than \$200.00 or more than \$1,000.00 and 1 or more of the following:

(*i*) Imprisonment for not less than 5 days or more than 1 year.

(*ii*) Community service for not less than 30 days or more than 90 days.

(c) If the violation occurs after 2 or more prior convictions, regardless of the number of years that have elapsed since any prior conviction, the person is guilty of a felony and must be sentenced to pay a fine of not less than \$500.00 or more than \$5,000.00 and to either of the following:

(*i*) Imprisonment under the jurisdiction of the department of corrections for not less than 1 year or more than 5 years.

(*ii*) Probation with imprisonment in the county jail for not less than 30 days or more than 1 year and community service for not less than 60 days or more than 180 days. Not less than 48 hours of the imprisonment imposed under this subparagraph must be served consecutively.

(d) A term of imprisonment imposed under [MCL 257.625(9)(b) or MCL 257.625(9)(c)] must not be suspended unless the defendant agrees to participate in a specialty court program and successfully completes the program." MCL 257.625(9).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Six points. See MCL 257.320a(1)(c). See Section 1.42 for more information on points.

- License suspension/revocation (length dependent on specific conviction and criminal history). See MCL 257.319(8)(a); MCL 257.319(14). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
 - Violation of MCL 257.625(1)(a) or MCL 257.625(1)(b) and no prior convictions within seven years: mandatory 180-day suspension; possible restricted license after 30 days under certain circumstances. See MCL 257.319(8)(a); MCL 257.319(14).
 - Violation of MCL 257.625(1)(c) and no prior convictions within seven years or no more than two convictions within 10 years: mandatory one year suspension; possible restricted license after 45 days under certain circumstances. See MCL 257.319(8)(g); MCL 257.319(14).
 - Any combination of two convictions within seven years for offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(c); MCL 257.303(4)(a).
 - Any combination of three convictions within 10 years for offenses listed in MCL 257.303(2)(g)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(g); MCL 257.303(4)(a).
- Vehicle immobilization (length dependent on specific conviction and criminal history). See MCL 257.625(9)(e); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: immobilization may be ordered for not more than 180 days. See MCL 257.625(9)(e); MCL 257.904d(1)(a).
 - Conviction under MCL 257.625(1) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(9)(e); MCL 257.904d(1)(c).
 - Conviction under MCL 257.625(1) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(9)(e); MCL 257.904d(1)(d).

- Vehicle forfeiture may be imposed for a violation of MCL 257.625(9)(b) or MCL 257.625(9)(c). See MCL 257.625(9)(f); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Ignition interlock device may be ordered as a condition of probation. See MCL 257.625(24); MCL 257.625k; MCL 257.625*l*. Ignition interlock device must be ordered if offender is issued a restricted license under MCL 257.319(8)(g). See MCL 257.319(8)(h)-(i). See Section 9.12 for more information on ignition interlock devices.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues¹²

1. Definition of Place Generally Accessible to Motor Vehicles

The trial court erred by dismissing the defendant's OWI charge where the defendant backed his vehicle out of his garage into his private driveway to a point in line with his house and then pulled his car back into the garage. People v Rea, 500 Mich 422, 425-426 (2017). Consulting the dictionary definitions of terms not defined by the MVC, the Court concluded that the phrase generally accessible in MCL 257.625(1) "means 'usually capable of being reached," and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles." Rea, 500 Mich at 429-432 (noting that it was error to conclude "that whether a place is 'generally accessible to motor vehicles' depends on whether the general public 'widely' or 'popularly' has permission to enter the location"). In this case, the defendant's private driveway was designed for vehicular travel, and "[a]reas designed for vehicular travel, are by their nature, areas a vehicle is usually capable of accessing." Id. at "Additionally, there [was] nothing on [the] 434-435. defendant's driveway that would *prevent* motor vehicles on the public street from turning into it." Id. at 435. "Given these facts, defendant's driveway is a place motor vehicles are usually capable of entering." Id. Therefore, the "defendant's driveway

¹² Some of the cases discussed in this subsection predate several statutory amendments to MCL 257.625, and references to the statute's provisions may be outdated. However, the amendments to the statute do not appear to affect the holdings discussed.

was generally accessible to motor vehicles under MCL 257.625(1)." *Rea*, 500 Mich at 435.

The district court "did not err by concluding that the area in which defendant's vehicle was found was generally accessible to motor vehicles" where the case involved "'a road that falls off into a ditch area a little bit, then kind of levels out in, essentially, a field." *People v Parrott*, 335 Mich App 648, 676-677 (2021) (the ditch and field were also in close proximity to an intersection). "Although . . . there were some obstacles, such as bushes lining the uncultivated field, it is clear that defendant was nonetheless able to access the field without extraordinary effort," and "there was nothing in the area that prevented defendant's vehicle from entering the field." *Id.* at 676-677.

2. Definition of Open to the General Public

The phrase "other place open to the general public" in MCL 257.625(1) refers to public accessibility and includes areas that invite and do not have any barriers to public access. *People v Hawkins*, 181 Mich App 393, 398-399 (1989). "The language of [MCL 257.625(1)] focuses upon the accessibility of the area to the public." *Hawkins*, 181 Mich App at 396. "For an area to be "open to public use" it does not have to be open to "everybody all the time." The essential feature of a public use is that it is not confined to privileged individuals or groups whose fitness or eligibility is gauged by some predetermined criteria, but is open to the indefinite public. It is the indefiniteness or unrestricted quality of potential users that gives a use its public character." *Hawkins*, 191 Mich App at 398-399, quoting *State v Boucher*, 207 Conn 612, 615 (1988) (citations omitted).

A shopping center parking lot was open to the general public under MCL 257.625(1) where there were no signs posted indicating that the parking lot was available for parking by the general public and no businesses were open in the shopping center at the time the defendant was driving there but there were also no signs posted restricting the use of the parking lot and testimony indicated that the lot was used after the stores were closed for various purposes. *Hawkins*, 181 Mich App at 395, 399.

A trailer park road was open to the general public under MCL 257.625(1) where the road was built and maintained privately but open 24 hours a day, and persons using the roadway as a shortcut were not prosecuted for trespassing. *City of Holland v Dreyer*, 184 Mich App 237, 239 (1990).

3. Definition of Vehicle

A personal electric scooter qualifies as a vehicle under MCL 257.625(1) because the scooter is "a device upon which a person [is] transported upon a highway." *People v Lyon*, 310 Mich App 515, 516-517 (2015) (the defendant was driving the scooter "along the paved portion of the 'curb lane" and "weaving into the traffic lane" on a public highway; when stopped by police he failed field sobriety tests and admitted he was intoxicated). The Court noted that "the definition of 'vehicle,' the term actually used in . . . MCL 257.625, is much more inclusive than the definition of 'motor vehicle[.]" *Lyon*, 310 Mich App at 519. "[T]he MVC govern[s a] defendant's conduct when he [or she] use[s a] scooter as a vehicle upon a highway[]" and thereby undertakes "the duties of a vehicle driver, which include refraining from driving while intoxicated[.]" *Id.* at 520-521.

4. Double Jeopardy Considerations¹³

A "defendant's convictions of both OWI[, MCL 257.625(1),] and OWI-injury[, MCL 257.625(5),] for the same intoxicated driving incident violates the multiple punishments prong of the [federal and state constitutions'] double jeopardy clauses." People v Miller, 498 Mich 13, 26 (2015). The "Court of Appeals erred by concluding that the Legislature did not 'evince a clear expression of any intent to allow . . . multiple punishments for the same offense[;]" rather, "in light of MCL 257.625(7)(d), [which specifically authorizes multiple convictions of and punishments for OWI with a minor in the car under MCL 257.625(7) and OWI-injury or OWI causing death,] the omission of a similar clause providing explicit authority to convict a defendant of multiple operating while intoxicated offenses arising out of the same incident in either MCL 257.625(1) or [MCL 257.625](5) is a clear indication that the Legislature did not intend for defendants to be convicted of and punished for OWI and OWI-injury arising out of the same incident." Miller, 498 Mich at 25-26.

5. Elements of OWI Generally

"OWI is a hybrid version of two offenses: MCL 257.625(1)(a) prohibits operating a motor vehicle under the influence of intoxicating liquor (OUIL) and MCL 257.625(1)(b) prohibits

¹³For a detailed discussion of double jeopardy, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 9.

operating with an unlawful bodily alcohol content (UBAL). Therefore, under MCL 257.625(1), OWI requires proof of three elements: (1) the defendant operated a motor vehicle (2) on a highway or other place open to the general public or generally accessible to motor vehicles (3) while under the influence of liquor or a controlled substance, or a combination of the two, *or* with [an unlawful bodily alcohol content]. Notably, the third element is disjunctive; that is, it can be satisfied in *either* of the two ways." *People v Hyde*, 285 Mich App 428, 447-448 (2009).

Note that element two, which addresses where a person is operating under the influence, "prohibits operating a vehicle while intoxicated in three types of locations: (1) upon a highway, (2) in a place open to the general public, or (3) in a place generally accessible^[14] to motor vehicles." *People v Rea*, 500 Mich 422, 428 (2017).

6. OWI is a Lesser Included Offense of OUIL

"With regard to the . . . 'under the influence' part of the Michigan OUIL statutory provision, the key instructive description of that offense comes from the decision in *People v* Lambert, 395 Mich 296 (1975)." Oxendine v Secretary of State, 237 Mich App 346, 353 (1999). In Lambert, 395 Mich at 305, our Supreme Court "indicated that OUIL would be committed if a defendant drove when the 'defendant's ability to drive was substantially and materially affected by consumption of intoxicating liquor." Oxendine, 237 Mich App at 354. "In contrast, the threshold for OWI is much lower." Id. "A defendant commits OWI by driving when the 'defendant's ability to drive was so weakened or reduced by consumption of intoxicating liquor that [the] defendant drove with less ability than would an ordinary, careful and prudent driver. Such weakening or reduction of ability to drive must be visible to an ordinary, observant person."" Id., quoting Lambert, 395 Mich at 305. "Accordingly, OWI is a lesser included offense of OUIL." Oxendine, 237 Mich App at 354.

¹⁴The phrase *generally accessible* "means 'usually capable of being reached[,]'" and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles." See Section 9.4(E)(1) for further discussion of this issue.

9.5 Permitting or Authorizing Certain Individuals to Drive—Section 625(2)

A. Statutory Authority

"The owner of a vehicle or a person in charge or in control of a vehicle shall not authorize or knowingly permit the vehicle to be operated on a highway or other place open to the general public or generally accessible to motor vehicles,¹⁵ including an area designated for the parking of motor vehicles, within [Michigan] by a person if any of the following apply:

(a) The person is under the influence of alcoholic liquor, a controlled substance, other intoxicating substance, or a combination of alcoholic liquor, a controlled substance, or other intoxicating substance.

(b) The person has an alcohol content of 0.08 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine or, beginning 5 years after the state treasurer publishes a certification under [MCL 257.625(28)], the person has an alcohol content of 0.10 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine.

(c) The person's ability to operate the motor vehicle is visibly impaired due to the consumption of alcoholic liquor, a controlled substance, or other intoxicating substance, or a combination of alcoholic liquor, a controlled substance, or other intoxicating substance." MCL 257.625(2).

B. Penalties

"A person who is convicted of violating [MCL 257.625(2)] is guilty of a crime as follows:

(a) Except as provided in subdivisions (b) and (c), a misdemeanor punishable by imprisonment for not more than 93 days or a fine of not less than \$100.00 or more than \$500.00, or both.

¹⁵In the context of MCL 257.625(1), the Court held that the phrase *generally accessible* "means 'usually capable of being reached[,]'" and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles[,]" which may "encompass[] [a] defendant's private driveway." *People v Rea*, 500 Mich 422, 430-431, 436 (2017). See Section 9.4(E)(1) for further discussion of the issue.

(b) If the person operating the motor vehicle violated [MCL 257.625(4)], a felony punishable by imprisonment for not more 5 years or a fine of not less than \$1,500.00 or more than \$10,000.00, or both.

(c) If the person operating the motor vehicle violated [MCL 257.625(5)], a felony punishable by imprisonment for not more than 2 years or a fine of not less than \$1,000.00 ore more than \$5,000.00, or both." MCL 257.625(10).

9.6 Operating While Visibly Impaired (OWVI)—Section 625(3)

A. Statutory Authority

"A person, whether licensed or not, shall not operate a vehicle on a highway or other place open to the general public or generally accessible to motor vehicles, ¹⁶ including an area designated for the parking of vehicles, within [Michigan] when, due to the consumption of alcoholic liquor, a controlled substance, or other intoxicating substance, or a combination of alcoholic liquor, a controlled substance, the person's ability to operate the vehicle is visibly impaired." MCL 257.625(3).

"If a person is charged with violating [MCL 257.625(1)], a finding of guilty under [MCL 257.625(3)] may be rendered." MCL 257.625(3).

B. Relevant Jury Instructions

- M Crim JI 15.2 addresses the common elements of operating while visibly impaired under MCL 257.625(3), and operating while intoxicated under MCL 257.625(1), discussed in Section 9.4.
- M Crim JI 15.4 addresses the elements of operating while visibly impaired under MCL 257.625(3).
- M Crim JI 15.5 addresses factors for the jury to consider when the defendant is charged with operating while visibly impaired and/or operating while intoxicated, which is discussed in Section 9.4.

¹⁶In the context of MCL 257.625(1), the Court held that the phrase *generally accessible* "means 'usually capable of being reached[,]'" and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles[,]" which may "encompass[] [a] defendant's private driveway." *People v Rea*, 500 Mich 422, 430-431, 436 (2017). See Section 9.4(E)(1) for further discussion of the issue.

C. Penalties

MCL 257.625(11) provides in part¹⁷ that "[i]f a person is convicted of violating [MCL 257.625(3)], all of the following apply:

(a) Except as otherwise provided, in subdivisions (b) and (c), the person is guilty of a misdemeanor punishable by 1 or more of the following:

(*i*) Community service for not more than 360 hours.

(*ii*) Imprisonment for not more than 93 days.

(*iii*) A fine of not more than \$300.00.

(b) If the violation occurs within 7 years of 1 prior conviction, the person must be sentenced to pay a fine of not less than \$200.00 or more than \$1,000.00, and 1 or more of the following:

(*i*) Imprisonment for not less than 5 days or more than 1 year.

(*ii*) Community service for not less than 30 days or more than 90 days.

(c) If the violation occurs after 2 or more prior convictions, regardless of the number of years that have elapsed since any prior conviction, the person is guilty of a felony and must be sentenced to pay a fine of not less than \$500.00 or more than \$5,000.00 and either of the following:

(*i*) Imprisonment under the jurisdiction of the department of corrections for not less than 1 year or more than 5 years.

(*ii*) Probation with imprisonment in the county jail for not less than 30 days or more than 1 year and community service for not less than 60 days or more than 180 days. Not less than 48 hours of the imprisonment imposed under this subparagraph must be served consecutively.

(d) A term of imprisonment imposed under subdivisions (b) or (c) must not be suspended unless the

¹⁷Additional sanctions found in MCL 257.625(11)(e)-(f) are discussed in Section 9.6(D).

defendant agrees to participate in a specialty court program and successfully completes the program."

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Four points. See MCL 257.320a(1)(i). See Section 1.42 for more information on points.
- License suspension/revocation (length dependent on specific conviction and criminal history). See MCL 257.319(8)(b); MCL 257.319(13). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
 - No prior convictions within seven years: mandatory 90-day suspension; period increased to 180 days if impairment was due to controlled substance or combination of alcohol and controlled substance; restricted license discretionary during all or a specified portion of suspension. See MCL 257.319(8)(b); MCL 257.319(14).
 - Any combination of two convictions within seven years for offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(c); MCL 257.303(4)(a).
 - Any combination of three convictions within 10 years for offenses listed in MCL 257.303(2)(g)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(g); MCL 257.303(4)(a).
- Vehicle immobilization (length dependent on specific conviction and criminal history). See MCL 257.625(11)(e); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: immobilization may be ordered for not more than 180 days. See MCL 257.625(11)(e); MCL 257.904d(1)(a).

- Conviction under MCL 257.625(3) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(11)(e); MCL 257.904d(1)(c).
- Conviction under MCL 257.625(3) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(11)(e); MCL 257.904d(1)(d).
- Vehicle forfeiture may be imposed for a violation of MCL 257.625(3). See MCL 257.625(11)(f); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues

"Although proof that a vehicle was being operated in an impaired manner, e.g., weaving from side to side, would ... greatly strengthen a prosecutor's case by indicating that a defendant's ability to drive was visibly impaired, the statute does not compel such proof to convict a defendant." People v Mikulen, 324 Mich App 14, 23-24 (2018). Rather, the statute "require[s] proof that a person's *ability* to operate a motor vehicle was visibly impaired[.]" *Id.* at 17. "A defendant commits [OWVI] by driving when the 'defendant's ability to drive was so weakened or reduced by consumption of intoxicating liquor that [the] defendant drove with less ability than would an ordinary, careful and prudent driver. Such weakening or reduction of ability to drive must be visible to an ordinary, observant person." Oxendine v Secretary of State, 237 Mich App 346, 354 (1999), quoting People v Lambert, 395 Mich 296, 305 (1975). See also Mikulen, 324 Mich App at 22 n 3. "[T]his evidentiary mandate compels a prosecutor to proffer evidence of a visual or observational nature, i.e., evidence describing or depicting actions, conduct, characteristics, or movements of the person during the pertinent time period, revealing an impaired ability relevant to operating a vehicle." Id. at 17. "[T]he focus is on whether the person's capacity to drive was impaired as could be observed by another." Id. at 23 (concluding that the OWVI charge was properly submitted to the jury where "there was evidence that defendant was operating a motor vehicle, that he had consumed alcohol just prior to driving, and that, due to the consumption of alcohol, defendant had glassy, bloodshot eyes and failed sobriety tests, as was visible to an observed by the arresting officer," and "the jury was permitted to

assess whether [the] defendant's ability to operate his vehicle was visibly impaired based on its viewing of the videotape of the stop and sobriety tests").

"[T]he MMMA^[18] does not supersede the OWVI statute." *People v Dupre*, 335 Mich App 126, 139 (2020). "Under the influence as used in MCL 333.26427(b)(4) is not limited in meaning to how that phrase is understood with regard to the OWI statute, MCL 257.625(1). A person may be considered under the influence of marijuana if it can be shown that consumption of marijuana had some effect on the person, such that it weakened or reduced the person's ability to drive such that the person drove with less ability than would an ordinary, careful, and prudent driver." *Dupre*, 335 Mich App at 139-140 (cleaned up). "[A] driver operating while visibly impaired appears to do so negligently, in violation of MCL 333.26427(b)(1)." *Dupre*, 335 Mich App at 138 ("discern[ing] no intent within the MMMA to immunize the visibly impaired driver from prosecution").

9.7 OWI or OWVI Causing Death—Section 625(4)

A. Statutory Authority

"A person, whether licensed or not, who operates a motor vehicle in violation of [MCL 257.625(1), MCL 257.625(3), or MCL 257.625(8)] and by the operation of that motor vehicle causes the death of another person is guilty of a crime[.]" MCL 257.625(4).¹⁹

B. Relevant Jury Instructions

- M Crim JI 15.11 addresses operating while intoxicated and operating while visibly impaired causing death.
- M Crim JI 15.11a addresses operating with any amount of a schedule 1 or 2 controlled substance in the body causing death.

C. Penalties

The penalties for violating MCL 257.625(4) are as follows:

"(a) Except as provided in subdivisions (b) and (c), the person is guilty of a felony punishable by imprisonment

¹⁸Michigan Medical Marihuana Act, MCL 333.26421 *et seq.*

¹⁹See *People v Schaefer*, 473 Mich 418 (2005), discussed in Section 8.4(E), which contains more information on causation.

for not more than 15 years or a fine not less than \$2,500.00 or more than \$10,000.00, or both[.]

(b) If the violation occurs while the person has an alcohol content of 0.17 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine, and within 7 years of a prior conviction, the person is guilty of a felony punishable by imprisonment for not more than 20 years or a fine of not less than \$2,500.00 or more than \$10,000.00, or both . . .

(c) If, at the time of the violation, the person is operating a motor vehicle in a manner proscribed under [MCL 257.653a²⁰] and causes the death of a police officer, firefighter, or other emergency response personnel, the person is guilty of a felony punishable by imprisonment for not more than 20 years or a fine of not less than \$2,500.00 or more than \$10,000.00 or both. This subdivision applies regardless of whether the person is charged with the violation of [MCL 257.653a]." MCL 257.625(4).

"A person may be charged with and convicted of any of the [offenses listed in MCL 769.36(1), including MCL 257.625(4),] for each death arising out of the same criminal transaction, and the court may order the terms of imprisonment to be served consecutively to each other[.]" MCL 769.36(1)(a).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(c). See Section 1.42 for more information on points.
- License revocation/denial (length dependent on specific conviction and criminal history). See MCL 257.303(2)(d); MCL 257.303(4)(a). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.

²⁰ Failing to give due care and caution when approaching and passing a stationary authorized emergency vehicle giving a visual signal. MCL 257.653a(1).

- Vehicle immobilization (length dependent on specific conviction and criminal history), if forfeiture is not ordered. See MCL 257.625(4); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: mandatory immobilization for not more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(4); MCL 257.904d(1)(b).
 - Conviction under MCL 257.625(4) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(4); MCL 257.904d(1)(c).
 - Conviction under MCL 257.625(4) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(4); MCL 257.904d(1)(d).
- Vehicle forfeiture may be imposed for a violation of MCL 257.625(4). See MCL 257.625(4); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues²¹

1. Elements of Section 625(4)

"The plain text of [MCL 257.625(4)] does not require that the prosecution prove the defendant's intoxicated state affected his or her operation of the motor vehicle." *People v Schaefer*, 473 Mich 418, 422 (2005), overruling *People v Lardie*, 452 Mich 231 (1996), "to the extent that *Lardie* held that [MCL 257.625(4)] requires the defendant's intoxicated driving to be a substantial cause of the victim's death[.]" "Indeed, [MCL 257.625(4)] requires no causal link at all between the defendant's intoxication and the victim's death." *Schaefer*, 473 Mich at 422. "The statute requires that the defendant's intoxicated manner of driving, must cause the victim's death." *Id.* "The defendant's

²¹ Some of the cases discussed in this subsection predate several statutory amendments to MCL 257.625, and references to the statute's provisions may be outdated. However, the amendments to the statute do not appear to affect the holdings discussed.

status as 'intoxicated' is a separate element of the offense of OUIL causing death." *Schaefer*, 473 Mich at 422. "It specifies the class of persons subject to liability under [MCL 257.625(4)]: intoxicated drivers." *Schaefer*, 473 Mich at 422. "Quite simply, by enacting [MCL 257.625(4)], the Legislature intended to punish 'operating while intoxicated,' not 'operating in an intoxicated manner." *Schaefer*, 473 Mich at 422.

While MCL 257.625(4) does not require a defendant's intoxicated driving to be a substantial cause of the victim's death, Schaefer, 473 Mich at 422, "[t]he plain language of the statutes that prohibit OWI causing death, MCL 257.625(1) and [MCL 257.625(4)], and the statutes that prohibit operating a motor vehicle with the presence of a schedule 1 controlled substance in one's body, causing death, MCL 257.625(4) and [MCL 257.625(8)], requires that the defendant's operation of a motor vehicle have *caused* the death of another person." People v Feezel, 486 Mich 184, 193 (2010). "[W]hile the victim's intoxication is not a defense, under the facts of [Feezel] it should have been a factor for the jury to consider when determining whether the prosecution proved beyond a reasonable doubt that [the] defendant's conduct was a . . . proximate cause of the victim's death, under MCL 257.625(4) and [MCL 257.625(8)]." Feezel, 486 Mich at 201-202. But see People v Bergman, 312 Mich App 471, 486 (2015) (holding that evidence of the victim's intoxication was not probative of causation and was properly excluded from trial where "[t]here was no evidence that [the victim] was not properly driving within his marked lane, or that [the victim's] vehicle would not have safely passed [the] defendant if [the] defendant had not crossed the center line in front of [the victim], presenting a serious and unexpected hazard[;] [t]hus, there was no evidence that [the victim] did anything that contributed to the accident in a way that would establish that he was negligent or grossly negligent and by his conduct was an intervening cause of the accident[]").

"[MCL 257.625(4)] provides that a defendant may be convicted when he or she 'operates a motor vehicle' while intoxicated and 'by the operation of that motor vehicle causes the death of another person" *People v Lechleitner*, 291 Mich App 56, 60 (2010). "[MCL 257.625(4)] does not require that the defendant's vehicle be in motion at the time of the accident, but rather that the victim's death be caused by the defendant's operation of the vehicle while intoxicated." *Lechleitner*, 291 Mich App at 60. In *Lechleitner*, 291 Mich App at 60, the "defendant was intoxicated, operated his vehicle, and crashed it, with the result that it sat in the middle of the freeway at night creating a risk of injury or death to others." The Michigan Supreme Court has "stated that 'operating' must be defined 'in terms of the danger the OUIL statute [MCL 257.625] seeks to prevent: the collision of a vehicle being operated by a person under the influence of intoxicating liquor with other persons or property." *Lechleitner*, 291 Mich App at 60, quoting *People v Wood*, 450 Mich 399, 404 (1995). "Accordingly, '[o]nce a person using a motor vehicle as a motor vehicle has put the vehicle in motion, or in a position posing a significant risk of causing a collision, such a person continues to operate it until the vehicle is returned to a position posing no such risk." *Lechleitner*, 291 Mich App at 60, quoting *Wood*, 450 Mich at 404-405. In *Lechleitner*, the Court of Appeals affirmed the defendant's conviction, "conclud[ing] that *Wood* remains good law and that the trial court properly followed it[.]" *Lechleitner*, 291 Mich App at 61.

2. Double Jeopardy²²

Involuntary manslaughter, MCL 750.321. "The offenses of involuntary manslaughter and OUIL causing death protect distinct societal norms, the amount of punishment for each statute does not involve a hierarchy of offenses, and each statute requires proof of an element that the other does not. Thus, pursuant to [*People v Price*, 214 Mich App 538 (1995)], [the] defendant's convictions and punishments under both statutes [did] not violate the Double Jeopardy Clauses of the United States and Michigan Constitutions." *People v Kulpinski*, 243 Mich App 8, 23 (2000).

Second-degree murder, MCL 750.317."Th[e] reasoning [in Kulpinski applies with equal force to dual convictions of second-degree murder and OUIL causing death. If the Legislature intended for the OUIL causing death statute to enforce societal norms that are distinct from the societal norms enforced by the involuntary manslaughter statute (grossly negligent conduct), it clearly also intended the OUIL statute to enforce societal norms other than those enforced by the second-degree murder statute (proscribing wanton conduct likely to cause death or great bodily harm)." People v Werner, 254 Mich App 528, 536 (2002), citing *Kulpinski*, 243 Mich App at 22-24 and Price, 214 Mich App at 543-544. "Moreover, the OUIL causing death statute and second-degree murder statute each contain an element not found in the other." Werner, 254 Mich App at 536. "The OUIL causing death statute includes the element of operating a motor vehicle with a specified blood

²²For a detailed discussion of double jeopardy, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 9.

alcohol level, but not the element of malice; the converse is true of the second-degree murder statute." *Id.* "Accordingly, [the] defendant's convictions of both second-degree murder and OUIL causing death [did] not violate the Double Jeopardy Clauses." *Id.* See also *People v Bergman*, 312 Mich App 471, 492 (2015) (holding the defendant's multiple convictions of OUIL causing death and second-degree murder did not violate the double jeopardy clauses).

Operating with a suspended license causing death, MCL 257.904(4). A defendant's multiple convictions of OUIL causing death and driving with a suspended license causing death do not violate the double jeopardy clauses of the United States and Michigan constitutions. *People v Bergman,* 312 Mich App 471, 492 (2015). "[T]he OUIL and suspended-license statutes enforce distinct societal norms, and their respective elements of intoxication while driving and lack of a valid operator's license are distinctive to each." *Id.* at 492 (citations omitted).

9.8 OWI or OWVI Causing Serious Impairment of a Body Function—Section 625(5)

A. Statutory Authority

"A person, whether licensed or not, who operates a motor vehicle in violation of [MCL 257.625(1), MCL 257.625(3), or MCL 257.625(8)] and by the operation of that motor vehicle causes a serious impairment of a body function of another person is guilty of a crime[.]" MCL 257.625(5).

B. Relevant Jury Instructions

- M Crim JI 15.12 addresses operating while intoxicated and operating while visibly impaired causing serious impairment of a body function.
- M Crim JI 15.12a addresses operating with any amount of a schedule 1 or 2 controlled substance in the body causing serious impairment of a body function.

C. Penalties

The penalties for violating MCL 257.625(5) are as follows:

"(a) Except as provided in subdivision (b), the person is guilty of a felony punishable by imprisonment for not

more than 5 years or a fine of not less than \$1,000.00 or more than \$5,000.00, or both. . . .

(b) If the violation occurs while the person has an alcohol content of 0.17 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine, and within 7 years of a prior conviction, the person is guilty of a felony punishable by imprisonment of not more than 10 years or a fine of not less than \$1,000.00 or more than \$5,000.00, or both. . . ." MCL 257.625(5)(a)-(b).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(c). See Section 1.42 for more information on points.
- License revocation/denial (length dependent on specific conviction and criminal history). See MCL 257.303(2)(d); MCL 257.303(4)(a). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
- Vehicle immobilization (length dependent on specific conviction and criminal history), if forfeiture is not ordered. See MCL 257.625(5); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: mandatory immobilization for not more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(5); MCL 257.904d(1)(b).
 - Conviction under MCL 257.625(5) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(5); MCL 257.904d(1)(c).
 - Conviction under MCL 257.625(5) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless

forfeiture is ordered under MCL 257.625n. See MCL 257.625(5); MCL 257.904d(1)(d).

- Vehicle forfeiture may be imposed for a violation of MCL 257.625(5). See MCL 257.625(5); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues

A "defendant's convictions of both OWI[, MCL 257.625(1),] and OWI-injury[, MCL 257.625(5),] for the same intoxicated driving incident violates the multiple punishments prong of the [federal and state constitutions'] double jeopardy clauses."²³ People v Miller, 498 Mich 13, 26 (2015). The "Court of Appeals erred by concluding that the Legislature did not 'evince a clear expression of any intent to allow ... multiple punishments for the same offense[;]" rather, "in light of MCL 257.625(7)(d), [which specifically authorizes multiple convictions of and punishments for OWI with a minor in the car under MCL 257.625(7) and OWI-injury or OWI causing death,] the omission of a similar clause providing explicit authority to convict a defendant of multiple operating while intoxicated offenses arising out of the same incident in either MCL 257.625(1) or [MCL 257.625](5) is a clear indication that the Legislature did not intend for defendants to be convicted of and punished for OWI and OWI-injury arising out of the same incident." Miller, 498 Mich at 25-26.

9.9 Zero Tolerance Violations—Section 625(6)

A. Statutory Authority

"A person who is less than 21 years of age, whether licensed or not, shall not operate a vehicle on a highway or other place open to the general public or generally accessible to motor vehicles,²⁴ including an area designated for the parking of vehicles, within [Michigan] if the person has any bodily alcohol content." MCL 257.625(6).

²³For a detailed discussion of double jeopardy, see the Michigan Judicial Institute's *Criminal Proceedings Benchbook, Vol. 1*, Chapter 9.

²⁴In the context of MCL 257.625(1), the Court held that the phrase *generally accessible* "means 'usually capable of being reached[,]'" and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles[,]" which may "encompass[] [a] defendant's private driveway." *People v Rea*, 500 Mich 422, 430-431, 436 (2017). See Section 9.4(E)(1) for further discussion of the issue.

"In a prosecution for a violation of [MCL 257.625(6)], the defendant bears the burden of proving that the consumption of alcoholic liquor was a part of a generally recognized religious service or ceremony by a preponderance of the evidence." MCL 257.625(23).

"Except for purposes of the enhancement described in [MCL 257.625(12)(b)], only 1 violation or attempted violation of [MCL 257.625(6)], a local ordinance substantially corresponding to [MCL 257.625(6)], or a law of another state substantially corresponding to [MCL 257.625(6)] may be used as a prior conviction." MCL 257.625(26).

B. Penalties

"If a person is convicted of violating [MCL 257.625(6)], all of the following apply:

(a) Except as otherwise provided in subdivision (b), the person is guilty of a misdemeanor punishable by 1 or both of the following:

(*i*) Community service for not more than 360 hours.

(*ii*) A fine of not more than \$250.00.

(b) If the violation occurs within 7 years of 1 or more prior convictions, the person may be sentenced to 1 or more of the following:

(*i*) Community service for not more than 60 days.

(*ii*) A fine of not more than \$500.00.

(*iii*) Imprisonment for not more than 93 days." MCL 257.625(12).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

• Four points. See MCL 257.320a(1)(i). See Section 1.42 for more information on points.

- License suspension/revocation (length dependent on specific conviction and criminal history). See MCL 257.303(2)(c); MCL 257.303(4)(a); MCL 257.319(8)(c). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
 - Violation of MCL 257.625(6) with no prior convictions within seven years: mandatory 30-day suspension; discretionary restricted license for all or part of the suspension period if otherwise eligible for a license. See MCL 257.319(8)(c); MCL 257.319(14).
 - Violation of MCL 257.625(6) with one or more prior convictions for that offense within seven years: mandatory 90-day suspension. See MCL 257.319(8)(d).
 - Any combination of two convictions within seven years for offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*) or a combination of one conviction for a violation or attempted violation of MCL 257.625(6) and one conviction for any offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*) within seven years: mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(c); MCL 257.303(4)(a).
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

D. Issues

For purposes of scoring PRV 5, MCL 777.55(2)(b), a defendant's "conviction of being a minor who operated a vehicle while having any bodily alcohol content [MCL 257.625(6)] constitute[s] a misdemeanor for operating a vehicle while 'under the influence' of alcohol or 'impaired by' alcohol." *People v Bulger*, 291 Mich App 1, 5 (2010). While a defendant's "prior conviction under the zero-tolerance provision did not require proof that he [or she] was actually under the influence of alcohol or was impaired by alcohol[,]" "the Legislature specifically chose to count prior violations of the zero-tolerance provision as prior convictions for the purposes of MCL 257.625[, see MCL 257.625(25)(b)]. The only limitation is that multiple violations of the zero-tolerance provision count as a single prior conviction[, MCL 257.625(26)]." *Bulger*, 291 Mich App at 5-6 (trial court properly assessed two points for PRV 5 where the defendant had a prior conviction of operating a vehicle as

a minor with any bodily alcohol content under the zero-tolerance provision, MCL 257.625(6)).

9.10 Child Endangerment—Section 625(7)

A. Statutory Authority

"A person, whether licensed or not, is subject to the following requirements:

(a) He or she shall not operate a vehicle in violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(4), MCL 257.625(5), or MCL 257.625(8)] while another person who is less than 16 years of age is occupying the vehicle.

* * *

(b) He or she shall not operate a vehicle in violation of [MCL 257.625(6)] while another person who is less than 16 years of age is occupying the vehicle...." MCL 257.625(7).

"[MCL 257.625(7)] does not prohibit a person from being charged with, convicted of, or punished for a violation of [MCL 257.625(4) or MCL 257.625(5)] that is committed by the person while violating [MCL 257.625(7)]." MCL 257.625(7)(d). "However, points shall not be assessed under [MCL 257.320a] for both a violation of [MCL 257.625(4) or MCL 257.625(5)] and a violation of [MCL 257.625(7)] for conduct arising out of the same transaction." MCL 257.625(7)(d).

B. Penalties

"A person who violates [MCL 257.625(7)(a)] is guilty of a crime punishable as follows:

(*i*) Except as provided in subparagraph (*ii*), a person who violates [MCL 257.625(7)(a)] is guilty of a misdemeanor and must be sentenced to pay a fine of not less than \$200.00 or more than \$1,000.00 and to 1 or more of the following:

(A) Imprisonment for not more than 1 year.

(B) Community service for not less than 30 days or more than 90 days.

(*ii*) If the violation occurs within 7 years of a prior conviction or after 2 or more prior convictions,

regardless of the number of years that have elapsed since any prior conviction, a person who violates [MCL 257.625(7)(a)] is guilty of a felony and must be sentenced to pay a fine of not less than \$500.00 or more than \$5,000.00 and to either of the following:

(A) Imprisonment under the jurisdiction of the department of corrections for not less than 1 year or more than 5 years.

(B) Probation with imprisonment in the county jail for not less than 30 days or more than 1 year and community service for not less than 60 days or more than 180 days. Not less than 48 hours of this imprisonment must be served consecutively.

(*iii*) A term of imprisonment imposed under [MCL 257.625(7)(a)(*ii*)(A)-(B)] must not be suspended unless the defendant agrees to participate in a specialty court program and successfully completes the program." MCL 257.625(7)(a).

"A person who violates [MCL 257.625(7)(b)] is guilty of a misdemeanor punishable as follows:

(*i*) Except as provided in subparagraph (*ii*), a person who violates [MCL 257.625(7)(b)] may be sentenced to 1 or more of the following:

- (A) Community service for not more than 60 days.
- (B) A fine of not more than \$500.00.
- (C) Imprisonment for not more than 93 days.

(*ii*) If the violation occurs within 7 years of a prior conviction or after 2 or more prior convictions, regardless of the number of years that have elapsed since any prior conviction, a person who violates [MCL 257.625(7)(b)] must be sentenced to pay a fine of not less than \$200.00 or more than \$1,000.00 and to 1 or more of the following:

(A) Imprisonment for not less than 5 days or more than 1 year. This term of imprisonment must not be suspended unless the defendant agrees to participate in a specialty court program and successfully completes the program.

(B) Community service for not less than 30 days or more than 90 days." MCL 257.625(7)(b).

C. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(c). See Section 1.42 for more information on points. "However, points shall not be assessed under [MCL 257.320a] for both a violation of [MCL 257.625(4) or MCL 257.625(5)] and a violation of [MCL 257.625(7)] for conduct arising out of the same transaction." MCL 257.625(7)(d).
- License suspension/revocation (length dependent on specific conviction and criminal history). See MCL 257.319(8)(e). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
 - Violation of MCL 257.625(7) if no prior convictions within seven years: mandatory 180-day suspension; discretionary restricted license after first 90 days of suspension period. See MCL 257.319(8)(e).
 - Any combination of two convictions within seven years for offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(c).
 - Any combination of three convictions within 10 years for offenses listed in MCL 257.303(2)(g)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(g); MCL 257.303(4)(a).
- Vehicle immobilization (length dependent on specific conviction and criminal history), unless forfeiture ordered under MCL 257.625n. See MCL 257.625(7)(c); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: immobilization may be ordered for not more than 180 days. See MCL 257.625(7)(c); MCL 257.904d(1)(a).
 - Conviction under MCL 257.625(7) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days,

unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(7)(c); MCL 257.904d(1)(c).

- Conviction under MCL 257.625(7) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(7)(c); MCL 257.904d(1)(d).
- Vehicle forfeiture may be imposed for a violation of MCL 257.625(7). See MCL 257.625(7)(c); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

9.11 Operating with the Presence of Certain Controlled Substances in Body—Section 625(8)

A. Statutory Authority

"A person, whether licensed or not, shall not operate a vehicle on a highway or other place open to the general public or generally accessible to motor vehicles,²⁵ including an area designated for the parking of vehicles, within [Michigan] if the person has in his or her body any amount of a controlled substance listed in schedule 1 under [MCL 333.7212], or a rule promulgated under [MCL 333.7212], or of a controlled substance described in [MCL 333.7214(a)(*iv*)]." MCL 257.625(8).

B. Relevant Jury Instruction

• M Crim JI 15.3a addresses operating with any amount of a schedule 1 or 2 controlled substance in his or her body.

C. Penalties

"If a person is convicted of violating [MCL 257.625(8)], all of the following apply:

 $^{^{25}}$ In the context of MCL 257.625(1), the Court held that the phrase *generally accessible* "means 'usually capable of being reached[,]'" and MCL 257.625(1) therefore "prohibits an intoxicated person from operating a vehicle in a place that is usually capable of being reached by self-propelled vehicles[,]" which may "encompass[] [a] defendant's private driveway." *People v Rea*, 500 Mich 422, 430-431, 436 (2017). See Section 9.4(E)(1) for further discussion of the issue.

(a) Except as otherwise provided in subdivisions (b) and (c), the person is guilty of a misdemeanor punishable by 1 or more of the following:

(*i*) Community service for not more than 360 hours.

(*ii*) Imprisonment for not more than 93 days[.]

(*iii*) A fine of not less than \$100.00 or more than \$500.00[.]

(b) If the violation occurs within 7 years of a prior conviction, the person must be sentenced to pay a fine of not less than \$200.00 or more than \$1,000.00 and 1 or more of the following:

(*i*) Imprisonment for not less than 5 days or more than 1 year.

(*ii*) Community service for not less than 30 days or more than 90 days.

(c) If the violation occurs after 2 or more prior convictions, regardless of the number of years that have elapsed since any prior conviction, the person is guilty of a felony and must be sentenced to pay a fine of not less than \$500.00 or more than \$5,000.00 and to either of the following:

(*i*) Imprisonment under the jurisdiction of the department of corrections for not less than 1 year or more than 5 years.

(*ii*) Probation with imprisonment in the county jail for not less than 30 days or more than 1 year and community service for not less than 60 days or more than 180 days. Not less than 48 hours of the imprisonment imposed under this subparagraph must be served consecutively.

(d) A term of imprisonment imposed under subdivision (b) or (c) must not be suspended unless the defendant agrees to participate in a specialty court program and successfully completes the program." MCL 257.625(9).

D. Sanctions

Only applicable sanctions are listed below; accordingly, if a particular sanction is omitted from this section, it is not applicable to this offense. The Offense Code Index for Traffic Violations

published by the secretary of state and sourced from the Michigan Department of State Court Manual includes a table detailing traffic offenses and applicable sanctions. See Section 1.41 for more information on abstracting procedures.

- Six points. See MCL 257.320a(1)(c). See Section 1.42 for more information on points.
- License suspension/revocation (length dependent on specific conviction and criminal history). See MCL 257.303(2)(c); MCL 257.319(8)(a). See Section 1.45 for more information on license suspension and Section 1.44 for more information on license revocation.
 - Violation of MCL 257.625(8) if no prior convictions within seven years: mandatory 180-day suspension; discretionary restricted license after first 30 days of suspension period if offender is otherwise eligible. See MCL 257.319(8)(a); MCL 257.319(14).
 - Any combination of two convictions within seven years for offenses listed in MCL 257.303(2)(c)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(c); MCL 257.303(4).
 - Any combination of three convictions within 10 years for offenses listed in MCL 257.303(2)(g)(*i*)-(*iii*): mandatory revocation of at least one year (length dependent on various factors). See MCL 257.303(2)(g); MCL 257.303(4).
- Vehicle immobilization (length dependent on specific conviction and criminal history). See MCL 257.625(9); MCL 257.904d(1). See Section 1.47 for more information on vehicle immobilization.
 - No prior convictions: immobilization may be ordered for not more than 180 days. See MCL 257.625(9)(e); MCL 257.904d(1)(a).
 - Conviction under MCL 257.625(8) within seven years after a prior conviction: mandatory immobilization for not less than 90 days or more than 180 days, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(9)(e); MCL 257.904d(1)(c).
 - Conviction under MCL 257.625(8) after two or more prior convictions: mandatory immobilization for not less than one year or more than three years, unless forfeiture is ordered under MCL 257.625n. See MCL 257.625(9)(e); MCL 257.904d(1)(d).

- Vehicle forfeiture may be imposed for a violation of MCL 257.625(8). See MCL 257.625(9)(e); MCL 257.625n. See Section 1.49 for more information on vehicle forfeiture.
- Registration denial is required under certain circumstances. See MCL 257.219(1)(c)-(d). See Section 1.50 for more information on registration denial.

E. Issues

"The Michigan Medical Marihuana Act (MMMA) prohibits the prosecution of registered patients who internally possess marijuana, but the act does not protect registered patients who operate a vehicle while 'under the influence' of marijuana." People v Koon, 494 Mich 1, 3 (2013). "The [MVC] prohibits a person from driving with any amount of a schedule 1 controlled substance, a list that includes marijuana, in his or her system." Id. However, "the MMMA's protection supersedes the [MVC's] prohibition and allows a registered patient to drive when he or she has indications of marijuana in his or her system but is not otherwise under the influence of marijuana." Id. Stated another way, "the [MVC's] zerotolerance provision, MCL 257.625(8), which is inconsistent with the MMMA, does not apply to the medical use of marijuana." Koon, 494 Mich at 7. "[T]he MMMA is inconsistent with, and therefore supersedes, MCL 257.625(8) unless a registered qualifying patient loses immunity because of his or her failure to act in accordance with the MMMA." Koon, 494 Mich at 8-9.

11-carboxy-THC, "a byproduct of metabolism created when the body breaks down the psychoactive ingredient of marijuana," is not a schedule 1 controlled substance because the Legislature did not intend for it to be a schedule 1 controlled substance under MCL 333.7212. People v Feezel, 486 Mich 184, 204-205, 207-212 (2010), overruling *People v Derror*, 475 Mich 316 (2006), to the extent that it conflicts with the holding in *Feezel*. See also *People v Stock*, 507 Mich 1008 (2021) (rejecting the prosecution's argument "for an interpretation of 'controlled substance' [for purposes of MCL 257.625(8)] that would include any metabolite of cocaine," and reversing the defendant's convictions for operating a motor vehicle while intoxicated causing death and causing serious impairment of a body function where the prosecution presented evidence "indicating the presence of an unidentified metabolite of cocaine in the defendant's urine," but "failed to present evidence that the presence of cocaine metabolites in the defendant's urine supports a reasonable inference that the defendant had cocaine in her body").

Effective December 6, 2018, Initiated Law 1 of 2018, MCL 333.27951 *et seq.*, created the Michigan Regulation and Taxation of Marihuana Act (MRTMA), the purpose of which "is to make marihuana legal

under state and local law for adults 21 years of age or older, to make industrial hemp legal under state and local law, and to control the commercial production and distribution of marihuana under a system that licenses, regulates, and taxes the businesses involved." MCL 333.27952.

Similar to the conduct limitations set forth in the MMMA,²⁶ the MRTMA "does not authorize . . . operating, navigating, or being in physical control of any motor vehicle, aircraft, snowmobile, off-road recreational vehicle, or motorboat while [consuming marihuana or] under the influence of marihuana[.]" MCL 333.27954(1)(a); MCL 333.27954(1)(g). "[S]moking marihuana within the passenger area of a vehicle upon a public way" is also prohibited. MCL 333.27954(1)(g). Notwithstanding, the MRTMA penalty provision does not penalize such conduct. MCL 333.27965.

As noted in *Koon*, the MVC continues to prohibit the operation of a vehicle if the operator has any amount of a schedule 1 substance, which includes marihuana, in his or her body. MCL 257.625(8); MCL 333.7212(1)(c). It is unclear if violations of the limited conduct set forth in the MRTMA will be pursued under the MVC. It is equally unclear if the *Koon* holding, which provides that the MMMA supersedes the MVC, will be extended to the MRTMA.²⁷

Except for a person engaged in certain described conduct, "a person under 21 years of age who possesses not more than 2.5 ounces of marihuana or who cultivates not more than 12 marihuana plants . . . is responsible for a civil infraction[.]" MCL 333.27965(3)(a). "MCL 257.625(8) criminalize[s] the 'use' of marijuana [when operating a motor vehicle], while [MCL 333.27965(3)] decriminalize[s] the 'possession' and 'cultivation' of marijuana for individuals under the age of 21. Michigan law recognizes a distinction between possessing marijuana, MCL 333.7403, and using marijuana, MCL 333.7404." People v Perry, ____ Mich App ___, ___ (2021). In Perry, defendant "was not charged with the possession or cultivation of marijuana. Rather, she was charged with operating a vehicle with 'any amount of a controlled substance' in her body. MCL 257.625(8). Using or consuming marijuana is a necessary step leading to the operation of a motor vehicle with marijuana in the driver's system, in violation of MCL 257.625(8); simple possession, however, is not." *Perry*, Mich App at ____. "Had the legislature intended to decriminalize the internal possession or use of marijuana for those under 21 it would

²⁶See MCL 333.26427(b)(4), which provides that the MMMA "does not authorize a person to . . . [o]perate, navigate, or be in actual physical control of any motor vehicle, aircraft, snowmobile, off-road recreational vehicle, or motorboat while under the influence of marihuana."

²⁷See SCAO Memorandum dated January 24, 2019, regarding frequently asked questions about the MRTMA.

presumably have placed a limit consistent with the amount a person could reasonably use or consume—much, much lower than the stated limit of 2.5 ounces." *Id.* at ____. "[W]hen a person is under the influence of marijuana or is consuming marijuana while operating a vehicle, the person is not afforded the same limitation on punishment as one who is under 21 and simply possesses less than 2.5 ounces or marijuana or cultivates 12 or fewer marijuana plants." *Id.* at ____. "[T]he MRTMA did not remove all criminal penalties for persons under the age of 21 who operate a motor vehicle with marijuana in their system, is under the influence of marijuana while driving, or consumes marijuana while operating a vehicle." *Perry*, ____ Mich App at ____.

For more general information on the MMMA and the MRTMA, see the Michigan Judicial Institute's *Controlled Substances Benchbook*, Chapter 8.

9.12 Ignition Interlock Device

A. Generally

"The manufacturer of an ignition interlock device shall design a warning label, and the person who has an ignition interlock device shall promptly affix that label to each ignition interlock device upon installation. The label shall contain a warning that any person tampering with, circumventing, or otherwise misusing the device is guilty of a misdemeanor punishable as provided by law." MCL 257.625*l*(1).

"A person who is only permitted to operate a motor vehicle equipped with an ignition interlock device shall not operate a motor vehicle on which an ignition interlock device is not properly installed." MCL 257.625*l*(2).

"A person who has an ignition interlock device installed and whose driving privilege is restricted shall not request, solicit, or allow any other person to blow into an ignition interlock device or to start a **vehicle** equipped with the device for the purpose of providing the person whose driving privilege is restricted with an operable vehicle." MCL 257.625*I*(3).

"A person shall not blow into an ignition interlock device or start a motor vehicle equipped with the device for the purpose of providing an operable vehicle to a person who has an interlock device installed and whose driving privilege is restricted." MCL 257.625*I*(4).

"A person shall not tamper with or circumvent the operation of an ignition interlock device." MCL 257.625/(5).

"A person who violates [MCL 257.625/(2), MCL 257.625/(3), MCL 257.625/(4), or MCL 257.625/(5)] is guilty of a misdemeanor punishable by imprisonment for not more than 6 months or a fine of not more than \$5,000.00, or both." MCL 257.625/(6).

B. Vehicle Impoundment

"If a law enforcement officer detains the operator of a motor vehicle for violating a law of [Michigan] or a local ordinance and the operator is a person required to only operate a motor vehicle with an ignition interlock device properly installed, but no ignition interlock device is properly installed on the motor vehicle, the law enforcement officer shall impound the motor vehicle." MCL 257.625/(7).

"If a motor vehicle impounded under [MCL 257.625/(7)] is individually or jointly owned by the operator, the law enforcement officer shall do all of the following:

(a) Immediately confiscate the motor vehicle registration plate and destroy it.

(b) Issue a temporary registration plate for the vehicle in the same manner prescribed by the secretary of state for temporary registration plates issued under [MCL 257.226a or MCL 257.226b].

(c) Place the temporary registration plate issued under subdivision (b) on the motor vehicle in the manner prescribed by the secretary of state.

(d) Notify the secretary of state through the law enforcement information network in a form prescribed by the secretary of state that the registration plate was destroyed and a temporary registration plate was issued to the motor vehicle." MCL 257.625*l*(7).

"A temporary registration plate issued under [MCL 257.625*I*] is valid until the charges for violating [MCL 257.625*I*(2)] are dismissed, the person pleads guilty or no contest to the charge, or the person is found guilty of or is acquitted of the charge." MCL 257.625*I*(8).

"If the motor vehicle impounded under [MCL 257.625*I*] is not owned individually or jointly by the operator, the law enforcement officer shall impound the motor vehicle by contacting a local towing agency." MCL 257.625*l*(9). "The motor vehicle shall only be returned to the registered owner." *Id*.

"The owner of a motor vehicle impounded under [MCL 257.6251] is liable for the expenses incurred in the removal and storage of the motor vehicle whether or not it is returned to him or her." MCL 257.6251(10). "The motor vehicle shall be returned to the owner only if the owner pays the expenses of removal and storage." *Id.* "If redemption is not made or the vehicle is not returned as described under [MCL 257.6251(10)], it shall be considered an abandoned vehicle and disposed of under [MCL 257.252a]." MCL 257.6251(10).

For information on vehicle impoundment generally, see Section 1.48.

9.13 Preliminary Chemical Breath Analysis (PBT) & Chemical Tests of Blood, Urine, or Breath

A. Preliminary Chemical Breath Analysis (PBT)

1. Submitting to or Refusing a Preliminary Chemical Breath Analysis Test (PBT)

"A peace officer who has reasonable cause to believe that a person was operating a vehicle on a public highway or other place open to the public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within [Michigan] and that the person by the consumption of alcoholic liquor may have affected his or her ability to operate a vehicle . . . or reasonable cause to believe that a person who is less than 21 years of age was operating a vehicle on a public highway or other place open to the public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within [Michigan] while the person had any bodily alcohol content as that term is defined in [MCL 257.625(6)], may require the person to submit to a preliminary chemical breath analysis test[(PBT)]." MCL 257.625a(2).

"A person who submits to a [PBT] remains subject to the requirements of [MCL 257.625c, MCL 257.625d, MCL 257.625e, and MCL 257.625f] for purposes of chemical tests described in those sections." MCL 257.625a(2)(c).

"Except as provided in [MCL 257.625a(5)²⁸], a person who refuses to submit to a [PBT] upon a lawful request by a peace

officer is responsible for a civil infraction." MCL 257.625a(2)(d).

2. Compliance with Administrative Rules

Mich Admin Code R 325.2655(2)²⁹ provides:

"A procedure that is used in conjunction with preliminary breath alcohol analysis must be approved by [LARA] and shall be in compliance with all of the following provisions:

(a) Except as provided otherwise in these rules, preliminary breath alcohol test instruments shall only be operated by appropriate class operators who have been certified in accordance with [Mich Admin Code R 325.2658.³⁰]

(b) A person may be administered a preliminary breath alcohol analysis on a preliminary breath alcohol test instrument only after the operator determines that the person has not smoked, regurgitated, or placed anything in his or her mouth for at least 15 minutes.

(c) Records shall be maintained at the preliminary breath alcohol test instrument location as prescribed by the department and copies shall be forwarded to the department as required."

Whether a "defendant's PBT results can be considered when determining whether there was probable cause to arrest" when it was not administered in compliance with Mich Admin Code R 325.2655^{31} depends on whether "the violation of the administrative rule was significant and calls into question the accuracy of the PBT." *People v Robe*, 337 Mich App 142, 145, 148 (2021).

²⁸"MCL 257.625a(5) concerns refusing to submit to a PBT while driving a commercial motor vehicle, which is beyond the scope of this benchbook.

²⁹The link to this resource was created using Perma.cc and directs the reader to an archived record of the page.

³⁰The link to this resource was created using Perma.cc and directs the reader to an archived record of the page.

³¹The link to this resource was created using Perma.cc and directs the reader to an archived record of the page.

"Because the administrative rule governing Breathalyzer tests is similar to the one controlling the administration of PBTs," the Robe Court viewed People v Boughner, 209 Mich App 397 (1995), and People v Wujkowski, 230 Mich App 181 (1998), as instructive. Robe, 337 Mich App at 146. In Boughner, "failure to comply with the 15-minutes observation rule sufficiently undermined the accuracy of the defendant's Breathalyzer test results to warrant the reversal of his plea-based conviction . . . [because] even though there was video of approximately 35 minutes before the Breathalyzer was administered, the operator of the Breathalyzer observed the defendant for no more than eight minutes. Moreover, the operator did not continuously observe the defendant for those eight minutes, and the defendant's hand was either on his face or in his mouth during the time in which he was videotaped." Robe, 337 Mich App at 146-147 (citation omitted). Contrast with *Wujkowski*, where "violation of the 15-minute observation rule did not warrant suppression of the Breathalyzer test results because there was only a de minimis, technical violation of the regulation," the operator observed the defendant for more than 15 minutes, and "[t]he alleged variance from the rule was a six-second period during which the operator walked away from the defendant to check the machine, and during that time there was another officer present while the operator left to check the machine." Robe, 337 Mich App at 147. In Robe, "the officer who administered the [PBT] only observed defendant for three minutes, and there [was] no evidence that anyone else observed defendant for the additional 12 minutes before the test was administered." Id. at 148. The Robe Court concluded that the facts of the case were more similar to Boughner and required suppression of the PBT result because "the violation of the administrative rule was significant and call[ed] into question the accuracy of the PBT." Robe, 337 Mich App at 148 (the Court did not consider whether there was probable cause to arrest absent the PBT result because the trial court did not address the issue, but indicated defendant could raise the issue on remand).

3. Arrest Authorized

"A peace officer may arrest a person based in whole or in part on the results of a [PBT]." MCL 257.625a(2)(a).

4. Admissibility at Trial

"The results of a [PBT] are admissible in a criminal prosecution for a crime enumerated in [MCL 257.625c(1)] or in an administrative hearing for 1 or more of the following purposes: (*i*) To assist the court or hearing officer in determining a challenge to the validity of an arrest. This subparagraph does not limit the introduction of other competent evidence offered to establish the validity of an arrest.

(*ii*) As evidence of the defendant's breath alcohol content, if offered by the defendant to rebut testimony elicited on cross-examination of a defense witness that the defendant's breath alcohol content was higher at the time of the charged offense than when a chemical test was administered under [MCL 257.625a(6)].

(*iii*) As evidence of the defendant's breath alcohol content, if offered by the prosecution to rebut testimony elicited on cross-examination of a prosecution witness that the defendant's breath alcohol content was lower at the time of the charged offense than when a chemical test was administered under [MCL 257.625(6)]." MCL 257.625a(2)(b).

MCL 257.625a(2)(b) "does not reflect an arbitrary exercise of a governmental authority" that "imposes an unconstitutional impediment on [a] defendant's right to present a complete defense" because "our Legislature has a legitimate interest in limiting collateral litigation and jury confusion concerning the result of comparatively unreliable breath tests" and "ensuring that reliable evidence is presented to the trier of fact in a criminal trial." People v Parrott, 335 Mich App 648, 658, 661, 663 (2021) (quotation marks and citation omitted). "MCL 257.625a(2)(b) reflects the exercise of legislative authority that is consistent with" those interests. *Parrott*, 335 Mich App at 661. "In MCL 257.625a, the Legislature equalized how both defendants and the prosecution can use PBT results. The plain language of the statute treats defendants and the prosecutors the same: neither may use PBT results unless the result rebuts certain testimony regarding the defendant's BAC at the time of the offense. The statute provides neither party an advantage. And even if [the Court] were to accept defendant's premise that prosecutors tend to have greater success when moving for the admission of PBT results, any imbalance is due to a defendant's cross-examination that opens the door to an unfavorable PBT result." Parrott, 335 Mich App at 664-665 (citation omitted; noting "a defendant's ill-advised crossexamination of a government witness [does not] render[] a neutral statute unconstitutionally disproportionate"). A defendant "is not deprived of the meaningful opportunity to

present a complete defense" because the defendant "may still present expert testimony that his BAC was lower at the time he was driving because his BAC was increasing" as time continued; defendant "simply may not use his PBT result to do so." *Id.* at 666.

B. Chemical Test

1. Implied Consent Upon Arrest

"A person who operates a vehicle upon a public highway or other place open to the general public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within [Michigan] is considered to have given consent to chemical tests of his or her blood, breath, or urine for the purpose of determining the amount of alcohol or presence of a controlled substance or other intoxicating substance, or any combination of them, in his or her blood or urine or the amount of alcohol in his or her breath in all of the following circumstances:

(a) If the person is arrested for a violation of [MCL 257.625(1), MCL 257.625(3), MCL 257.625(4), MCL 257.625(5), MCL 257.625(6), MCL 257.625(7), or MCL 257.625(8)], [MCL 257.625a(5)], or [MCL 257.625m] or a local ordinance substantially corresponding to [MCL 257.625(1), MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)], [MCL 257.625(3), MCL 257.625(6), or MCL 257.625(8)], [MCL 257.625a(5)], or [MCL 257.625m].

(b) If the person is arrested for a violation of [MCL 257.601d], [MCL 257.626(3) or MCL 257.626(4)], or manslaughter, or murder resulting from the operation of a motor vehicle, and the peace officer had reasonable grounds to believe the person was operating the vehicle in violation of [MCL 257.625]." MCL 257.625c(1).

"A person who is afflicted with hemophilia, diabetes, or a condition requiring the use of an anticoagulant under the direction of a physician is not considered to have given consent to the withdrawal of blood." MCL 257.625c(2).

"The tests shall be administered as provided in [MCL 257.625a(6)]." MCL 257.625c(3).

2. Advice

"A person arrested for a crime described in [MCL 257.625c(1)] shall be advised of all of the following:

(*i*) If he or she takes a chemical test of his or her blood, urine, or breath administered at the request of a peace officer, he or she has the right to demand that a person of his or her own choosing administer 1 of the chemical tests.^[32]

(*ii*) The results of the test are admissible in a judicial proceeding as provided under [the MVC] and will be considered with other admissible evidence in determining the defendant's innocence or guilt.

(*iii*) He or she is responsible for obtaining a chemical analysis of a test sample obtained at his or her own request.

(*iv*) If he or she refuses the request of a peace officer to take a test described in subparagraph (*i*), a test shall not be given without a court order, but the peace officer may seek to obtain a court order.

(*v*) Refusing a peace officer's request to take a test described in subparagraph (*i*) will result in the suspension of his or her operator's . . . license and vehicle group designation or operating privilege and in the addition of 6 points to his or her driver record." MCL 257.625a(6)(b).

3. Submitting to or Refusing a Chemical Test

"A chemical test described in [MCL 257.625a(6)] shall be administered at the request of a peace officer having reasonable grounds to believe the person has committed a crime described in [MCL 257.625c(1)]." MCL 257.625a(6)(d).

³²However, the defendant cannot choose a specific analyst at the Michigan State Police (MSP) laboratory to administer the chemical test. See *People v Green*, 310 Mich App 249, 255 (2015) (reversing the trial court's order compelling the MSP laboratory to retest evidence at the defendant's request and holding that MCL 257.625a(6), which provides the defendant with an identical right to independent testing in the context of motor vehicles, does not permit a defendant to choose a forensic scientist at the MSP laboratory to administer a chemical test because "there is no indication that the MSP laboratory ... offers chemical testing services to private individuals or is able to bill for such services[]"). "[T]he trial court lacks authority to compel a state agency to perform services it does not offer." *Green*, 310 Mich App at 255. Further, "MCR 6.201(A)(6) does not provide the trial court with the authority to order the MSP to retest its own evidence. Rather, it merely provides the court with the authority to provide [the] *defendant* the opportunity to test any tangible physical evidence." *Green*, 310 Mich App at 256-257.

"A person who takes a chemical test administered at a peace officer's request as provided in [MCL 257.625a] shall be given a reasonable opportunity to have a person of his or her own choosing administer 1 of the chemical tests described in [MCL 257.625a(6)(d)] within a reasonable time after his or her detention." MCL 257.625a(6)(d).³³ "The test results are admissible and shall be considered with other admissible evidence in determining the defendant's innocence or guilt." *Id*.

"If the person charged is administered a chemical test by a person of his or her own choosing, the person charged is responsible for obtaining a chemical analysis of the test sample." MCL 257.625a(6)(d).

"If a person refuses the request of a peace officer to submit to a chemical test offered under [MCL 257.625a(6)], a test shall not be given without a court order, but the officer may seek to obtain the court order." MCL 257.625d(1).³⁴

"A written report shall immediately be forwarded to the secretary of state by the peace officer." MCL 257.625d(2). "The report shall state that the officer had reasonable grounds to believe that the person had committed a crime described in [MCL 257.625c(1)], and that the person had refused to submit to the test upon the request of the peace officer and had been advised of the consequences of the refusal." MCL 257.625d(2).

a. Requesting a Hearing

"If a person refuses to submit to a chemical test pursuant to [MCL 257.625d], the peace officer shall immediately notify the person in writing that within 14 days of the date of the notice the person may request a hearing as provided in [MCL 257.625f]." MCL 257.625e(1).

³³However, the defendant cannot choose a specific analyst at the Michigan State Police (MSP) laboratory to administer the chemical test. See *People v Green*, 310 Mich App 249, 255 (2015) (reversing the trial court's order compelling the MSP laboratory to retest evidence at the defendant's request and holding that MCL 257.625a(6), which provides the defendant with an identical right to independent testing in the context of motor vehicles, does not permit a defendant to choose a forensic scientist at the MSP laboratory to administer a chemical test because "there is no indication that the MSP laboratory . . . offers chemical testing services to private individuals or is able to bill for such services[]"). "[T]he trial court lacks authority to compel a state agency to perform services it does not offer." *Green*, 310 Mich App at 255. Further, "MCR 6.201(A)(6) does not provide the trial court with the authority to order the MSP to retest its own evidence. Rather, it merely provides the court with the authority to provide [the] *defendant* the opportunity to test any tangible physical evidence." *Green*, 310 Mich App at 256-257.

³⁴See Section 9.13(B)(7) for additional information regarding constitutional issues of search warrants.

"The notice shall specifically state that failure to request a hearing within 14 days will result in the suspension of the person's license or permit to drive." MCL 257.625e(2). "The notice shall also state that there is not a requirement that the person retain counsel for the hearing, although counsel would be permitted to represent the person at the hearing." *Id*.

"If a person who refuses to submit to a chemical test pursuant to [MCL 257.625d] does not request a hearing within 14 days after the date of notice pursuant to [MCL 257.625e], the secretary of state shall impose the following license sanctions:

(a) If the person was operating a vehicle other than a commercial motor vehicle, suspend or deny the person's operator's . . . license or permit to drive, or nonresident operating privilege, for 1 year or, for a second or subsequent refusal within 7 years, for 2 years. If the person is a resident without a license or permit to operate a vehicle in [Michigan], the secretary of state shall not issue the person a license or permit for 1 year or, for a second or subsequent refusal within 7 years, for 2 years." MCL 257.625f(1).

If the person requests a hearing, MCL 257.322 governs the procedures for that administrative hearing. MCL 257.625f(2). "If the person who requested a hearing does not prevail, the secretary of state shall impose the following license sanctions after the hearing:

(a) If the person was operating a vehicle other than a commercial motor vehicle, suspend or deny issuance of a license or driving permit or a nonresident operating privilege of the person for 1 year or, for a second or subsequent refusal within 7 years, for 2 years. If the person is a resident without a license or permit to operate a vehicle in [Michigan], the secretary of state shall not issue the person a license or permit for 1 year or, for a second or subsequent refusal within 7 years, for 2 years." MCL 257.625f(7).

b. Circuit Court Appeal

"The person may file a petition in the circuit court of the county in which the arrest was made to review the suspension or denial as provided in [MCL 257.323]." MCL 257.625f(7)(a). "If the person who requested the hearing prevails, the peace officer who filed the report under [MCL 257.625d] may, with the consent of the prosecuting attorney, file a petition in the circuit court of the county in which the arrest was made to review the determination of the hearing officer as provided in [MCL 257.323]." MCL 257.625f(8).

4. Requirements for Collecting Sample/Specimen for Chemical Test

"A sample or specimen of urine or breath must be taken and collected in a reasonable manner. Only a licensed physician, or an individual operating under the delegation of a licensed physician under [MCL 333.16215], qualified to withdraw blood and acting in a medical environment, may withdraw blood at a peace officer's request to determine the amount of alcohol or presence of a controlled substance or other intoxicating substance in the person's blood, as provided in [MCL 257.625a(6)(c)]. Liability for a crime or civil damages predicated on the act of withdrawing or analyzing blood and related procedures does not attach to a licensed physician or individual operating under the delegation of a licensed physician who withdraws or analyzes blood or assists in the withdrawal or analysis in accordance with [the MVC] unless the withdrawal or analysis is performed in a negligent manner." MCL 257.625a(6)(c).

5. Disclosure

"If a chemical test described in [MCL 257.625a(6)] is administered, the test results must be made available to the **person** charged or the person's attorney on the written request to the prosecution, with a copy of the request filed with the court. The prosecution shall furnish the results at least 2 days before the day of the trial. The prosecution shall offer the test results as evidence in that trial. Failure to fully comply with the request bars the admission of the results into evidence by the prosecution." MCL 257.625a(8).

Similarly, after an accident, chemical tests may be taken and must be disclosed as required by statute. See MCL 257.625a(6)(e)-(f).

Under MCL 257.625a(6)(e), if a chemical analysis of a driver's blood has been taken pursuant to that provision (taking blood after accident for medical purposes), "[t]he medical facility or person performing the chemical analysis shall disclose the results of the analysis to a prosecuting attorney who requests the results for use in a criminal prosecution as provided in [MCL 257.625a(6)(e)]. A medical facility or person disclosing information in compliance with [MCL 257.625a(6)(e)] is not civilly or criminally liable for making the disclosure." MCL 257.625a(6)(e). "[MCL $257.625a(6)(e)^{35}$] is constitutional under US Const, Am IV and Const 1963, art 1, § 11, and the Equal Protection Clauses of US Const, Am XIV and Const 1963, art 1, § 2." People v Perlos, 436 Mich 305, 309 (1990). Although the Perlos holding construed a former version of this statute, the subsequent statutory amendments do not appear to disturb this holding.

Under MCL 257.625a(6)(f), a person's blood "must be withdrawn in a manner directed by the medical examiner" following an accident where the driver dies. "The medical examiner shall give the results of the chemical analysis of the sample to the law enforcement agency investigating the accident and that agency shall forward the results to the department of state police." *Id*.

6. Admissibility Issues

"The amount of alcohol or presence of a controlled substance or other intoxicating substance in a driver's blood or urine or the amount of alcohol in a person's breath at the time alleged as shown by chemical analysis of the person's blood, urine, or breath is admissible into evidence in any civil or criminal proceeding and is presumed to be the same as at the time the person operated the vehicle." MCL 257.625a(6)(a).

"The results of the test are admissible in a judicial proceeding as provided under [the MVC] and will be considered with other admissible evidence in determining the defendant's innocence or guilt." MCL 257.625a(6)(b)(*ii*).

"A person's refusal to submit to a chemical test as provided in [MCL 257.625a(6)] is admissible in a criminal prosecution for a crime described in [MCL 257.625c(1)] only to show that a test was offered to the defendant, but not as evidence in determining the defendant's innocence or guilt." MCL 257.625a(9). "The jury must be instructed accordingly." *Id*. M

³⁵ Formerly MCL 257.625a(9).

Crim JI 15.9 addresses the defendant's decision to forgo chemical testing.

"The provisions of [MCL 257.625a(6)] relating to chemical testing do not limit the introduction of any other admissible evidence bearing on any of the following questions:

(a) Whether the person was impaired by, or under the influence of, alcoholic liquor, a controlled substance or other intoxicating substance, or a combination of alcoholic liquor, a controlled substance, or other intoxicating substance.

(b) Whether the person had an alcohol content of 0.08 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine or, beginning 5 years after the state treasurer publishes a certification under [MCL 257.625(28)], the person had an alcohol content of 0.10 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine.

(c) If the person is less than 21 years of age, whether the person had any bodily alcohol content within his or her body." MCL 257.625a(7).

"If a chemical test described in [MCL 257.625a(6)] is administered, the test results must be made available to the person charged or the person's attorney on the written request to the prosecution, with a copy of the request filed with the court." MCL 257.625a(8). "The prosecution shall furnish the results at least 2 days before the day of the trial. The prosecution shall offer the test results as evidence in that trial. Failure to fully comply with the request bars the admission of the results into evidence by the prosecution." *Id*.

7. Constitutional Issues

"[T]he Fourth Amendment permits warrantless breath tests incident to arrests for drunk driving," and a state may criminally prosecute a driver for refusing a warrantless breath test;³⁶ "[t]he impact of breath tests on privacy is slight, and the need for [blood alcohol concentration (BAC)] testing is great." *Birchfield v North Dakota*, 579 US 438, 474 (2016). However, "[b]ecause breath tests are significantly less intrusive than blood tests and in most cases amply serve law enforcement

³⁶ Note that Michigan does not currently criminalize an individual's refusal to submit to a preliminary chemical breath analysis (PBT); refusal to submit is a civil infraction. See MCL 257.625a(2)(d).

interests, ... a blood test[] may [not] be administered as a search incident to a lawful arrest for drunk driving," and "motorists cannot be deemed to have consented to submit to a blood test on pain of committing a criminal offense." *Id.* at 476-478 (concluding that one of the three petitioners in the case "was threatened with an unlawful search" under a state law making it a crime to refuse a warrantless blood draw, and that "the search he refused [could not] be justified as a search incident to his arrest or on the basis of implied consent") (emphasis added).

The "defendant's stated fear of economic consequences that would stem from the suspension of his license under the implied-consent law" did not render his consent to submit to a blood draw invalid where he "admitted during [an] evidentiary hearing that he fully understood his choices under the implied-consent law and made an informed, reasoned decision." *People v Stricklin*, 327 Mich App 592, 599, 603 (2019) (defendant did not challenge the constitutionality of Michigan's implied consent law, but instead argued that "the threat of [licensing] sanctions affected the voluntariness of his... consent" "because defendant drove for a living and feared the impact that losing his license would have on his economic livelihood"). "Having to make a choice between two undesirable options does not render defendant's express consent to the blood draw coercive and involuntary." *Id*. at 603.

"[T]he natural metabolization of alcohol in the bloodstream [does not] present[] a *per se* exigency that justifies an exception to the Fourth Amendment's warrant requirement for nonconsensual blood testing in all drunk-driving cases." *Missouri v McNeely*, 569 US 141, 145 (2013). "[C]onsistent with general Fourth Amendment principles . . . exigency in this context must be determined case by case based on the totality of the circumstances." *Id.* See also *Birchfield*, 579 US at 474-475 (citing *McNeely*, 569 US 141, and noting that "[n]othing prevents the police from seeking a warrant for a blood test when there is sufficient time to do so in the particular circumstances or from relying on the exigent circumstances exception to the warrant requirement when there is not"). See MCL 257.625d(1).

In a plurality opinion³⁷, the United States Supreme Court held that "in a narrow . . . category of cases . . . in which the driver is unconscious and therefore cannot be given a breath

³⁷ "A plurality opinion of the United States Supreme Court . . . is not binding precedent. *Texas v Brown*, 460 US 730, 737 (1983)." *People v Beasley*, 239 Mich App 548, 559 (2000).

test, . . . the exigent circumstances rule almost always permits a blood test without a warrant." *Mitchell v Wisconsin*, 588 US ____,

____ (2019). "[E]xigency exists when (1) BAC evidence is dissipating and (2) some other factor creates pressing health, safety, or law enforcement needs that would take priority over a warrant application. Both conditions are met when a drunk-driving suspect is unconscious[.]" *Id.* at ____.

A search warrant to perform chemical testing should not be invalidated unless "material misstatements or omissions *necessary to the finding of probable cause* have been made." *People v Czuprynski*, 325 Mich App 449, 471 (2018) (citation omitted). A search warrant remains valid even if it contains some incorrect information, or fails to include exculpatory information, if the incorrect or omitted information does not negate a finding of probable cause. *Id.* at 470.

"Reliance on a warrant is reasonable even if the warrant is later invalidated for lack of probable cause, except under three circumstances: (1) if the issuing magistrate or judge is misled by information in the affidavit that the affiant either knew was false or would have known was false except for his or her reckless disregard of the truth; (2) if the issuing judge or magistrate wholly abandons his or her judicial role; or (3) if an officer relies on a warrant based on a 'bare bones' affidavit so lacking in indicia of probable cause as to render official belief in its existence entirely unreasonable." *Czuprynski*, 325 Mich at 472, citing *United States v Leon*, 468 US 897, 915, 923 (1984); *People v Goldston*, 470 Mich 523, 531 (2004).

"[B]lood [that] has been lawfully collected for analysis may be analyzed without infringing on additional privacy interests or raising separate Fourth Amendment concerns." People v Woodard, 321 Mich App 377, 390-391 (2017). "[O]nce police procured a sample of [the] defendant's blood pursuant to her consent, she had no reasonable expectation of privacy in the blood alcohol content of that sample and it could be examined for that purpose without her consent[;]" "the subsequent analysis of the blood did not constitute a separate search and [the] defendant simply had no Fourth Amendment basis on which to object to the analysis of the blood for the purpose for which it was drawn." Id. at 396. "[W]ithdrawal of consent after the search has been completed does not entitle a defendant to the return of evidence seized during the course of a consent search because those items are lawfully in the possession of the police; and, by the same token, a defendant who consents to the search in which evidence is seized cannot, by revoking consent, prevent the police from examining the lawfully obtained evidence." Id. at 394-395.

9.14 Roadside Drug Testing

"The department of state police may establish a pilot program in [Michigan] for roadside drug testing to determine whether an individual is operating a vehicle while under the influence of a controlled substance in violation of [MCL 257.625]." MCL 257.625t(1). Pilot programs will run for one calendar year. MCL 257.625t(2). The department of state police is responsible for developing a written policy for the implementation of the pilot program and the administration of roadside drug testing, and promulgating administrative rules to implement the pilot program. MCL 257.625t(3)-(4).

A. Preliminary Oral Fluid Analysis

Qualified officers participating in the roadside drug testing pilot program will use preliminary oral fluid analysis to determine whether an individual is operating a vehicle while under the influence of a controlled substance. See MCL 257.625r.³⁸

1. Authority to Require Analysis

In order to require a person to submit to a preliminary oral fluid analysis, a peace officer must be:

- a certified drug recognition expert; and
- in a county participating in the roadside drug testing pilot. MCL 257.625r(1).

Before conducting a preliminary oral fluid analysis, the peace officer must have reasonable cause to believe either that

- "a person was operating a vehicle upon a highway or other place open to the public or generally accessible to motor vehicles, including an area designated for the parking of vehicles, within this state and that the person by the consumption of a controlled substance, may have affected his or her ability to operate a vehicle," or
- "a person had in his or her body any amount of a controlled substance listed in schedule 1 under . . . MCL 333.7212, or a rule promulgated under that

³⁸Note that MCL 257.625r also discusses specific procedures and rules applicable to operators of commercial motor vehicles who are subjected to preliminary oral fluid analysis as part of the roadside drug testing pilot. See MCL 257.625r(6)-(9). Discussion of commercial motor vehicles is outside the scope of this benchbook.

section, or of a controlled substance described in [MCL 333.7214(a)(*iv*).]" MCL 257.625r(1).

2. Chemical Test Requirements Apply

"A person who submits to a preliminary oral fluid analysis remains subject to the requirements of [MCL 257.625a, MCL 257.625c, MCL 257.625d, MCL 257.625e, and MCL 257.625f] for purposes of chemical tests described in those sections." MCL 257.625r(4). See Section 9.13 for a discussion of chemical testing.

3. Civil Liability

Refusal to submit to a preliminary oral fluid analysis upon a lawful request by a peace officer is a civil infraction. MCL 257.625r(5).

B. Authority to Arrest

"A peace officer who is certified drug recognition expert... in a county participating in the roadside drug testing pilot program under [MCL 257.625t] may arrest a person in whole or in part upon the results of a preliminary oral fluid analysis." MCL 257.625r(2).

C. Admissibility of Preliminary Oral Fluid Analysis

"The results of a preliminary oral fluid analysis are admissible in a criminal prosecution for a crime enumerated in [MCL 257.625c(1)] or in an administrative hearing for 1 or more of the following purposes:

(a) To assist the court or hearing officer in determining a challenge to the validity of an arrest. This subdivision does not limit the introduction of other competent evidence offered to establish the validity of an arrest.

(b) As evidence of the presence or nonpresence of a controlled substance in the defendant's oral fluid if offered by the defendant to rebut testimony elicited on cross-examination of a defense witness that a preliminary oral fluid analysis of the defendant's oral fluid showed the presence of a controlled substance that was not found to be present when a chemical test of the defendant's blood or urine was administered under [MCL 257.625a].

(c) As evidence of the presence or nonpresence of a controlled substance in the defendant's oral fluid if offered by the prosecution to rebut testimony elicited on cross-examination of a prosecution witness that a preliminary oral fluid analysis of the defendant's oral fluid showed no presence of a controlled substance that was found to be present when a chemical test of the defendant's blood or urine was administered under [MCL 257.625a]." MCL 257.625r(3).

9.15 Testimony Regarding Field Sobriety Tests

"A person who is qualified by knowledge, skill, experience, training, or education, in the administration of standardized field sobriety tests, including the horizontal gaze nystagmus (HGN) test, shall be allowed to testify subject to showing of a proper foundation of qualifications. This section does not preclude the admissibility of a nonstandardized field sobriety test if it complies with the Michigan rules of evidence." MCL 257.625s.

9.16 Drug Recognition Evaluator or Expert Testimony

Drug Recognition Evaluator or Expert (DRE) testimony is not automatically admissible, and the trial court must still make a determination whether a DRE officer is qualified to offer expert testimony. See People v Bowden, 344 Mich App 171, 175 n 2 (2022) (noting that even if a person's "certification designates him to be a drug recognition 'expert,' that label has no bearing on whether he may properly testify as an expert for purposes of MRE 702"). In Bowden, "the prosecution filed a motion in the district court requesting the court to 'declare [a deputy involved in the traffic stop with the defendant] an expert in the field of Drug Evaluation and Classification and be allowed to testify, and provide an expert opinion, as a [DRE]."" Id. at 177 (the defendant was charged with operating while intoxicated on the basis of marijuana use). However, "the prosecution did not present any evidence in the district court to show that the DRE protocol had been validated as a reliable method for demonstrating a person's *level of impairment* due to marijuana or the degree to which a person's driving abilities could be diminished by any given level of marijuana." Id. at 189. Although the "studies on which the prosecution relied demonstrated the DRE protocol's level of accuracy with respect to determining whether a particular type of substance was *present* in a person's blood," "neither of the submitted reports purported to even address the question of how particular levels of marijuana impacted a person's ability to drive or rendered a person 'impaired." Id. at 189. Accordingly, the Court concluded that "the determination under the DRE protocol that a person

is 'impaired' and unable to safely drive a car appears to be ultimately based on the DRE officer's subjective judgment, and there is no evidence in this record that the ability of a person to make such a judgment based on the application of the DRE protocol has been tested to demonstrate the accuracy and validity of reaching such a conclusion on a person's level of impairment due to marijuana." Id. at 189. The Bowden Court noted that the "prosecution did not present any evidence that the DRE protocol has been tested, or has a known error rate, with respect to the purpose for which the prosecution intended to use the results of the protocol in this case—to provide evidence of defendant's level of impairment and impaired driving ability." Id. at 190. The Court commented that there was simply no evidence "to support that the DRE protocol [could] reliably be used to detect the degree or level of intoxication caused by marijuana and determine whether that level of intoxication has made the person unable to safely drive a motor vehicle." Id. at 191. Because the prosecution "failed to establish any valid connection between the use of the DRE protocol and a conclusion regarding the degree to which a person's driving ability was diminished by the use of marijuana," the Bowden Court held that "the prosecution failed to meet its burden to establish the reliability, and thus the admissibility, of the proposed expert testimony." Id. at 191.

9.17 Obtaining or Transferring a Vehicle to Circumvent Vehicle or License Sanctions

A. Unlawful Acquisition of a Vehicle

1. Unlawful Purchase or Lease of a Vehicle to Circumvent Immobilization

"A person shall not purchase or lease another vehicle or an interest in another vehicle with the intent to circumvent the restrictions created by immobilization of a vehicle under [the MVC]." MCL 257.233(2).

"A person who violates [MCL 257.233(2)] is guilty of a misdemeanor punishable by imprisonment for not more than 1 year or a fine of not more than \$1,000.00, or both." MCL 257.233(5).

2. Unlawful Acquisition of a Vehicle Following License Suspension, Revocation, or Denial

"A person whose operator's . . . license is suspended, revoked, or denied for, or who has never been licensed by [Michigan] and was convicted for, a third or subsequent violation of [MCL 257.625 or MCL 257.625m], o[r] a local ordinance substantially

corresponding to [MCL 257.625 or MCL 257.625m], or a law of another state substantially corresponding to [MCL 257.625 or MCL 257.625m], or for a fourth or subsequent suspension or revocation under [MCL 257.904] shall not purchase, lease, or otherwise acquire a motor vehicle during the suspension, revocation, or denial period." MCL 257.233(6).

"A person who violates [MCL 257.233(6)] is guilty of a misdemeanor punishable by imprisonment for not more than 93 days or a fine of not more than \$100.00, or both." MCL 257.233(6).

3. Assigned Holder of Plates Fails to Produce Old Registration Certificate or Certificate of Title Upon Application for New Registration Certificate

"If the assigned holder of registration plates applies for a new registration certificate, the application shall be accompanied either by the old registration certificate or by a certificate of title showing the person to be the assigned holder of the registration plates for which the old registration certificate had been issued." MCL 257.233(7).

"A person who fails or neglects to fulfill the requirements of [MCL 257.233(7)] is guilty of a misdemeanor punishable by imprisonment for not more than 93 days or a fine of not more than \$100.00, or both." MCL 257.233(7).

B. Unlawful Transfer of a Vehicle

1. Transfer to Avoid Forfeiture

"A person shall not transfer or attempt to transfer ownership or right of possession of a vehicle subject to forfeiture or ordered forfeited under [the MVC] with the intent to avoid the forfeiture of that vehicle." MCL 257.233(3).

"A person who violates [MCL 257.233(3)] is guilty of a misdemeanor punishable by imprisonment for not more than 1 year or a fine of not more than \$1,000.00, or both." MCL 257.233(5).

2. Transfer of a Vehicle Subject to Vehicle Sanctions to a Person Exempt from Use Tax

"During the time a vehicle is subject to a temporary registration plate, vehicle forfeiture, immobilization, registration denial, or the period from adjudication to immobilization of forfeiture under [the MVC], a person shall not without a court order transfer or assign the title or an interest in the vehicle to a person who is not subject to payment of a use tax under [MCL 205.93]." MCL 257.233(4).

"The following transfers or purchases are not subject to use tax:

(a) A transaction or a portion of a transaction if the transferee or purchaser is the spouse, mother, father, brother, sister, child, stepparent, stepchild, stepbrother, stepsister, grandparent, grandchild, legal ward, or a legally appointed guardian with a certified letter of guardianship, of the transferor.

(b) A transaction or a portion of a transaction if the transfer is a gift to the beneficiary in the administration of an estate.

(c) If a vehicle, ORV, manufactured housing, aircraft, snowmobile, or watercraft that has once been subjected to the Michigan sales or use tax is transferred in connection with the organization, reorganization, dissolution, or partial liquidation of an incorporated or unincorporated business and the beneficial ownership is not changed.

(d) If an insurance company licensed to conduct business in [Michigan] acquires ownership of a late model distressed vehicle as defined in [MCL 257.12a], through payment of damages in response to a claim or when the person who owned the vehicle before the insurance company reacquires ownership from the company as part of the settlement of a claim." MCL 205.93(3).

"A person who violates [MCL 257.233(4)] is guilty of a misdemeanor punishable by imprisonment for not more than 1 year or a fine of not more than \$1,000.00, or both." MCL 257.233(5).

Glossary

A

Abandoned vehicle

• For purposes of MCL 257.252a, *abandoned vehicle* means "any of the following:

(a) A vehicle that has remained on private property without the consent of the owner.

(b) A vehicle that has remained on public property for a period of not less than 48 hours, or on a state trunk line highway as described in . . . MCL 247.651, as follows:

(*i*) If a valid registration plate is affixed to the vehicle, for a period of not less than 18 hours.

(*ii*) If a valid registration plate is not affixed to the vehicle.

(c) A vehicle, other than a late-model vehicle, to which all of the following apply:

(*i*) An insurance company has not acquired ownership of the vehicle under [MCL 257.217c].

(*ii*) The vehicle cannot be disposed of under [MCL 257.248c].

(*iii*) The vehicle has remained in the custody of a vehicle salvage pool or broker site without the consent of the vehicle salvage pool operator or the broker for a period of not less than 60 days." MCL 257.252a(2).

Alcoholic liquor

• For purposes of the MVC, *alcoholic liquor* means "any liquid or compound, whether or not medicated, proprietary, patented, and by whatever name called, containing any amount of alcohol including any liquid or compound described in [MCL 436.1105]." MCL 257.1d.

Any bodily alcohol content

 For purposes of MCL 257.625(6), MCL 267.625a(7)(c), and MCL 777.48, any bodily alcohol content means "either of the following:

(a) An alcohol content of 0.02 grams or more but less than 0.08 grams per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine or, beginning 5 years after the state treasurer publishes a certification under [MCL 257.625(28)], the person has an alcohol content of 0.02 grams or more but less than 0.10 grams per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine.

(b) Any presence of alcohol within a person's body resulting from the consumption of alcoholic liquor, other than consumption of alcoholic liquor as a part of a generally recognized religious service or ceremony." MCL 257.625(6); MCL 257.625a(7)(c).

• For purposes of MCL 436.1703, *any bodily alcohol content* means "either of the following:

(*i*) An alcohol content of 0.02 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine.

(*ii*) Any presence of alcohol within a person's body resulting from the consumption of alcoholic liquor, other than consumption of alcoholic liquor as a part of a generally recognized religious service or ceremony." MCL 436.1703(17)(a).

Appearance ticket

• For purposes of MCL 764.9c to MCL 764.9g, appearance ticket means "a complaint or written notice issued and subscribed by a police officer or other public servant authorized by law or ordinance to issue it directing a designated person to appear in a designated local criminal court at a designated future time in connection with his or her alleged commission of a designated violation or violations of state law or local ordinance." MCL 764.9f(1).

Approved sponsor

• For purposes of MCL 257.320d, *approved sponsor* means "a sponsor of a basic driver improvement course that is approved by the secretary of state under [MCL 257.320d(12)] and whose approved status is not suspended or revoked under [MCL 257.320d(16)]." MCL 257.320d(18).

Authorized activity

• For purposes of MCL 257.310e, *authorized activity* "means any of the following:

(a) A school or a school-sanctioned event or activity. For purposes of this subdivision, school means a public or private school, including a home school.

(b) A sporting event or activity, or extracurricular event or activity, that is not school-sanctioned but that is part of an official sports league or association or an official extracurricular club, or that is paid for as a service offered by a business specializing in those events or activities or training for those events or activities.

(c) A class or program of vocational instruction offered by a college, community college, nonprofit association, or unit of government or by a business specializing in vocational training.

(d) An event or activity sponsored by a religious organization that is tax-exempt under federal law.

(e) Transporting an individual in need of immediate emergency care or personal protection to a health care professional, hospital, police station, domestic violence shelter, or public safety location." MCL 257.310e(15).

Authorized emergency vehicle

• For purposes of the MVC, *authorized emergency vehicle* means any one of the following:

"(a) Vehicles of the fire department, police vehicles, ambulances, privately owned motor vehicles of volunteer or paid fire fighters, or volunteer members of an emergency rescue unit if authorized by the chief of an organized fire department, a county sheriff, or the director of the department of state police, or privately owned motor vehicles of volunteer or paid members of a life support agency licensed by the department of licensing and regulatory affairs if authorized by the life support agency.

(b) For purposes of [MCL 257.698(5)(c)] during an emergency, a vehicle owned and operated by a federally recognized nonprofit charitable organization that is used exclusively for assistance during that emergency.

(c) For purposes of [MCL 257.653a], a road service vehicle giving a visual signal by means of a flashing, rotating, or oscillating red or amber light. As used in this subdivision, 'road service vehicle' means a vehicle that is clearly marked and readily recognizable as a vehicle used to assist disabled vehicles." MCL 257.2(1).

Automated driving system

• For purposes of the MVC, *automated driving system* means "hardware and software that are collectively capable of performing all aspects of the dynamic driving task for a vehicle on a part-time or full-time basis without any supervision by a human operator." MCL 257.2b(1).

Automated motor vehicle

For purposes of the MVC, automated motor vehicle means "a motor vehicle on which an automated driving system has been installed, either by a manufacturer of automated driving systems or an upfitter that enables the motor vehicle to be operated without any control or monitoring by a human operator. Automated motor vehicle does not include a motor vehicle enabled with 1 or more active safety systems or operator assistance systems, including, but not limited to, a system to provide electronic blind spot assistance, crash avoidance, emergency braking, parking assistance, adaptive cruise control, lane-keeping assistance, lane departure warning, or traffic jam and queuing assistance, unless 1 or more of these technologies alone or in combination with other systems enable the vehicle on which any active safety systems or operator assistance systems are installed to operate without any control or monitoring by an operator." MCL 257.2b(2).

Automated speed enforcement system

• For the purposes of the MVC, *automated speed enforcement system* "means an electronic traffic sensor system that does both of the following:

(a) Automatically detects a vehicle exceeding the posted speed limit using a lidar system or another technology that must not be either of the following:

(*i*) A radar system.

(*ii*) Less capable than a lidar system.

(b) Produces a recorded image of the vehicle described in [MCL 257.2c(1)(a)] that shows all of the following:

(*i*) A clear and legible identification of the vehicle's registration plate.

(*ii*) The location where the recorded image was taken.

(*iii*) The date and time when the recorded image was taken." MCL 257.2c(1).

Automated speed enforcement system operator

• For purposes of the MVC, *automated speed enforcement system operator* "means an individual trained and certified to operate and monitor an automated speed enforcement system by the automated speed enforcement unit within the state transportation department." MCL 257.2c(2).

Automated technology

• For purposes of the MVC, *automated technology* "means technology installed on a motor vehicle that has the capability to assist, make decisions for, or replace a human operator." MCL 257.2b(3).

B

Bicycle

• For purposes of the Michigan Vehicle Code, *bicycle* means "a device propelled by human power upon which a person may ride, having either 2 or 3 wheels in a tandem or tricycle

arrangement, all of which are over 14 inches in diameter." MCL 257.4.

С

Camera-based violation

• For the purposes of MCL 257.682, *camera-based violation* "means a violation of [MCL 257.682(1)] based solely on a photograph captured or a video recorded by a stop-arm camera system." MCL 257.682(9)(a).

Cancellation

• For purposes of the MVC, *cancellation* means "that a license or registration is annulled and terminated because of some error or defect or because the licensee or registrant is no longer entitled to such license or registration, but the cancellation of a license or registration is without prejudice and application for a new license or registration may be made at any time after such cancellation." MCL 257.5a.

Certified drug recognition expert

• For purposes of MCL 257.43b, MCL 257.625r, and MCL 257.625t, *certified drug recognition expert* "means a law enforcement officer trained to recognize impairment in a driver under the influence of a controlled substance rather than, or in addition to, alcohol." MCL 257.625t(6)(a). See also MCL 257.43b (defining *certified drug recognition expert* as the term is defined in MCL 257.625t); MCL 257.625r(1) (defining *certified drug recognition expert* as the term is defined in MCL 257.625t).

Chauffeur

- For purposes of the MVC, *chauffeur*.
- "(1) . . . means any of the following:

(a) A person who operates a motor vehicle as a motor carrier under the motor carrier act, . . . MCL 475.1 to [MCL] 479.42, or a motor carrier of passengers as defined in . . . MCL 474.103.^[1]

(b) A person who is employed for the principal purpose of operating a motor vehicle with a GVWR of 10,000 pounds or more.

(c) A person who operates a bus or school bus.

(2) For purposes of subsection (1)(b), a person shall be considered to be employed for the principal purpose of operating a motor vehicle when the person's employment customarily involves transporting for gain or hire any merchandise for display, sale, or delivery.

(3) 'Chauffeur' does not include any of the following:

(a) A farmer or an employee of a farmer operating a vehicle exclusively in connection with the farming operations of the farmer.

(b) A fire fighter or a member of a fire department operating an ambulance.

(c) Emergency medical services personnel operating an ambulance. As used in this subdivision, 'emergency medical services personnel' means that term as defined in . . . MCL 333.20904.

(d) State transportation department employees whose work consists of operating vehicles with a gross vehicle weight rating of 10,000 pounds or more for the purpose of transporting highway and bridge maintenance materials and supplies for all aspects of state trunkline maintenance, including winter maintenance and facilities maintenance.

(e) County road commission employees and other employees of local units of government who do not drive their own vehicles and whose work consists of hauling road building materials and supplies for the road commission or for other municipal purposes.

(f) A person operating a motor vehicle for a volunteer program who only receives reimbursement for the costs of operating the motor vehicle.

(g) A person who operates a motor home for personal pleasure.

¹Effective March 21, 2017, 2016 PA 349 amended MCL 474.103 to remove the definition of *motor carrier of passengers* and to instead define only *motor carrier*. MCL 257.6 has not been amended to reflect the changes made by 2016 PA 349.

(h) A parent or parent's designee for the purpose of transporting pupils to or from school and school related events.

(i) A transportation network company driver.

- (j) A limousine driver.
- (k) A taxicab driver." MCL 257.6.

Citation

• For purposes of the MVC, *citation* is defined as "a complaint or notice upon which a police officer shall record an occurrence involving 1 or more vehicle law violations by the person cited." MCL 257.727c(1).

Civil Infraction

• For purposes of the MVC, *civil infraction* is defined as "an act or omission prohibited by law which is not a crime as defined in [MCL 750.5], and for which civil sanctions may be ordered." MCL 257.6a.

Civil infraction action

• For purposes of the MVC, *civil infraction action* is defined as "a civil action in which the defendant is alleged to be responsible for a civil infraction." MCL 257.741(1).

Clear vision area

• The Michigan Administrative Rules define *clear vision area* as "land acquired or used by the agency having jurisdiction over a highway, for the purpose of maintaining unobstructed vision." Michigan Administrative Rule 247.201(5).

Commercial motor vehicle

• For purposes of the MVC, *commercial motor vehicle* means "a motor vehicle or combination of motor vehicles used in commerce to transport passengers or property if 1 or more of the following apply:

(a) It is designed to transport 16 or more passengers, including the driver.

(b) It has a gross vehicle weight rating or gross vehicle weight, whichever is greater, of 26,001 pounds or more.

(c) It has a gross combination weight rating or gross combination weight, whichever is greater, of 26,001 pounds or more, inclusive of towed units with a gross vehicle weight rating or gross vehicle weight, whichever is greater, of more than 10,000 pounds.

(d) A motor vehicle carrying hazardous material and on which is required to be posted a placard as defined and required under 49 CFR parts 100 to 199." MCL 257.7a(1).²

Commercial quadricycle

• For purposes of the MVC, *commercial quadricycle* means "a vehicle that satisfies all of the following:

(a) The vehicle has fully operative pedals for propulsion entirely by human power.

(b) The vehicle has at least 4 wheels and is operated in a manner similar to a bicycle.

(c) The vehicle has at least 6 seats for passengers.

(d) The vehicle is designed to be occupied by a driver and powered either by passengers providing pedal power to the drive train of the vehicle or by a motor capable of propelling the vehicle in the absence of human power.

(e) The vehicle is used for commercial purposes.

(f) The vehicle is operated by the owner of the vehicle or an employee of the owner of the vehicle." MCL 257.7b.

Complaint

• For purposes of the Code of Criminal Procedure, *complaint* means "a written accusation, under oath or upon affirmation, that a felony, misdemeanor, or ordinance violation has been committed and that the person named or described in the accusation is guilty of the offense. MCL 761.1(c).

Controlled substance

²Note that a commercial motor vehicle "does not include a vehicle used exclusively to transport personal possessions or family members for nonbusiness purposes." MCL 257.7a(2).

- For purposes of the MVC, *controlled substance* means "a controlled substance or controlled substance analogue as defined in [MCL 333.7104][.]" MCL 257.8b. MCL 333.7104(3) defines *controlled substance* as "a drug, substance, or immediate precursor included in schedules 1 to 5 of part 72[of the Public Health Code]."
- For purposes of MCL 257.43b, MCL 257.625c, and MCL 257.625t, *controlled substance* means "that term as defined in ... MCL 333.7104." MCL 257.43b; MCL 257.625c(4)(a); MCL 257.625t(6)(b). MCL 333.7104(3) defines *controlled substance* as "a drug, substance, or immediate precursor included in schedules 1 to 5 of part 72[of the Public Health Code]."

Conviction

• For purposes of the MVC, *conviction* means "any of the following:

(a) A final conviction, the payment of a fine, a plea or guilty or nolo contendere if accepted by the court, or a finding of guilt for a criminal law violation or a juvenile adjudication, probate court disposition, or juvenile disposition for a violation that if committed by an adult would be a crime, regardless of whether the penalty is rebated or suspended.

(b) A conviction defined in federal law under 49 CFR 383.5, regarding the operation of a commercial motor vehicle or the operation of a noncommercial motor vehicle operated by a person licensed to operate a commercial motor vehicle." MCL 257.8a.

Co-occurring disorder

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *co-occurring disorder* "means having 1 or more disorders relating to the use of alcohol or other controlled substances of abuse as well as any serious mental illness, serious emotional disturbance, or developmental disability. A diagnosis of co-occurring disorders occurs when at least 1 disorder of each type can be established independent of the other and is not simply a cluster of symptoms resulting from 1 disorder." MCL 600.1099b(a).

County jail

• For purposes of the MVC, *county jail* "shall be construed to mean the county jail of any county where the violation of any of the provisions of [the MVC] occur." MCL 257.9.

County road commission

 For purposes of MCL 257.628(1), county road commission "means the board of county road commissioners elected or appointed under [MCL 224.6], or, in the case of a charter county with a population of 1,500,000 or more with an elected county executive that does not have a board of county road commissioners, the county executive." MCL 257.628(1).

Court records

• For purposes of the Michigan Court Rules, *court records* "are defined by MCR 8.119 and [MCR 1.109(A)]. Court records are recorded information of any kind that has been created by the court or filed with the court in accordance with Michigan Court Rules. Court records may be created using any means and may be maintained in any medium authorized by [the Michigan Court Rules] provided those records comply with other provisions of law and [the Michigan Court Rules]." MCR 1.109(A)(1).

D

Data

For purposes of MCR 1.109(A)(1), in which the term *court records* is defined, *data* "refers to any information entered in the case management system that is not ordinarily reduced to a document but that is still recorded information, and any data entered into or created by the statewide electronic-filing system." MCR 1.109(A)(1)(b)(*iii*).

Delinquency proceeding

• "[A] proceeding concerning an offense by a juvenile, as defined in MCR 3.903(B)(3)." MCR 3.903(A)(5).

Department

• For purposes of the MVC, *department* means "the department of state." MCL 257.12.

Departure

• For purposes of MCL 769.31 and MCL 769.34, *departure* "means a sentence imposed that is not within the

appropriate minimum sentence range established under the sentencing guidelines set forth in [MCL 777.1 *et seq.*]." MCL 769.31(a).

Developmental disability

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *developmental disability* "means that term as defined in . . . [MCL 330.1100a]." MCL 600.1099b(c).

Disabled person

 For purposes of sections MCL 257.675, MCL 257.803d, and MCL 257.803h of the MVC, *disabled person* means "a person who is determined by a physician, a physician assistant, a physical therapist, or an optometrist as specifically provided in [MCL 257.675, MCL 257.803d, or MCL 257.803h, respectively,] licensed to practice in [Michigan] to have 1 or more of the following physical characteristics:

(a) Blindness as determined by an optometrist, a physician, or a physician assistant.

(b) Inability to walk more than 200 feet without having to stop and rest.

(c) Inability to do both of the following:

(*i*) Use 1 or both legs or feet.

(*ii*) Walk without the use of a wheelchair, walker, crutch, brace, prosthetic, or other device, or without the assistance of another person.

(d) A lung disease from which the person's forced expiratory volume for 1 second, when measured by spirometry, is less than 1 liter, or from which the person's arterial oxygen tension is less than 60 mm/hg of room air at rest.

(e) A cardiovascular condition that causes the person to measure between 3 and 4 on the New York heart classification scale, or that renders the person incapable of meeting a minimum standard for cardiovascular health that is established by the American Heart Association and approved by the department of public health.

(f) An arthritic, neurological, or orthopedic condition that severely limits the person's ability to walk.

(g) The persistent reliance upon an oxygen source other than ordinary air." MCL 257.675(22); MCL 257.803d(2); MCL 257.803h(3).

Disabled person or person with disabilities

• Generally,³ for purposes of the MVC, *disabled person* or *person with disabilities* means "a person who is determined by a physician, a physician assistant, or an optometrist as specifically provided in [MCL 257.19a] licensed to practice in [Michigan] to have 1 or more of the following physical characteristics:

(a) Blindness as determined by an optometrist, a physician, or a physician assistant.

(b) Inability to walk more than 200 feet without having to stop and rest.

(c) Inability to do both of the following:

(*i*) Use 1 or both legs or feet.

(*ii*) Walk without the use of a wheelchair, walker, crutch, brace, prosthetic, or other device, or without the assistance of another person.

(d) A lung disease from which the person's forced expiratory volume for 1 second, when measured by spirometry, is less than 1 liter, or from which the person's arterial oxygen tension is less than 60 mm/hg of room air at rest.

(e) A cardiovascular condition that causes the person to measure between 3 and 4 on the New York heart classification scale, or that renders the person incapable of meeting a minimum standard for cardiovascular health that is established by the American heart association and approved by the department of public health.

(f) An arthritic, neurological, or orthopedic condition that severely limits the person's ability to walk.

(g) The persistent reliance upon an oxygen source other than ordinary air." MCL 257.19a.

³Note that some statutes within the MVC contain their own, specific definition of *disabled person* as indicated above.

Distressed vehicle

For purposes of the MVC, *distressed vehicle* means "a vehicle that has a major component part that has been wrecked, destroyed, damaged, stolen, or missing to the extent that the total estimated cost of repairs to rebuild or reconstruct the vehicle, including parts and labor, is equal to or exceeds 75% of the actual cash value of the vehicle in its predamaged condition. The estimated costs of the repair parts shall be determined by using the current published retail cost of original manufacturer equipment parts or an estimate of the actual cost of the repair parts. The estimated labor costs shall be computed by using the hourly rate and time allocations which are reasonable and commonly assessed in the repair industry in the community where the repairs are performed. For the purposes of [MCL 257.12a], 'actual cash value' means the retail dollar value of a vehicle as determined by an objective vehicle evaluation using local market resources such as dealers or want ads or by an independent vehicle evaluation or appraisal service or by a current issue of a nationally recognized used vehicle guide for financial institution appraisal purposes in [Michigan]." MCL 257.12a.

Documents

 For purposes of MCR 1.109(A)(1), in which the term *court records* is defined, *documents* "include, but are not limited to, pleadings, orders, and judgments." MCR 1.109(A)(1)(a)(*i*).

Drag race

• For purposes of MCL 257.626a, d*rag race* "means the operation of 2 or more vehicles from a point side by side at accelerating speeds in a competitive attempt to out-distance each other over a common selected course or where timing is involved or where timing devices are used in competitive accelerations of speeds by participating vehicles." MCL 257.626a(4).

Driver

• For purposes of the MVC, *driver* means "every person who drives or is in actual physical control of a vehicle." MCL 257.13.

DWI/sobriety court

• For purposes of MCL 600.1084, *DWI/sobriety court* "means the specialized court docket and programs established within judicial circuits and districts throughout this state that are designed to reduce recidivism among alcohol offenders and that comply with the 10 guiding principles of DWI courts as promulgated by the National Center for DWI Courts." MCL 600.1084(9)(a).

Ε

Electric bicycle

- For purposes of the Michigan Vehicle Code, *electric bicycle* "means a device upon which an individual may ride that satisfies all of the following:
 - (a) The device is equipped with all of the following:
 - (*i*) A seat or saddle for use by the rider.
 - (*ii*) Fully operable pedals for human propulsion.
 - (*iii*) An electric motor of not greater than 750 watts.
 - (b) The device falls within 1 of the following categories:

(*i*) Class 1 electric bicycle. As used in this subparagraph, 'class 1 electric bicycle' means an electric bicycle that is equipped with an electric motor that provides assistance only when the rider is pedaling and that disengages or ceases to function when the electric bicycle reaches a speed of 20 miles per hour.

(*ii*) Class 2 electric bicycle. As used in this subparagraph, 'class 2 electric bicycle' means an electric bicycle that is equipped with a motor that propels the electric bicycle to a speed of no more than 20 miles per hour, whether the rider is pedaling or not, and that disengages or ceases to function when the brakes are applied.

(*iii*) Class 3 electric bicycle. As used in this subparagraph, 'class 3 electric bicycle' means an electric bicycle that is equipped with a motor that provides assistance only when the rider is pedaling and that disengages or ceases to function when the electric bicycle reaches a speed of 28 miles per hour." MCL 257.13e.

Electric skateboard

• For purposes of the Michigan Vehicle Code, *electric skateboard* "means a wheeled device that has a floorboard designed to be stood upon when riding that is no more than 60 inches long and 18 inches wide, is designed to transport only 1 person at a time, has an electrical propulsion system with power of no more than 2,500 watts, and has a maximum speed on a paved level surface of not more than 25 miles per hour. An electric skateboard may have handlebars and, in addition to having an electrical propulsion system with power of no more than 2,500 watts, may be designed to also be powered by human propulsion." MCL 257.13f.

Emergency scene

• For purposes of MCL 257.601b, *emergency scene* means "a traffic accident, a serious incident caused by weather conditions, or another occurrence along a highway or street for which a police officer, firefighter, or emergency medical personnel are summoned to aid an injured victim." MCL 257.601b(5)(a).

F

Felony

- For purposes of MCL 750.223(3)(a), *felony* is "a violation of a law of [Michigan], or of another state, or of the United States that is punishable by imprisonment for 4 years or more." MCL 750.223(3)(a).
- For purposes of MCL 750.224f, a *felony* is "a violation of a law of [Michigan], or of another state, or of the United States that is punishable by imprisonment for a term exceeding 1 year, or an attempt to violate such a law." MCL 750.224f(10)(b).
- For purposes of the Code of Criminal Procedure, *felony* "means a violation of a penal law of [Michigan] for which the offender, upon conviction, may be punished by imprisonment for more than 1 year or an offense expressly designated by law to be a felony." MCL 761.1(f).

Felony in which a motor vehicle was used

• For purposes of MCL 257.303, MCL 257.319, and MCL 257.732, a *felony in which a motor vehicle was used* "means a felony during the commission of which the individual operated a motor vehicle and while operating the vehicle presented real or potential harm to individuals or property and 1 or more of the following circumstances existed: (a) The vehicle was used as an instrument of the felony. (b) The vehicle was used to transport a victim of the felony. (c) The vehicle was necessary for the commission of the felony." MCL 257.303(7); MCL 257.319(2)(c);⁴ MCL 257.732(6).

G

Glove compartment

• For purposes of MCL 257.624a, *glove compartment* means "a recess with a hinged and locking door in the dashboard of a motor vehicle." MCL 257.624a(6)(a).

Golf cart

• For purposes of MCL 257.657*a*, *golf cart* "means a vehicle designed for transportation while playing the game of golf." MCL 257.657a(21).

Gross negligence

• For purposes of MCL 257.618a(6), gross negligence means that term as it is defined by MCL 257.606a. MCL 257.618a(6). MCL 257.606a(2) defines gross negligence as "conduct so reckless as to demonstrate a substantial lack of concern for whether an injury results."

Η

Highway

• For purposes of the MVC, *highway* means "the entire width between [] boundary lines of every way publicly

⁴The definition in MCL 257.319(2)(c) is substantively identical to the definition quoted, and set forth in MCL 257.303(7) and MCL 257.732(6), except that instead of using (a)-(d) to set forth the four circumstances, it uses (h)-(iv).

maintained when any part thereof is open to the use of the public for purposes of vehicular travel." MCL 257.20; MCL 257.64.

Historic snowmobile

• For purposes of MCL 257.628b, *historic snowmobile* "means that term as defined in [MCL 324.82101]." MCL 257.628b(4)(a).

Hospital highway zone

• For purposes of MCL 257.627(7), *hospital highway zone* "means a portion of state trunk line highway maintained by the state transportation department that has a posted speed limit of at least 50 miles per hour and has 2 or fewer lanes for travel in the same direction, traverses along property owned by a hospital, contains an ingress and egress point from hospital property, and extends not more than 1,000 feet beyond the boundary lines of hospital property in both directions in a municipality." MCL 257.627(7).

Hold

 For purposes of MCL 257.602b, *hold* means "to physically support with any part of the hands, arms, or shoulders." MCL 257.602b(12)(a).

I

Ignition interlock device

• For purposes of the MVC and MCL 600.1084, *ignition interlock device* means "an alcohol concentration measuring device that prevents a motor vehicle from being started at any time without first determining through a deep lung sample the operator's alcohol level, calibrated so that the motor vehicle cannot be started if the breath alcohol level of the operator, as measured by the test, reaches a level of 0.025 grams per 210 liters of breath, and to which all of the following apply:

> (a) The device meets or exceeds the model specifications for breath alcohol ignition interlock devices (BAIID), 78 FR 26849 - 26867 (May 8, 2013) or any subsequent model specifications.

(b) The device utilizes alcohol-specific electrochemical fuel sensor technology.

(c) As its anticircumvention method, the device installation uses a positive-negative-positive air pressure test requirement, a midtest hum tone requirement, or any other anticircumvention method or technology that first becomes commercially available after July 31, 2007 and that is approved by the department as equally or more effective." MCL 257.20d; MCL 600.1084(9)(b).

Implement of husbandry

• For purposes of the MVC, *implement of husbandry* means "a vehicle or trailer in use for the exclusive function of serving agricultural, horticultural, or livestock operations" and "includes a farm tractor, self-propelled application-type vehicle, farm wagon, farm trailer, a vehicle or trailer adapted for lifting or carrying another implement of husbandry being used in agricultural production, or any substantially similar equipment used to transport products necessary for agricultural production." MCL 257.21.

Intoxicating substance

• For purposes of MCL 257.625 and MCL 257.625c, *intoxicating substance* means "any substance, preparation, or a combination of substances and preparations other than alcohol or a controlled substance, that is either of the following:

(*i*) Recognized as a drug in any of the following publications or their supplements:

(A) The official United States pharmacopoeia.

(B) The official homeopathic pharmacopoeia of the United States.

(C) The official national formulary.

(*ii*) A substance, other than food, taken into a person's body, including, but not limited to, vapors or fumes, that is used in a manner or for a purpose for which it was not intended, and

that may result in a condition of intoxication." MCL 257.625(25)(a). See also MCL 257.625c(4)(b).

J

Judgment

• For purposes of the MVC, *judgment* means "any judgment which shall become final by expiration without appeal of the time within which an appeal might have been perfected, or by final affirmation on appeal, rendered by a court of competent jurisdiction of any state or of the United States, upon a cause of action arising out of the ownership, maintenance, or use of any motor vehicle, for damages, including damages for care and loss of services, because of bodily injury to or death of any person, or for damages because of injury to or destruction of property, including the loss of use thereof, or upon a cause of action on an agreement of settlement for such damages." MCL 257.23.

Juvenile

• For purposes of the Juvenile Code, MCL 712A.1 *et seq.*, until September 30, 2021, a *juvenile* is "a person who is less than 17 years of age who is the subject of a delinquency petition. Beginning October 1, 2021, 'juvenile' means a person who is less than 18 years of age who is the subject of a delinquency petition." MCL 712A.1(1)(i).

Juvenile mental health court

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *juvenile mental health court* "means all of the following:

(*i*) A court-supervised treatment program for juveniles who are diagnosed by a mental health professional with having a serious emotional disturbance, co-occurring disorder, or developmental disability.

(*ii*) Programs designed to adhere to the 7 common characteristics of a juvenile mental health court as described under [MCL 600.1099c(3)].

(*iii*) Programs designed to adhere to the 10 essential elements of a mental health court promulgated by the

Bureau of Justice Assistance, or amended, that include all of the following characteristics:

(A) A broad-based group of stakeholders representing the criminal justice system, the juvenile justice system, the mental health system, the substance abuse treatment system, any related systems, and the community guide the planning and administration of the court.

(B) Eligibility criteria that address public safety and a community's treatment capacity, in addition to the availability of alternatives to pretrial detention for juveniles with mental illnesses, and that take into account the relationship between mental illness and a juvenile's offenses, while allowing the individual circumstances of each case to be considered.

(C) Participants are identified, referred, and accepted into mental health courts, and then linked to community-based service providers as quickly as possible.

(D) Terms of participation are clear, promote public safety, facilitate the juvenile's engagement in treatment, are individualized to correspond to the level of risk that each juvenile presents to the community, and provide for positive legal outcomes for those individuals who successfully complete the program.

(E) In accordance with the Michigan indigent defense commission act, [MCL 780.981–MCL 780.1003], provide legal counsel to juvenile respondents to explain program requirements, including voluntary participation, and guide juveniles in decisions about program involvement. Procedures exist in the juvenile mental health court to address, in a timely fashion, concerns about a juvenile's competency whenever they arise.

(F) Connect participants to comprehensive and individualized treatment supports and services in the community and strive to use, and increase the availability of, treatment and services that are evidence based.

(G) Health and legal information are shared in a manner that protects potential participants'

confidentiality rights as mental health consumers and their constitutional rights. Information gathered as part of the participants' court-ordered treatment program or services is safeguarded from public disclosure in the event that participants are returned to traditional court processing.

(H) A team of criminal justice, if applicable, juvenile justice, and mental health staff and treatment providers receives special, ongoing training and assists mental health court participants to achieve treatment and criminal and juvenile justice goals by regularly reviewing and revising the court process.

(I) Criminal and juvenile justice and mental health staff collaboratively monitor participants' adherence to court conditions, offer individualized graduated incentives and sanctions, and modify treatment as necessary to promote public safety and participants' recovery.

(J) Data are collected and analyzed to demonstrate the impact of the juvenile mental heath court, its performance is assessed periodically, procedures are modified accordingly, court processes are institutionalized, and support for the court in the community is cultivated and expanded." MCL 600.1099b(e).

L

Law enforcement agency

• For the purposes of MCL 257.682, *law enforcement agency* "means any of the following:

(*i*) The department of state police.

(*ii*) The county sheriff's office.

(*iii*) The police department of a local unit of government.

(*iv*) Any other governmental law enforcement agency in [Michigan]." MCL 257.682(9)(b).

Law of another state

• For purposes of the MVC, *law of another state* means "a law or ordinance enacted by another state or by a local unit of government in another state." MCL 257.24c.

Leased vehicle

• For purposes of the MVC, *leased vehicle* means "a motor vehicle for which a person is granted possession for a contracted period of time and in return for a contracted sum." MCL 257.24a.

Leased vehicle owner

• For purposes of MCL 257.675b, *leased vehicle owner* means "a person in the business of renting or leasing leased vehicles or an affiliate of the person, if the person or the affiliate is the registered owner of a standing or parked leased vehicle involved in a violation of a local ordinance or state statute." MCL 257.675b(4)(b).

License

• For purposes of the MVC, *license* means "any driving privileges, license, temporary instruction permit, commercial learner's permit, or temporary license issued under the laws of [Michigan] pertaining to the licensing of persons to operate motor vehicles." MCL 257.25.

Limousine

• For purposes of the Michigan Vehicle Code, *limousine* "means that term as defined in [MCL 257.2102]." MCL 257.25c. MCL 257.2102(d) provides that *limousine* "means a self-propelled motor vehicle used in the carrying of passengers and the baggage of the passengers for hire with a seating capacity of 8 passengers or fewer, including the driver. Limousine does not include a commercial vehicle. Limousine also does not include a vehicle operated by any of the following:

(*i*) A county, city, township, or village as provided by law, or other authority incorporated under 1963 PA 55, MCL 124.351 to [MCL] 124.359.

(*ii*) An authority incorporated under the metropolitan transportation authorities act of 1967, 1967 PA 204, MCL 124.401 to [MCL] 124.426, or that operates a transportation service pursuant to an interlocal

agreement under the urban cooperation act of 1967, 1967 (Ex Sess) PA 7, MCL 124.501 to [MCL] 124.512.

(*iii*) Operating under a contract entered into under 1967 (Ex Sess) PA 8, MCL 124.531 to [MCL] 124.536, or 1951 PA 35, MCL 124.1 to [MCL] 124.13.

(*iv*) An authority incorporated under the public transportation authority act, 1986 PA 196, MCL 124.451 to [MCL] 124.479, or a nonprofit corporation organized under the nonprofit corporation act, 1982 PA 162, MCL 450.2101 to [MCL] 450.3192, that provides transportation services.

(*v*) An authority financing public improvements to transportation systems under the revenue bond act of 1933, 1933 PA 94, MCL 141.101 to [MCL] 141.140.

(*vi*) A person that is only operating limousines to provide the transportation of passengers for funerals.

(*vii*) An employer that is only using the vehicle, or on whose behalf the vehicle is being used, to transport its employees to and from their place of employment."

Limousine driver

• For purposes of the Michigan Vehicle Code, *limousine driver* "means that term as defined in [MCL 257.2102]." MCL 257.25d. MCL 257.2102(f) provides that *limousine driver* "means an individual who uses a limousine to provide transportation services to potential passengers."

Local unit of government

• For the purposes of MCL 257.682, *local unit of government* "means a state university or college or a county, city, village, or township." MCL 257.682(9)(c).

Μ

Manufacturer

• For purposes of the MVC, *manufacturer* means "a person, firm, corporation or association engaged in the manufacture of new motor vehicles, trailers or trailer coaches or semi-trailers, as a regular business." MCL 257.28.

Manufacturer of automated driving systems

• For purposes of the MVC, manufacturer of automated technology means "a manufacturer or subcomponent system producer recognized by the secretary of state that develops or produces automated driving systems or automated vehicles."MCL 257.2b(10).

Mental health professional

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *mental health professional* "means an individual who is trained and experienced in the area of mental illness or developmental disabilities and who is 1 of the following:

(*i*) A physician.

(ii) A psychologist.

(*iii*) A registered professional nurse licensed or otherwise authorized to engage in the practice of nursing under part 172 of the public health code, ... MCL 333.17201 to [MCL] 333.17242].

(*iv*) A licensed master's social worker licensed or otherwise authorized to engage in the practice of social work at the mater's level under part 185 of the public health code, ... MCL 333.18501 to [MCL] 333.18518.

(*v*) A licensed professional counselor licensed or otherwise authorized to engage in the practice of counseling under part 181 of the public health code, ... MCL 333.18101 to [MCL] 333.18117.

(*vi*) A marriage and family therapist licensed or otherwise authorized to engage in the practice of marriage and family therapy under part 169 of the public health code, ... MCL 333.16901 to [MCL] 333.16915." MCL 600.1099b(f).

Minor child

• For purposes of MCL 257.304(4)(*xi*), *minor child* "means an individual who is less than 18 years of age." MCL 257.304(4)(*xi*).

Misdemeanor

• For purposes of the Code of Criminal Procedure, *misdemeanor* "means a violation of a penal law of [Michigan] that is not a felony or a violation of an order,

rule, or regulation of a state agency that is punishable by imprisonment or a fine that is not a civil fine." MCL 761.1(n).

Mobile electronic device

• For purposes of MCL 257.602b, *mobile electronic device* means "an electronic device that is not permanently installed in a motor vehicle, including, but not limited to, a device capable of text messaging, voice communication, entertainment, navigation, accessing the internet, or producing email. Mobile electronic device does not include either of the following:

(*i*) A radio designed for the Citizens Band Service or the Amateur Radio Service of the Federal Communications Commission or a commercial 2-way radio communications device or equipment permanently installed in a motor vehicle.

(*ii*) A medical device that is designed to be worn, including, but not limited to, an insulin pump." MCL 257.602b(12)(b).

Mobile home park

• For purposes of MCL 257.627, *mobile home park* is defined in MCL 125.2302. See MCL 257.627(2)(a). MCL 125.2302(j) defines *mobile home park* as "a parcel or tract of land under the control of a person upon which 3 or more mobile homes are located on a continual, nonrecreational basis and which is offered to the public for that purpose regardless of whether a charge is made for the parcel or tract of land, together with any building, structure, enclosure, street, equipment, or facility used or intended for use incident to the occupancy of a mobile home."

Motor vehicle

• For purposes of the MVC, a *motor vehicle* "means every vehicle that is self-propelled, but for purposes of [MCL 257.401 *et seq.*], motor vehicle does not include industrial equipment such as a forklift, a front-end loader, or other construction equipment that is not subject to registration under [the MVC]. Motor vehicle does not include a power-driven mobility device when that power-driven mobility device is being used by an individual with a mobility disability. Motor vehicle does not include an electric patrol vehicle being operated in compliance with the electric

patrol vehicle act, 1997 PA 55, MCL 257.1571 to [MCL] 257.1577. Motor vehicle does not include an electric personal assistive mobility device. Motor vehicle does not include an electric carriage. Motor vehicle does not include a commercial quadricycle. Motor vehicle does not include an electric bicycle. Motor vehicle does not include an electric skateboard." MCL 257.33.

Motor vehicle manufacturer

- For purposes of the MVC, *motor vehicle manufacturer* "means a person that has manufactured and distributed motor vehicles in the United States that are certified to comply with all applicable federal motor vehicle safety standards and that has submitted appropriate manufacturer identification information to the National Highway Traffic Safety Administration as provided in 49 CFR part 566." MCL 257.2b(12).
- As used in MCL 257.2b, MCL 257.665a, and MCL 257.665b only, *motor vehicle manufacturer* "also includes a person that satisfies all of the following:

(a) Has manufactured automated motor vehicles in the United States that are certified to comply with all applicable federal motor vehicle safety standards.

(b) Has operated automated motor vehicles using a test driver and with an automated driving system engaged on public roads in the United States for at least 1,000,000 miles.

(c) Has obtained an instrument of insurance, surety bond, or proof of self-insurance in the amount of at least \$10,000,000.00, and has provided evidence of that insurance, surety bond, or self-insurance to the department in a form and manner required by the department." MCL 257.2b(12).

Moving violation

- For purposes of MCL 257.601b and MCL 257.601c, moving violation means "an act or omission prohibited under [the MVC] or a local ordinance substantially corresponding to [the Michigan Vehicle Code] that occurs while a person is operating a motor vehicle, and for which the person is subject to a fine." MCL 257.601b(5)(b); MCL 257.601c(3).
- For purposes of MCL 257.601d and MCL 257.907, *moving violation* means "an act or omission prohibited under [the

MVC] or a local ordinance substantially corresponding to [the MVC] that involves the operation of a motor vehicle, and for which a fine may be assessed." MCL 257.601d(4); MCL 257.907(16).

Ν

Nonresident operating privilege

• For purposes of the MVC, *nonresident operating privilege* means "the privilege conferred upon a nonresident by the laws of [Michigan] pertaining to the operation by him [or her] of a motor vehicle, or the use of a motor vehicle owned by him [or her], in [Michigan]." MCL 257.35.

0

Offense by a juvenile

"[A]n act that violates a criminal statute, a criminal ordinance, a traffic law, or a provision of MCL 712A.2(a) or (d)." MCR 3.903(B)(3).

On-demand automated motor vehicle network

• For purposes of the MVC, *on-demand automated motor vehicle network* "means a digital network or software application used to connect passengers to automated motor vehicles, not including commercial motor vehicles, in participating fleets for transportation between points chosen by passengers, for transportation between locations chosen by the passenger when the automated motor vehicle is operated by the automated driving system." MCL 257.2b(13).

Other on-track equipment

• For purposes of the MVC, *other on-track equipment* "means any car, rolling stock, or other device that, alone or coupled to another device, is operated on stationary rails." MCL 257.36c.

Operate

• For purposes of the MVC, *operate* or *operating* means:

"1 or more of the following:

"(a) Being in actual physical control of a vehicle. [MCL 257.35a(a)] applies regardless of whether or not the person is licensed under [the MVC] as an operator or chauffeur.

(b) Causing an automated motor vehicle to move under its own power in automatic mode upon a highway or street regardless of whether the person is physically present in that automated motor vehicle at that time. [MCL 257.35a(b)] applies regardless of whether the person is licensed under [the MVC] as an operator or chauffeur." MCL 257.35a.

 For purposes of MCL 257.602b, *operate* means "to drive or assume physical control of a motor vehicle on a public way, street, road, or highway, including operation while temporarily stationary because of traffic, road conditions, a traffic light, or a stop sign. Operate does not include a motor vehicle that is lawfully parked." MCL 257.602b(12)(c).

Operating while intoxicated

• For purposes of MCL 257.625, *operating while intoxicated* means "any of the following:

(a) The person is under the influence of alcoholic liquor, a controlled substance, or other intoxicating substance or a combination of alcoholic liquor, a controlled substance, or other intoxicating substance.

(b) The person has an alcohol content of 0.08 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine or, beginning 5 years after the state treasurer publishes a certification under [MCL 257.625(28)], the person has an alcohol content of 0.10 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine.

(c) The person has an alcohol content of 0.17 grams or more per 100 milliliters of blood, per 210 liters of breath, or per 67 milliliters of urine." MCL 257.625(1).

Operating while intoxicated offense

• For purposes of MCL 764.9c, *operating while intoxicated offense* "means a violation of any of the following:

(*i*)... MCL 257.625 [or MCL] 257.625m.

(*ii*) A local ordinance substantially corresponding to a violation listed in [MCL 764.9c(9)(b)(i)].

(*iii*) A law of an Indian tribe substantially corresponding to a violation listed in [MCL 764.9c(9)(b)(i)].

(*iv*) A law of another state substantially corresponding to a violation listed in [MCL 764.9c(9)(b)(*i*)].

(*v*) A law of the United States substantially corresponding to a violation listed in [MCL 764.9c(9)(b)(*i*)]." MCL 764.9c(9)(b).

Operator

• For purposes of the MVC, *operator* means "a person, other than a chauffeur, who does either of the following:

(a) Operates a motor vehicle upon an highway or street.

(b) Operates an automated motor vehicle upon a highway or street." MCL 257.36.

Other licensed or certified substance use disorder professional

 For purposes of MCL 257.625b(5), other licensed or certified substance use disorder professional "means an individual or organization licensed or credentialed in [Michigan] to treat substance use disorders, including individuals certified by the Michigan certification board for addiction professionals and individuals who have training in providing assessments for alcohol dependency." MCL 257.625b(5).

Owner

• For purposes of the MVC, *owner* means any of the following:

"(a) Any person, firm, association, or corporation renting a motor vehicle or having the exclusive use thereof, under a lease or otherwise, for a period that is greater than 30 days.

(b) Except as otherwise provided in [MCL 257.401a], a person who holds the legal title of a vehicle.

(c) A person who has the immediate right of possession of a vehicle under an installment sale contract." MCL 257.37.

Ρ

Parking

• For purposes of the MVC, parking means "standing a vehicle, whether occupied or not, upon a highway, when not loading or unloading except when making necessary repairs." MCL 257.38.

Parking violation notice

For purposes of the MVC, *parking violation notice* means "a notice, other than a citation, directing a person to appear at a parking violations bureau in the city, village, or township in which, or of the college or university for which, the notice is issued and to pay the fine and costs, if any, prescribed by ordinance for the parking or standing of a motor vehicle in violation of the ordinance." MCL 257.742(10)(a).

Participant

- For purposes of MCR 4.101(F)(4), *participant* is defined in MCR 2.407(A)(1). MCR 2.407(A)(1) states that *participants* "include, but are not limited to, parties, counsel, and subpoenaed witnesses, but do not include the general public."
- For purposes of Chapter 10C of the Revised Judicature Act of 1961, *participant* "means a juvenile who is admitted into a juvenile mental health court." MCL 600.1099b(g).

Participating fleet

• For purposes of the MVC, *participating fleet* "means any of the following:

(a) Vehicles that are equipped with automated driving systems that are operating on the public roads and highways of [Michigan] in a SAVE project as provided in [MCL 257.665b].

(b) Vehicles that are supplied or controlled by a motor vehicle manufacturer, and that are equipped with

automated driving systems that are operating on the public roads and highways of [Michigan] in an ondemand automated motor vehicle network." MCL 257.2b(14).

Passenger area

 For purposes of the MVC, *passenger area* means "the area designated to seat the operator and passengers of a motor vehicle while it is in operation and any area that is readily accessible to the operator or a passenger while in his or her seating position, including the glove compartment." MCL 257.624a(6)(b).

Pedestrian

• For purposes of the MVC, *pedestrian* means "any person afoot. Pedestrian includes an individual with a mobility disability who is using a power-driven mobility device." MCL 257.39.

Person

• For purposes of the MVC, *person* means "very natural person, firm, copartnership, association, or corporation and their legal successors." MCL 257.40.

Personal vehicle

• For purposes of the Michigan Vehicle Code, *personal vehicle* "means that term as defined in [MCL 257.2102]." MCL 257.40c. MCL 257.2102(h) provides that *personal vehicle* "means a motor vehicle with a seating capacity of 8 passengers or fewer, including the driver, that is used by a transportation network company driver that satisfies both of the following:

(*i*) The vehicle is owned, leased, or otherwise authorized for use by the transportation network company driver.

(*ii*) The vehicle is not a taxicab, limousine, or commercial vehicle."

Portable signal preemption device

• For purposes of MCL 257.616a, *portable signal preemption device* "means a device that, if activated by a person, is capable of changing a traffic control signal to green out of sequence." MCL 257.616a(5)(a).

Power-driven mobility device

• For purposes of the MVC, *power-driven mobility device* "means a mobility device powered by a battery, fuel, or other engine and used by an individual with a mobility disability for the purpose of locomotion. Notwithstanding any other provisions of [the MVC], the requirements of [the MVC] apply to a power-driven mobility device while that device is being operated on a street, road, or highway in this state." MCL 257.43c.

Preliminary chemical breath analysis

• For purposes of the MVC, *preliminary chemical breath analysis* "means the on-site taking of a preliminary breath test from the breath of a person for the purpose of detecting the presence of any of the following within the person's body:

(a) Alcoholic liquor.

(b) A controlled substance, as that term is defined in . . . MCL 333.7104.

(c) Any other intoxicating substance, as that term is defined in [MCL 257.625].

(d) Any combination of the substances listed in subdivisions (a) to (c)." MCL 257.43a.

Preliminary oral fluid analysis

• For purposes of the MVC, *preliminary oral fluid analysis* "means the on-site taking of a preliminary oral fluid test, performed by a certified drug recognition expert . . . from the oral fluid of a person for the purpose of detecting the presence of a controlled substance" MCL 257.43b.

Present

• For purposes of MCL 257.627c, *present* "means located in proximity to a roadway that is not protected by a guardrail or barrier." MCL 257.627c(10).

Prior conviction

• For purposes of MCL 257.625, *prior conviction* means "a conviction for any of the following, whether under a law of [Michigan], a local ordinance substantially corresponding to a law of [Michigan], a law of the United States substantially corresponding to a law of [Michigan], or a law

of another state substantially corresponding to a law of [Michigan], subject to [MCL 257.625(27)⁵]:

(*i*) Except as provided in [MCL 257.625(26)⁶], a violation or attempted violation of any of the following:

(A) [MCL 257.625], except a violation of [MCL 257.625(2)], or a violation of any prior enactment of [MCL 257.625] in which the defendant operated a vehicle while under the influence of intoxicating or alcoholic liquor or a controlled substance, or a combination of intoxicating or alcoholic liquor and a controlled substance, or while visibly impaired, or with an unlawful bodily alcohol content.

(B) [MCL 257.625m].

(C) Former [MCL 257.625b].

(*ii*) Negligent homicide, manslaughter, or murder resulting from the operation of a vehicle or an attempt to commit any of those crimes.

(*iii*) [MCL 257.601d] or [MCL 257.626(3) or MCL 257.626(4)]." MCL 257.625(25)(b).

• For purposes of MCL 257.904d, prior conviction means:

(a) Subject to [MCL 257.904d(9) and MCL 257.904d(10)], 'prior conviction' means a conviction for any of the following, whether under a law of [Michigan], a local ordinance substantially corresponding to a law of [Michigan], or a law of another state substantially corresponding to a law of [Michigan]:

(*i*) Except as otherwise provided in [MCL 257.904d(10)], a violation or attempted violation of any of the following:

(A) [MCL 257.625], except a violation of [MCL 257.625(2)], or a violation of any prior enactment of [MCL 257.625] in which the defendant operated a vehicle while under the influence of intoxicating or

⁵MCL 257.625(27) states that "[i]f 2 or more convictions described in [MCL 257.625(25)] are convictions for violations arising out of the same transaction, only 1 conviction must be used to determine if the person has a prior conviction."

⁶MCL 257.625(26) states that "[e]xcept for purposes of the enhancement described in [MCL 257.625(12)(b)], only 1 violation or attempted violation of [MCL 257.625(6)], a local ordinance substantially corresponding to [MCL 257.625(6)], or a law of another state substantially corresponding to [MCL 257.625(6)] may be used as a prior conviction."

alcoholic liquor or a controlled substance, or a combination of intoxicating or alcoholic liquor and a controlled substance, or while visibly impaired, or with an unlawful bodily alcohol content.

(B) [MCL 257.625m].

(C) Former [MCL 257.625b].

(*ii*) Negligent homicide, manslaughter, or murder resulting from the operation of a vehicle or an attempt to commit any of those crimes.

(*iii*) Beginning October 31, 2010, a violation of [MCL 257.601d] or [MCL 257.626(3) or MCL 257.626(4)]." MCL 257.904d(8)(a).

"If 2 or more convictions described in [MCL 257.904d(8)(a)] are convictions for violations arising out of the same incident, only 1 conviction shall be used to determine whether the person has a prior conviction." MCL 257.904d(9).

"Only 1 violation or attempted violation of [MCL 257.625(6)], a local ordinance substantially corresponding to [MCL 257.625(6)], or a law of another state substantially corresponding to [MCL 257.625(6)] may be used as a prior conviction." MCL 257.904d(10).

Prior judgment

• For purposes of MCL 436.1703, *prior judgment* "means a conviction, juvenile adjudication, finding of responsibility, or admission of responsibility for any of the following, whether under a law of this state, a local ordinance substantially corresponding to a law of this state, a law of the United States that substantially corresponds to a law of this state, or a law of another state that substantially corresponds to a law of this state:

(*i*) [MCL 436.1703⁷] or [MCL 436.1701] or [MCL 436.1707].

(*ii*) . . . MCL 257.624a, [MCL] 257.624b, [or MCL] 257.625.

(*iii*) . . . MCL 324.80176, [MCL] 324.81134, [or MCL] 324.82127.

⁷ Misdemeanor violations of MCL 436.1703(1) that are successfully deferred, discharged, and dismissed under MCL 436.1703(3) are still considered prior judgments for purposes of MCL 436.1703(1)(c). MCL 436.1703(4).

(*iv*) . . . MCL 750.167a [or MCL] 750.237." MCL 436.1703(17)(d).

Program

• For purposes of MCL 600.1084, *program* "means the specialty court interlock program created under [MCL 600.1084]." MCL 600.1084(9)(c).

Prosecuting attorney

• For purposes of the MVC, *prosecuting attorney*, "except as the context otherwise requires, means the attorney general, the prosecuting attorney of a county, or the attorney representing a local unit of government." MCL 257.45a.

Provider

For purposes of MCL 257.625b(5), provider "means an individual with prescribing authority under the public health code, . . . MCL 333.1101 to [MCL] 333.25211, who regularly communicates with the treatment team during the defendant's recovery and who has training or experience that demonstrates the provider's ability to treat and manage patients with alcohol dependency." MCL 257.625b(5).

Public transit vehicle

• For purposes of MCL 257.698(5)(j), *public transit vehicle* means "a motor vehicle, other than a station wagon or passenger van, with a gross vehicle weight rating of more than 10,000 pounds." MCL 257.698(5)(j).

Public school academy

 For purposes of MCL 257.682, public school academy "means a public school academy established under part 6a [of the Revised School Code (RSC)] and, except as used in part 6a [of the RSC], also includes an urban high school academy established under part 6c [of the RSC], a school of excellence established under part 6e [of the RSC], and a strict discipline academy established under [MCL 380.1311b to MCL 380.1311m]." MCL 257.682(9)(d); MCL 380.5(7).

R

Record of sale

 For purposes of MCL 257.252a(1), record of sale means "either a photocopy of the reassigned title or a form or document that includes the name, address, driver license number, and signature of the person to whom the vehicle is sold and the purchase price and date of sale of the vehicle." MCL 257.240(2)(b).

Registration

• For purposes of the MVC, *registration* means "a registration certificate, plate, adhesive tab, or other indicator of registration issued under [the MVC] for display on a vehicle." MCL 257.50.

Resident

• For purposes of the MVC, *resident* means "every person who resides in [Michigan] and establishes that he or she is legally present in the United States. This definition applies to the provisions of [the MVC] only." MCL 257.51a.

Revocation

• For purposes of the MVC, *revocation* means "that the operator's or chauffeur's license and privilege to operate a motor vehicle on the public highways are terminated and shall not be renewed or restored until the later of the following:

(a) The expiration of not less than 1 year after the license was revoked.

(b) The expiration of not less than 5 years after the date of a subsequent revocation occurring within 7 years after the date of a prior revocation." MCL 257.52(1).

Roadway

• For purposes of the MVC, *roadway* means "that portion of a highway improved, designed, or ordinarily used for vehicular travel. In the event a highway includes 2 or more

separate roadways, the term 'roadway[,]'[] as used herein, shall refer to any such roadway separately, but not to all such roadways collectively." MCL 257.55.

S

SAVE project

 For purposes of the MVC, SAVE project "means an initiative that authorizes eligible motor vehicle manufacturers to make available to the public on-demand automated motor vehicle networks as provided in [MCL 257.665b]." MCL 257.2b(15).

School district

 For the purposes of MCL 257.682, school district "means a general powers school district organized under [the Revised School Code], regardless of previous classification, a community district, or a school district of the first class," MCL 380.6(1); MCL 257.682(9)(d).

School bus zone

• For purposes of MCL 257.601b, *school bus zone* means "the area lying within 20 feet of a school bus that has stopped and is displaying 2 alternately flashing red lights at the same level, except as described in [MCL 257.682(2)⁸]." MCL 257.601b(5)(c).

School zone

- For purposes of MCL 257.601b, *school zone* is "that term as defined in [MCL 257.627a]." MCL 257.601b(5)(d).
- For purposes of MCL 257.627a, *school zone* means "school property on which a school building is located and the adjacent property. A school zone extends not more than 1,000 feet from the school property line in any direction. If 2 or more schools occupy the same property or adjacent properties, 1 of the following applies, as applicable:

(*i*) If the hours of instruction at the schools are the same, then a single combined school zone shall be established.

⁸ MCL 257.682(2) describes certain highways where operators are not required to stop for a school bus.

(*ii*) If the hours of instruction at the schools are different, overlapping school zones shall be established." MCL 257.627a(1)(c).

Secretary of State

• For purposes of the MVC, *secretary of state* means "the secretary of state of [Michigan], acting directly or through his [or her] duly authorized deputy, investigators, agents and employees." MCL 257.58.

Security interest

• For purposes of the MVC, *security interest* means that term "as defined in the uniform commercial code . . . [MCL 440.1101 *et seq.*]." MCL 257.58b.

Segment 1

• For purposes of the Driver Education Provider and Instructor Act, *segment 1* means "a teen driver education course that meets the requirements in [MCL 256.657]." MCL 256.627(f).

Segment 2

 For purposes of the Driver Education Provider and Instructor Act, *segment 2* means "a teen driver education course that meets the requirements in [MCL 256.659]." MCL 256.627(g).

Serious emotional disturbance

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *serious emotional disturbance* "means that term as defined in . . . [MCL 330.1100d]." MCL 600.1099b(h).

Serious mental illness

• For purposes of Chapter 10C of the Revised Judicature Act of 1961, *serious mental illness* "means that term as defined in . . . [MCL 330.1100d]." MCL 600.1099b(i).

Serious impairment of a body function

- For purposes the MVC, *serious impairment of a body function* "includes, but is not limited to, one or more of the following:
- (a) Loss of a limb or loss of use of a limb.

(b) Loss of a foot, hand, finger, or thumb or loss of use of a foot, hand, finger, or thumb.

(c) Loss of an eye or ear or loss of use of an eye or ear.

(d) Loss or substantial impairment of a bodily function.

(e) Serious visible disfigurement.

(f) A comatose state that lasts for more than 3 days.

(g) Measurable brain or mental impairment.

(h) A skull fracture or other serious bone fracture.

(i) Subdural hemorrhage or subdural hematoma.

(j) Loss of an organ." MCL 257.58c.

• For purposes of MCL 750.479a, *serious impairment of a body function* "includes, but is not limited to, 1 or more of the following:

(a) Loss of a limb or loss of use of a limb.

(b) Loss of a foot, hand, finger, or thumb or loss of use of a foot, hand, finger, or thumb.

(c) Loss of an eye or ear or loss of use of an eye or ear.

(d) Loss or substantial impairment of a bodily function.

(e) Serious visible disfigurement.

(f) A comatose state that lasts for more than 3 days.

(g) Measurable brain or mental impairment.

(h) A skull fracture or other serious bone fracture.

(i) Subdural hemorrhage or subdural hematoma.

(j) Loss of an organ." MCL 257.58c; MCL 750.479a(9)(b).

Serious misdemeanor

• For purposes of MCL 769.5, *serious misdemeanor* "means that term as defined in . . . MCL 780.811." MCL 769.5(7).

Serious offense involving a motor vehicle

• For purposes of MCL 257.321a, *serious offense involving a motor vehicle* "means a felony or misdemeanor punishable by at least 93 days in jail, during the commission of which

the individual operated a motor vehicle in a manner that presented real or potential harm to a person or property and 1 or more of the following circumstances apply to the offense:

(a) The motor vehicle was used as an instrument of the offense.

(b) The motor vehicle was used to transport a victim of the offense.

(c) The motor vehicle was used to flee the scene of the offense.

(d) The motor vehicle was necessary for the commission of the offense." MCL 257.321a(10).

Snowmobile

• For purposes of MCL 257.628b, *snowmobile* "means a vehicle that meets all of the following requirements:

(i) Is engine-driven.

(ii) Has an endless belt tread.

(iii) Was originally manufactured solely for operation over snow.

(iv) Is steered by 2 sled-type runners or skis that are in contact with the snow and the distance between the centers of the spindles of which is, except for a historic snowmobile, not more than 48 inches." MCL 257.628b(4)(b).

Snowplow

• For purposes of MCL 257.643b, *snowplow* "means a vehicle that is all of the following:

(a) Operating on behalf of this state, a county, a county road commission, or a municipality.

(b) Actively engaging in snow and ice removal, snowplowing, salting, or sanding.

(c) Operating lights as described in [MCL 257.698(5)(d)]." MCL 257.643b(5).

Social networking site

• For purposes of MCL 257.602b, *social networking site* means "any web-based service that allows individuals to construct

a profile within a founded system and communicate with other users of the site for social or amusement purposes." MCL 257.602b(12)(d).

Specialty court

• For purposes of MCL 600.1084, *specialty court* "means any of the following:

(*i*) A drug treatment court.

(ii) A DWI/sobriety court.

(*iii*) A hybrid of the programs under [MCL 600.1084(9)(d)(*i*)-(*ii*)].

(*iv*) A mental health court, as that term is defined in [MCL 600.1090].

(*v*) A veterans treatment court, as that term is defined in [MCL 600.1200]. MCL 600.1084(9)(d).

Specialty court program (or specialty court)

• For purposes of the Michigan Vehicle Code, *specialty court program* (or *specialty court*) "means any of the following:

(*i*) A drug treatment court, as that term is defined in . . . MCL 600.1060, in which the participant is an adult.

(*ii*) A DWI/sobriety court, as that term is defined in . . . MCL 600.1084.

(*iii*) A hybrid of the programs under [MCL 257.83(a)(*i*)-(a)(*ii*)].

(*iv*) A mental health court, as that term is defined in . . . MCL 600.1090.

(v) A veterans treatment court, as that term is defined in . . . MCL 600.1200." MCL 257.83(a).

Specialty court interlock program

• For purposes of the Michigan Vehicle Code, *specialty court interlock program* "means a program as that term is defined in . . . [MCL 600.1084]." MCL 257.83(b).

Standardized field sobriety test

• For purposes of the MVC, *standardized field sobriety test* "means 1 of the standardized tests validated by the

National Highway Traffic Safety Administration. A field sobriety test is considered a standardized field sobriety test under [MCL 257.62a] if it is administered in substantial compliance with the standards prescribed by the National Highway Traffic Safety Administration." MCL 257.62a.

State

• For purposes of the MVC, *state* means "any state, territory, or possession of the United States, Indian country as defined in 18 USC 1151, the District of Columbia, the Dominion of Canada, or any province or territory of the Dominion of Canada." MCL 257.65.

Stop-arm camera system

• For the purposes MCL 257.682, *stop-arm camera system* "means a system of 2 or more cameras affixed to a school bus that meets all of the following requirements:

(a) Is synchronized to automatically record video or 1 or more sequenced photographs of a vehicle that fails to stop for a school bus or passes a school bus in violation of . . . MCL 257.682.

(b) Is capable of capturing images of all of the following:

(*i*) The vehicle.

(*ii* The registration plate on the rear of the vehicle.

(*iii*) A distance of not less than 200 feet in front of the school bus.

(c) Records the date, time, and location on a video recorded or photograph captured by the system." MCL 257.682(9)e); MCL 257.1805(8).

Street

• For purposes of the MVC, *street* means "the entire width between . . . boundary lines of every way publicly maintained when any part thereof is open to the use of the public for purposes of vehicular travel." MCL 257.20; MCL 257.64.

Streetcar

• For purposes of the MVC, *streetcar* "means a car other than a railroad train for transporting persons or property

operated on stationary rails, including a streetcar operated as part of a street railway system." MCL 257.63.

Streetcar track

• For purposes of the MVC, *streetcar track* "means a stationary rail owned by a street railway that is part of a street railway system." MCL 257.63a.

Street railway

• For purposes of the MVC, *street railway* "means that term as defined [in] . . . MCL 125.4507." MCL 257.64a.

Street railway system

• For purposes of the MVC, *street railway system* "means that term as defined [in] . . . MCL 125.4507." MCL 257.64b.

Suspension

• For purposes of the MVC, *suspension* means "that the driver's license and privilege to drive a motor vehicle on the public highways are temporarily withdrawn but only during the period of the suspension." MCL 257.66.

Т

Taxicab

• For purposes of the Michigan Vehicle Code, *taxicab* "means that term as defined in [MCL 257.2102]." MCL 257.67c. MCL 257.2102(i) provides that *taxicab* "means a motor vehicle with a seating capacity of 8 passengers or fewer, including the driver, that is equipped with a roof light and that carries passengers for a fee usually determined by the distance traveled. Taxicab does not include a commercial vehicle."

Taxicab driver

• For purposes of the Michigan Vehicle Code, *taxicab driver* "means that term as defined in [MCL 257.2102]." MCL 257.67d. MCL 257.2102(k) provides that *taxicab driver* "means an individual who uses a *taxicab* to provide transportation services to potential passengers."

Temporary traffic control device

• For the purposes of MCL 257.79d, *temporary traffic control device* "means a traffic control device that is installed for a limited time period during construction, maintenance, public utility work, reconstruction, repair, resurfacing, or surveying as described in [MCL 257.79d(1)]." MCL 257.79d(2).

Traffic control device

- For purposes of the MVC, *traffic control device* means "all signs, signals, markings, and devices not inconsistent with [the Michigan Vehicle Code] placed or erected by authority of a public body or official having jurisdiction, for the purpose of regulating, warning or guiding traffic." MCL 257.70.
- For purposes of MCL 750.377d, traffic control device "means a sign, signal, electronic traffic control sign or signal, marking, light post, railroad sign or signal, or device not inconsistent with the [MVC], placed or erected by authority of a public body or official having jurisdiction, for the purpose of regulating, warning, or guiding traffic, maintaining highway safety, or providing information to motor vehicle operators." MCL 750.377d(3).

Traffic offense

• For purposes of MCL 762.11, *traffic offense* "means a violation of [the MVC], or a violation of a local ordinance substantially corresponding to [the MVC], that involves the operation of a vehicle and, at the time of the violation, is a felony or a misdemeanor." MCL 762.11(7)(b).

Trailway municipal civil infraction

• For purposes of the Revised Judicature Act, *trailway municipal civil infraction* "means a civil infraction involving the operation of a vehicle on a recreational trailway at a time, in a place, or in a manner prohibited by ordinance." MCL 600.113(1)(g).

Transportation network company

 For purposes of the Michigan Vehicle Code, transportation network company "means that term as defined in [MCL 257.2102]." MCL 257.76a. MCL 257.2102(*I*) provides that transportation network company "means a person operating in [Michigan] that uses a digital network to connect transportation network company riders to transportation network company drivers who provide transportation network company prearranged rides. Transportation network company does not include a taxi service, transportation service arranged through a transportation broker, ridesharing arrangement, or transportation service using fixed routes at regular intervals."

Transportation network company digital network

• For purposes of the Michigan Vehicle Code, *digital network* "means that term as defined in [MCL 257.2102]." MCL 257.12c. Although MCL 257.2102 does not define *digital network*, MCL 257.2102(m) defines *transportation network company digital network* to mean "an online-enabled application, website, or system offered or utilized by a transportation network company that enables the prearrangement of rides with transportation network company drivers."

Transportation network company driver

• For purposes of the Michigan Vehicle Code, *transportation network company driver* "means that term as defined in [MCL 257.2102]." MCL 257.76b. MCL 257.2102(n) provides that *transportation network company driver* "means an individual who satisfies all of the following:

(*i*) Receives connections to potential passengers and related services from a transportation network company in exchange for payment of a fee to the transportation network company.

(*ii*) Uses a personal vehicle to offer or provide transportation network company prearranged rides to transportation network company riders upon connection through a digital network controlled by a transportation network company in return for compensation or payment of a fee."

Transportation network company prearranged ride

• For purposes of the Michigan Vehicle Code, *transportation network company prearranged ride* "means that term as defined in [MCL 257.2102]." MCL 257.76c. MCL 257.2102(o) provides that *transportation network company prearranged ride* "means the provision of transportation by a transportation network company driver to a transportation network company rider, beginning when a transportation network company driver accepts a ride requested by a transportation network company rider through a digital network controlled by a transportation network company, continuing while the transportation network company driver transports the requesting transportation network company rider, and ending when the last requesting transportation network company rider departs from the personal vehicle. Transportation network company prearranged ride does not include a shared-expense carpooling or vanpooling arrangement or transportation provided using a taxicab, limousine, or other vehicle."

Transportation network company rider

• For purposes of the Michigan Vehicle Code, *transportation network company rider* "means that term as defined in [MCL 257.2102]." MCL 257.76d. MCL 257.2102(p) provides that *transportation network company rider* "means an individual who uses a transportation network company's digital network to connect with a transportation network company driver who provides a transportation network company prearranged ride to the transportation network company rider in the transportation network company driver's personal vehicle between points chosen by the transportation network company rider."

U

Upfitter

 For purposes of the MVC, *upfitter* "means a person that modifies a motor vehicle after it was manufactured by installing an automated driving system in that motor vehicle to convert it to an automated motor vehicle. Upfitter includes a subcomponent system producer recognized by the secretary of state that develops or produces automated driving systems." MCL 257.2b(16).

Use

• For purposes of MCL 257.602c(1), and in addition to the requirements of MCL 257.602b, *use* means "to initiate a call, answer a call, or listen to or engage in verbal communication through the cellular telephone." MCL 257.602c(1).

Use a mobile electronic device

- For purposes of MCL 257.602b other than MCL 257.602b(2), *use a mobile electronic device* means "using a mobile electronic device to do any task, including, but not limited to, any of the following:
 - (*i*) Send or receive a telephone call.

(*ii*) Send, receive, or read a text message.

(*iii*) View, record, or transmit a video.

(*iv*) Access, read, or post to a social networking site." MCL 257.602b(13).

• For purposes of MCL 257.602b(2), *use a mobile electronic device* means "1 or more of the following:

(a) using a mobile electronic device to do any task, including, but not limited to, any of the following:

(*i*) Send or receive a telephone call.

(*ii*) Send, receive, or read a text message.

(*iii*) View, record, or transmit a video.

(*iv*) Access, read, or post to a social networking site.

(b) Reaching for a mobile electronic device in a manner that requires a driver to maneuver so that the driver is no longer in a seated driving position, restrained by a seat belt that is installed as required by 49 CFR 393.93 and adjusted in accordance with the vehicle manufacturer's instructions." MCL 257.602b(2).

V

Vehicle

 For purposes of the MVC, *vehicle* means "every device in, upon, or by which any person or property is or may be transported or drawn upon a highway, except devices exclusively moved by human power or used exclusively upon stationary rails or tracks and except, only for the purpose of titling and registration under [the MVC], a mobile home as defined in [MCL 125.2302]." MCL 257.79.

Vehicle immobilization

 For purposes of MCL 257.904d, vehicle immobilization means "requiring the motor vehicle involved in the violation immobilized in a manner provided in [MCL 257.904e]." MCL 257.904d(8)(b).

Vehicular access point

• For purposes of MCL 257.627, *vehicular access point* means "a driveway or intersecting roadway." MCL 257.627(18)(b).

Videoconferencing

• For purposes of Subchapter 2.400 of the Michigan Court Rules, *videoconferencing* "means the use of an interactive technology, including a remote digital platform, that sends video, voice, and/or data signals over a transmission circuit so that two or more individuals or groups can communicate with each other simultaneously using video codecs, monitors, cameras, audio microphones, and audio speakers. It includes use of a remote video platform through an audio-only option." MCR 2.407(A)(2).

W

Work zone

- For purposes of the MVC, *work zone* "means a portion of a street or highway open to vehicular traffic and adjacent to a barrier, berm, lane, or shoulder of a street or highway within which construction, maintenance, public utility work, reconstruction, repair, resurfacing, or surveying is being conducted by 1 or more individuals and that meets any of the following:
 - (a) Is between both of the following:
 - (*i*) A sign notifying the beginning of work.

(*ii*) An 'end road work' sign or, if no sign is posted, the last temporary traffic control device before the normal flow of traffic resumes.

(b) Is between a 'begin work convoy' sign and an 'end work convoy' sign.

(c) If a moving or stationary vehicle or equipment exhibiting a rotating beacon or strobe light is used, is between both of the following points:

(*i*) A point that is 150 feet behind the rear of the vehicle or equipment or that is the point from which the beacon or strobe light is first visible on the street or highway behind the vehicle or equipment, whichever is closer to the vehicle or equipment.

(*ii*) A point that is 150 feet in front of the front of the vehicle or equipment or that is the point from which the beacon or strobe light is first visible on the street or highway in front of the vehicle or equipment, whichever is closer to the vehicle or equipment." MCL 257.79d.

Subject Matter Index

A

Attempts 1-3 general statute 1-5 MVC provisions 1-3

С

Chemical tests admissibility 9-57 advice 9-53 blood 9-48 breath 9-48 constitutional issues 9-58 disclosure 9-56 implied consent 9-52 refusal 9-53 requirements for sample collection 9-56 submission 9-53 urine 9-48 Citation defendant's options 1-19 admit responsibility with explanation 1-20 admit responsibility without explanation 1-20 deny responsibility 1-21 failure to answer 1-28 misdemeanor 1-42 sworn complaint 1-19 technical requirements 1-12 contents 1-12 signed under oath 1-14 **Civil infraction** actions time guidelines 1-10 assessments 1-30 waiver 1-36 authority to conduct 1-10 district court 1-10 magistrates 1-11

municipal judge 1-10 civil fines 1-30 waiver 1-36 contempt proceedings 1-38 costs 1-30 waiver 1-36 default judgment 1-28 setting aside judgment 1-29 education programs 1-39 formal hearings 1-26 appeals 1-27 determining responsibility 1-27 discovery 1-26 notice 1-27 procedure 1-26 representation 1-26 subpoenas 1-27 holding or using a mobile electronic device while driving 4-47 informal hearings 1-22 appeals 1-24 conducting 1-23 determining responsibility 1-24 discovery 1-23 notice 1-23 representation 1-23 subpoenas 1-23 initiating an action 1-15 authority to issue citation 1-15 filing citation or parking notice violation 1-17 serving citation or parking notice violation 1-17 jurisdiction juveniles 2-3 moving careless driving 4-2 failing to remove motor vehicle from roadway after an accident 4-4 failing to stop for school bus 4-5 failure to yield 4-21 following a fire truck too closely 4-8 interference with control of vehicle 4-14 interference with operation of vehicle 4-14 interference with view of vehicle 4-14 Kelsey's Law 4-53 negligent driving 4-2 overtaking and passing 4-15 bicycles 4-16 vehicles 4-15 permitting minor to ride in pickup truck bed 4-18 railroad crossings 4-19

```
right-of-way 4-21
   signaling 4-51
   signals 4-42
   signs 4-42
   speed violations 4-26
       citation requirements 4-35
       evidence 4-35
          radar, vascar, or laser 4-35
          witness 4-35
       statutory authority
          general speed limit 4-33
          lawful highway speeds 4-27
          maximum and minimum speed limits 4-33
          modified speed limits 4-34
   stop and go 4-42
   turning 4-51
   using a cellular telephone while operating 4-53
   wrong side 4-55
   wrong way 4-55
nonmoving
   abandoning a vehicle 3-14
   equipment violations 3-19
   failing to assure title transfer upon sale 3-8
   failing to change address on registration or title 3-8
   failure to change address on license 3-2
   invalid or no registration plate 3-9
   no proof of insurance 3-10
   no proof of registration 3-12
   operating in violation of graduated liensing requirements 3-3
   parking 3-22
   registration plate violations 3-13
   safety belt violations 3-15
   soliciting business at the scene of a motor vehicle accident or disabled vehi-
          cle 3-30
   standing 3-22
   stopping 3-22
nonresidents 1-19
rehabilitation programs 1-39
treatment programs 1-39
```

D

Delayed Sentencing 1-6 Driver improvement course 1-78 cost 1-81 eligibility 1-80 secretary of state responsibilities 1-78 Drug recognition evaluator

testimony 9-63 DWI/sobriety court 9-13

F

Felony costs 1-58 felony in which a motor vehicle was used 1-59 general criminal penalties 1-55 initiating an action 1-54 jurisdiction 1-53 minimum costs 1-57 moving failure to stop at signal of police officer 8-2 failure to use due care and caution causing injury or death to emergency response personnel 8-7 fleeing and eluding 8-2 leaving the scene of an accident resulting in death 8-9 leaving the scene of an accident resulting in serious impairment of a body function 8-9 malicious destruction of trees, shrubs, plants, or soil with a motor vehicle 8-27 manslaughter 8-29 moving violation causing death in school bus zone 8-15 moving violation causing death in work zone 8-15 moving violation causing death of operator of implement of husbandry 8-16 operating a motor vehicle while unlicensed 8-17 operating a motor vehicle with suspended or revoked license 8-17 reckless driving causing death 8-22 reckless driving causing serious impairment of a body function 8-25 unlawfully driving away an automobile 8-31 nonmoving allowing a person whose license is suspended or revoked to operate a motor vehicle and causing serious impairment of a body function or death 7 - 2allowing an unlicensed person to drive and causing serious impairment of body function or death 7-2 altering documents 7-4 buying a portable signal preemption device 7-5 false certification 7-8 false statement in application for certificate of title 7-9 false statement in assignment of title 7-9 falsifying documents 7-4 forging documents 7-4 odometer tampering 7-11 offering for sale a stolen, false, or counterfeit certificate of insurance 7-13 police officer knowingly making materially false statement in citation 7-12 police officer perjury 7-12

possessing a stolen, false, or counterfeit certificate of insurance 7-13 possession of stolen vehicle 7-9 selling a portable signal preemption device 7-5 selling a stolen, false, or counterfeit certificate of insurance 7-13 using a portable signal preemption device 7-5
Felony sentencing 1-55 calculating the minimum sentencing range 1-56 departures 1-57
Field sobriety tests testimony 9-63

I

Ignition interlock device 9-46 vehicle impoundment 9-47 Investigative stops 1-9

J

Jurisdiction civil infraction 1-9 **Juveniles** age calculation 2-2costs 2-11 court rules 2-3 crime victim's rights assessment 2-11 custody 2-4 investigative stops 2-4 jurisdiction civil infraction 2-3 criminal traffic violations 2-2 continued jurisdiction 2-2 procedures failure to respond to a motor vehicle violation 2-9 Michigan Penal Code 2-5 traffic cases 2-5 commencing proceedings 2-6 notice 2-7 felony in which a motor vehicle was used 2-7 preliminary inquiries and hearings 2-8 restitution 2-11 setting aside traffic adjudications 2-11 Traffic cases Holmes Youthful Trainee Act 2-12 venue 2-3

L

Licensing sanctions

abstracts 1-62 authority 1-61 license denial 1-65 license revocation 1-66 license suspension 1-67 appeal to secretary of state 1-71 improper suspension 1-71 mandaotry suspension for failure to answer citation/notice to appear or failure to comply with order/judgment 1-68 mandatory suspension under 319 1-67 mandatory suspension under 904 1-70 notice to individual 1-68 notice to secretary of state 1-69 multiple sanctions for the same offense 1-61 nonresidents 1-71 out-of-state offenses 1-71 points 1-64 vehicle forfeiture 1-76 vehicle immobilization 1-72

M

```
Misdemeanor
   arraignment 1-47
   citation
       accepting a plea 1-44
       failure to appear 1-45
       requirements 1-43
       restrictions on issuance 1-46
   costs 1-50
   criminal penalty 1-48
   failure to comply with sentence 1-52
   initiating action by citation 1-42
   jurisdiction 1-40
       magistrates 1-41
   minimum costs 1-50
   moving
       disobeying the direction of a police officer who is regulating traffic 6-2
       drag racing 6-3
       driving in violation of a restricted license 6-4
       driving or moving an unregistered vehicle 6-5
       driving with an invalid license 6-6
       failing to stop for school crossing guard 6-7
       failing to yield to handicapped individual 6-8
      improper operation of a golf cart 6-8
```

improper passing of a stationary emergency vehicle 4-9 improper use of emergency lights 4-11 improper use of emergency vehicle 6-11 joyriding 6-26 moving violation causing death 6-17 moving violation causing injury to person in school bus zone 6-13 moving violation causing injury to person in work zone 6-13 moving violation causing injury to person operating farm equipment 6-14 moving violation causing serious impairment of a body function 6-15 operating a motor vehicle while unlicensed 6-21 operating a motor vehicle without a license in possession 6-22 operating motor vehicle while license is suspended or revoked 6-18 reckless driving 6-23 reckless driving causing miscarriage, stillbirth, or death 6-25 unlawful use of an automobile without intent to steal 6-26 nonmoving allowing another person to operate a motor vehicle with a revoked license 5-10 allowing another person to operate a motor vehicle with a suspended license 5-10 allowing another to drive or move an unregistered vehicle 5-16 allowing another to operate in violation of the MVC 5-10 altering a license or using such license with intent to commit a misdemeanor 5-14 altering certificate of title or using such certificate with intent to commit a crime 5-20 causing or permitting an unlicensed minor to drive 3-2 counterfeiting a license or using such license with intent to commit a misdemeanor 5-14 counterfeiting certificate of title or using such certificate with intent to commit a crime 5-20 duplicating a license or using such license with intent to commit a misdemeanor 5-14 duplicating certificate of title or using such certificate with intent to commit a crime 5-20 failing to answer citation 5-22 failing to apply for registration and certificate of title 5-17 failing to give information and aid at the scene of an accident 5-3 failing to report accident involving death 5-5 failing to report accident involving personal injury 5-5 failing to report accident involving property damage of more than one thousand dollars 5-5 failing to report an accident 5-9 failing to report an accident resulting in personal injury 5-7 failing to report an accident resulting in vehicle damage 5-8 failing to transfer title 5-18 failure to appear in court 5-22 failure to comply with an order or judgment 5-22

failure to disclose odometer mileage 5-23 falsifying a citation 5-24 forging a license or using such license with intent to commit a misdemeanor 5 - 14forging certificate of title or using such certificate with intent to commit a crime 5-20 forging proof of insurance 5-19 garage or repair shop failing to report evidence of accident 5-6 garage or repair shop failing to report evidence of bullet 5-6 improper use of diabled person identification 5-24 improperly disposing of a citation 5-24 leaving the scene of an accident 5-9 leaving the scene of an accident resulting in damage to fixtures that are upon or adjacent to a highway 5-6 leaving the scene of an accident resulting in personal injury 5-7 leaving the scene of an accident resulting in vehicle damage 5-8 minor consuming alcoholic liquor 5-28 furnishing fraudulent identification to purchase alcoholic liquor 5-28 having any bodily alcohol content 5-28 possessing alcoholic liquor 5-28 purchasing alcoholic liquor 5-28 using fraudulent identification to purchase alcoholic liquor 5-29 possessing a counterfeit license 5-12 possessing a duplicate license 5-12 possessing a forged license 5-12 possessing a reproduced license 5-12 possessing alcohol in a motor vehicle by a person less than 21 years of age 3-31. 5-27 possessing an altered license 5-12 possessing open alcohol in a motor vehicle 5-34 producing false evidence of motor vehicle insurance 5-19 reporting a false address change to the secretary of state 5-13 reproducing a license or using such license with intent to commit a misdemeanor 5-14 reproducing certificate of title or using such certificate with intent to commit a crime 5-20 temporary registration violations 5-20 traffic control device 5-26 transporting alcohol in a motor vehicle by a person less than 21 years of age 3-31, 5-27 transporting open alcohol in a motor vehicle 5-34 unlawful display of license 5-14 unlawful lending or use of certificate 5-21 unlawful lending or use of permit 5-21 unlawful lending or use of plate 5-21 unlawful lending or use of registration 5-21 unlawful lending or use of title 5-21

unlawful use of license 5-14 rebuttable presumption regarding jail sentence 1-49 rebuttable presumption regarding probation sentence 1-49 summons 1-47

0

```
Obtaining a vehicle to circumvent sanctions 9-64
   unlawful title acquisition 9-64
Operating while intoxicated 9-15
   causing death 9-28
   causing serious impairment of a body function 9-33
   DWI/sobriety court 9-13
   jury instructions 9-15
   penalties 9-16
   sanctions 9-17
   special procedures 9-2
      advice to accused before accepting plea 9-4
      appeals to circuit court 9-13
      arraignment 9-3
      final adjudication 9-5
      issuance of a restricted license 9-7
      order screening and assessment for rehabilitative services 9-5
      pleas 9-4
      pretrial conference 9-3
      warrantless arrest 9-2
   specialty court interlock program 9-13
   statutory authority 9-15
Operating while visibly impaired 9-24
   causing death 9-28
   causing serious impairment of a body function 9-33
   special procedures 9-2
      advice to accused before accepting plea 9-4
      appeals to circuit court 9-13
      arraignment 9-3
      final adjudication 9-5
      issuance of a restricted license 9-7
      order screening and assessment for rehabilitative services 9-5
      pleas 9-4
      pretrial conference 9-3
      warrantless arrest 9-2
```

Ρ

Parking violations

technical requirements 1-12 notice 1-14 **People v Konopka, 309 Mich App 345 (2015)** 1-58

Pleas

under advisement 1-8 felony cases 1-8 misdemeanor 1-8 traffic exception 1-8 **Preliminary chemical breath analysis** administrative rules 9-49 admissibility 9-50 arrest 9-50 refusal 9-48 submission 9-48 **preliminary chemical breath analysis** 9-48 **Probation** 1-6

R

Registration denial 1-78 Restitution 1-8 Roadside drug testing 9-61 authority to arrest 9-62 preliminary oral fluid analysis 9-61 admissibility 9-62

S

Section 625 offenses

```
child endangerment 9-38
   permitting or authorizing certain individuals to drive 9-23
   special procedures 9-2
       advice to accused before accepting plea 9-4
       appeals to circuit court 9-13
       arraignment 9-3
      final adjudication 9-5
      issuance of a restricted license 9-7
       order screening and assessment for rehabilitative services 9-5
       pleas 9-4
      pretrial conference 9-3
      warrantless arrest 9-2
   zero tolerance offenses
      controlled substances 9-41
   zero tolerance violations
       persons less than 21 years of age 9-35
Specialty court interlock program 9-13
Summary proceedings
   filings by incarcerated individuals 1-3
```

Т

Transferring a vehicle to circumvent sanctions 9-64

transfer of vehicle subject to sanctions to person exempt from use tax 9-65 unlawful title acquisition 9-64 unlawful transfer to avoid forfeiture 9-65

V

Venue

civil infraction 1-10

Table of Authorities Index Traffic Benchbook - Fourth Edition

Tables of Authority

Michigan Statutes Michigan Rules of Evidence Constitutional Authority Michigan Criminal Jury Instructions Table of Authorities Index Traffic Benchbook - Fourth Edition

Cases

A

Ahmed v Tokio Marine America Ins Co, 337 Mich App 1 (2021) 6-28 Alleyne v United States, 570 US 99 (2013) 1-57 Almendarez–Torres v United States, 523 US 224 (1998) 1-57 Apprendi v New Jersey, 530 US 466 (2000) 1-56

B

Birchfield v North Dakota, 579 US 438 (2016) 9-58

С

California v Byers, 402 US 424 (1971) 8-14 *City of Holland v Dreyer,* 184 Mich App 237 (1990) 9-20 *City of Monroe v Jones,* 259 Mich App 443 (2003) 3-26 *City of Plymouth v McIntosh,* 291 Mich App 152 (2010) 1-44 *Cole v Eckstein,* 202 Mich App 111 (1993) 4-35

D

Dunn v Detroit Inter-Ins Exch, 254 Mich App 256 (2002) 1-iv

F

Flanagin v Kalkaska Co Rd Comm, 319 Mich App 633 (2017) 6-13 *Ford Motor Credit Co v Detroit,* 254 Mich App 626 (2003) 3-28

G

Graham v Foster, 500 Mich 23 (2017) 1-iv

Η

Hamilton v Gordon, 135 Mich App 289 (1984) 3-3 *Hicks v Bacon,* 26 Mich App 487 (1970) 4-35

I

In re Hague, 412 Mich 532 (1982) 1-iv *In re Nale Estate,* 290 Mich App 704 (2010) 8-30 *In re Wilson,* 113 Mich App 113 (1982) 2-8 Table of Authorities: Cases Traffic Benchbook - Fourth Edition

K

Kansas v Glover, ____ US ____ (2020) 6-20

L

Landon v Titan Ins Co, 251 Mich App 633 (2002) 6-27, 8-32

M

Mason v Wurth, 181 Mich App 129 (1989) 3-27 *Missouri v McNeely,* 569 US 141 (2013) 9-59 *Mitchell v Wisconsin,* 588 US (2019) 9-60 *Monaco v Home-Owners Ins Co,* 317 Mich App 738 (2016) 5-10

Ν

Neander v Clampett, 344 Mich 292 (1955) 4-25 *New Jersey v TLO,* 469 US 325 (1985) 2-4

0

Oxendine v Secretary of State, 237 Mich App 346 (1999) 9-22, 9-27

P

People of the City of Auburn Hills v Mason, ____ Mich App ____ (2024) 1-49 *People v Acosta-Baustista*, 296 Mich App 404 (2012) 6-20, 8-19, 8-21 *People v Beasley*, 239 Mich App 548 (2000) 9-59 People v Bensch, 328 Mich App 1 (2019) 1-6 People v Bergman, 312 Mich App 471 (2015) 8-13, 8-21, 9-31, 9-33 *People v Blondia*, 69 Mich App 554 (1976) 4-39 *People v Bonilla-Machado*, 489 Mich 412 (2011) 1-56 People v Boscaglia, 419 Mich 556 (1984) 7-11 People v Boughner, 209 Mich App 397 (1995) 9-50 *People v Bowden,* ____ Mich App ____ (2022) 9-63 *People v Boynton*, 185 Mich App 669 (1990) 1-6 People v Branch, 202 Mich App 550 (1993) 8-5, 8-12 People v Bulger, 291 Mich App 1 (2010) 9-4, 9-37 *People v Burton,* 252 Mich App 130 (2002) 1-4 *People v Cain*, 495 Mich 874 (2013) 8-33 People v Cameron, 319 Mich App 215 (2017) 1-50, 1-51, 1-57 *People v Carll*, 322 Mich App 690 (2018) 6-24, 8-25, 8-26 *People v Carson*, 220 Mich App 662 (1996) 1-iv *People v Chandler*, 75 Mich App 585 (1976) 4-35 *People v Chapo*, 283 Mich App 360 (2009) 1-16 *People v Crear*, 242 Mich App 158 (2000) 1-iv

People v Crosby, 82 Mich App 1 (1978) 6-27, 8-32 People v Cross, 281 Mich App 737 (2008) 1-24, 1-27 People v Cunningham (Cunningham I), 301 Mich App 218 (2013) 1-58 People v Cunningham (Cunningham II), 496 Mich 145 (2014) 1-58 *People v Cunningham,* 496 Mich 145 (2014) 1-52 People v Czuprynski, 325 Mich App 449 (2018) 6-16, 9-60 People v Derror, 475 Mich 316 (2006) 9-44 *People v Dubis,* 158 Mich App 504 (1987) 1-6 People v Dumback, 330 Mich App 631 (2019) 8-10, 8-11, 8-13 People v Dunbar, 449 Mich 60 (2016) 3-14 *People v Dupre*, 335 Mich App 126 (2020) 9-28 People v Estabrooks, 175 Mich App 532 (1989) 1-16 People v Feezel, 486 Mich 184 (2010) 8-12, 9-31, 9-44 People v Ferency, 133 Mich App 526 (1984) 1-19, 1-24, 1-27, 4-35 People v Goldston, 470 Mich 523 (2004) 9-60 *People v Goodin*, 257 Mich App 425 (2003) 5-4, 8-14 People v Grayer, 235 Mich App 737 (1999) 8-6 *People v Grayer*, 252 Mich App 349 (2002) 8-6 People v Green, 260 Mich App 710 (2004) 8-6, 8-7 People v Green, 310 Mich App 249 (2015) 9-53, 9-54 *People v Harbour,* 76 Mich App 552 (1977) 7-10 People v Hawkins, 181 Mich App 393 (1989) 9-20 People v Hayward, 127 Mich App 50 (1983) 6-27, 8-32 People v Hoaglin, 262 Mich 162 (1933) 8-12 *People v Houseman*, 128 Mich App 17 (1983) 7-12 *People v Hrlic*, 277 Mich App 260 (2007) 4-53 *People v Hyde,* 285 Mich App 428 (2009) 9-22 People v James, 326 Mich App 98 (2018) 1-iv *People v Jensen*, 162 Mich App 171 (1987) 7-10 *People v Johnson*, 366 Mich App688_ (2021) 1-50 *People v Jones*, 443 Mich 88 (1993) 1-5 People v Jones, 497 Mich 155 (2014) 8-24 *People v Kelly*, 231 Mich App 627 (1998) 1-54 *People v Konopka (On Remand)*, 309 Mich App 345 (2015) 1-51 *People v Konopka*, 309 Mich App 345 (2015) 1-51, 1-59 People v Koon, 494 Mich 1 (2013) 9-44 *People v Kulpinski*, 243 Mich App 8 (2000) 9-32 People v Lambert, 395 Mich 296 (1975) 9-22, 9-27 People v Langworthy, 416 Mich 630 (1982) 1-5 People v Lardie, 452 Mich 231 (1996) 9-30 People v Laur, 128 Mich App 453 (1983) 6-27 People v Lechleitner, 291 Mich App 56 (2010) 9-31 People v Lockridge, 498 Mich 358 (2015) 1-56, 1-57 *People v Lyon*, 310 Mich App 515 (2015) 5-35, 9-21 People v Martinez, 211 Mich App 147 (1995) 2-12 People v Martinson, 161 Mich App 55 (1987) 8-14 *People v McCann*, 314 Mich App 605 (2016) 1-7 *People v McIntosh*, 23 Mich App 412 (1970) 4-43 People v McKinley, 255 Mich App 20 (2003) 6-23 *People v McLott*, 70 Mich App 524 (1976) 1-6 People v Mendoza, 468 Mich 527 (2003) 8-30

Table of Authorities: Cases Traffic Benchbook - Fourth Edition

People v Mikulen, 324 Mich App 14 (2018) 9-27 People v Milbourn, 435 Mich 630 (1990) 1-57 People v Miller, 482 MIch 540 (2008) 1-iv *People v Miller*, 498 Mich 13 (2015) 9-21, 9-35 *People v Morton*, 384 Mich 38 (1970) 7-10 People v Noble, 238 Mich App 647 (1999) 5-4, 8-14 *People v Nunley*, 491 Mich 686 (2012) 8-21, 8-22 *People v Oliver*, 242 Mich App 92 (2000) 8-15 _ (2023) <mark>8-24</mark> People v Otto, _ _ Mich App _ People v Pace, 311 Mich App 1 (2015) 6-16, 6-17 *People v Parrott*, 335 Mich App 648 (2021) 9-20, 9-51 People v Peltola, 489 Mich 174 (2011) 1-56 People v Perlos, 436 Mich 305 (1990) 9-57 *People v Perry,* ____ Mich App ____ (2021) 9-45 People v Pitts, 222 Mich App 260 (1997) 3-21 People v Price, 214 Mich App 538 (1995) 9-32 People v Rea, 500 Mich 422 (2017) 9-19, 9-20, 9-22, 9-23, 9-24, 9-35, 9-41 *People v Rice (On Remand)*, 235 Mich App 429 (1999) 1-55 *People v Rice,* 318 Mich App 688 (2017) 1-55 People v Robe, 337 Mich App 142 (2021) 9-49 People v Ross, 204 Mich App 310 (1994) 7-10 People v Sanders (Robert) (After Remand), 298 Mich App 105 (2012) 1-58 People v Sanders (Robert), 296 Mich App 710 (2012) 1-58 *People v Sartor*, 235 Mich App 614 (1999) 5-4 People v Schaefer, 473 Mich 418 (2005) 8-12, 8-20, 8-21, 9-28, 9-30 *People v Schmidt*, 196 Mich App 104 (1992) 5-5 *People v Shenoskey*, 320 Mich App 80 (2017) 1-50, 1-57 People v Sherman, 497 Mich 1025 (2015) 8-33 *People v Simmons*, 316 Mich App 322 (2016) 3-14 *People v Smith (Ryan)*, 496 Mich 133 (2014) 1-6 People v Snow, 386 Mich 586 (1972) 1-56 People v Steanhouse, 500 Mich 453 (2017) 1-55, 1-57 People v Stock, 507 Mich 1008 (2021) 9-44 *People v Strickland*, 79 Mich App 454 (1977) 8-14 People v Stricklin, 327 Mich App 592 (2019) 9-59 *People v Thousand*, 465 Mich 149 (2001) 1-5 *People v Tierney*, 266 Mich App 687 (2005) 1-54 People v Tims, 449 Mich 83 (1995) 8-31 People v Turner, 92 Mich App 485 (1979) 1-6 People v VanEss, Mich App (2024) **1-41** People v Werner, 254 Mich App 528 (2002) 9-32 People v Williams, 236 Mich App 610 (1999) 3-22 *People v Wood*, 276 Mich App 669 (2007) 8-6 *People v Wood*, 450 Mich 399 (1995) 9-32 People v Woodard, 321 Mich App 377 (2017) 9-60 *People v Woolfolk*, 304 Mich App 450 (2014), 2-2 *People v Woolfolk*, 497 Mich 23 (2014) 2-2 *People v Wujkowski*, 230 Mich App 181 (1998) 9-50 Placek v Sterling Hts, 405 Mich 638 (1979) 4-25

R

Rife v Colestock, 297 Mich 194 (1941) 4-43 *Rosen v Beh,* 272 Mich 487 (1935) 4-56

S

Shoniker v English, 254 Mich 76 (1931) 4-43 Smith v Whitehead, 342 Mich 542 (1955) 4-56 Soule v Grimshaw, 266 Mich 117 (1934) 4-56 Stabler v Copeland, 304 Mich 1 (1942) 4-43 State v Boucher, 207 Conn 612 (1988) 9-20 Stein v Home-Onwers Ins Co, 303 Mich App 382 (2013) 1-iv

Т

Terry v Detroit, 226 Mich App 418 (1997) 6-13 *Texas v Brown,* 460 US 730 (1983) 9-59

U

United States v Gaudin, 515 US 506 (1995) 7-11, 7-13 *United States v Leon,* 468 US 897 (1984) 9-60

W

Wayne Co Prosecutor v Wayne Circuit Judge, 154 Mich App 216 (1986) 1-55 *Whitcraft v Wolfe,* 148 Mich App 40 (1985) 5-24

Y

Young v Flood, 182 Mich App 538 (1990) 4-56

Table of Authorities: Cases Traffic Benchbook - Fourth Edition

Michigan Statutes

MCL 15.231 4-31 MCL 24.201 3-15, 3-17 MCL 24.328 3-15, 3-17 MCL 28.243 2-9 MCL 124.1 Glossary-24 MCL 124.13 Glossary-24 MCL 124.351 Glossary-23 MCL 124.359 Glossary-23 MCL 124.401 Glossary-23 MCL 124.426 Glossary-23 MCL 124.451 Glossary-24 MCL 124.479 Glossary-24 MCL 124.501 Glossary-24 MCL 124.512 Glossary-24 MCL 124.531 Glossary-24 MCL 124.536 Glossary-24 MCL 125.2302 Glossary-26, Glossary-48 MCL 125.2302(j) Glossary-26 MCL 125.3101 4-27 MCL 125.3702 4-27 MCL 125.4507 Glossary-44 MCL 141.101 Glossary-24 MCL 141.140 Glossary-24 MCL 205.93 9-66 MCL 205.93(3) 1-74, 9-66 MCL 224.6 Glossary-11 MCL 247.651 4-27, Glossary-1 MCL 247.675 4-27 MCL 256.627 3-4 MCL 256.627(f) Glossary-39 MCL 256.627(g) Glossary-39 MCL 256.657 Glossary-39 MCL 256.659 Glossary-39 MCL 257.1 1-11, 1-41, 4-42, 7-10 MCL 257.1d Glossary-2 MCL 257.2(1) Glossary-4 MCL 257.2b Glossary-27 MCL 257.2b(1) Glossary-4 MCL 257.2b(2) Glossary-4 MCL 257.2b(3) Glossary-5 MCL 257.2b(10) Glossary-25 MCL 257.2b(12) Glossary-27 MCL 257.2b(13) Glossary-28 MCL 257.2b(14) Glossary-32 MCL 257.2b(15) Glossary-38 MCL 257.2b(16) Glossary-47 MCL 257.2c(1) Glossary-5

MCL 257.2c(2) Glossary-5 MCL 257.4 Glossary-6 MCL 257.5a Glossary-6 MCL 257.6 Glossary-7, Glossary-8 MCL 257.6a Glossary-8 MCL 257.6b(c) 1-24, 1-27 MCL 257.6b(d) 1-28 MCL 257.7a Glossary-9 MCL 257.7a(1) Glossary-9 MCL 257.7a(2) Glossary-9 MCL 257.7b Glossary-9 MCL 257.8a 1-7, Glossary-10 MCL 257.8b Glossary-10 MCL 257.9 Glossary-10 MCL 257.12 Glossary-11 MCL 257.12a 9-66, Glossary-14 MCL 257.12c Glossary-46 MCL 257.13 Glossary-14 MCL 257.13e Glossary-15 MCL 257.13f Glossary-16 MCL 257.19a 3-25, 3-27, Glossary-13 MCL 257.20 Glossary-18, Glossary-43 MCL 257.20d Glossary-19 MCL 257.21 Glossary-19 MCL 257.23 Glossary-20 MCL 257.24a Glossary-23 MCL 257.24c Glossary-23 MCL 257.25 Glossary-23 MCL 257.25c Glossary-23 MCL 257.25d Glossary-24 MCL 257.28 Glossary-24 MCL 257.33 7-11, Glossary-27 MCL 257.35 Glossary-28 MCL 257.35a Glossary-29 MCL 257.35a(a) Glossary-29 MCL 257.35a(b) Glossary-29 MCL 257.36 Glossary-30 MCL 257.36c Glossary-28 MCL 257.37 Glossary-31 MCL 257.38 Glossary-31 MCL 257.39 Glossary-32 MCL 257.40 Glossary-32 MCL 257.40c Glossary-32 MCL 257.43a Glossary-33 MCL 257.43b Glossary-6, Glossary-10, Glossary-33 MCL 257.43c Glossary-33 MCL 257.45a Glossary-36 MCL 257.50 Glossary-37 MCL 257.51a Glossary-37 MCL 257.52(1) Glossary-37 MCL 257.55 Glossary-38

MCL 257.58 Glossary-39 MCL 257.58b Glossary-39 MCL 257.58c Glossary-40 MCL 257.62a Glossary-43 MCL 257.63 Glossary-44 MCL 257.63a Glossary-44 MCL 257.64 Glossary-18, Glossary-43 MCL 257.64a Glossary-44 MCL 257.64b Glossary-44 MCL 257.65 Glossary-43 MCL 257.66 Glossary-44 MCL 257.67c Glossary-44 MCL 257.67d Glossary-44 MCL 257.70 Glossary-45 MCL 257.76a Glossary-45 MCL 257.76b Glossary-46 MCL 257.76c Glossary-46 MCL 257.76d Glossary-47 MCL 257.79 Glossary-48 MCL 257.79d Glossary-45, Glossary-50 MCL 257.79d(1) Glossary-45 MCL 257.79d(2) Glossary-45 MCL 257.79d(a) 4-29 MCL 257.83(a) Glossary-42 MCL 257.83(b) Glossary-42 MCL 257.201 1-63, 1-65, 1-73, 3-9 MCL 257.204a 9-5 MCL 257.204b 1-3, 1-4, 1-5 MCL 257.204b(1) 1-4 MCL 257.204b(2) 1-4 MCL 257.215 5-17, 6-5 MCL 257.216 1-73, 3-9 MCL 257.216(1) 5-16, 6-5 MCL 257.216(2) 5-16, 6-5 MCL 257.217(1) 5-17 MCL 257.217c Glossary-1 MCL 257.219(1) 1-78, 6-20, 8-19, 9-19, 9-27, 9-30, 9-35, 9-37, 9-41, 9-44 MCL 257.222(6) 5-20, 7-10 MCL 257.223 1-37, 3-12, 3-13 MCL 257.223(1) 3-12 MCL 257.223(2) 3-13 MCL 257.223(3) 3-12 MCL 257.223(4) 3-12 MCL 257.225 3-13 MCL 257.225(2) 3-13, 3-14 MCL 257.225(7) 3-13 MCL 257.226a 9-47 MCL 257.226b 9-47 MCL 257.226b(1) 5-20 MCL 257.226b(2) 5-21 MCL 257.228(1) 3-8

MCL 257.233(2) 9-64 MCL 257.233(3) 9-65 MCL 257.233(4) 9-66 MCL 257.233(5) 9-64, 9-65, 9-66 MCL 257.233(6) 9-65 MCL 257.233(7) 9-65 MCL 257.233a(1) 5-23, 5-24 MCL 257.233a(6) 5-23, 7-11, 7-12 MCL 257.233a(11) 7-12 MCL 257.233a(15) 5-24, 7-12 MCL 257.233a(17) 5-23 MCL 257.234(1) 5-18 MCL 257.234(3) 5-18 MCL 257.240 1-32, 3-15 MCL 257.240(1) 3-8 MCL 257.240(2) 3-8, 3-9, Glossary-37 MCL 257.240(3) 1-32, 3-9 MCL 257.240(4) 3-9 MCL 257.248c Glossary-1 MCL 257.252a 1-76, 9-48, Glossary-1 MCL 257.252a(1) 3-14, 3-15, Glossary-37 MCL 257.252a(2) Glossary-1 MCL 257.252d 3-31 MCL 257.254 7-9, 7-10, 7-11 MCL 257.255 3-10 MCL 257.255(1) 3-9, 3-10 MCL 257.255(4) 3-10 MCL 257.256(1) 5-21 MCL 257.256(2) 5-21 MCL 257.257(1) 7-4 MCL 257.257(2) 7-5 MCL 257.257(3) 7-5 MCL 257.257(4) 7-5 MCL 257.259 3-9 MCL 257.301 3-2, 5-15 MCL 257.301(1) 6-6 MCL 257.301(2) 6-6 MCL 257.301(3) 6-6 MCL 257.301(6) 6-6 MCL 257.303 1-60, 3-3, 3-4, 6-16, 6-18, 9-7, 9-9, 9-10, 9-11, Glossary-17 MCL 257.303(1) 1-65, 5-16, 5-28, 5-35, 8-5, 8-29, 8-32, 9-8 MCL 257.303(2) 1-60, 1-65, 1-66, 1-67, 6-14, 6-15, 6-24, 6-25, 7-7, 8-5, 8-9, 8-12, 8-16, 8-17, 8-19, 8-23, 8-26, 8-29, 8-30, 8-32, 9-7, 9-8, 9-18, 9-26, 9-29, 9-34, 9-37, 9-40, 9-43 MCL 257.303(4) 1-66, 1-67, 8-5, 8-9, 8-12, 8-16, 8-17, 8-19, 8-23, 8-26, 8-30, 9-18, 9-26, 9-29, 9-34, 9-37, 9-40, 9-43 MCL 257.303(6) 1-66 MCL 257.303(7) 1-59, 1-67, 7-7, Glossary-17 MCL 257.303a 1-61, 1-62, 1-71 MCL 257.304 1-66, 1-75, 1-78, 9-8, 9-9, 9-10, 9-11, 9-15 MCL 257.304(1) 9-7, 9-9

```
MCL 257.304(2) 9-7, 9-11
MCL 257.304(3) 9-7
MCL 257.304(4) 9-9, Glossary-25
MCL 257.304(5) 9-9
MCL 257.304(6) 9-10
MCL 257.304(7) 9-9, 9-10
MCL 257.304(8) 9-10
MCL 257.304(9) 9-11
MCL 257.304(10) 9-12
MCL 257.304(11) 9-13
MCL 257.310 3-3
MCL 257.310(7) 5-14
MCL 257.310(10) 5-12
MCL 257.310(16) 5-12
MCL 257.310e 3-4, 4-53, Glossary-3
MCL 257.310e(1) 3-3
MCL 257.310e(3) 3-4
MCL 257.310e(4) 3-4, 3-6, 3-7
MCL 257.310e(5) 3-5, 3-7
MCL 257.310e(6) 3-5, 3-6, 3-7
MCL 257.310e(7) 3-4, 3-5, 3-6, 3-7
MCL 257.310e(8) 3-6, 3-7
MCL 257.310e(9) 3-7
MCL 257.310e(10) 3-7
MCL 257.310e(14) 1-65, 3-7, 3-8
MCL 257.310e(15) Glossary-3
MCL 257.311 1-47, 1-64, 1-65, 6-22, 6-23
MCL 257.311a 1-47, 1-48, 6-22
MCL 257.312 6-4
MCL 257.312(1) 6-4
MCL 257.312(3) 6-5
MCL 257.312(4) 6-4
MCL 257.312(5) 6-4
MCL 257.312a(4) 1-32
MCL 257.315 5-13
MCL 257.315(1) 3-3
MCL 257.315(3) 3-3
MCL 257.315(4) 5-13
MCL 257.315(5) 5-13
MCL 257.317(1) 1-72
MCL 257.318(1) 1-71
MCL 257.319 1-60, 1-67, 1-68, 1-71, 2-7, 2-8, 9-7, 9-8, 9-9, Glossary-17
MCL 257.319(1) 1-67
MCL 257.319(2) 1-59, 1-60, 6-16, 6-18, 7-5, 7-7, 8-5, 8-32, Glossary-17
MCL 257.319(3) 5-8, 6-14, 6-15, 6-24, 8-29
MCL 257.319(4) 9-8
MCL 257.319(5) 5-16, 5-18, 7-9, 7-10
MCL 257.319(6) 6-27
MCL 257.319(7) 5-28, 5-35
MCL 257.319(8) 9-11, 9-18, 9-19, 9-26, 9-37, 9-40, 9-43
MCL 257.319(9) 9-8
```

MCL 257.319(12) 1-67 MCL 257.319(13) 1-68, 9-26 MCL 257.319(14) 1-68, 9-18, 9-26, 9-37, 9-43 MCL 257.319(15) 1-68 MCL 257.319(16) 1-68 MCL 257.319(21) 1-71 MCL 257.319b 1-71 MCL 257.320(1) 9-8 MCL 257.320a 1-34, 1-49, 1-64, 1-65, 1-78, 1-79, 1-80, 1-81, 3-19, 4-3, 4-9, 4-15, 4-18, 4-19, 4-21, 4-25, 4-40, 4-41, 4-42, 4-46, 4-53, 4-57, 6-13, 8-15, 8-16, 8-19, 8-31, 9-38, 9-40 MCL 257.320a(1) 1-60, 1-64, 3-8, 4-3, 4-8, 4-9, 4-10, 4-14, 4-15, 4-18, 4-19, 4-21, 4-25, 4-42, 4-46, 4-50, 4-53, 4-54, 4-57, 5-7, 5-9, 5-28, 5-35, 6-2, 6-3, 6-5, 6-7, 6-8, 6-12, 6-14, 6-16, 6-18, 6-19, 6-22, 6-24, 6-25, 6-27, 7-3, 7-7, 8-5, 8-9, 8-11, 8-16, 8-17, 8-19, 8-23, 8-26, 8-28, 8-30, 8-31, 9-17, 9-26, 9-29, 9-34, 9-36, 9-40, 9-43 MCL 257.320a(2) 1-65, 3-8, 4-50, 4-54 MCL 257.320a(3) 1-65 MCL 257.320a(4) 1-65 MCL 257.320a(5) 1-65 MCL 257.320a(9) 1-65 MCL 257.320c 3-12 MCL 257.320d 1-64, 1-78, 1-80, Glossary-3 MCL 257.320d(1) 1-78, 1-79, 1-81 MCL 257.320d(2) 1-79 MCL 257.320d(3) 1-78, 1-79, 1-80, 1-81 MCL 257.320d(4) 1-79, 1-81 MCL 257.320d(5) 1-79 MCL 257.320d(6) 1-79, 1-80 MCL 257.320d(7) 1-81 MCL 257.320d(8) 1-81 MCL 257.320d(9) 1-81 MCL 257.320d(10) 1-80 MCL 257.320d(12) Glossary-3 MCL 257.320d(16) Glossary-3 MCL 257.320d(17) 1-78, 1-79, 1-80, 1-81 MCL 257.320d(18) Glossary-3 MCL 257.321a 1-19, 1-29, 1-46, 1-68, 2-9, 5-23, 8-18, Glossary-40 MCL 257.321a(1) 1-68, 1-69, 1-80, 5-22, 9-8 MCL 257.321a(2) 1-68, 1-69, 5-22, 5-23, 9-8 MCL 257.321a(3) 1-70 MCL 257.321a(4) 1-70 MCL 257.321a(5) 1-70 MCL 257.321a(10) Glossary-41 MCL 257.321b 5-23 MCL 257.322 9-9, 9-11, 9-55 MCL 257.322(5) 9-11 MCL 257.322(6) 9-11, 9-12 MCL 257.322a 9-12 MCL 257.323 9-56 MCL 257.323c 9-8 MCL 257.324(1) 5-15

```
MCL 257.324(2) 5-15
MCL 257.325 3-2
MCL 257.326 5-10
MCL 257.328 1-37, 3-11, 5-19
MCL 257.328(1) 1-37, 1-38, 1-64, 3-10, 3-11, 3-12
MCL 257.328(2) 1-37, 3-10, 3-11
MCL 257.328(3) 1-37, 1-38, 3-11
MCL 257.328(4) 3-12
MCL 257.328(6) 5-19
MCL 257.329(1) 7-13
MCL 257.329(2) 7-13
MCL 257.329(3) 7-13
MCL 257.401 Glossary-26
MCL 257.401a Glossary-30
MCL 257.501 1-73, 5-19
MCL 257.601 8-2
MCL 257.601b 1-61, Glossary-16, Glossary-27, Glossary-38
MCL 257.601b(1) 1-34, 1-48
MCL 257.601b(2) 1-49, 6-13, 8-16
MCL 257.601b(3) 8-15, 8-16
MCL 257.601b(5) Glossary-16, Glossary-27, Glossary-38
MCL 257.601c 1-61, Glossary-27
MCL 257.601c(1) 6-14, 8-17
MCL 257.601c(2) 8-16, 8-17
MCL 257.601c(3) Glossary-27
MCL 257.601d 1-46, 1-47, 6-15, 6-16, 6-17, 9-52, Glossary-27, Glossary-34, Glossary-
       35
MCL 257.601d(1) 6-17
MCL 257.601d(2) 6-15, 6-16
MCL 257.601d(3) 6-15, 6-17
MCL 257.601d(4) Glossary-28
MCL 257.602 6-2, 8-7
MCL 257.602a 1-61, 6-2, 8-2, 8-6
MCL 257.602a(1) 8-3
MCL 257.602a(2) 8-4, 8-6
MCL 257.602a(3) 8-4
MCL 257.602a(4) 8-4
MCL 257.602a(5) 8-4, 8-6
MCL 257.602a(6) 8-6
MCL 257.602b 4-47, 4-48, 4-50, 4-53, Glossary-18, Glossary-26, Glossary-29, Glossary-
      41, Glossary-47, Glossary-48
MCL 257.602b(1) 1-32, 4-47, 4-49, 4-50
MCL 257.602b(2) 1-32, 4-47, 4-49, 4-50, Glossary-48
MCL 257.602b(3) 4-48
MCL 257.602b(4) 1-32, 4-49
MCL 257.602b(5) 1-32, 4-50
MCL 257.602b(6) 1-32, 4-49, 4-50
MCL 257.602b(7) 4-47
MCL 257.602b(10) 4-49
MCL 257.602b(12) Glossary-18, Glossary-26, Glossary-29, Glossary-42
MCL 257.602b(13) Glossary-48
```

MCL 257.602c 1-65, 4-53, 4-54 MCL 257.602c(1) 4-53, 4-54, Glossary-47 MCL 257.602c(2) 4-54 MCL 257.602c(4) 4-53 MCL 257.602c(6) 4-53 MCL 257.603 6-11, 6-12, 6-13 MCL 257.603(2) 6-11 MCL 257.603(3) 6-12 MCL 257.603(4) 6-12 MCL 257.603(5) 6-12 MCL 257.606a Glossary-17 MCL 257.606a(2) Glossary-17 MCL 257.608 4-42 MCL 257.611 4-46 MCL 257.612 4-24, 4-46 MCL 257.612(1) 4-44, 4-45 MCL 257.612(4) 6-8 MCL 257.613d 6-7 MCL 257.613d(1) 6-7 MCL 257.613d(2) 6-8 MCL 257.614 4-46 MCL 257.614(1) 4-45 MCL 257.616a 7-6, 7-7, Glossary-32 MCL 257.616a(1) 7-8 MCL 257.616a(2) 7-6, 7-7 MCL 257.616a(5) Glossary-32 MCL 257.617 1-42, 1-46, 1-61, 5-3, 8-10, 8-12, 8-13, 8-14 MCL 257.617(1) 8-10, 8-11, 8-13 MCL 257.617(2) 8-11 MCL 257.617(3) 8-10, 8-11, 8-12, 8-13 MCL 257.617a 8-14, 8-15 MCL 257.617a(1) 5-7 MCL 257.617a(2) 5-7 MCL 257.618 5-8, 8-14, 8-15 MCL 257.618(2) 5-8 MCL 257.618a 4-4, 4-5 MCL 257.618a(1) 4-4 MCL 257.618a(2) 4-4 MCL 257.618a(3) 4-4 MCL 257.618a(4) 4-5 MCL 257.618a(5) 4-5 MCL 257.618a(6) 4-5, Glossary-17 MCL 257.618a(7) 4-5 MCL 257.619 1-42, 1-46, 5-3, 5-4, 5-7, 5-8, 8-9, 8-10, 8-12, 8-14 MCL 257.619(a) 5-7, 5-8, 8-9 MCL 257.619(b) 5-7, 5-8 MCL 257.620 5-9 MCL 257.621(a) 5-6 MCL 257.622 5-5 MCL 257.623 5-6 MCL 257.624a 1-64, 1-73, 5-34, 5-35, 5-36, Glossary-17, Glossary-35

- MCL 257.624a(1) 5-34
- MCL 257.624a(2) 5-34
- MCL 257.624a(3) 5-35
- MCL 257.624a(5) 5-34
- MCL 257.624a(6) Glossary-17, Glossary-32
- MCL 257.624b 1-64, 1-73, 3-31, 5-27, Glossary-35
- MCL 257.624b(1) 3-32, 5-27, 5-28
- MCL 257.624b(2) 5-28
- MCL 257.624b(4) 3-32
- MCL 257.624b(5) 3-32
- MCL 257.625 1-4, 1-41, 1-44, 1-68, 1-74, 1-76, 1-77, 1-78, 5-22, 8-2, 9-2, 9-4, 9-11, 9-12, 9-15, 9-19, 9-21, 9-30, 9-32, 9-37, 9-52, 9-61, 9-64, 9-65, Glossary-19, Glossary-29, Glossary-30, Glossary-33, Glossary-34, Glossary-35
- MCL 257.625(1) 1-46, 1-47, 1-48, 1-62, 1-69, 5-22, 9-3, 9-4, 9-5, 9-6, 9-7, 9-14, 9-15, 9-16, 9-18, 9-19, 9-20, 9-21, 9-22, 9-23, 9-24, 9-28, 9-31, 9-33, 9-35, 9-38, 9-41, 9-52, Glossary-29
- MCL 257.625(2) 1-72, 9-4, 9-23, Glossary-34
- MCL 257.625(3) 1-46, 1-47, 1-48, 1-62, 9-3, 9-4, 9-5, 9-7, 9-14, 9-16, 9-24, 9-25, 9-27, 9-28, 9-33, 9-38, 9-52
- MCL 257.625(4) 1-47, 8-20, 8-21, 9-4, 9-5, 9-24, 9-28, 9-29, 9-30, 9-31, 9-38, 9-40, 9-52
- MCL 257.625(5) 1-47, 9-4, 9-5, 9-21, 9-24, 9-33, 9-34, 9-35, 9-38, 9-40, 9-52
- MCL 257.625(6) 1-46, 1-48, 1-62, 1-69, 5-22, 5-35, 9-3, 9-4, 9-5, 9-7, 9-14, 9-35, 9-36, 9-37, 9-38, 9-48, 9-51, 9-52, Glossary-2, Glossary-34, Glossary-35
- MCL 257.625(7) 1-47, 1-48, 9-3, 9-4, 9-5, 9-21, 9-35, 9-38, 9-39, 9-40, 9-41, 9-52
- MCL 257.625(8) 1-46, 1-48, 1-62, 1-69, 5-22, 9-3, 9-4, 9-5, 9-28, 9-31, 9-33, 9-38, 9-41, 9-43, 9-44, 9-45, 9-52
- MCL 257.625(9) 9-16, 9-17, 9-18, 9-19, 9-42, 9-43, 9-44
- MCL 257.625(10) 9-24
- MCL 257.625(11) 9-25, 9-26, 9-27
- MCL 257.625(12) 9-36, Glossary-34
- MCL 257.625(16) 9-4
- MCL 257.625(17) 1-4
- MCL 257.625(23) 9-36
- MCL 257.625(24) 9-19
- MCL 257.625(25) 9-37, Glossary-20, Glossary-34
- MCL 257.625(26) 9-36, 9-37, Glossary-34
- MCL 257.625(27) Glossary-34
- MCL 257.625(28) 9-23, 9-58, Glossary-2, Glossary-29
- MCL 257.625a 1-14, 9-51, 9-54, 9-62, 9-63
- MCL 257.625a(1) 9-2
- MCL 257.625a(2) 9-48, 9-49, 9-50, 9-51, 9-58
- MCL 257.625a(5) 9-48, 9-49, 9-52
- MCL 257.625a(6) 9-51, 9-52, 9-53, 9-54, 9-56, 9-57, 9-58
- MCL 257.625a(7) 9-58, Glossary-2
- MCL 257.625a(8) 9-56, 9-58
- MCL 257.625a(9) 9-57
- MCL 257.625b Glossary-34, Glossary-35
- MCL 257.625b(1) 9-3
- MCL 257.625b(2) 9-3, 9-4
- MCL 257.625b(3) 9-5
- MCL 257.625b(4) 5-28, 9-5

MCL 257.625b(5) 9-5, 9-6, Glossary-30, Glossary-36 MCL 257.625b(6) 9-13 MCL 257.625c 9-48, 9-62, Glossary-10, Glossary-19 MCL 257.625c(1) 9-50, 9-52, 9-53, 9-54, 9-57, 9-62 MCL 257.625c(2) 9-52 MCL 257.625c(3) 9-52 MCL 257.625c(4) Glossary-10, Glossary-20 MCL 257.625d 9-48, 9-54, 9-55, 9-56, 9-62 MCL 257.625d(1) 9-54, 9-59 MCL 257.625d(2) 9-54 MCL 257.625e 9-48, 9-55, 9-62 MCL 257.625e(1) 9-54 MCL 257.625e(2) 9-55 MCL 257.625f 9-8, 9-48, 9-54, 9-62 MCL 257.625f(1) 9-55 MCL 257.625f(2) 9-55 MCL 257.625f(7) 9-55, 9-56 MCL 257.625f(8) 9-56 MCL 257.625k 9-19 MCL 257.6251 1-76, 9-19, 9-47, 9-48 MCL 257.625l(1) 9-46 MCL 257.625l(2) 1-75, 9-46, 9-47 MCL 257.625l(3) 9-46, 9-47 MCL 257.625l(4) 9-46, 9-47 MCL 257.625l(5) 9-47 MCL 257.625l(6) 9-47 MCL 257.625l(7) 9-47 MCL 257.625l(8) 9-47 MCL 257.625l(9) 9-48 MCL 257.625l(10) 9-48 MCL 257.625m 1-41, 1-44, 1-62, 1-65, 1-78, 9-3, 9-4, 9-5, 9-52, 9-64, 9-65, Glossary-30, Glossary-34, Glossary-35 MCL 257.625n 1-77, 9-13, 9-18, 9-19, 9-27, 9-30, 9-34, 9-35, 9-40, 9-41, 9-43, 9-44 MCL 257.625n(1) 1-76, 8-23, 8-26 MCL 257.625n(2) 1-76 MCL 257.625n(3) 1-76 MCL 257.625n(4) 1-76 MCL 257.625n(5) 1-77 MCL 257.625n(6) 1-77 MCL 257.625n(10) 1-77 MCL 257.625n(11) 1-77 MCL 257.625n(12) 1-77 MCL 257.625r 9-61, Glossary-6 MCL 257.625r(1) 9-61, 9-62, Glossary-6 MCL 257.625r(2) 9-62 MCL 257.625r(3) 9-63 MCL 257.625r(4) 9-62 MCL 257.625r(5) 9-62 MCL 257.625r(6) 9-61 MCL 257.625s 9-63 MCL 257.625t 9-62, Glossary-6, Glossary-10

```
MCL 257.625t(1) 9-61
MCL 257.625t(2) 9-61
MCL 257.625t(3) 9-61
MCL 257.625t(6) Glossary-6, Glossary-10
MCL 257.626 1-47, 1-48, 1-61, 1-68, 1-69, 4-3, 5-22, 6-23, 8-22, 8-24
MCL 257.626(1) 6-23, 8-25
MCL 257.626(2) 4-3, 6-23, 8-22, 8-25
MCL 257.626(3) 8-26, 9-52, Glossary-34, Glossary-35
MCL 257.626(4) 8-23, 8-24, 9-52, Glossary-34, Glossary-35
MCL 257.626(5) 8-24
MCL 257.626a Glossary-14
MCL 257.626a(1) 6-3
MCL 257.626a(2) 6-3, 6-4
MCL 257.626a(3) 6-3, 6-4
MCL 257.626a(4) Glossary-14
MCL 257.626b 1-80, 4-2, 4-3, 4-41
MCL 257.627 1-34, 1-80, 4-26, 4-29, 4-30, 4-33, 4-40, 4-41, 4-42, Glossary-26, Glossary-
      49
MCL 257.627(1) 4-26, 4-27
MCL 257.627(2) 4-28, Glossary-26
MCL 257.627(3) 4-28
MCL 257.627(4) 4-29
MCL 257.627(5) 1-34, 4-28, 4-40, 4-42
MCL 257.627(6) 1-80, 4-29, 4-41, 4-42
MCL 257.627(7) 1-34, 4-34, 4-42, Glossary-18
MCL 257.627(8) 4-34
MCL 257.627(9) 1-80, 4-34
MCL 257.627(10) 4-34
MCL 257.627(16) 4-26
MCL 257.627(17) 4-34
MCL 257.627(18) Glossary-49
MCL 257.627a 1-80, Glossary-38
MCL 257.627a(1) Glossary-39
MCL 257.627a(2) 4-32
MCL 257.627c 1-32, 1-35, 4-29, 4-30, 4-31, 4-32, Glossary-33
MCL 257.627c(1) 4-29, 4-30
MCL 257.627c(2) 1-33, 4-30, 4-31
MCL 257.627c(3) 4-31
MCL 257.627c(4) 4-31
MCL 257.627c(5) 4-31, 4-32
MCL 257.627c(6) 4-32
MCL 257.627c(7) 4-30
MCL 257.627c(10) Glossary-33
MCL 257.628 4-29, 4-33
MCL 257.628(1) 4-33, 4-34, Glossary-11
MCL 257.628b 4-33, Glossary-18, Glossary-41
MCL 257.628b(2) 4-33
MCL 257.628b(3) 4-33
MCL 257.628b(4) Glossary-18, Glossary-41
MCL 257.629b(1) 4-34
MCL 257.629c 1-34, 1-64, 4-40, 4-41, 4-42
```

MCL 257.629c(1) 1-34, 4-40, 4-42 MCL 257.629c(2) 1-34, 4-40, 4-42 MCL 257.629c(3) 1-34, 4-40, 4-42 MCL 257.632 4-30, 6-11, 6-13 MCL 257.633(1) 4-35 MCL 257.634 4-55, 4-57 MCL 257.634(1) 4-55 MCL 257.634(2) 4-55 MCL 257.635 4-57 MCL 257.635(1) 4-56 MCL 257.636 4-15, 4-16 MCL 257.636(1) 4-15, 4-17 MCL 257.636(2) 4-16 MCL 257.636(3) 4-16 MCL 257.636(4) 4-15, 4-16 MCL 257.637 4-15, 4-16, 4-17 MCL 257.637(1) 4-16 MCL 257.637(2) 4-16 MCL 257.637(3) 4-16 MCL 257.637(4) 4-16, 4-17 MCL 257.639(1) 4-17 MCL 257.640 4-16, 4-17 MCL 257.641 4-57 MCL 257.641(2) 4-56 MCL 257.642 4-17 MCL 257.642(1) 4-55, 4-56 MCL 257.643 4-17 MCL 257.643a 4-15 MCL 257.643b 4-16, 4-17, Glossary-41 MCL 257.643b(1) 4-16 MCL 257.643b(2) 4-16 MCL 257.643b(3) 4-16 MCL 257.643b(4) 4-16 MCL 257.643b(5) Glossary-41 MCL 257.644 4-56 MCL 257.645 4-57 MCL 257.647 4-51, 4-52 MCL 257.647(1) 4-51, 4-52 MCL 257.648 4-52, 4-53, 6-9 MCL 257.648(1) 4-52 MCL 257.648(2) 4-52 MCL 257.648(3) 4-52 MCL 257.648(5) 4-52 MCL 257.649 4-21, 4-23, 4-46 MCL 257.649(1) 4-21, 4-24 MCL 257.649(2) 4-21, 4-24 MCL 257.649(3) 4-21 MCL 257.649(4) 4-21, 4-22 MCL 257.649(5) 4-22 MCL 257.649(6) 4-22, 4-24 MCL 257.649(7) 4-21, 4-23

```
MCL 257.649(8) 4-23, 4-24, 4-43
MCL 257.649(9) 4-23, 4-24
MCL 257.649(10) 4-21
MCL 257.649a 4-21
MCL 257.649a(1) 4-24
MCL 257.649a(2) 4-21
MCL 257.650 4-24, 4-25
MCL 257.652(1) 4-24
MCL 257.653 4-24, 4-25
MCL 257.653a 1-61, 4-24, 8-9, 9-29, Glossary-4
MCL 257.653a(1) 4-9, 4-10, 8-8, 9-29
MCL 257.653a(2) 4-10, 4-41, 8-9
MCL 257.653a(3) 4-10, 8-8, 8-9
MCL 257.653a(4) 4-10, 8-8
MCL 257.653a(5) 4-10
MCL 257.654 4-24
MCL 257.655 4-21
MCL 257.657a 6-10, Glossary-17
MCL 257.657a(1) 6-8
MCL 257.657a(5) 6-10
MCL 257.657a(9) 6-9
MCL 257.657a(12) 6-10
MCL 257.657a(17) 6-10
MCL 257.657a(18) 6-10
MCL 257.657a(19) 6-10
MCL 257.657a(20) 6-10
MCL 257.657a(21) 6-10, 6-11, Glossary-17
MCL 257.658 1-65
MCL 257.658b 6-11
MCL 257.665a Glossary-27
MCL 257.665b Glossary-27, Glossary-31, Glossary-38
MCL 257.667 4-20
MCL 257.667(1) 4-20
MCL 257.667(2) 4-20
MCL 257.667(3) 4-20
MCL 257.668 4-20
MCL 257.668(1) 4-20
MCL 257.668(2) 4-20
MCL 257.668(4) 4-20
MCL 257.671(3) 4-24
MCL 257.672 3-27, 3-28
MCL 257.672(1) 3-22
MCL 257.674 3-22, 3-27, 3-28
MCL 257.674(1) 1-33, 3-24, 3-26, 3-27, 3-30
MCL 257.674(n) 3-27
MCL 257.674a 3-27, 3-28
MCL 257.674a(1) 3-24
MCL 257.675 3-25, 3-27, 3-28, 5-24, 5-25, Glossary-12
MCL 257.675(5) 3-27
MCL 257.675(6) 3-25, 3-26
MCL 257.675(8) 3-24
```

MCL 257.675(13) 5-26 MCL 257.675(14) 5-24, 5-25 MCL 257.675(15) 3-25, 5-25 MCL 257.675(22) Glossary-13 MCL 257.675a 3-28 MCL 257.675b 3-29, 3-30, Glossary-23 MCL 257.675b(1) 3-28, 3-29 MCL 257.675b(2) 3-29 MCL 257.675b(3) 3-29 MCL 257.675b(4) Glossary-23 MCL 257.675c(1) 3-28, 3-29 MCL 257.675c(2) 3-28 MCL 257.675c(3) 3-30 MCL 257.675c(4) 3-28 MCL 257.676 3-26, 3-28 MCL 257.676(1) 3-26, 3-27 MCL 257.676a(3) 1-33 MCL 257.676c 1-33, 3-30, 3-31 MCL 257.676c(1) 3-31 MCL 257.677(1) 4-14 MCL 257.677(2) 4-14 MCL 257.679(1) 3-26, 4-8 MCL 257.682 1-10, 1-13, 1-33, 1-35, 1-80, 4-7, Glossary-6, Glossary-22, Glossary-24, Glossary-36, Glossary-38, Glossary-43 MCL 257.682(1) 4-5, 4-6, 4-7, Glossary-6 MCL 257.682(2) 4-7, Glossary-38 MCL 257.682(3) 4-6 MCL 257.682(4) 1-33, 4-6, 4-7 MCL 257.682(5) 4-7 MCL 257.682(6) 4-6 MCL 257.682(7) 4-7 MCL 257.682(8) 4-7 MCL 257.682(9) Glossary-6, Glossary-22, Glossary-24, Glossary-37, Glossary-38, Glossary-43 MCL 257.682a 3-20 MCL 257.682b 4-18 MCL 257.682b(1) 4-18 MCL 257.682b(2) 4-18 MCL 257.682c 4-12 MCL 257.683 1-37, 3-19, 3-20, 3-21 MCL 257.683(1) 3-19 MCL 257.683(2) 3-20 MCL 257.683(4) 3-20 MCL 257.683(5) 3-20 MCL 257.683(6) 3-19 MCL 257.684 3-20 MCL 257.685 3-20, 6-10 MCL 257.686 3-20, 3-22, 6-10 MCL 257.686(1) 3-22 MCL 257.686(2) 3-22 MCL 257.687 3-21

```
MCL 257.688 3-21
MCL 257.688(1) 3-21
MCL 257.689 3-21
MCL 257.690 3-21
MCL 257.691 3-21
MCL 257.693 3-20
MCL 257.694 3-20
MCL 257.695 3-20
MCL 257.696 3-20
MCL 257.697 3-20, 3-21, 6-10
MCL 257.697b 6-10
MCL 257.698 3-19, 3-20, 4-12, 4-13, 8-7
MCL 257.698(5) 4-13, Glossary-4, Glossary-36, Glossary-41
MCL 257.698(6) 4-13
MCL 257.698(7) 4-12
MCL 257.698(9) 4-13
MCL 257.700 3-20
MCL 257.701 3-20
MCL 257.702 3-20
MCL 257.703 3-21
MCL 257.704 3-20, 6-11
MCL 257.705 3-20
MCL 257.706 3-20, 6-11
MCL 257.707 3-20
MCL 257.707d 3-19
MCL 257.708 3-20
MCL 257.708a 3-21, 6-11
MCL 257.708b 3-21
MCL 257.709 3-20, 3-21, 3-22
MCL 257.710 3-21
MCL 257.710a 6-11
MCL 257.710c 3-20
MCL 257.710d 1-33, 1-38, 1-65, 3-15, 3-16, 3-17, 3-18, 3-19
MCL 257.710d(1) 3-15, 3-17
MCL 257.710d(2) 3-15, 3-16
MCL 257.710d(4) 3-16
MCL 257.710d(5) 3-16
MCL 257.710d(6) 3-17
MCL 257.710d(9) 3-17
MCL 257.710e 1-33, 1-64, 3-16, 3-17, 3-18, 3-19, 6-11
MCL 257.710e(1) 3-18
MCL 257.710e(2) 3-18
MCL 257.710e(3) 3-17
MCL 257.710e(5) 3-17
MCL 257.710e(6) 3-17
MCL 257.710e(13) 3-19
MCL 257.711 3-19, 3-20, 3-21
MCL 257.717 1-65
MCL 257.719 1-65
MCL 257.719a 1-65
MCL 257.721 3-21
```

MCL 257.721(3) 3-21 MCL 257.723 1-65 MCL 257.727 1-42, 1-43, 1-47, 2-5 MCL 257.727(a) 1-43, 1-46 MCL 257.727(b) 1-46, 1-47 MCL 257.727(c) 1-47 MCL 257.727(d) 1-47 MCL 257.727c 1-13, 1-15, 1-18, 1-19, 1-44 MCL 257.727c(1) 1-12, 1-44, Glossary-8 MCL 257.727c(2) 1-12, 1-44 MCL 257.727c(3) 1-14 MCL 257.728 1-48, 2-5, 7-12 MCL 257.728(1) 1-42, 1-43, 1-46 MCL 257.728(5) 1-43 MCL 257.728(8) 1-14, 1-43 MCL 257.728a 1-18 MCL 257.728a(1) 1-18 MCL 257.728a(2) 1-18 MCL 257.728d 5-24 MCL 257.728e 1-44, 1-45 MCL 257.729 1-36, 1-52 MCL 257.732 1-7, 1-29, 1-62, 1-64, 1-70, 2-6, 2-10, 3-12, Glossary-17 MCL 257.732(1) 1-7, 1-23, 1-26, 1-62, 2-10, 4-10, 4-14, 5-7, 5-9, 5-28, 5-35, 6-2, 6-3, 6-5, 6-7, 6-8, 6-12, 6-14, 6-16, 6-18, 6-19, 6-22, 6-24, 6-25, 6-27 MCL 257.732(3) 1-63 MCL 257.732(4) 1-60, 1-63, 2-7 MCL 257.732(5) 5-32 MCL 257.732(6) 1-59, 7-7, Glossary-17 MCL 257.732(7) 1-60 MCL 257.732(8) 1-60, 2-7 MCL 257.732(9) 1-60 MCL 257.732(16) 1-62, 1-63, 1-64, 1-65 MCL 257.732(18) 1-63 MCL 257.732(21) 1-8 MCL 257.732(22) 1-7, 1-64 MCL 257.741(1) 1-15, Glossary-8 MCL 257.741(2) 1-9, 1-10, 1-11, 1-23, 1-26, 2-3 MCL 257.741(3) 1-10, 1-13 MCL 257.741(4) 1-10, 1-15, 1-18 MCL 257.741(5) 1-9, 2-3 MCL 257.742 1-10, 1-12, 1-13, 1-15, 1-16, 1-19, 4-32, 7-12 MCL 257.742(1) 1-15, 1-16, 1-17 MCL 257.742(2) 1-15 MCL 257.742(3) 1-15, 1-16, 1-17 MCL 257.742(4) 1-15 MCL 257.742(5) 1-17 MCL 257.742(6) 1-17 MCL 257.742(7) 1-14, 1-15, 1-18 MCL 257.742(9) 1-15, 1-18 MCL 257.742(10) Glossary-31 MCL 257.743 1-15, 1-18, 1-44

```
MCL 257.743(1) 1-13
MCL 257.743(2) 1-13
MCL 257.743(3) 1-14, 1-17
MCL 257.743(4) 1-14
MCL 257.744 1-19
MCL 257.744a 7-12, 7-13
MCL 257.745 1-19
MCL 257.745(1) 1-19
MCL 257.745(2) 1-20, 1-21
MCL 257.745(3) 1-20, 1-21, 1-29
MCL 257.745(4) 1-21, 1-29
MCL 257.745(5) 1-21, 1-22, 1-26
MCL 257.746 1-11, 1-22, 1-29
MCL 257.746(1) 1-11, 1-23, 4-35
MCL 257.746(2) 1-23
MCL 257.746(3) 1-23, 1-24
MCL 257.746(4) 1-24
MCL 257.746(5) 1-11, 1-24, 1-25
MCL 257.747 1-19, 1-22, 1-29
MCL 257.747(1) 1-11, 1-26
MCL 257.747(2) 1-26
MCL 257.747(3) 1-27
MCL 257.747(4) 1-26
MCL 257.747(5) 1-27
MCL 257.748 1-29
MCL 257.749 1-16
MCL 257.749(1) 1-19
MCL 257.749(2) 1-19
MCL 257.749(3) 1-19
MCL 257.749(4) 1-19
MCL 257.803d 3-25, 5-24, 5-25, Glossary-12
MCL 257.803d(2) Glossary-13
MCL 257.803f 3-25, 5-24, 5-25
MCL 257.803h Glossary-12
MCL 257.803h(3) Glossary-13
MCL 257.806 5-18
MCL 257.806(1) 5-18
MCL 257.806(3) 5-18
MCL 257.809 3-10
MCL 257.901 8-10
MCL 257.901(1) 1-3, 5-3, 6-2, 8-7
MCL 257.901(2) 1-48, 5-4, 5-5, 5-6, 5-7, 5-8, 5-9, 5-10, 5-15, 5-17, 5-18, 5-21, 5-23, 5-
      27, 5-34, 6-2, 6-3, 6-4, 6-6, 6-7, 6-8, 6-10, 6-12, 6-22
MCL 257.901(3) 1-3
MCL 257.901a 6-23
MCL 257.902 1-55, 7-4, 7-8, 7-12, 7-13
MCL 257.903(1) 7-8
MCL 257.903(2) 7-8
MCL 257.903(3) 7-8
MCL 257.904 1-4, 1-61, 1-74, 1-75, 1-76, 1-77, 1-78, 5-10, 5-11, 5-12, 6-18, 6-19, 6-20,
      6-21, 7-2, 7-3, 8-17, 8-19, 8-20, 8-21, 9-65
```

MCL 257.904(1) 1-75, 5-12, 6-18, 6-20, 6-21, 8-18, 8-19, 8-20, 8-21, 8-22 MCL 257.904(2) 1-72, 5-10, 5-12, 6-20, 7-4 MCL 257.904(3) 1-75, 5-11, 5-12, 6-18, 6-19, 6-20, 8-22 MCL 257.904(4) 8-18, 8-19, 8-20, 8-21, 8-29, 8-32, 9-33 MCL 257.904(5) 8-18, 8-20 MCL 257.904(7) 7-2, 7-3 MCL 257.904(9) 5-11, 6-19 MCL 257.904(10) 1-70, 6-19, 9-8 MCL 257.904(15) 5-12, 6-20, 7-3, 8-19 MCL 257.904(19) 5-12, 6-20, 7-3, 8-19 MCL 257.904a 6-21 MCL 257.904b 1-76, 6-20 MCL 257.904b(1) 1-75, 6-20 MCL 257.904b(2) 1-76 MCL 257.904b(3) 1-76 MCL 257.904b(4) 1-76 MCL 257.904b(5) 1-76 MCL 257.904d 1-72, 1-73, 1-75, 9-13, Glossary-34, Glossary-49 MCL 257.904d(1) 1-72, 1-73, 1-75, 8-23, 8-26, 9-18, 9-26, 9-27, 9-30, 9-34, 9-35, 9-40, 9-41, 9-43 MCL 257.904d(2) 1-72 MCL 257.904d(3) 1-72 MCL 257.904d(4) 1-72, 5-12 MCL 257.904d(5) 1-75 MCL 257.904d(6) 1-73 MCL 257.904d(7) 1-72, 1-73 MCL 257.904d(8) Glossary-35, Glossary-49 MCL 257.904d(9) Glossary-34, Glossary-35 MCL 257.904d(10) Glossary-34, Glossary-35 MCL 257.904d(11) 1-75 MCL 257.904d(12) 1-75 MCL 257.904d(13) 1-75 MCL 257.904e 1-73, Glossary-49 MCL 257.905(1) 5-19 MCL 257.907 1-20, 1-21, 1-24, 1-27, 1-31, 1-32, 1-33, 1-34, 1-35, 1-36, 3-18, 3-28, 3-30, 4-4, 4-7, 4-40, 4-42, Glossary-27 MCL 257.907(2) 1-31, 1-32, 1-33, 1-35, 1-36, 1-38, 1-39, 3-9, 3-18, 3-19, 3-30, 3-31, 4-7, 4-49, 4-50 MCL 257.907(4) 1-35, 1-36, 1-39, 3-31 MCL 257.907(5) 1-39 MCL 257.907(7) 1-30 MCL 257.907(8) 1-31 MCL 257.907(9) 1-37, 3-21 MCL 257.907(11) 1-33, 1-38, 3-18, 3-19 MCL 257.907(12) 1-36, 1-39 MCL 257.907(13) 1-37, 3-13 MCL 257.907(14) 1-37, 3-10, 3-11, 3-12 MCL 257.907(15) 1-37 MCL 257.907(16) Glossary-28 MCL 257.907a 1-35 MCL 257.908(1) 1-38

MCL 257.908(3) 1-38 MCL 257.908(4) 1-39 MCL 257.908(5) 1-39 MCL 257.908(6) 1-39 MCL 257.908(7) 1-39 MCL 257.909 1-36 MCL 257.909(1) 1-35, 4-7 MCL 257.909(2) 1-35 MCL 257.909(3) 1-35 MCL 257.909(4) 1-35 MCL 257.909(5) 1-35 MCL 257.909(6) 1-35 MCL 257.923 1-11, 1-41 MCL 257.1110 9-8 MCL 257.1571 Glossary-27 MCL 257.1577 Glossary-27 MCL 257.1805(8) Glossary-43 MCL 257.1820 1-35, 4-6 MCL 257.1857 4-20 MCL 257.2102 Glossary-23, Glossary-24, Glossary-32, Glossary-44, Glossary-45, Glossary-46, Glossary-47 MCL 257.2102(d) Glossary-23 MCL 257.2102(f) Glossary-24 MCL 257.2102(h) Glossary-32 MCL 257.2102(i) Glossary-44 MCL 257.2102(k) Glossary-44 MCL 257.2102(1) Glossary-45 MCL 257.2102(m) Glossary-46 MCL 257.2102(n) Glossary-46 MCL 257.2102(o) Glossary-46 MCL 257.2102(p) Glossary-47 MCL 267.625a(7) Glossary-2 MCL 275.675c(3) 3-30 MCL 275.698 4-9 MCL 324.80176 Glossary-35 MCL 324.81134 Glossary-35 MCL 324.82101 Glossary-18 MCL 324.82105a(2) 9-8 MCL 324.82127 Glossary-35 MCL 330.1100a Glossary-12 MCL 330.1100d Glossary-39 MCL 333.1101 Glossary-36 MCL 333.6230 5-29, 5-30 MCL 333.7104 Glossary-10, Glossary-33 MCL 333.7104(3) Glossary-10 MCL 333.7212 9-41, 9-44, 9-61 MCL 333.7212(1) 9-45 MCL 333.7214(a) 9-41, 9-62 MCL 333.7403 9-45 MCL 333.7404 9-45 MCL 333.16215 9-56

MCL 333.16901 Glossary-25 MCL 333.16915 Glossary-25 MCL 333.17201 Glossary-25 MCL 333.17242 Glossary-25 MCL 333.18101 Glossary-25 MCL 333.18117 Glossary-25 MCL 333.18501 Glossary-25 MCL 333.18518 Glossary-25 MCL 333.20904 Glossary-7 MCL 333.25211 Glossary-36 MCL 333.26421 9-28 MCL 333.26427(b) 9-28, 9-45 MCL 333.27951 9-44 MCL 333.27952 9-45 MCL 333.27954(1) 9-45 MCL 333.27965 9-45 MCL 333.27965(3) 9-45 MCL 380.5(7) Glossary-37 MCL 380.6(1) Glossary-38 MCL 380.1311b Glossary-36 MCL 380.1311m Glossary-37 MCL 436.1101 5-27 MCL 436.1105 Glossary-2 MCL 436.1701 Glossary-35 MCL 436.1703 5-27, 5-29, 5-31, 5-32, 5-33, 5-34, Glossary-2, Glossary-35 MCL 436.1703(1) 1-63, 1-73, 5-27, 5-28, 5-29, 5-30, 5-31, 5-32, 5-33, 5-35, Glossary-35 MCL 436.1703(2) 1-63, 1-73, 5-29, 5-30, 5-31 MCL 436.1703(3) 5-29, 5-31, Glossary-35 MCL 436.1703(4) Glossary-35 MCL 436.1703(5) 5-29, 5-30 MCL 436.1703(6) 5-32 MCL 436.1703(7) 5-32, 5-33 MCL 436.1703(8) 5-34 MCL 436.1703(9) 5-33 MCL 436.1703(10) 5-33 MCL 436.1703(12) 5-33 MCL 436.1703(13) 5-34 MCL 436.1703(14) 5-34 MCL 436.1703(16) 5-31 MCL 436.1703(17) Glossary-2, Glossary-36 MCL 436.1705 5-31 MCL 436.1707 Glossary-35 MCL 436.2303 5-27 MCL 440.1101 Glossary-39 MCL 450.2101 Glossary-24 MCL 450.3192 Glossary-24 MCL 474.103 Glossary-6, Glossary-7 MCL 475.1 Glossary-6 MCL 479.42 Glossary-6 MCL 500.3101 1-38, 1-64, 3-10, 3-11, 3-12, 6-10 MCL 500.3102 1-38, 1-64, 3-11

MCL 500.3177 9-8 MCL 552.601 1-72 MCL 552.650 1-72 MCL 600.113(1) Glossary-45 MCL 600.113(2) 1-3, 1-9 MCL 600.181 1-36 MCL 600.225 2-2 MCL 600.1007 2-10 MCL 600.1021(1) 1-40, 1-53 MCL 600.1060 Glossary-42 MCL 600.1084 9-13, Glossary-15, Glossary-18, Glossary-36, Glossary-42 MCL 600.1084(1) 9-13 MCL 600.1084(2) 9-13 MCL 600.1084(3) 9-13 MCL 600.1084(4) 9-14 MCL 600.1084(5) 9-13 MCL 600.1084(6) 9-15 MCL 600.1084(7) 9-11, 9-15 MCL 600.1084(8) 9-15 MCL 600.1084(9) Glossary-15, Glossary-19, Glossary-36, Glossary-42 MCL 600.1090 Glossary-42 MCL 600.1091(1) 1-41, 1-53 MCL 600.1094(1) 1-41, 1-54 MCL 600.1099b(a) Glossary-10 MCL 600.1099b(c) Glossary-12 MCL 600.1099b(e) Glossary-22 MCL 600.1099b(f) Glossary-25 MCL 600.1099b(g) Glossary-31 MCL 600.1099b(h) Glossary-39 MCL 600.1099b(i) Glossary-39 MCL 600.1099c 1-40, 1-53 MCL 600.1099c(3) Glossary-20 MCL 600.1099f(1) 1-40, 1-53 MCL 600.1200 Glossary-42 MCL 600.1517 2-2 MCL 600.8301(2) 1-9, 2-3 MCL 600.8311(a) 1-40 MCL 600.8311(d) 1-53 MCL 600.8311(e) 1-40 MCL 600.8311(f) 1-53 MCL 600.8312 1-10 MCL 600.8312(6) 1-10 MCL 600.8381(4) 1-50 MCL 600.8511 1-42, 1-52 MCL 600.8511(b) 1-41, 1-44 MCL 600.8512 1-11, 1-52 MCL 600.8512(1) 1-11 MCL 600.8512(3) 1-11 MCL 600.8512(4) 1-11 MCL 600.8512a 1-52 MCL 600.8512a(a) 1-12

MCL 600.8513 1-52 MCL 600.8515 1-25 MCL 600.8541(1) 1-12 MCL 600.8703(2) 2-3 MCL 600.8717(4) 1-20, 1-22, 1-26 MCL 600.8719 4-6 MCL 600.8721 4-6 MCL 712A.1 2-5, Glossary-20 MCL 712A.1(1) Glossary-20 MCL 712A.2(a) 1-40, 2-2, 2-3, 2-5, 2-11, Glossary-28 MCL 712A.2(d) 2-2, 2-3 MCL 712A.2b 2-5, 2-9 MCL 712A.2b(b) 2-7 MCL 712A.2b(c) 2-9 MCL 712A.2b(d) 2-10 MCL 712A.2b(e) 2-10 MCL 712A.2c 2-9 MCL 712A.2d(2) 2-5 MCL 712A.4 2-5 MCL 712A.11(4) 2-2 MCL 712A.12 2-6, 2-7 MCL 712A.13 2-6, 2-7 MCL 712A.14(1) 2-4 MCL 712A.14(2) 2-4 MCL 712A.17c(1) 2-6 MCL 712A.18 2-6, 2-9, 2-10 MCL 712A.18(1) 2-9, 2-10 MCL 712A.18(3) 2-11 MCL 712A.18(7) 2-11 MCL 712A.18(8) 2-9 MCL 712A.18(9) 2-9, 2-10 MCL 712A.18(10) 2-9 MCL 712A.18(11) 2-10 MCL 712A.18e 2-11 MCL 712A.18e(1) 2-11 MCL 712A.18e(10) 2-12 MCL 712A.18e(17) 2-11 MCL 712A.18t 2-11 MCL 712A.18t(10) 2-11 MCL 712A.30 2-11 MCL 712A.30(2) 2-11 MCL 712A.31 2-11 MCL 730.551 1-40 MCL 750.1 2-5 MCL 750.5 Glossary-8 MCL 750.8 6-25 MCL 750.90e 6-25 MCL 750.92 1-3, 1-4, 1-5 MCL 750.92(1) 1-5 MCL 750.167a Glossary-36 MCL 750.223(3) Glossary-16

```
MCL 750.224f Glossary-16
MCL 750.224f(10) Glossary-16
MCL 750.237 Glossary-36
MCL 750.317 9-32
MCL 750.321 1-61, 8-29, 8-30, 8-31, 9-32
MCL 750.377d 5-26, Glossary-45
MCL 750.377d(1) 5-26, 5-27
MCL 750.377d(2) 5-26
MCL 750.377d(3) Glossary-45
MCL 750.382 8-27, 8-28
MCL 750.382(1) 8-27, 8-28
MCL 750.413 1-4, 2-11, 6-27, 8-31, 8-32
MCL 750.414 6-26, 6-27, 8-32
MCL 750.479a 8-2, 8-3, 8-6, 8-7, Glossary-40
MCL 750.479a(1) 8-3
MCL 750.479a(2) 8-4, 8-6
MCL 750.479a(3) 8-4
MCL 750.479a(4) 8-3, 8-4
MCL 750.479a(5) 8-4, 8-6
MCL 750.479a(6) 8-5
MCL 750.479a(7) 8-5
MCL 750.479a(8) 8-6
MCL 750.479a(9) Glossary-40
MCL 750.503 1-55
MCL 750.520b 5-33
MCL 750.520g 5-33
MCL 761.1(c) Glossary-9
MCL 761.1(f) Glossary-16
MCL 761.1(n) Glossary-26
MCL 762.1 1-53
MCL 762.11 2-12, Glossary-45
MCL 762.11(3) 2-12
MCL 762.11(7) Glossary-45
MCL 764.1a 1-54
MCL 764.1a(2) 1-54
MCL 764.1e 1-44
MCL 764.1f 2-5
MCL 764.3(1) 1-45
MCL 764.3(2) 1-45
MCL 764.3(3) 1-45
MCL 764.3(4) 1-46
MCL 764.9a(1) 1-47
MCL 764.9c 1-42, 1-46, 5-31, 9-2, Glossary-2, Glossary-29
MCL 764.9c(1) 1-42, 9-2
MCL 764.9c(4) 1-42, 9-2
MCL 764.9c(5) 1-42, 9-2
MCL 764.9c(8) 1-43, 9-3
MCL 764.9c(9) Glossary-30
MCL 764.9f(1) Glossary-2
MCL 764.9g 1-45, Glossary-2
MCL 764.9g(1) 1-45
```

MCL 764.13 1-42, 1-47, 9-2 MCL 764.27 2-5 MCL 765.6 6-22 MCL 767.39 8-5 MCL 768.5(5) 1-53 MCL 768.32(1) 8-24 MCL 769.1a(2) 1-8 MCL 769.1j(1) 1-50, 1-57 MCL 769.1j(3) 1-57 MCL 769.1k 1-51, 1-52, 1-58, 1-59 MCL 769.1k(1) 1-50, 1-51, 1-52, 1-57, 1-58, 1-59 MCL 769.1k(2) 1-52 MCL 769.1k(4) 1-52 MCL 769.1k(5) 1-52 MCL 769.1k(10) 1-52 MCL 769.5 Glossary-40 MCL 769.5(3) 1-49 MCL 769.5(4) 1-49 MCL 769.5(6) 1-53 MCL 769.5(7) Glossary-40 MCL 769.31 Glossary-11 MCL 769.31(a) Glossary-12 MCL 769.34 Glossary-11 MCL 769.34(2) 1-55 MCL 769.34(3) 1-57 MCL 769.34(6) 1-51, 1-52, 1-59 MCL 769.36(1) 8-6, 8-13, 8-29, 9-29 MCL 771.1 1-7 MCL 771.1(1) 1-6 MCL 771.1(2) 1-6 MCL 771.3(1) 1-50, 1-57 MCL 777.1 1-57, Glossary-12 MCL 777.5 1-56 MCL 777.5(a) 1-56 MCL 777.21 1-56 MCL 777.21(1) 1-56 MCL 777.22 1-56 MCL 777.22(1) 1-56 MCL 777.22(5) 1-56 MCL 777.48 Glossary-2 MCL 777.55(2) 9-37 MCL 777.61 1-56 MCL 777.69 1-56 MCL 780.64 6-22 MCL 780.766(2) 1-8 MCL 780.781 2-7 MCL 780.794 2-11 MCL 780.794(2) 1-8 MCL 780.795 2-11 MCL 780.811 Glossary-40 MCL 780.826(2) 1-8

TOA: MCLs - 24

MCL 780.901(f) 2-11 MCL 780.905(3) 2-11 MCL 780.981 Glossary-21 MCL 780.1003 Glossary-21

Μ

MCR 1.109(A) 11, 14 MCR 1.109(D) 30 MCR 1.109(E) 15, 17, 43 MCR 1.110 31 MCR 1.112 3 MCR 2.301(A) 23, 26 MCR 2.407 23, 26 MCR 2.407(A) 31, 49 MCR 2.407(B) 21, 22, 23, 26 MCR 2.501(C) 24, 27 MCR 2.603(D) 30 MCR 2.612 30 MCR 3.606(F) 39 MCR 3.901 4 MCR 3.901(A) 3 MCR 3.901(B) 4 MCR 3.902(A) 4 MCR 3.903(A) 2, 5, 8, 11 MCR 3.903(B) 2, 4, 5, 11, 28 MCR 3.907 7, 8 MCR 3.912(A) 9 MCR 3.913(A) 9 MCR 3.920 7 MCR 3.921 7 MCR 3.925(F) 11 MCR 3.926(A) 3 MCR 3.926(B) 3 MCR 3.926(D) 3 MCR 3.931(A) 6, 7 MCR 3.931(B) 7 MCR 3.931(C) 6 MCR 3.931(D) 9 MCR 3.932(A) 8 MCR 3.932(C) 6 MCR 3.933(B) 6, 7, 8 MCR 3.934(A) 8 MCR 3.935 4, 8 MCR 3.935(A) 8 MCR 3.935(B) 8 MCR 3.943 9 MCR 3.943(E) 10 MCR 3.950 9 MCR 3.991 4 MCR 3.993 4 MCR 4.101 21, 22, 23, 24, 26, 27 MCR 4.101(A) 12, 13, 15, 17, 18 MCR 4.101(B) 20, 21, 22, 28, 29 MCR 4.101(C) 24 MCR 4.101(D) 29, 30 MCR 4.101(E) 20, 21, 22

MCR 4.101(F) 21, 22, 23, 24, 26, 27, 31 MCR 4.101(G) 31, 36, 68 MCR 4.101(H) 20, 21, 24, 25, 27, 28 MCR 4.401(B) 11 MCR 4.401(C) 12, 25, 42 MCR 4.401(D) 25 MCR 5.113 3 MCR 6.001(A) 6, 8 MCR 6.001(B) 8 MCR 6.008(A) 40 MCR 6.008(B) 53 MCR 6.008(C) 53 MCR 6.008(D) 53 MCR 6.008(E) 41, 54 MCR 6.102(A) 54 MCR 6.102(C) 54 MCR 6.102(D) 54 MCR 6.103 45 MCR 6.103(A) 45 MCR 6.103(B) 45 MCR 6.103(C) 45 MCR 6.104 53 MCR 6.110 53 MCR 6.201(A) 53, 54 MCR 6.302 6 MCR 6.302(F) 6, 8 MCR 6.425(D) 39, 50, 52, 58 MCR 6.430 9 MCR 6.430(A) 9 MCR 6.615 44 MCR 6.615(A) 42, 43 MCR 6.615(B) 45, 46 MCR 6.615(D) 40 MCR 7.104 25 MCR 7.104(D) 28 MCR 7.113 28 MCR 7.215(J) iv MCR 8.110(C) 2 MCR 8.110(D) 8 MCR 8.119 10, 11

Michigan Rules of Evidence

MRE 401 4-35 MRE 403 4-35 MRE 702 9-63 Table of Authorities: Michigan Rules of Evidence Traffic Benchbook - Fourth Edition

Constitutional Authority

Michigan Constitutional Authority

Const 1963, art 1, § 2 9-57 Const 1963, art 1, § 11 9-57 Const 1963, art 1, § 11 1-9, 2-4 Const 1963, art 3, § 2 1-50, 1-57 Const 1963, art 4, § 32 1-50, 1-57 Const 1963, art 4, §24 7-10 Const 1963, art 6, §13 1-53

U.S. Constitutional Authority

US Const, Am IV 1-9, 2-4, 9-57 US Const, Am XIV 9-57 Table of Authorities: MI & US Constitutions Traffic Benchbook - Fourth Edition

Michigan Criminal Jury Instructions

M Crim JI 13.6 8-3, 8-4 M Crim JI 15.1 9-15 M Crim JI 15.2 9-16, 9-24 M Crim JI 15.3 9-16, 9-41 M Crim JI 15.4 9-24 M Crim JI 15.5 9-16, 9-24 M Crim JI 15.6 9-16 M Crim JI 15.7 9-16 M Crim JI 15.9 9-57 M Crim JI 15.11 9-28 M Crim JI 15.12 9-33 M Crim JI 15.14 5-7, 5-8, 8-11 M Crim JI 15.15 6-23 M Crim JI 15.16 8-23 M Crim JI 15.17 8-25 M Crim JI 15.18 6-17 M Crim JI 15.19 6-15, 6-16 M Crim JI 15.20 6-18 M Crim JI 15.23 5-11 M Crim JI 15.24 7-2 M Crim JI 15.25 7-2 M Crim JI 16.8 8-29 M Crim JI 16.10 8-29 M Crim JI 24.1 8-31 M Crim JI 24.2 6-26 M Crim JI 24.4 6-26, 6-27, 8-31, 8-32 M Crim JI 24.6 7-9 M Crim II 24.7 7-9

Table of Authorities: Michigan Criminal Jury Instructions Traffic Benchbook - Fourth Edition